

# Professional interpreters as a subgroup of highly proficient bilinguals: a Dynamic Systems Theory (DST) approach

Anastasia Barsukova

Ph.D

June 2017

London Metropolitan University

- CONTRACTOR AND	CALL THE PROPERTY OF THE PROPE
Dess	18/9/17
Fund	
Control no	
Collection/ Loan type	sthrt REF
Class	418.02
Accession Di-	3011783001
AND STREET, ST.	the same of the sa

#### Acknowledgements

I am very grateful to my supervisory team, Udo Jörg and Alex Krouglov, for their continuing support and advice throughout this process. I would also like to thank Christopher Thiéry, whose pioneering work on the bilingualism of interpreters served as an inspiration for this project, and who very generously shared his only copy of his doctoral thesis with me when I was still an MA student.

I am also deeply indebted to my parents, Igor Barsukov and Lena Barsukova, without whose encouragement and belief that I could get this project done, it would certainly not have been completed. There are numerous other people who supported and encouraged me during my time as a Ph.D student. I would particularly like to thank Gay, for being proud of me for doing this, it means a lot, and Nada, for putting up with me in the difficult times.

Finally, I sincerely thank all of the interpreters who took part in this study. Without your generosity, there would be no project.

#### Abstract

Professional interpreters are often described as "exceptional" bilinguals, however, there is very little information in the literature about the particulars of this kind of bilingual proficiency, or the conditions and life experiences resulting in its development (Valdés and Angelelli, 2003). Information about the bilingual proficiency of interpreters tends to be based on the guidelines and definitions of professional organisations, such as the AIIC A and B language classifications used in conference interpreting. However, with the exception of Thiéry's (1975) inquiry into "true bilingualism", the implications of these classifications have not been investigated in terms of what they correspond to in practice with respect to interpreters' bilingualism. Furthermore, the bilingualism of interpreters who specialise in other professional domains, such as the public services, has not yet been studied or described. At the same time, it is maintained that the contribution of studies looking to describe the bilingualism of interpreters and its development is not limited to the discipline of interpreting only, and may provide important insights into aspects of individual bilingualism more generally (Valdés and Angelelli, 2003).

The present project contrasts three populations of professional interpreters, AIIC AA and AB conference interpreters, and public service interpreters (PSIs) on the UK National Register. The comparison takes into consideration biographical and perceptional factors pertaining to bilingualism, and the interrelationships between biographies and perceptions. The three populations of interpreters were considered as a group of highly proficient bilinguals characterised by a set of additional properties, such as strong language maintenance motivation and intensive use of both languages. Data on linguistic biographies, attitudes and perceptions were collected from the three groups by means of a large-scale questionnaire (N = 400). Applying the principles of Dynamic Systems Theory (DST), the data were analysed in terms of the relationships between components of the bilinguals' external and internal environment.

An analysis of the findings for the three groups together provided evidence in support of the DST view of bilingualism with respect to the development of the bilingual language system over a lifetime. A comparison of the AIIC AA and AB interpreters revealed differences between the groups with respect to both biographical factors and perceptions, which were discussed in terms of their implications for bilingualism. In particular, the findings appeared to be in line with the literature on age and context of SLA and ultimate L2 attainment. At the same time, individual variation was observed within both the AA and AB groups, which may be considered evidence against a critical period for SLA. In addition, childhood linguistic factors were found to be important in shaping notions of identity and attachment to the languages among the two groups. The PSI and AIIC AB groups were found to be similar in terms of linguistic biographies, but with the present dominant linguistic environment of the PSI group skewed significantly towards the L2.

A comparison of PSI and AB groups in terms of perceptional factors pointed to the influence of the contextual factor of linguistic environment on bilinguals' internal perceptions. In particular, the linguistic environment was found to have an effect on perceptions of cross-linguistic interference, as well as language confidence.

Several directions for further research at the intersection of interpreting and bilingualism were identified on the basis of the results of the study. More generally, it is proposed that professional interpreters represent a valuable population for research into aspects of individual bilingualism, in particular in the context of a DST approach, and that the inclusion of interpreters of different types (e.g. conference, PSI) in such studies would only serve to enrich and broaden the scope of the research.

# **Table of Contents**

Acknowledgements	j
Abstract	ii
Table of Contents	iv
1. Introduction	1
1.1 Introduction	1
1.2 Thesis aims and outline	2
1.3 Terminology	4
2. Bilingualism	7
2.1 Defining individual bilingualism	8
2.1.1 The problem of defining the native	9
2.1.2 Competence vs. language use	11
2.2 Describing individual bilingualism	13
2.3 The bilingual person	26
2.3.1 Bilinguals as unique speaker-hearers	26
2.3.2 Biculturalism and bilingualism	29
2.4 Degree of bilingualism and advanced bilingual proficiency	30
2.4.1 Measuring bilingualism	31
2.4.2 Highly proficient bilinguals	32
3. Bilingualism as a Dynamic System	35
3.1 Dynamic systems theory (DST): mathematical origins	35
3.1.1 Dynamical systems theory	36
3.1.2 Chaos theory	36
3.1.3 Complex systems	37
3.2 DST in the wider context of cognition and development	38
3.3 A DST approach to second language development	41
3.3.1 Rationale for adopting a DST framework	42
3.3.2 Dynamic models of second language development	44
3.3.3 Examples of DST studies of second language development	47
3.4 A DST approach to bilingualism: implications for the present project	49
3.4.1 A dynamic view of bilingualism	49
3.4.2 The bilingual profile as a dynamic concept	51

4. Interpreting and Bilingualism	
4.1 What is interpreting?	54
4.1.1 Definitions and domains	
4.1.2 Conference and public service interpreting: some key distinctions	
4.2 Interpreting and the study of bilingualism	
4.2.1 Cross-disciplinary research: overview of main directions and recent studies	
4.2.2 "True bilingualism": Thiéry's (1975) investigation of the bilingualism of AIIC AA	67
interpreters	
4.3 Interpreters as a group of highly proficient bilinguals	77
5. Research Questions and Methodology	79
5.1 Research questions and hypotheses	79
5.1.1 Research questions	
5.1.2 Hypotheses	80 82
5.2 Design of the study	87
5.2.1 Theoretical framework	87
5.2.2 Data collection instrument: large-scale questionnaire	88
5.2.3 Participants: AIIC AA, AB and UK PSI interpreters	
5.3 Data processing and analysis	
5.3.1 Initial data processing and software used	
5.3.2 Coding of respondents' working languages	
5.3.3 Statistical methods used and presentation of results	93
6. Results	94
6.1 Data collection results	94
6.1.1 Response rate	
6.1.2 General description of the populations	94 95
6.2 Questions pertaining to the AIIC AA group	98
6.2.1 Two mother tongues and the AIIC AA classification	98
6.2.2 Perceptions of the true bilingualism definition	100
6.2.3 Further investigation of results: a closer look at anomalous groups	103
6.3 Description and comparison of the AHC AA and AB groups	
6.3.1 Mother tongue identity, biographical factors and language use	115
6.3.2 Attitudes, perceptions and self-ratings	124
6.3.3 Self-ascribed mother tongue identity vs. identity ascribed by others among the AB	132
group	
6.4 Description of the PSI group and comparison with the AIIC AB group	137
6.4.1 Mother tongue identity, biographical factors and language use	
6.4.2 Attitudes, perceptions and self-ratings	

7. Discussion	153
7.1 Interpreters as a group of highly proficient bilinguals	15.
7.1.1 Biographical factors and language use	
7.1.2 Attitudes and perceptions	15:
7.2 True bilingualism and the AIIC AA and AB classifications	15'
7.2.1 Comparison of AIIC AA and AB groups and implications for bilingualism	15
7.2.2 The true bilingualism definition and the AIIC AA classification	165
7.2.3 AIIC AA interpreters who obtained their AA classification at a later date	16
7.3 Self-ascribed and externally-ascribed mother tongue identity in two languages	170
7.3.1 AIIC AA interpreters who do not consider that they have two mother tongues	170
7.3.2 AIIC AB interpreters who consider that they do have two mother tongues	174
7.4 Comparison of PSI and AB conference interpreters as bilinguals	170
7.4.1 Biographical factors and language use	176
7.4.2 Attitudes and perceptions	178
8. Conclusion	184
8.1 Summary of key findings	184
8.1.1 Interpreters as highly proficient bilinguals	184
8.1.2 Characteristics of AIIC AA interpreters and their implications	185
8.1.3 Factors associated with self-ascribed mother tongue identity in two languages	
8.1.4 Implications of the comparison of the AIIC AB and PSI groups for bilingualism	192
8.2 Directions for future research	194
8.2.1 Questions pertaining to the AIIC AA and AB groups	194
8.2.2 Questions raised by the comparison of the PSI and AIIC AB groups	197
8.3 Strengths and limitations of the study	199
References	201
Appendix I: Questionnaires: AIIC AA, AIIC AB and PSI groups	210
Appendix II: Comparison of responses given by the subgroup of AIIC AA interpreters	242
who do not consider both A languages to be mother tongues with those of the rest of the	
AA group	
Appendix III: Comparison of responses given by the subgroup of AIIC AA interpreters	250
who were awarded their AA classification after joining AIIC with those of the rest of the	
AA group	
Appendix IV: Comparison of responses given by the subgroup of AIIC AB interpreters	258
who consider both of their languages to be mother tongues with those given by the rest of	
the AB group	

#### 1. Introduction

#### 1.1 Introduction

The present project focuses on a group of bilinguals who are often referred to in the literature as "unique" or "exceptional" (e.g. Babcock and Vallesi, 2017; Valdés and Angelelli, 2003) – that of professional interpreters. While interest in this group within the field of bilingualism appears to be on the rise, as evidenced by the recent dedication of a special issue of the *International Journal of Bilingualism* (16:2, June 2012) to the topic of simultaneous interpreting, research at the intersection of the two disciplines has mostly focused on the cognitive processes involved in interpreting, or the psycholinguistic characteristics of interpreters. What has received far less attention is the nature of interpreters' bilingualism more generally – their lived experiences, biographical and personal characteristics. Indeed, as pointed of by Valdés and Angelelli (2003), literature on interpreters and interpreting has tended to assume a "perfect grasp" or "mastery" of the working languages, yet very little is known about the particulars of this assumed language proficiency, or the conditions that have led to its development. Furthermore, it must be noted that research on the bilingualism of interpreters has focused almost exclusively on conference interpreters, and the bilingualism of interpreters who work in other settings, such as the public services, has not yet been studied or described in a systematic way.

Information about the bilingual proficiency of interpreters tends to be based on the guidelines and definitions of professional organisations. In particular, within the field of conference interpreting, the International Association of Conference Interpreters (AIIC) classifies interpreters' active working languages as either A or B, where an A language is the "mother tongue or its strict equivalent", whereas a B language is taken to be a foreign language in which the interpreter is "perfectly fluent" (AIIC, 2017; see Chapter 4 for details). However, there has only been one detailed inquiry to date into the implications of these language classifications in terms of bilingualism: the work of Christopher Thiéry (1975) on AIIC interpreters with two A languages (i.e. two "mother tongues"). In his unpublished 1975 doctoral thesis on the topic Thiéry, himself an AIIC AA interpreter, described the bilingualism of a sample of this population in terms of biographical, personal and attitudinal factors (see Thiéry, 1978; 1982 for summary). Thiéry's aim was to understand what the AIIC AA classification corresponds to in terms of bilingualism and the characteristics and life experiences of this group, defined by him as "true bilinguals" (see Thiéry, 1978). Thiéry's pioneering study is unique not only because it represents the first attempt to probe assumptions about interpreters' bilingual proficiency, but it remains the only detailed description of the bilingualism of professional interpreters with respect to both biographical and attitudinal factors.

It is considered that the limited amount of research into interpreters' bilingualism and its development is an important gap in the literature at the interface of bilingualism and interpreting, and such studies have the potential to make a contribution to both disciplines. According to Valdés and Angelelli (2003), investigations of this kind can provide important information on such aspects of individual bilingualism as the contexts of language acquisition that affect ultimate second language attainment and, in particular, contribute to debates on the critical period in second language acquisition. As regards interpreting studies, inquiries into the linguistic backgrounds of different groups of interpreters may offer insights into questions such as who interprets in which situations, and what the characteristics of these interpreters are (*ibid*). In addition to these potential contributions, it is argued here that descriptive studies of the bilingualism of interpreters may provide other insights into the bilingualism of groups who are considered to be highly proficient in both languages, including into the relationships between external and internal (personal) contextual factors, and the dynamics between self-ascribed identity in the two languages and that ascribed by others.

#### 1.2 Thesis aims and outline

The present dissertation aims to investigate some of the above questions by describing and comparing three populations of professional interpreters as bilinguals in terms of biographical factors, attitudes and perceptions. The specific populations included in the study are AIIC AA and AB conference interpreters, and public service interpreters (PSIs) on the UK National Register (NRPSI). By extending the scope beyond the domain of conference interpreting, the project aims to investigate the general question of what the characteristics of professional interpreters are as a group of highly proficient bilinguals, regardless of professional setting. At the same time, the division of the populations of interpreters into the above three groups makes it possible to explore several specific questions identified on the basis of the review of the literature, including questions raised by Thiéry's (1975) inquiry into the AIIC AA language classification, as well a number of the questions identified by Valdés and Angelelli (2003) (see Chapter 5, Section 5.1.1 for full list of research questions). In addition, the findings are examined in terms of their implications for bilingualism, with reference to recent theoretical developments in the field, in particular, the dynamic systems theory (DST) approach to second language development.

The thesis is divided into two main sections: a theoretical part (Chapters 2-4), which reviews the relevant literature, and an experimental part (Chapters 5-8), which sets out the research questions and study design, presents the findings and their analysis. **Chapter 2** focuses on individual bilingualism, explaining how the term is defined in the context of the present dissertation and introducing some fundamental issues and concepts. In particular, the multidimensional nature of bilingualism is discussed, and the dimensions that are relevant for the description of individual

bilingualism are identified. The Chapter also covers issues related to the bilingual person, including the unique nature of bilingual competence (e.g. Grosjean, 1989) and the relationship between bilingualism and biculturalism. Finally, the complex issue of degree of bilingualism is discussed, and it is explained in what sense bilingualism can be "measured"; in light of this, it is clarified what is meant by a high level of bilingual proficiency in the context of the present study. Chapter 3 builds on the theory of bilingualism set out in the preceding Chapter, by introducing the theoretical framework of DST, which makes it possible to take into account the external and internal contextual factors that are relevant for the description of individual bilingualism, as well as their interaction over time. This Chapter is structured so at to, first of all, explain the origins of DST, which has its roots in mathematics, and how this approach has been applied in a more general way in the sphere of cognition and development. It is next explained what DST means when it is applied to the study of bilingualism, and why it is considered a relevant and useful set of principles in this context. Some specific examples of DST studies looking at aspects of individual bilingualism are considered, before explaining the implications of adopting this approach for the present project. Chapter 4 concerns interpreting and bilingualism. First of all, interpreting is defined and the key domains are outlined, with a focus on the distinction between the settings of conference and public service interpreting. Next, an overview of the literature at the intersection of interpreting and bilingualism is presented, identifying the main trends and the areas that remain unexplored. In particular, it is observed that limited research has been conducted into the nature of interpreters' "exceptional" bilingualism and its development (Valdés and Angelelli, 2003). As a unique example of such a study, Thiéry's (1975) inquiry into "true bilingualism" is discussed in detail, and the questions raised by this study are identified. The Chapter concludes with an analysis of what is known about interpreters as a group of bilinguals, and a bilingual profile of this population is proposed on the basis of the theory set out in the preceding chapters.

Chapter 5 formulates the research questions to be investigated as part of the present project, and sets out the hypotheses on the basis of the literature covered in the theoretical part of the thesis. The study design is explained and described, as are the participants involved in the study. Details are provided on technical aspects of data processing and analysis, including the software and statistical methods used. Chapter 6 sets out the results for all three groups of interpreters. This Chapter is intended to be a presentation of the findings only, and does not include any analysis of the results on the basis of the literature. Chapter 7 discusses the implications of the findings in light of the research questions and hypotheses set out in Chapter 5, with reference to the literature. The structure of the discussion follows the order in which the research questions are presented.

Chapter 8 summarises the key findings and their implications, identifying the main conclusions drawn from the study. A number of directions for further research are identified, with reference to some specific methodological recommendations. Finally, some of the strengths and limitations of the present study are discussed.

# 1.3 Terminology

The following is a list of some of the key terms and abbreviations that are used frequently throughout the present dissertation.

AIIC - International Association of Conference Interpreters (see Chapter 4).

**Attrition** – defined here as the "non-pathological decrease in proficiency in a language that had previously been acquired by an individual" (Schmid and Köpke, 2004) (see Chapter 2, Section 2.2).

**Bilingualism (bilingual)** – in the context of the present dissertation, the term "bilingualism" is taken to mean individual bilingualism. Bilinguals are defined here as "those who use two or more languages (or dialects) in their everyday lives" (Grosjean, 2010: 4) (see Chapter 2, Section 2.1.2).

**Code-switching** - defined here as the alternate use of two languages by a bilingual, i.e. when a speaker changes to another language for a word or phrase, and then reverts back to the original language (Grosjean, 2010) (see Chapter 2, Section 2.2).

**DMM** – Dynamic Model of Multilingualism (Herdina and Jessner, 2002). A model of multilingualism developed on the basis of DST principles (see Chapter 3, Section 3.3.2).

**DST** – dynamic systems theory. A theory concerned with the study of change over time, which has its origins in mathematics. In the context of applied linguistics, DST refers not to a single theory, but to a set of general principles, including continuous temporal evolution, complete interconnectedness and sensitive dependence on initial conditions (see Chapter 3).

Highly proficient bilingual – bilingual proficiency is taken to be a relative concept, and hence where the bar is set for 'high' proficiency is also relative (e.g. Goto Butler, 2013). Within the context of the present dissertation, the term is defined as a function of language use and refers, in particular, to bilinguals who use both languages regularly in both formal (social, family) and informal (professional, educational) settings (see Chapter 2, Section 2.4.2 for full characterisation).

**Interference** – refers to cross-linguistic interference between a bilingual's languages. Here, interference is taken to mean the involuntary influence of one language on the other (after Grosjean, 1982) (see Chapter 2, Section 2.3.1).

Interpreting (interpreter) – interpreting is defined broadly as "interlingual, intercultural oral or signed mediation, enabling communication between individuals or groups who do not share, or do not choose to share, the same language(s)" (Pöchhacker and Shlesinger, 2002: 2-3). However, in the context of the present dissertation, the term "interpreter" is used to refer to professional spoken language interpreters specifically (see Chapter 4).

Interpreter's A and B languages – used in the context of conference interpreting to refer to the interpreter's active working languages. According to the AIIC, an A language is the interpreter's "mother tongue or its strict equivalent", while the B language is a foreign language in which the interpreter is "perfectly fluent" (see Chapter 4). The terms "AA interpreter" and "AB interpreter" are used to refer to AIIC interpreters with these language combinations.

L1 – a bilingual's first language.

L2 – a bilingual's second language.

Mother tongue – defined broadly as an individual's "first linguistic experience during the formative years of language development" (Hamers and Blanc, 2000: 2). As used here, the term "mother tongue" is taken to refer to an individual's first language. However, it is recognised that this term carries certain emotional connotations and, in particular, associations with childhood (see Chapter 2, Section 2.1.1).

Native language – as used here, the term "native language" is taken to refer to an individual's first language. No assumptions are made regarding proficiency in this language, or the individual's cultural identity. It is also not assumed that individuals can have only one native language (see Chapter 2, Section 2.1.1).

Native-like – in a general sense, the term native-like competence is taken to refer to speakers who are distinguishable from native speakers in small ways (Gass and Glew, 2013) (see Chapter 2, Section 2.1.1 for more technical definition and discussion).

NRPSI – UK National Register of Public Service Interpreters (see Chapter 4).

SLA – second language acquisition.

SLD – second language development. A general term used to refer to changes in the bilingual language system over time, which include the acquisition, attrition and maintenance of the languages (see Chapter 3, Section 3.3).

**PSI** – public service interpreter or public service interpreting (see Chapter 4).

True bilingualism – definition proposed by Thiéry (1975) as a more formal equivalent of the AIIC AA language. A true bilingual is defined as "someone who is taken to be one of themselves by the members of two different linguistic communities, at roughly the same social and cultural level" (Thiéry, 1978: 146) (see Chapter 4, Section 4.2.2).

#### 2. Bilingualism

What is bilingualism? At first glance, the concept appears fairly straightforward. The Oxford English Dictionary, for example, defines bilingualism simply as "fluency in or use of two languages". This initially seems intuitive; however, even a little probing reveals that this apparent simplicity is, in fact, deceptive. Indeed, such are the complexities involved in defining bilingualism that, currently, no universally agreed-upon definition exists, with the concept interpreted differently in different fields, such as second language acquisition (SLA), psychology or education (Gass and Glew, 2013: 270).

What is it that makes bilingualism so problematic to define, and where do the issues arise? To illustrate the challenges involved, consider the two popular definitions below:

"Bilingualism (noun): the ability to speak two languages equally well" (Collins English Dictionary online, 2017);

"Bilingualism (noun):

- 1. the ability to speak two languages
- 2. the frequent use (as by a community) of two languages
- 3. the political or institutional recognition of two languages"

(Merriam Webster Dictionary online, 2017)

Immediately, questions begin to arise as to what, exactly, is covered by the term "bilingualism". Does it refer to individuals, or to communities? Is speech an inherent part of bilingualism and, if so, where does that leave nonverbal language, such as sign language? Does bilingualism require equal proficiency in two languages, or is habitual use of two languages sufficient? Finally, what exactly is meant by speaking a language "well"? If bilingualism is to be defined as a function of competence or fluency, it is not clear where one draws the line – in other words, where bilingualism 'starts' (Edwards, 2013: 5).

In fact, bilingualism can refer to both spoken language and nonverbal language: people who are able to use a sign language in addition to spoken language are bilingual, as are deaf people who use two different types of sign language (e.g. Gianico and Altarriba, 2013). Bilingualism can be approached from both social and individual perspectives. Hamers and Blanc (2000) even propose a terminological distinction to differentiate between the two, with the term "bilingualism" used to refer to the study of languages in contact at the societal level, and "bilinguality" – the individual level. Since the term "bilingualism", in its most general sense, can refer to all of these things – the individual language system, interpersonal relations and intergroup relations - the study of bilingualism is vast, encompassing numerous disciplines, such as sociology, psychology,

linguistics, cultural studies, neurolinguistics and education. Indeed, a multidisciplinary approach is often required to understand the complex, mutlidimensional (Hamers and Blanc, 2000) phenomenon that is bilingualism.

Even though the focus of the present thesis is individual, spoken language bilingualism, the social and cultural context cannot be ignored. The group and individual levels are inextricably linked and influence each other at all levels. For example, individual attitudes towards a language can shape the linguistic landscape at a community level (e.g. "English-only" movements in America, see Peréa and García Coll, 2013). On the other hand, individual pragmatic competence is shaped by implicit cultural pragmatic conventions (e.g. Paradis, 2004: 18), while government policies on bilingual education, for example, have a direct impact on individuals' linguistic environment (e.g. Garcia, 2013), affecting how and in what setting languages are used. Furthermore, individual bilingualism is about much more than the ability to speak or write in two (or more) languages — rather, it is a "socially based communication system" (Vega, 2013: 186), and as such cannot be understood outside the social and cultural context in which communication takes place (e.g. Larsen-Freeman, 2012).

The aim of this Chapter is to introduce some fundamental definitions and concepts related to individual bilingualism, which will form the basis of the subsequent theoretical and methodological chapters. First of all, bilingualism is defined and the key dimensions of bilingualism are outlined. In particular, the idea of a bilingual profile – a way of describing bilingualism – is introduced. Next, attention is turned to the bilingual person and the unique nature of bilingual competence, and it is argued, with reference to recent findings pointing to the interconnected nature of the bilingual language system, that bilinguals are not two monolinguals in one person (Grosjean, 1989). Finally, the complex issue of degree of bilingualism is discussed, in terms of what it means to say that a bilingual is "advanced" or "highly proficient". The debate on measuring or defining bilingual proficiency is still ongoing in applied linguistics; indeed, a recent special issue of the International Journal of Bilingualism (15(2), 2011) has been dedicated to this topic (see Daller, 2011 for introduction and overview). There are no simple answers, and an in-depth inquiry into this topic is beyond the scope of the present dissertation. Therefore, the last part of the present Chapter limits itself to a clarification of what is meant by highly proficient bilingualism in the context of the present dissertation, on the basis of the literature surveyed, and is not intended as a general definition of advanced bilingual proficiency.

## 2.1 Defining individual bilingualism

Definitions of bilingualism can essentially be divided into two types – those based on competence and those based on language use. The former correspond to earlier attempts to define bilingualism,

whereas the latter are in line with a more functional view of language, concerned instead with the purpose bilingualism serves and the context it functions in. Such an approach is the one that currently prevails in the field and, as explained in this Section, this is the view adopted in the present dissertation.

However, since the earliest definitions of bilingualism focused on competence in the two languages and, in particular, used the native as a standard against which this competence is to be measured, some attention needs to be given, first of all, to the fundamental concept of the native speaker.

# 2.1.1 The problem of defining the native

The idea of the native speaker is another concept the apparent simplicity of which is deceptive. The term is widely used in applied linguistics, yet no single, universally-accepted definition exists (see Faez, 2011 for an overview of different approaches to defining the native speaker concept). Part of the problem is down to the fact that the term "native speaker" actually encompasses not one, but several concepts, as well as numerous underlying assumptions. For example, the idea of language inheritance is implicit in the concept of the native speaker, that is, a native speaker is often defined as someone who has acquired the language in question from birth (Gass and Glew, 2013: 266). On the other hand, the term native speaker presupposes language expertise – it is often taken to refer to a group of people who "have a special control over a language, insider knowledge about 'their' language" (Davies, 2003: 1). Furthermore, the idea of the native speaker includes the notion of belonging to a linguistic community, being part of a culture or social group, both in terms of the individual's own opinion about what his/her native language is, as well as external validation – acceptance/ascription by others (e.g. Davies, 2003).

Problems arise when either the concept of the native speaker is reduced to only one of the dimensions, or when assumptions are made about what the various dimensions imply. For example, it cannot be assumed that language inheritance implies expertise. This is clearly not always the case: being born into a group does not automatically imply being able to speak the language well (e.g. Rampton, 1990). Moreover, some people acquire a language from birth and subsequently forget it – it is unlikely that such individuals would be considered native speakers of that language. Therefore, the concept of the native speaker cannot be defined as a function of language inheritance alone. Language acquisition and use is far removed from the theoretical "ideal speaker-listener in a completely homogeneous speech-community" (Chomsky, 1969: 3). It is much more complex, with most individuals acquiring different languages to varying degrees over a lifetime. In situations where individuals find themselves in different linguistic environments over a lifetime, where the use of their native language is limited and accompanied by the acquisition and intensive use of a different language, it is not always clear what the native language is, or even who a native speaker

is (Gass and Glew, 2013). Add to this the different varieties of a given language, as well as the cultural aspects of language use, and the situation becomes more complex still (Faez, 2011).

Possibly the most problematic aspect of the native speaker concept is the idea of native competence. If it were possible to define native competence, then a native speaker could be defined as someone who meets this standard in the language in question. The problem is, however, that native competence cannot be defined in practice. Theoretically, native competence is taken to mean "whatever it is that native speakers possess", implying that any native speaker can be taken to stand for all native speakers (see, for example, Cook, 2013 for discussion). However, in practice, competence varies greatly even within a monolingual population, thus making native competence impossible to define (e.g. Hamers and Blanc, 2000) and entirely relative. In other words, due to this individual variation, it is not possible to set a global standard that all monolingual native speakers meet, which can be used as an operational definition of native competence. Even if a single parameter were taken as a measure, such as grammar or vocabulary size, competence would still vary greatly in a monolingual population, depending on a variety of factors, such as education or aptitude (see DeKeyser, 2013 for discussion on variability among native speakers). The issue is so complex, that there does not currently exist an operational definition of even minimal native-level competence, or what it is that differentiates the native from the non-native (Faez, 2011). However, despite the fact that native competence evades definition in a general sense, the idea of the native speaker as an ideal or standard-setter prevails nonetheless, both in real life situations and applied linguistics (Aneja, 2016).

Thus, as no definition of native competence exists, a native speaker can't be defined on the basis of linguistic competence alone. What is more, any definition on this basis would essentially reduce the native speaker to a theoretical, abstract entity, entirely detached from the social context in which s/he functions. Such an approach is problematic, as illustrated by the following example. In an analysis of several case studies of international speakers of English, Brutt-Griffler and Samimy (2001) found that native speaker identity is determined by others on the basis of social factors - "preconceived notions of what a 'native speaker' should look like or sound like", and not within a linguistic construct of the native speaker. In other words, while linguistic factors such as accent may be used by members of a community to make judgements about whether someone is a native speaker of their particular language (or dialect) or not, social and cultural factors are at least of equal importance. Furthermore, in the same study it was found that self-ascribed native speaker identities do not necessarily correspond to external validations by the community (*ibid*). This suggests that neither self-ascription nor ascription by others can be used alone to define native speaker identity.

Taken together, the foregoing implies that the concept of the native speaker is too complex to be defined as a function of a single variable. It is also too complex to be considered a static concept. The most satisfactory approach to thinking about the native speaker, therefore, appears to be as a function of several key overlapping components, such as (relative) language expertise, language inheritance and language affiliation (Rampton, 1990), or proficiency, self-ascription and ascription by others (Davies, 2003). That is, the idea of the native speaker does not make sense outside of a particular context. It is a function of a set of internal (affiliation) and external (proficiency as perceived by others; acceptance by others) factors. Furthermore, defined as such, the idea of the native speaker is dynamic, as the various components, with the exception of language inheritance, can change over time.

Finally, before proceeding to discuss the topic of bilingualism, some terminological issues need to be addressed. As the present thesis concerns bilinguals - those who know and use more than one language - it is often necessary to distinguish between the individual's first and second language, for the purpose of discussion and comparison. Thus a certain degree of simplification is required. For this reason, the term "native language" as used in the context of the present thesis will be taken to mean an individual's first language, or L1. No assumptions will be made regarding proficiency in this language, relative or otherwise, or the individual's social or cultural identity (self-ascribed or ascribed by others). It is also not assumed that individuals can have only one first language, but such cases will be mentioned separately to avoid confusion. A separate mention is needed for the related term "mother tongue", which has been widely recognised as somewhat problematic (Hamers and Blanc, 2000: 2). At the psychological level, it can be defined as an individual's "first linguistic experience during the formative years of language development" (ibid), and it is often taken to be synonymous with the native language. The problem with using the term "mother tongue" in this way is that it is a less neutral and much more emotionally loaded term, invoking images of childhood, mothers, and the associated emotional response (e.g. Hoffmann, 1991). For this reason, throughout the present thesis, the term "mother tongue" shall be taken to also refer to an individual's first language, however, in light of the stronger emotional/cultural connotations it implies, the term will not be used unless it is specifically intended to invoke these associations, or unless it forms part of another definition (for example, the AIIC language classifications discussed in Chapter 4).

#### 2.1.2 Competence vs. language use

As previously stated, the earliest definitions of bilingualism tended to focus on competence in the two languages. One of the most famous early examples of such a definition is that proposed by Bloomfield, who defines bilingualism as the "native-like control of two languages" (1935: 56). The term "native-like" or "near-native" competence is used to refer to speakers who are distinguishable

from native speakers in small ways (Gass and Glew, 2013: 266). In a more technical, experimental sense, native-like (near-native) competence is defined in terms of a control group of native speakers, i.e. performance in the language in question must fall within the range of monolingual control subjects in order to be considered native-like (Birdsong, 2005). Therefore, the idea of native-like control of two languages is something that, like native competence itself, is entirely relative, depending on what it is being compared to, which parameters are included in the comparison, and where the cut-off point for "native-like control" is set by the researcher. Furthermore, setting the bar at "native-like control" seems arbitrary and excludes a vast group of people - those who fall short of the prescribed (ambiguous) standard in their two languages, yet are not monolingual. Setting such a standard also raises the question of whether one can cease to be bilingual: if the prescribed standard is attained, and then subsequently lost in either language, due to reduced use of that language, can the individual then still be considered bilingual? Finally, implicit in Bloomfield's definition is the idea that using the native monolingual standard as a benchmark against which to measure bilingualism is appropriate. In other words, it is assumed that bilingual competence is not a unique phenomenon, but that bilinguals effectively function as two monolinguals in one person. However, this does not appear to be the case, with evidence from studies on cross-linguistic influence in bilinguals seeming to support Grosjean's (1989) famous claim that bilinguals are not the sum of two monolinguals in one person (see Section 2.3.1 below for details and discussion).

Bloomfield's (1935) definition is an example of a maximalist viewpoint, and later competence-based definitions of bilingualism became much broader and more inclusive. For example, Haugen (1953) defined bilinguals as individuals who can produce complete meaningful utterances in a second language (cited in Goto Butler, 2013), while Macnamara defined bilinguals as "persons who possess at least one of the language skills [speaking, writing, listening, reading] even to a minimal degree in their second language" (Macnamara, 1967: 59-60). Yet, even with this more inclusive approach, definitions based on competence are not generally considered operational by most researchers (Hamers and Blanc, 2000). Firstly, this is because the definitions remain ambiguous, since the "minimal degree" of skill required is not specified, and it is not entirely clear what is meant by a complete meaningful utterance. Secondly, and perhaps most crucially, all competence-based definitions of bilingualism effectively reduce bilingualism to the linguistic dimension only, and isolate language and the bilingual person from the context in which they function, viewing both as abstract entities (Hoffmann, 1991). What the bilingual uses his or her languages for - a key consideration in the study of bilingualism (e.g. Mackey, 2000) - is entirely neglected when the focus is on competence alone.

The alternative approach is to focus not on what a bilingual individual can do (competence), but on what s/he uses her/his languages for, and in what setting. In other words, to define bilingualism as

the use of two languages (e.g. Grosjean, 2010). Such a viewpoint immediately situates the individual in his/her cultural and social context, thereby avoiding the problem of reducing the bilingual person to the purely linguistic dimension. Moreover, defining bilingualism on the basis of language use eliminates the requirement to define competence in an absolute way. For these reasons, this is the approach currently adopted by many researchers in the fields of bilingualism and SLA. The definition of bilingualism that shall be adopted in the present thesis is that proposed by Grosjean:

"Bilinguals are those who use two or more languages (or dialects) in their everyday lives" (Grosjean, 2010: 4).

What are the implications of defining bilingualism in this much more general way? An immediate corollary of the above definition is what Grosjean has called the "complementarity principle", namely: "Bilinguals usually acquire and use their languages for different purposes, in different domains of life, with different people. Different aspects of life often require different languages" (Grosjean, 2010: 29). This highlights the complexity of bilingualism – the great number of possibilities, combinations of "diverse individual and linguistic conditions ... nested in larger societal contexts which cause varying degrees and types of language contact" (Goto Butler, 2013: 112). Furthermore, defining bilingualism in this way – as the use of two or more languages in different contexts and for different purposes - emphasises the fluid nature of bilingualism. Instead of being viewed as a static phenomenon, it is conceived as something that is undergoing constant change over time, due to the continuous interaction of sets of external and internal environmental factors (see Section 2.2 below for details; also, Chapter 3). This recognition of the dynamic nature of bilingualism and the importance of contextual factors in SLA is in line with recent developments in the field of applied linguistics, calling for a dynamic systems or complexity theory approach to bilingualism (see, for example, Larsen-Freeman, 2012). This idea of viewing bilingualism as a (complex) dynamic system is discussed in detail in Chapter 3. However, first of all, the key contextual factors that shape bilingualism over time need to be identified, in order to narrow down and make operational the very broad definition of bilingualism set out above.

# 2.2 Describing individual bilingualism

It has long been recognised that bilingualism is "multidimensional" (e.g. Hamers and Blanc, 2000: 25). In other words, that there are a number of key contextual and individual factors that contribute to and shape bilingualism. These include aspects such as the age and context of language acquisition, (relative) competence in the two languages, the bilingual's linguistic environment and cultural identity (see Hamers and Blanc, 2000; Goto Butler, 2013, for a summary of the dimensions historically considered relevant by researchers). Over the years, researchers have attempted to

classify bilinguals on the basis of one or a combination of these dimensions, resulting in classifications such as early/late bilinguals (age of acquisition); balanced/dominant bilinguals (relative competence) and simultaneous/sequential bilinguals (order in which the languages were acquired) to name but a few (e.g. Goto Butler, 2013). The problem with such an approach is that the multidimensional phenomenon of bilingualism is reduced to only a couple of its key dimensions, resulting in an oversimplification and subsequent failure to capture its complexity.

Acknowledging that bilingualism is multidimensional implies that it can only be understood by taking into account all of the significant factors simultaneously, including both contextual, personal and linguistic dimensions (Hamers and Blanc, 2000: 25). At the same time, some degree of simplification is needed, otherwise the number of interacting factors would be overwhelming. A solution is to identify the key factors - those that are widely recognised as relevant by scholars in the field – and describe the specific bilingual situation in question on the basis of these factors, thus producing a bilingual profile of the individual or group. While some details and terminology may vary, there is a general consensus as to which dimensions are considered relevant for such a description, in other words, which factors are key in shaping individual bilingualism. These can be broken down into the following broad and overlapping categories: (1) age and context of language acquisition; (2) (relative) competence in the two languages; (3) state of the language system, that is, whether each language is being acquired, maintained or forgotten; (4) domains and context of language use; (5) the bilingual's external environment (linguistic, cultural, social); (6) the bilingual's internal environment (e.g. motivation, attitude towards the languages, identity, education, aptitude, personality) (see, for example, Goto Butler, 2013; Grosjean, 2010; Hamers and Blanc, 2000; Hoffmann, 1991).

The bilingual profile therefore represents a way of describing bilingualism at a specific moment in time. The various components are not separate – they interact and influence each other, shaping the individual's bilingualism over time. Examples of some the ways in which the various contextual, personal and psychological factors identified interact are given below, with an emphasis on the global effects of these factors on the development of bilingualism over time that have been identified in the literature.

#### (1) Age and context of language acquisition

The effect of age on L2 acquisition has received much attention in SLA and applied linguistics more generally, yet many questions remain unanswered. The age factor has been approached from various perspectives, in particular in terms of its effect on the route of acquisition, the rate of acquisition and the ultimate attainment of L2 proficiency, with the latter accounting for the majority of the research (see Goto Butler, 2013 for summary of main findings). Perhaps the most

contentious issue in terms of the influence of age of SLA on ultimate L2 attainment is whether there exists a "critical period" for L2 acquisition — defined as a limited developmental window during which the attainment of native-like¹ proficiency in the language is possible (e.g. Goto Butler, 2013: 123). The concept of a critical period, if it exists, implies a specific age range during which the capacity to learn a second language declines, and a biological reason for this decline (see, for example, DeKeyser, 2013, for historical discussion and summary of core issues). However, currently, there is no consensus as to the start and end points of the critical period, nor any biological evidence of its existence, and both of these facts are cited as arguments against the critical period hypothesis by some researchers (e.g. Birdsong, 2005). Therefore, the more neutral and general term "age effects" is preferred, as it does not carry the same theoretical implications as the idea of a critical period (DeKeyser, 2013: 443).

What researchers do agree on are two main things: firstly, that there is a strong negative correlation between the age of L2 acquisition and many aspects of grammar and pronunciation, and secondly that some individuals do pass for native speakers in everyday interaction, despite learning the language in adulthood (DeKeyser, 2013: 445). In particular, some studies have shown that late learners of an L2 are able to attain native-like proficiency, even when it comes to accent (e.g. Bongaerts et al., 2000). Taken together, what do these findings imply in terms of the effect of age of SLA on the development of the bilingual language system over time and, in particular, what predictions can be made on the basis of the age factor? First of all, the strong negative correlation between age of acquisition and ultimate attainment throughout the lifespan is undeniable according to the literature. Thus, at a global level, the probability of attaining native-like proficiency in the L2 does decrease with age. However, the fact that some individuals do seem to attain this level of proficiency despite learning the language later in life suggests that there is always the potential to attain native-like proficiency, in other words, one is not forever constrained by age effects. Furthermore, the importance of the effect of individual and contextual factors on L2 attainment cannot be underestimated (e.g. Birdsong, 2006). Indeed, some studies suggest that, in certain cases at least, some of these factors play a greater role than age in terms of learners' ultimate L2 attainment. For example, Pfenninger and Singleton (2016) found that motivation had a greater effect than age of onset in a study of 200 Swiss learners of English as a foreign language. Therefore, while the age of SLA can be used to make predictions about individual bilingualism at a global (population) level, it is important not to make assumptions about individuals on the basis of general trends.

The context of acquisition - in particular, the type of input and interaction the learner is exposed to - has also been found to be a significant factor in second language learning (e.g. Mackey, Abbuhl

<sup>&</sup>lt;sup>1</sup> While it is the view of the author that bilinguals cannot be reduced to two monolinguals in one person, comparing bilinguals with monolingual natives can be useful in some contexts (e.g. Birdsong, 2005).

and Gass, 2012; see also Hummel, 2014 for an overview of language learning contexts). Research on this topic has ranged from the very specific, such as the effect of particular classroom or study techniques on L2 acquisition (see, for example, Williams, 2012 for an overview of classroom research in SLA), to the influence of broader categories of learning contexts on aspects of second language acquisition. For example it has been found that study abroad contexts tend to favour the acquisition of L2 pragmatics, compared with contexts where learners do not spend time in their L2 environment (e.g. Bardovi-Harlig, 2012). While a discussion of the impact of different learning contexts on SLA is beyond the scope of the present dissertation, it can generally be surmised that naturalistic learning contexts (i.e. learning by immersion) provide not only more opportunities to use the language in question (better quality and quantity of input), but also put the learner in contact with the L2 culture (e.g. Hummel, 2014). For these reasons, learning by immersion in the target country is generally considered to be the ideal language learning setting (e.g. Sieloff Magnan and Lafford, 2012).

However, the age and context of L2 acquisition do not only influence the dimension of competence. How and in what setting the bilingual's languages are acquired, and the bilingual's childhood linguistic environment more generally, have a profound impact on the individual's identity, as well as emotional response and attitude towards his/her languages. For example, key considerations include whether the speech communities of both languages were present in the bilingual's linguistic environment in childhood (Hamers and Blanc, 2000), as well as whether the languages in question were acquired in the home, or in a formal, educational and emotionally neutral setting (e.g. Martinovic and Altarriba, 2013; Dewaele, 2010). The former is important because it shapes the individual's cultural identity, and the shared childhood cultural references with other members of that culture. The latter is significant because the context of SLA, as well as whether the L2 was acquired early or late in life, can play a key role in the way an individual's emotions are represented, with the native language often used to code childhood experiences, which shape future thoughts and feelings (Martinovic and Altarriba, 2013: 294). In support of this, studies have shown that bilinguals tend to prefer to answer questions in their native language when discussing emotional topics, and they also tend to demonstrate greater affect when speaking in their native language, (ibid). In particular, Dewaele found that bilinguals typically experience swear/taboo words with greater force in their L1, and tend to perceive the phrase I love you as having more emotional weight in their L1 compared with other languages (2004 and 2008 respectively). However, other variables were also found to be important in both studies, such as the age and context of L2 acquisition, (self-perceived) language dominance and L2 proficiency (ibid). For example, in the 2004 study, Dewaele observed that those who acquired the L2 in a formal instructed setting gave lower ratings on emotional force to swear/taboo words in the L2 compared with those who acquired the language in question in a naturalistic setting. The emotional force of such words in the L1 was also found to weaken when the L1 was no longer the dominant language

of the speaker, whereas the scores on emotional force for L2 were higher among participants who acquired the L2 at a younger age, or were more proficient in the language. These findings suggest that, while the childhood linguistic environment plays a key role in shaping bilinguals' emotional response to their languages, this relationship between language and emotions continues to evolve over the lifespan, under the influence of a variety of contextual factors.

What is more, research on bilingual memory has shown that aspects of bilinguals' experience seem to be language-specific in terms of how these experiences are retrieved (Javier, 2007: 34). Further, if the events in question were emotionally distressing, it appears that this emotional response is also associated with the language in which this situation was experienced by the individual. For example, Javier (2007) discusses several such cases, including the case of a patient who refused to speak her native language, Spanish, choosing instead to speak English and adopt an identity that reflects her idea of a successful woman in an English-speaking community. In her mind, Spanish was associated with negative events in her past, when she was emotionally neglected and abandoned by her mother (*ibid*). This highlights the strong link between language, culture and identity. Along similar lines, Javier (2007) mentions cases of adopted children who report not to remember their native language, or anything of their experience associated with that language. Such cases show that the context in which a language was acquired – in particular, if acquisition took place in early life – forms profound emotional associations with the language in question. This, in turn, has an indirect impact on proficiency in the languages, as negative associations with one language can result in its subsequent disuse, as well as on the bilingual's choice of identity.

## (2) Competence in the two languages

Attention has already been drawn to the fact that defining and measuring competence in a language is highly problematic, even in the simpler case of monolingual speakers (see Section 2.1.1 above). Bilingualism, of course, involves two linguistic (and pragmatic) competences and the relationship between them, introducing further levels of complexity. The problematic topic of measuring bilingualism, and in what sense this is possible, is discussed in Section 2.4 of this Chapter. The aim of this Section is to introduce some general ideas and terminology, and emphasise the importance of taking contextual factors into account when approaching the issue of bilingual competence.

The idea of bilingual competence can be approached from two related perspectives. First of all, in terms of the relationship between the two languages, i.e. which of the bilingual's languages is stronger and which is weaker, or if they can be taken to be more or less equal. Historically, researchers have been interested in classifying this relationship between the two languages in terms of language 'balance' and 'dominance' (e.g. Hamers and Blanc, 2000: 27). As pointed out by Hamers and Blanc, a balance between the two languages does not imply a high level of proficiency

in the languages in question – merely that the level of proficiency attained is more or less equal in both (*ibid*). Secondly, bilingual competence can be approached from the point of view of degree of bilingualism, i.e. how proficient the bilingual is in relation to other bilinguals. As argued in Section 2.4 below, the notion of degree of bilingualism is entirely relative and, in particular, where the bar is set for 'high proficiency' is often arbitrary (e.g. Goto Butler, 2013: 115).

Both relative competence and degree of bilingualism only make sense in context, when it is taken into account what the bilingual uses his/her languages for, and in what setting. For example, relative competence in the languages is highly domain-specific, and it has been found that bilinguals can be balanced in some domains, but not in others, which is known as 'domain specificity' (e.g. Grosjean, 2010; 1989). Similarly, a bilingual who only uses one of his/her languages in informal contexts may be considered highly proficient in informal social settings, but not necessarily in settings where a more formal register is required. Thus, bilingual competence is entirely relative, depending not only on which aspects of competence are taken into account (e.g. pragmatic, linguistic, grammar, vocabulary), but also in terms of the domains of language use considered. What is more, since bilingual proficiency is shaped by the dynamic factors of domains and context of language use, proficiency in each language and the balance/dominance relationship between them also fluctuate over time (see Section 2.4 below; also, Chapter 3).

## (3) The state of the language system

This refers to whether each of the bilingual's languages is being acquired, maintained or forgotten. A combination of these things is usually taking place at any given moment in time, depending on the influence of the bilingual's external and internal environment (see (5) and (6) below for details). For example, a bilingual who has moved to a country where the dominant linguistic environment is his/her L2 and contact with the L1 is limited is likely to be experiencing both L2 acquisition and L1 attrition, at least with respect to some aspects of each language system.

A key related concept is that of the 'stability' of the language system, in other words, whether the system has reached a (temporary) state of equilibrium, or if it is undergoing drastic change and restructuring. This idea of changes of state over time, and why it is conceptually important for the study of bilingualism, is covered in Chapter 3. However, to illustrate what is meant, a good example to consider is a learner's interlanguage (IL). Here, the IL is defined as "a linguistic system situated at some point between the native language and the target language" (Schrauf, 2013: 111). While the language in question is being actively acquired, the IL can generally be considered a very unstable system, as reflected in the high degree of variability in the learner's performance (e.g. Larsen-Freeman, 1997). On the other hand, it has been observed that a learner's IL does not continue developing indefinitely and, at some stage, a plateau is reached, known in SLA as the

phenomenon of 'fossilization' (Selinker, 1972). This concept was originally defined by Selinker, based on his observation that the majority of second language learners do not achieve "native-like" competence in their second language, as follows: "fossilization, a mechanism which ... underlies surface linguistic material which speakers will tend to keep in their IL productive performance, no matter what the age of the learner or the amount of instruction he receives in the TL" (*ibid*: 229). Fossilization in SLA is characterised as a "permanent lack of mastery of a target language", despite a favourable learning environment and adequate motivation to improve (Han and Singleton, 2004: 4; see also Han, 2012 for an overview of the historical perspective and summary of core issues). While the permanence of this cessation of learning has not been conclusively demonstrated (Ortega, 2013: 135) and, indeed, it is argued in Chapter 3 of the present dissertation that there are no permanent states in language development, a fossilized IL is certainly very resistant to change, and as such is an example of a highly stable state reached by the language system (e.g. Larsen-Freeman, 1997).

## (4) Domains and context of language use

By virtue of the definition of bilingualism as the use of two (or more) languages, what the bilingual uses his/her languages for, with whom and in what setting defines bilingualism and shapes its development over the lifespan. We have already seen how the domains of language use impact relative competence in each language, resulting in domain specificity. In (5) below, it is explained how external contextual factors, such as the bilingual's linguistic and cultural environment, affect various aspects of the bilingual language system. To avoid repetition, this section is used to draw attention to several important aspects of language use which are not covered expressly in section (5) on contextual factors.

First of all, the frequency of language use is a key consideration, due to what is known as the activation threshold hypothesis (e.g. Paradis, 2004). According to the activation threshold hypothesis, bilinguals' ability to retrieve both lexical items and grammatical structures is affected by how frequently the particular item is used (activated) in each language. The 'activation threshold' of an item is therefore considered to be a function of frequency: the less often an item is called upon, the more resources are subsequently required to retrieve it (*ibid*: 28). It has been shown that disuse of a language system tends to affect lexical items in the first instance, but over time grammatical knowledge is also affected (e.g. Köpke, 2007; Paradis, 2004). For this reason, frequency of language is considered to be an important predictor of L1 attrition (e.g. Paradis, 2007; Schmid, 2007), where attrition is defined as the "non-pathological decrease in proficiency in a language that had previously been acquired by an individual" (Schmid and Köpke, 2004). However, as explained below, language use itself is a complex factor, and the type of language

contact – not just frequency – needs to be taken into account when looking at the effect of this factor on bilingual language development.

Another important aspect of the context of bilingual language use is the idea of the bilingual's 'language mode' proposed by Grosjean (2008). Grosjean argues that language mode is a continuum, ranging from monolingual mode, in which bilingual speakers communicate in one of their languages only, avoiding conscious transfer of elements from the other language, to bilingual mode, in which elements of both languages are used and bilinguals are free to 'code-switch' between languages and blend aspects of the two languages (ibid). Code-switching is defined here as the alternate use of two languages by a bilingual, i.e. when a speaker changes to another language for a word or phrase, and then reverts back to the original language (Grosjean, 2010: 51-2). Between these two ends of the continuum various intermediary points exist, depending on such factors as the interlocutor, context, subject of discourse and function of the interaction (Grosjean, 2010). The idea of language mode refers to the various degrees of activation of the two languages, from both being activated (bilingual mode) and one being activated to a greater extent than the other (monolingual mode) (Grosjean, 2008). This idea of activation/suppression of the languages is referred to again in Section 2.3.1. However, the key point made here is that whether the bilingual tends to use a language in bilingual mode or monolingual mode is likely to have an impact on some aspects of competence in the language in question. For example, if a bilingual usually uses a language with other bilinguals with the same language combination in an informal setting, s/he can borrow words from the other language if they are more readily available. Over time, this may lead to slower lexical retrieval or reduced vocabulary size in the language, as predicted by the activation threshold hypothesis (see also Grosjean and Py, 1991; Schmid, 2007 for discussion). On the other hand, bilinguals who have to use a language in the monolingual mode will need to search for lexical equivalents in that language in order to communicate, resulting in better lexical retrieval in the language in question. Furthermore, who the bilingual uses his/her languages with does not only impact proficiency or language maintenance, but may have an effect on the bilingual's attitude and emotional response to the languages in question. For example, in his study exploring the emotional weight of the phrase I love you in bilinguals' different languages, Dewaele (2008) found that the nature of the network of interlocutors in the L2, as well as frequency of use, had a significant effect on the emotional weight attributed to the phrase by participants.

Finally, it needs to be taken into account whether the bilingual has an active or passive command of the languages in question, and what the bilingual is able to do in each language. Not only is this an important classification of bilingualism in itself (e.g. receptive/productive bilingualism; see, for example, Goto Butler, 2013), but the level of education attained in each language and, in particular, literacy plays a key role in determining how the language system will behave in certain contexts. For example, in L1 attrition research, literacy has been identified as an important predictor of L1

attrition, as being literate in the L1 provides more opportunities to use the language, and may have a positive impact on motivation to maintain it (e.g. Cherciov, 2011, Köpke, 2007).

# (5) The bilingual's external environment

As has already been mentioned in the examples discussed above, external contextual factors – linguistic, cultural, social - influence the development of the bilingual language system, in terms of both acquisition and attrition. The importance of contextual factors in both SLA and attrition research is becoming more and more apparent, and interest in these contextual factors and their influence on language development is increasing (e.g. Larsen-Freeman, 2012; de Bot, 2007; see also Chapter 3). Several examples of the effect of contextual factors on the development of bilingualism over time have been discussed in this Chapter. This Section addresses some the ways in which changes in the bilingual's external environment shape language development over time, either directly (e.g. by affecting exposure to one or both of the languages) or indirectly (e.g. by influencing one of the internal factors, such as motivation or attitude towards one of the languages; see (6) below on the bilingual's internal environment). Given the interconnectedness of internal and external contextual factors, it is impossible to keep the two discussions entirely separate. For this reason, aspects of the bilingual's internal environment will also be mentioned here, but from the point of view of how they have been shaped by the external environment. Additionally, to avoid repetition, some examples of the interplay between environmental and personal factors are not mentioned here, but instead discussed in (6) below.

What must be borne in mind from the outset is the complexity of individual patterns of external contextual factors that influence language development. Firstly, the external environment can be split into three major interrelated components: the bilingual's linguistic, cultural and social environment. Some of the effects of external contextual factors on language development are direct. For example, the dominant linguistic environment of the bilingual's country of residence directly affects access and exposure to the bilingual's languages. Individual pragmatic capacity is shaped by cultural and social pragmatic conventions (e.g. Paradis, 2004: 18), and hence the sociocultural environment exerts a direct influence on pragmatic competence in the languages. Other external contextual factors may have an indirect effect on language development. For example, social attitudes to a language can have a significant impact on the degree of attainment of that language by individuals, due to their effect on individuals' motivation to acquire the language in question (e.g. Paradis, 2004: 26-7; see (6) below for details). Furthermore, the factors of time and age need to be taken into account when considering the bilingual's external environment, in other words, how long the bilingual has spent in a given linguistic environment, and at what age s/he was first exposed to this environment. These considerations are especially important in studies of L1-speaking migrants living in their L2 environment, where the length of residence (LOR) and

age of arrival (AOA) need to be taken into account. In particular, a recurrent finding in the literature is that migrants' proficiency in the L1 is strongly influenced by whether their arrival in the new linguistic environment took place before or after the onset of puberty (e.g. Schmid, Köpke and de Bot, 2013).

What is more, external contextual factors that shape the bilingual's linguistic and sociocultural context cannot be reduced only to the language(s) and culture(s) of the bilingual's country of residence at a given time, although these are important considerations. We have already seen that frequency and context of language use are key in shaping the development of bilingualism over time. For this reason, any changes in the external environment that have an impact on the frequency of use of the bilingual's languages have to be taken into account as part of the context. As pointed out by de Bot, specific language-related major life events are relevant for the study of language development over time (2007: 57). These events may be unequivocally important, such as migration, but may also appear fairly minor on the larger scale of life, yet may be highly significant for the development of the language system, such as having a pen friend abroad, attending a bilingual kindergarten or going on an international school exchange (ibid). These events may also influence language development by both increasing frequency of use, as well as by affecting internal attitudinal factors. For example, Paradis refers to anecdotal evidence, which supports the "strong facilitating effect of learning a language for a significant other" (2004: 27). The situation is complicated further by the relationship between language and memory, and emotion (see Javier, 2007 and (1) above). Thus, some events that could be significant for the development of bilingualism may not be directly language-related in the sense that they do not have an immediate impact on the frequency or context of language use at all. However, the events may be highly emotional and experienced in a particular language, possibly forming negative or positive associations with the language, which may have an effect on subsequent language use or disuse. In particular, the childhood linguistic environment and associated memories may strongly influence the bilingual's attitude to the languages in question, and subsequent desire to maintain a language or cultural identity (see (1) above; see also Schmid, Köpke and de Bot, 2013).

The foregoing illustrates the complexity of external contextual factors that may have an impact – either directly or indirectly – on the development of bilingualism over time. This complexity, combined with the influence of other factors, such as biographies (e.g. childhood linguistic environment, age of acquisition; see, for example, Birdsong, 2014 on the effect of age of acquisition on language dominance patterns over the lifespan) and internal factors (e.g. attitudes, motivation, identity – see (6) below) means that the predictive power of external environmental factors taken in isolation is somewhat limited. Even factors that could reasonably be expected to exert a strong influence on language development, such as the length of residence in the L2 environment, or amount of contact with the L1 in emigration contexts, have not been found to be

conclusive predictors of L1 attrition and maintenance (see, for example, Schmid, 2007 for overview of findings on the role of L1 contact in L1 attrition; Cherciov, 2011 for impact of LOR on L1 attrition). This having been said, there is some evidence to suggest that the dominant linguistic environment of the bilingual's country of residence exerts such a strong influence on language development, that its effects manifest themselves despite attempts to counteract them. For example, in a study of L1 attrition among Russian-English bilinguals – graduate students teaching Russian as a foreign language at a US university - Isurin (2007) found that changes to the L1 occurred at the lexical and grammatical levels, despite extensive exposure to the L1 and high levels of professional motivation to maintain the L1 intact. Additionally, Laufer (2003) found that LOR in the L2 environment had a strong negative influence on both retention of L1 collocational knowledge and lexical diversity in the L1. These effects were observed despite a good initial level of L1 knowledge and continued (daily) contact with the L1. The age of arrival in the L2 country was not found to play a significant role (ibid). Thus, while the effect of external factors on the development of bilingualism is highly individual and context-specific, it can tentatively be expected that, at a global level, the factor of dominant linguistic environment of the country of residence, as well as whether the change in linguistic environment occurred before or after puberty, are likely to have the greatest influence, even in the presence of mitigating factors.

Finally, it is worth mentioning that, in emigration contexts, the effect of the culture of the country of residence and, in particular, the factor of length of residence in the country in question, on individual identity is also highly variable and not always in line with intuitive expectations. For example, it is reasonable to suppose that the longer an L1-speaking migrant spends living in his/her L2 country, the more s/he will feel part of the L2 culture, perhaps identifying to a greater extent with the L2 culture over time. However, findings of studies investigating, among other factors, the effect of LOR in the L2 environment on identity are often contradictory, highlighting the individual and unpredictable nature of the relationship between contextual factors and bilingual identity. For example, as previously mentioned, Dewaele (2008) found that length of residence in the L2speaking country was found to increase the emotional weight of the phrase I love you in the L2 as perceived by bilingual migrants. Dewaele argues that the increased emotional weight assigned to the phrase in the L2 could be seen as a conceptual shift towards the L2 for this particular emotional script, and this shift occurred only after a long period of socialisation in the L2 (ibid). On the other hand, in a study of first language attrition in German migrants living in the Netherlands, Prescher (2007) found that, as the duration of immigration increased, the more the migrants tried to return to their original identity and language. Also, as time went on, the participants became "more confident about their origin, culture and mother tongue and more critical about the culture, language and mentality of the guest country" (ibid: 201). Taken together, the results of the two studies demonstrate that the development of bilingual identity, just like language development, is not easily predicted on the basis of isolated factors.

## (6) The bilingual's internal environment

A number of internal contextual factors have been identified as important in terms of their impact on bilingual language development over time, with respect to language acquisition, maintenance and attrition. The factors that are considered key include cognitive ability (intelligence, language aptitude), educational level and attitudinal factors (motivation, attitude towards the languages and cultures in question, identity) (see, for example, Köpke, 2007; Schmid and Dusseldorp, 2010, for extralinguistic factors considered important for language attrition and maintenance; Larsen-Freeman and Cameron, 2008; Herdina and Jessner, 2002 for bilingual language development more generally). The factor of education has already been discussed briefly, in terms of its indirect impact on language development over the lifespan (see (4) above). Language aptitude has been studied extensively in SLA (see Skehan, 2012 for historical context and current perspective), but has received less attention in research on attrition and language maintenance. Indeed, while aptitude is considered an important predictor of foreign language attainment, its impact on language maintenance and attrition remains unclear. For example, Cherciov (2011) argues that greater language aptitude will lead not only to higher foreign language proficiency, but also prevent first language attrition. On the other hand, she notes that such a prediction contradicts other theories, such as the 'trade-off hypothesis', which predicts a greater degree of L1 attrition among highly proficient L2 speakers (Cherciov, 2011; see also Köpke, 2007).

Attitudinal factors have received a great deal of attention in SLA research, and it is generally recognised that these sets of factors are among the most important predictors of foreign language attainment (e.g. Ben-Rafael and Schmid, 2007). In recent years, there has been a growing interest in the role of attitudinal factors in L1 attrition and language maintenance, and it has been found that such factors are strong predictors of language attrition and maintenance in emigration contexts (e.g. Cherciov, 2013; Ben-Rafael and Schmid, 2007). Of these attitudinal factors, which can roughly be divided into the categories of motivation (to acquire or maintain a language), attitude towards the languages/linguistic communities, and identity (cultural, linguistic, social), it is motivation that is generally considered the most important, and which has received the most attention in SLA (see, for example, Ushioda and Dörnyei, 2012, for historical context and overview of current issues). Indeed, as pointed out by Paradis, it is the motor that drives the learning and use of both the first and second language; without some motivation, no speech event would take place (2004: 223). As argued by Herdina and Jessner (2002) (see also Chapter 3) in their Dynamic Model of Multilingualism (DMM), language maintenance requires a degree of conscious effort. Thus, there needs to be some motivation present in order to expend this effort. Indeed, the impact of other attitudinal factors on language learning and maintenance could perhaps be down to their influence on motivation. For example, a positive attitude towards a language or linguistic community increases motivation to learn or maintain the language in question, and seek out contact with

members of the group. Attitudinal factors are not only interconnected, but, as previously mentioned, are shaped by the external environment. Paradis (2004) cites some powerful examples of the role of motivation (and counter-motivation) in foreign language learning, and how this is influenced by attitudes, which in turn are shaped by social and political factors. For instance, Paradis (2004) describes the case of English-speaking students of French in Montreal in the late 1950s and 1960s. Despite learning the language since Grade 1 (age 6), by Grade 11 (age 16) the vast majority of students could hardly understand a word of French, and certainly not a full sentence in a normal conversation. On the other hand, it was noticed that students who arrived from Poland or Pakistan in Grade 7 with no prior knowledge of French were doing much better than their classmates, who had studied French since Grade 1. The difference was down to how the language was perceived by the two groups, with the English-speaking residents of Montreal perceiving French as a low-status language, whereas French was considered a prestigious language by the immigrants. This had an impact on their motivation to acquire the language (*ibid*: 26-7). Another case (originally reported by Kecskés and Papp, 2002; referred to in Paradis, 2007) describes the situation of students of Russian as a foreign language in Hungary during the time of Soviet occupation. It was found that these students had a significantly poorer command of Russian compared than those who studied English as a foreign language had of English, even though the students of Russian had been exposed to their L2 for longer than the students of English had been to theirs (ibid: 123). These examples illustrate how sociocultural and socioeconomic, or political factors influence individual attitudes towards a language/linguistic community, which in turn has a strong effect on motivation to acquire the language (or not).

Finally, while motivation is considered to be the best predictor of success in the acquisition of a foreign language (e.g. Paradis, 2007: 123), and attitudinal factors are recognised as strong predictors of L1 attrition in emigration contexts (e.g. Cherciov, 2013), in terms of predicting the effect these factors may have on the development of bilingualism, the context needs to be taken into account. In other words, it is not enough to think of motivation only in terms of degree; the reasons behind this motivation also play an important role. As a reflection of this complexity and the importance of context, in the sphere of SLA different kinds of motivation have been identified, such as instrumental motivation, which is the desire to learn a language for practical advantages (e.g. professional), and integrative motivation - the desire to acquire a language because of a genuine interest in the people and culture associated with the language (classifications proposed by Gardner and Lambert, 1972; see Paradis, 2004: 27; Ushioda and Dörnyei, 2012: 396-7). Integrative motivation has been found to be particularly important in foreign language acquisition (e.g. Paradis, 2004). However, these distinctions are not only important in terms of the degree to which they propel an individual to acquire/maintain a language. We have already seen that who the language(s) are used with, and in what 'mode' (see (4) above), shapes competence in the language(s). Thus, with respect to motivation, the purpose of learning/maintaining the language is

key to understanding the potential effect of this factor. For example, professional demands may require that a language be maintained in its 'pure' form, avoiding code-switching, borrowing and other forms of language mixing, such as in the case of spoken language interpreters, or language teachers. On the other hand, if a bilingual's motivation is to be part of a linguistic community, maintaining the 'purity' of the language may not be such an important consideration. For instance, in emigration contexts, a bilingual may feel a strong attachment towards the L1 and want to maintain ties with the language and community. However, s/he may maintain contact with the bilingual L1 community in the L2 country, using the L1 in bilingual mode. Furthermore, the bilingual's identity may also have an impact on his/her desire, or lack thereof, to assimilate with the native speakers of the country, and the language may be used to assert his/her identity – either as an L1 speaker, or L1/L2 bilingual. For example, in a study of South Korean bilinguals of English and Korean, Choi (2015) found that participants did not want to acquire native-like competence in English, preferring instead to assert their bilingual identity through their use of the language. This demonstrates how the bilingual's sense of identity, and the image s/he wishes to project, influences language use, as well as motivation to acquire the L2 to a native-like standard.

#### 2.3 The bilingual person

The foregoing has served to illustrate that bilingualism is a complex phenomenon, which is shaped over the lifespan by sets of interacting linguistic and extralinguistic factors. Predicting the effect of a single factor on the development of individual bilingualism is difficult, because of the principles of individual variation (see Chapter 3 for details) and the contributions of the other significant factors. What is more, as argued below, it is not just the different external and internal environmental factors that interact and influence each other: the bilingual's languages and cultures are also not separate. The implication of this is that bilingual competence is a unique phenomenon, which is not reducible to two separate linguistic competences, or two separate cultures. This idea is particularly important when it comes to assessing bilingual proficiency, as explained in Section 2.4.

# 2.3.1 Bilinguals as unique speaker-hearers

The view of the bilingual person adopted in the present dissertation is in line with that proposed by Grosjean, that the bilingual cannot be reduced to the sum of two complete or incomplete monolinguals, but rather is an integrated whole, with a unique and specific linguistic configuration (e.g. Grosjean, 1989). Similar views have been expressed by other researchers, such as Baetens Beardsmore, who notes the "specificity of bilingual behaviour" (1986: 86), and Cook, who defines the concept of "multi-competence" (e.g. Cook, 2003). The first argument for adopting such a position stems from the function of the bilingual's languages. As previously mentioned, the bilingual person usually needs and uses the two languages for different purposes, in different

domains of life and with different interlocutors. The level of competence developed in each language is therefore likely be different and highly domain-specific. Thus, as pointed out by Grosjean (1989), the bilingual's communicative competence cannot be assessed by considering only one of the languages, but must be studied through the bilingual's total language repertoire, reflecting the use of the languages in everyday life.

The second argument is based on the observation that the bilingual's languages are not separate systems: they interact and influence each other, thus the bilingual's language system is inherently different to that of a monolingual speaker. Evidence for this comes from studies of cross-linguistic activation in bilinguals. In other words, if we take the example of the concept of language mode (see (4) above), even if a bilingual is functioning in monolingual mode, s/he does not "become" a monolingual. This is because the other language can never be deactivated fully, which results in the observable bilingual phenomenon of interference. There are numerous types and sub-categories of interference/transfer (see, for example, Grosjean, 2012). Throughout the present thesis, interference is taken to mean the involuntary influence of one language on the other (after Grosjean, 1982; Hoffmann, 1991), to distinguish it from contact phenomena such as code-switching, borrowing etc., which are assumed to be more conscious. An overview of the literature of cross-linguistic influence and activation in bilinguals is given by Goto Butler (2013), providing evidence in support of the interconnectedness of the bilingual language system, but several examples are discussed below to illustrate the key points. In particular, the bilingual mental lexicon and bilingual language processing have received much attention, and several models have been proposed, such as the BIA/BIA+ (see Djikstra and van Heuven, 2002), the Revised Hierarchical Model (see Kroll et al., 2010 for a critical overview; also, Brysbaert & Duyck, 2010) and the Three-Store Hypothesis (Paradis, 2004). While there is currently no model that can account for all aspects of bilingual language processing, research so far has produced overwhelming evidence in favour of an integrated bilingual lexicon and non-selective access (see, for example, Djikstra and van Heuven, 2002; Brysbaert and Duyck, 2010 for an overview; also Meuter, 2009 for neurolinguistic evidence). There is also evidence for shared bilingual syntactic representations, as shown by structural priming tasks (e.g. Kantola and van Gompel, 2011). Further, studies suggest that interference occurs regardless of proficiency (e.g. Degani, Prior and Tokowicz, 2011; Sunderman and Kroll, 2006; see also Cook, 2003 for a selection of studies on the influence of the L2 on the L1). Additionally, evidence from fMRI studies suggests that the bilingual brain cannot avoid language conflict, even when the task requires only target language knowledge (van Heuven, Schriefers, Dijkstra and Hagoort, 2008). Interestingly, it has also been shown that language activation in bilinguals is affected by extra-linguistic factors. For example, a recent study conducted by Molnar et al. (2015), appears to support the view that bilingual language processing is highly contextspecific. The study focused on a single factor, interlocutor identity, and it was found that interlocutor identity affects language activation in bilinguals.

Furthermore, it has been demonstrated that the two languages influence each other not only at the level of lexis, grammar and phonology, but also in terms of pragmatic competence, and it is generally recognised that L1 pragmatics affects L2 pragmatics (see, for example, Bardovi-Harlig, 2012). Recent studies have also found that this influence does not only go in one direction, but both of the bilingual's languages influence each other. For example, a study conducted by Su (2010) examining the requesting behaviour of Chinese learners of English as a foreign language showed that the participants' requesting behaviour was affected in their two languages, indicating that bi-directional transfer can occur at the level of pragmatics in foreign language learners. In addition, it has been shown that L1 speech perception, and not just speech production, can be affected by the presence of an L2. For instance, a study of sentence parsing among fluent Spanish-English bilinguals showed a difference in parsing preferences between the bilingual group and the monolingual control (Dussias, 2001). Further, Dussias and Sagarra (2007) found that linguistic environment is a significant factor affecting speech perception, demonstrating that L2 exposure influences sentence parsing in Spanish-English bilinguals.

Taken together, the foregoing implies that the very presence of a second language has altered the language system as a whole. This cross-linguistic influence occurs across all levels of language (lexical, syntactic, phonological, pragmatic) and affects both speech production and perception. Bilinguals are not able to deactivate either one of their languages, even when the task demands the use of one language only (i.e. when they are operating in monolingual mode). Thus, the nature of the bilingual language system is essentially different to the monolingual one, and the presence of this involuntary bidirectional influence means that bilingual performance will always differ from monolingual performance, regardless of the level of proficiency achieved (see also Herdina and Jessner, 2002; Chapter 3).

That said, it is important to note that the above does not imply that the bilingual language system is qualitatively different from that of a monolingual in the sense that there is any kind of physical change, or additional 'mechanism' that is present in bilinguals, but absent in monolinguals. This argument is presented concisely by Runnqvist et al. who, having reviewed the evidence, conclude that there are no qualitative differences between bi- and monolinguals, in particular, there is no "system dedicated to language control" (2013: 244) that is unique to bilinguals. Along similar lines, Paradis argues that there is no need for any kind of "input/output switch" that enables bilinguals to switch between languages (2004: 206). Indeed, Paradis (2004) contends that any bilingual process has a monolingual equivalent, and thus does not require any kind of separate mechanism. For example, switching between languages can be likened to switching between different registers in monolinguals, or the skill of translation is akin to paraphrasing. However, despite this absence of qualitative differences, it cannot be said that a bilingual is 'two monolinguals in one person',

because the presence of a second language has affected the bilingual's performance in, and perception of, both languages, as evidenced by the studies cited above. As argued below, the same can be said for the bilingual's two (or more) cultures.

### 2.3.2 Biculturalism and bilingualism

A concept that is closely related to bilingualism is that of biculturalism, although one does not necessarily imply the other (e.g. Grosjean, 2015). As in the case of bilingualism, there is no single definition of a bicultural person and, also similarly to bilingualism, existing definitions tend to focus either on cultural knowledge/competence or use (interacting in two or more cultures) (see Grosjean, 2015 for an overview). For reasons that are analogous to those cited in the case of bilingualism (see Section 2.1.3 above; also, Grosjean, 2015), biculturalism here shall be defined as a function of use, or interaction in two (or more) cultures. After Grosjean, bicultural individuals are taken to be those who are characterised by the following three traits: "1. They take part, to varying degrees, in the life of two or more culture. 2. They adapt, at least in part, their attitudes, behaviours, values, languages etc. to these cultures. 3. They combine and blend aspects of the cultures involved" (2008: 214).

Defined in this way, the concept of biculturalism has several implications, most of which have equivalents in bilingualism. In particular, biculturalism is a dynamic phenomenon, which changes over the lifespan, shaped by life events and personal choices (e.g. moving country, being part of different social networks, changing jobs etc.) A bicultural individual need not feel equally a part of both cultures, or have equivalent cultural knowledge of each – one of the cultures can be dominant, and this relationship between the different cultures can fluctuate over time. Along similar lines to the bilingual's language modes, Grosjean (2008) points out that bicultural individuals also adapt to the specific situation and context they find themselves in, such as the interlocutor or country. However, this does not mean that a bicultural person can function as a 'monocultural', because it is impossible to deactivate all traits of the other culture, even when in a monocultural environment (ibid: 215). Thus, bicultural individuals are not 'two monoculturals in one person'. Further evidence for this claim comes from studies looking at the effect of biculturalism - and, more generally, multiculturalism - on personality and identity. For example, in a study examining the impact of multiculturalism/multilingualism on personality among Third Culture Kids<sup>2</sup> (TCKs) living in London, Dewaele and van Oudenhoven (2009) found evidence to suggest that personality is shaped by cultural and social factors. In particular, the group of TCKs were found to score higher

-

<sup>&</sup>lt;sup>2</sup> The definition adopted here is that proposed by Pollock and Van Reken (2009: 13): "A Third Culture Kid (TCK) is a person who has spent a significant part of his or her developmental years outside the parents' culture. The TCK frequently builds relationships to all of the cultures, while not having full ownership in any. Although elements from each culture may be assimilated into the TCK's life experience, the sense of belonging is in relationship to others of similar background."

on the dimensions of open-mindedness and cultural empathy, but lower on emotional stability, compared to the control.

Finally, a key aspect of biculturalism that needs to be considered is identity, in particular, the conflict that can sometimes arise between self-ascribed identity and acceptance by others. This is similar to the discussion of native speaker identity (see Section 2.1.1 above), where self-ascribed identity and the identity ascribed to an individual by others do not necessarily match. Furthermore, because two cultures are involved, there is a double categorisation - by members of both cultural communities. These categorisations can sometimes be contradictory, for example, an individual can be categorised as a member of culture A by members of culture B, and as a member of culture B by members of culture A (Grosjean, 2015). Grosjean also points out that such categorisations are often absolute, that is, cultures do not tend to readily accept that a person can be part of their culture as well as another (ibid). What is more, these categorisations are not based solely on linguistic competence or cultural knowledge, but also on factors such as physical appearance or attitudes (Grosjean, 2008: 2015). This is again similar to the idea of native speaker identity, which is often determined by others on the basis of extralinguistic factors, such as what a native speaker of a given language should look like (see Section 2.1.1). In other words, some factors on which acceptance is based are entirely outside the individual's control. It can therefore be difficult for the individual to reach a decision regarding their cultural identity, and accepting one's biculturalism can be "a long and arduous process" for some (Grosjean, 2015: 583). On the other hand, some individuals embrace their multicultural and multilingual identity, and in fact emphasise it, for example through language use (see, for example, Choi, 2015, discussed above). In particular, research on identity among TCKs, and indeed the very definition of this population proposed by Pollock and Van Reken (2009), often concludes that this population views multiculturalism and multilingualism as their identity (e.g. Tannenbaum and Tseng, 2015). This idea of belonging, acceptance/rejection and cultural identity, both in terms of the individual's own sense of identity and the identity attributed to the individual by others, is a key consideration in the next section on degree of bilingualism.

# 2.4 Degree of bilingualism and advanced bilingual proficiency

It has been argued that bilinguals form a highly heterogeneous group, and bilingualism itself is a complex and multidimensional phenomenon, encompassing linguistic, socio-cultural and psychological dimensions. This is what makes bilingual competence so difficult to define and measure. On the one hand, there are degrees of language proficiency, which can be considered to lie on a continuum between zero proficiency and "native-like" attainment (Schrauf, 2013). Thus, if we accept that there are different degrees of bilingual competence, we must also accept that bilingualism is, in some sense at least, measurable (Hoffmann, 1991: 22). On the other hand, the

multidimensional and context-dependent nature of bilingualism means that conceptualising and measuring bilingual language proficiency (and language proficiency more generally) is, and remains, one of the central issues in SLA (see Daller, 2011; Goto Butler, 2013: 116-122 for summary of core issues). What does it mean to say that someone is "more bilingual" than someone else? Clearly, there is a difference between, for example, someone who has just started learning a second language, and a person who uses two languages professionally on a daily basis, such as a spoken language interpreter (see Chapter 4). But how one might define and measure this difference in proficiency is entirely relative, depending on the dimensions taken into consideration (e.g. linguistic, pragmatic, cultural, identity etc.) Furthermore, we have seen that bilingual competence cannot be reduced to dual (complete or incomplete) monolingualism, which further complicates attempts to measure bilingualism. The aim of this final section is to, first of all, give a brief overview of how the problem of measuring bilingualism has been approached in the past, noting the shortcomings of some of these approaches. Against this background, focus will shift to the specific population of interest in the present project, that of highly proficient bilinguals, and it will be explained how this group is defined in the context of the present dissertation.

# 2.4.1 Measuring bilingualism

Historically, there have been various approaches and tests that have been designed for the purpose of measuring bilingualism. For example, Hamers and Blanc (2000) provide a succinct overview of the different measures of bilingualism used. These can be roughly divided into the following types: (1) tests of competence in a second language (i.e. foreign language testing); (2) tests of competence in the native language; (3) tests designed to compare the two languages, i.e. tests aimed at measuring language dominance or balance; (4) language biographies and self-evaluation; (5) tests of bilingual specificity (i.e. bilingual phenomena such as code-switching and transfer; bilingual production); (6) measures of affective correlates of bilingualism (i.e. the bilingual's attitude to the two languages) and cultural identity. Tests vary greatly in terms of construction, with some measuring only one factor, such as size of vocabulary or grammar, whereas others focus on how the language is used (e.g. verbal description tasks); some language dominance tests have also focused on measuring reaction speeds and comparing these in the two languages.

Of the above-mentioned, tests of competence are particularly problematic, as they tend to over-simplify bilingual competence, reducing it both in terms of the dimensions measured and domains taken into consideration (e.g. Hamers and Blanc, 2000). As previously discussed, neglecting to take into account the domains and context of language use can lead to incorrect interpretation of results, such as in the case of tests of bilingual balance/dominance. Another issue with tests of competence in a language is that many tests take monolingual standards as the norm against which bilingualism is measured. This is known in SLA is the "monolingual bias" (e.g. Goto Butler, 2013: 119).

However, comparisons with monolingual native controls can be useful in some contexts (e.g. Birdsong, 2005). Furthermore, testing can often be artificial, removing language from the context in which it is used, and as a consequence disregarding the social, cultural and pragmatic aspects of language use. That said, the various tests are useful tools in the right context. For example, testing specific aspects of what a language learner knows, such as grammar or vocabulary, and comparing these to a set standard is a useful approach in certain contexts. Measures of lexical richness, in particular appear to be useful tests of language dominance (see, for example, Treffers-Daller, 2011). When studying L1 attrition, for instance, it makes sense to take monolingual native language use as a benchmark against which to measure the extent to which the L1 has changed. What must be borne in mind is that no single test can measure bilingual proficiency in its entirety, only some set aspects of it. This means that bilingualism cannot be measured in a way that is absolute. This paradox - the existence of degrees of bilingualism, but no absolute way of distinguishing between these degrees - means that it is not at all evident how to distinguish between populations of bilinguals in terms of their relative proficiency, i.e. who can and cannot be considered a "highly proficient" bilingual. The way in which this concept is approached in the present dissertation is outlined below.

# 2.4.2 Highly proficient bilinguals

Given that there is no single approach to conceptualising and measuring the degree of bilingualism, and that the concept itself is relative, depending on which dimension(s) are considered relevant, it follows that where the bar is set for 'high' proficiency is also relative. For example, depending on the study, highly proficient or advanced bilinguals can be taken to be individuals who have achieved a certain score on a specific test, studied a language for a certain number of years, or completed a degree in their second language. High proficiency in the native language is often taken for granted in such cases. Therefore, like bilingualism itself, the concept of a population of highly proficient bilinguals only makes sense in context, and when accompanied by the specific bilingual profile of the group in question.

On the basis of the literature reviewed, it can be surmised that any definition or description of highly proficient bilinguals cannot be based on fixed standards of competence, as this is not definable in practice, and is entirely relative (see Section 2.1 above). Thus, highly proficient bilinguals are most appropriately characterised in terms of what they use their languages for. In particular, it is contended that there needs to be some reference to the broadness of the domains covered by each language, and what is considered sufficient for the purposes of the definition. For example, whether it is enough to use one of the languages regularly and fluently, but in the context of the home/family only, or other limited settings. A related consideration is whether the languages are used mostly in bilingual or monolingual mode, and if high proficiency implies, or should imply,

the ability to use the languages without code-switching if necessary (i.e. there needs to be sufficient vocabulary in each language to be able to function in monolingual mode, in a wide range of domains). Note that this does not suggest that bilingual speech must (or, indeed, can) be free from interference, or that bilinguals alternately function as monolinguals.

Furthermore, it seems reasonable to require some degree of external validation, of recognition by others that the individual is a member of this particular group – the group of highly proficient bilinguals. To illustrate this point, it is worth considering the "true bilingualism" definition proposed by Christopher Thiéry (1975). Originally developed in order to define a very specific group of highly proficient bilinguals - professional conference interpreters with two native languages. Thiéry defined a true bilingual as "someone who is taken to be one of themselves by the members of two different linguistic communities, at roughly the same social and cultural level" (1978: 146). This definition is discussed in detail in Chapter 4, but it is relevant to mention it here briefly as it addresses a similar kind of problem, that of defining a group of highly proficient bilinguals in a way that is operational, without reference to fixed standards of competence. The definition hinges on the identity ascribed to bilinguals by others, i.e. whether they are perceived as members of the linguistic community or not. This in itself can be problematic, for instance because, as pointed out earlier in the discussion on native speaker identity, judgements of identity are often made on the basis of non-linguistic factors such as appearance (see Section 2.1 above). Additionally, it is implicit in the definition that such high level bilinguals want to pass for native speakers in both of their languages, which may be the case in the specific context examined by Thiéry (see Chapter 4), but is not necessarily the case in general (e.g. Choi, 2015). While Thiéry's definition refers to a very specific group of "extreme" bilinguals, the idea of external validation without reference to competence benchmarks, or explicit comparisons with monolinguals, is important for characterising highly proficient bilinguals more generally. The other component of identity, which is not covered by the true bilingualism definition is self-ascribed identity. In the context of finding some way of distinguishing between highly proficient bilinguals and bilinguals in general, a key consideration is whether the bilinguals consider themselves highly proficient, or fluent, in their languages (i.e. identify as highly proficient bilinguals). This is considered relevant because it has generally been found that self-ratings of fluency tend to correspond to more objective, external ratings (e.g. Hamers and Blanc, 2000).

In light of the foregoing considerations and the literature reviewed earlier in this Chapter, the following characterisation of highly proficient bilinguals is proposed as compatible with the aims of the present dissertation. As indicated at the outset, the proposed characterisation not intended to be an all-encompassing or absolute definition of highly proficient bilinguals, but rather as an explanation of what the term is understood to mean within the context of the present study. Indeed, it describes a specific subgroup of highly proficient bilinguals – those who use their languages in a

professional or educational setting in addition to social/informal language use. The rationale for this restriction is as follows. First of all, as argued above, it is not possible to define high bilingual proficiency in an absolute way, and thus some additional constraints are needed. The condition of language use in a professional setting is a means of ensuring that the languages are used in a variety of contexts, and implies that a certain standard of proficiency is met, without reference to competence and, crucially, without imposing monolingual standards on bilinguals. Secondly, the group of bilinguals that forms the basis of the present inquiry – professional spoken language interpreters – satisfies this condition by definition.

Thus, within the context of the present thesis, a highly proficient bilingual is taken to be someone who:

- (i) uses two (or more) languages on a regular basis and in a variety of contexts, including in both informal (social or family) and formal (professional or educational) settings;
- (ii) is perceived by interlocutors as being fluent in the language(s) in question;
- (iii) identifies him/herself as a highly proficient bilingual, i.e. would describe him/herself as fluent in the two languages.

Criterion (i) is intended to imply that the bilingual is able to function in monolingual mode in both languages and in a variety of domains, without needing to borrow lexical items from the other language, as well as the ability to use both formal and informal registers as appropriate. Furthermore, the regular use in varied settings supposes a degree of cultural and pragmatic competence in both languages. Criterion (ii) is essentially a weaker version of the condition imposed by Thiéry (1975), and (iii) concerns self-ascribed identity. Professional spoken language interpreters, as a population, satisfy several additional conditions, and these are discussed in Chapter 4, where a specific bilingual profile for this group is drawn up. Before attention is turned to this specific group, however, additional theoretical considerations concerning individual bilingualism more generally need to be introduced.

### 3. Bilingualism as a Dynamic System

The term "dynamic systems theory" (DST) essentially refers to the study of the change of systems over time and has its origins in mathematics. Within the context of applied linguistics, DST refers to the view that language behaves like a complex (chaotic) dynamic system, in the sense that it exhibits the key properties of such a system, including continuous change over time and individual variation (see Section 3.3 below for details). It must be noted that there is a lack of terminological clarity in the literature on DST in applied linguistics, with a variety of terms, such as dynamic systems, complex (adaptive) systems, complexity theory, chaos, nonlinear systems, used almost interchangeably to refer to the same basic set of principles. In line with de Bot, the term DST will be used as a standard term taken to encompass this entire set of notions (de Bot, 2008). As we shall see, in the field of applied linguistics, DST refers not to a single theory, but rather to "a set of tools and approaches" (de Bot et. al., 2007: 167). Thus, adopting a DST view of bilingualism, or language development more generally, does not mean that one is rejecting all previous theories of language development, but instead accepting all theories that are compatible with DST principles, and viewing them as part of a coherent framework. The rationale for adopting such an approach and its advantages are discussed in Section 3.3 of this Chapter.

The main objective of this Chapter is to explain what DST means in the context of applied linguistics, and what its implications are for the study and description of individual bilingualism. To this end, the Chapter is structured as follows. First of all, the mathematical origins of DST are outlined, in order to explain where these principles have come from. Building on this, it is explained how these principles have been applied in the more general context of cognition and development, before focusing on DST in applied linguistics. The rationale for adopting a DST approach to the study of second language development is explained. This is followed by an outline of three key DST models of second language development and an overview of recent DST studies in this field, in order to demonstrate the varied applications of the theory. To conclude, the general principles of DST are applied to the specific aims of the present project. The view of individual bilingualism outlined in Chapter 2 is consolidated in light of DST principles, the concept of the bilingual profile introduced in Chapter 2 is extended on the basis of these principles and Van Geert's (1995) growth model in particular, and the methodological implications of adopting a DST approach are discussed.

#### 3.1 Dynamic systems theory (DST): mathematical origins

The term "dynamic system" as used in the cognitive and developmental sciences in fact refers to a complex dynamic system which exhibits chaotic behaviour, and these three related mathematical

concepts are outlined separately below. Their combined implications for the study of cognition and development are subsequently discussed in Section 3.2.

#### 3.1.1 Dynamical systems theory

Dynamical systems theory is a purely mathematical construct, which concerns the study of the temporal evolution of systems, or sets of interconnected components. A dynamical system essentially consists of three components: a state (or phase) space, which represents the set of all possible states of the system, a set of times, and a rule that describes the evolution of the system over time (e.g. Giunti, 1995). Put simply, a dynamical system is therefore a system in which states evolve from earlier ones according to a fixed law (Lorenz, 1993).

Dynamical systems can either be discrete or continuous, depending on whether time is taken to be discrete (e.g. the integers) or continuous (e.g. the reals). In mathematics, discrete dynamical systems are known as mappings and are modelled using difference equations, whereas continuous dynamical systems are technically called flows, and are modelled using differential equations. Actual physical examples of such systems include a swinging pendulum, a rolling rock or a wave (e.g. Lorenz, 1993).

Dynamical systems have their origins in Newtonian mechanics and the mathematics of differential equations, which have been used to model various physical systems. The difficulty is that most differential equations have no exact solutions, and it was not until 1881, when Henri Poincaré introduced the idea of the *qualitative* study of differential equations, that modern dynamical systems theory was born. Instead of seeking a formula for each solution, he proposed to study the set of all solutions all time and all initial conditions simultaneously (e.g. Norton, 1995). This signified an important change in viewpoint, an interest in the global behaviour of a system over long periods of time and the emergence of new themes, such as stability, recurrence of states and periodicity. Instead of seeking precise solutions, qualitative questions were being posed, including questions pertaining to the stability of the system and effects of perturbations (e.g. Norton, 1995). Certain dynamical systems were found to exhibit an additional property, resulting in so-called chaotic behaviour. Thus chaos theory was born.

# 3.1.2 Chaos theory

Chaos theory can also be traced back to Poincaré in the 1890s. Broadly speaking, a dynamical system is called chaotic if it exhibits *sensitive dependence on initial conditions* (e.g. Lorenz, 1993). Chaotic, however, does not mean random. Chaotic systems are, in fact, deterministic (i.e. their evolution is determined by a fixed rule), but they do not appear to be. In other words, in chaotic

systems, "the present state completely or almost completely determines the future, but does not appear to do so" (Lorenz, 1993: 8). This is due to their sensitive dependence on initial conditions. It is worth noting that "initial conditions" does not necessarily refer to the conditions that existed when the system was created - they can be, but can also be the conditions at the start of any stretch of time that is of interest to the researcher (Lorenz, 1993). An immediate consequence of this sensitive dependence is a limited ability to predict the future development of the system. An oftencited example of this is weather prediction. However, chaos does not imply that prediction is impossible, but it does mean a great degree of variation, and also that simple systems are capable of producing very complicated outputs (e.g. Ottino, 2003).

An important related concept, which is rather difficult to define in general terms, is that of an attractor. One way of thinking of attractors is as the set of "states of any system that ... occur again and again, or are approximated again and again, more and more closely" (Lorenz, 1993: 41). In other words, an attractor is a set of values towards which a system tends to evolve, for a wide variety of starting conditions. One frequently cited physical example is a swinging pendulum: each swing of the pendulum resembles the previous one, i.e. the pendulum does not suddenly swing wildly one moment and gently the next (Lorenz, 1993). Another is a marble rolling around in a bowl - in this case, the attractor is the fixed point corresponding to the marble's resting position at the bottom (Norton, 1995). Attractors are important because they represent the long-term states and evolution of the system. The emergence of such attractor states means that, despite the great degree of individual variation inherent in chaotic systems due to the sensitive dependence on initial conditions property, patterns also emerge in the course of development.

# 3.1.3 Complex systems

Complex systems theory has its roots in chaos theory, and complex systems are used to model processes in a diverse range of disciplines, from economics and meteorology to sociology and the cognitive sciences. Examples of complex systems include the human nervous system, a developing infant, the global economy and the internet. The applications of complex systems theory are therefore vast.

What exactly is a complex system? No single definition exists, however, the term is generally used to describe phenomena that share some common properties (Foote, 2007: 410). A rough descriptive definition of a complex system is as follows:

"A complex system is a system with a large number of elements, building blocks or agents, capable of exchanging stimuli with one another and with their environment. The interaction between elements may occur only with immediate neighbors or with distant ones; the agents can be all

identical or different; they may move in space or occupy fixed positions, and can be in one state or multiple states. The common characteristic of all complex systems is that they display organization without any external organizing principle being applied. In the most elaborate examples, the agents can learn from past history and modify their states accordingly. Adaptability and robustness are often the by-product. Part of the system may be altered, and the system may still be able to function." (Ottino, 2003: 293).

A more technical, but more or less equivalent definition is proposed by Foote (2007), who uses the term complex system to refer to phenomena, which exhibit the following characteristics: "(i) They are inherently complicated or intricate, in that they have factors such as the number of parameters affecting the system or the rules governing interactions of components of the system; (ii) they are rarely completely deterministic, and state parameters or measurement data may only be known in terms of probabilities; (iii) mathematical models of the system are usually complex and involve nonlinear, ill-posed, or chaotic behavior; and (iv) the systems are predisposed to unexpected outcomes (so-called "emergent behavior")." (Foote, 2007: 410).

Whichever definition is preferred, the implications are the same. A complex system cannot be understood by studying its components in isolation. "The very essence of the system lies in the interaction between parts and the overall behaviour that emerges from the interactions" (Ottino, 2003: 293). This emergent behaviour, i.e. behaviour that does not depend on the individual parts or agents, but on their interactions with one another, means that even systems made up of simple parts can behave in a complex way. Therefore, a reductionist approach cannot be applied to the study of complex systems. It is worth emphasising the distinction between the complex and the merely complicated. Many mechanical systems are highly complex due to the large number of parts involved. However, these systems are not complex, as they lack the ability to self-organise and adapt to solve problems (e.g. Ottino, 2003).

# 3.2 DST in the wider context of cognition and development

As previously stated, within the cognitive and developmental sciences, dynamic systems theory encompasses the three mathematical concepts of dynamical systems, chaos and complexity theory. In other words, in this context, the DST approach is based on the assumption that cognitive processes exhibit the broad properties of complex chaotic dynamic systems. Expressly, these are: continuous evolution over time, sensitive dependence on initial conditions (and hence nonlinearity and individual variation), complete interconnectedness of subsystems, self-organisation (resulting in adaptability and "emergent" behaviour), and the emergence of attractor states in the course of development (see, for example, de Bot, 2008; de Bot et al., 2013, 2007; Port and Van Gelder, 1995; and Thelen, 1995; Thelen and Smith, 1994 for an overview of the DST approach to cognition and

development). Therefore, it is argued that the DST framework is a biologically more appropriate approach for the description and exploration of cognitive systems, compared to the more traditional computational approaches.

DST emerged as an approach in the cognitive and developmental sciences a little over 20 years ago, with the first comprehensive overview of the dynamical approach to cognition, Mind as Motion, Port and Van Gelder (eds.), published in 1995. Since then, while it can be argued that DST is still in its infancy in this field (Spencer et al., 2011), its contributions and successes cannot be denied. For example, DST has revolutionised the field of motor development, largely due to the work of Thelen et al. (Spencer et al., 2011). It has had a significant impact on research in the field of cognitive development. Furthermore, aspects such as emotional, social and personality development, as well as creativity, are starting to be explored using DST principles (see Lewis, 2000; Spencer et al., 2011 and 2012; Howe and Lewis, 2005 for an overview of the major contributions of DST in the cognitive and developmental sciences; Gruber, 2012 for an example of DST principles applied to creativity; Lewis, 2005 for an example of the application of DST to emotion theory and neurobiology). Indeed, DST has recently been proposed as a general framework that can integrate the many different aspects of human development (Lewis, 2000).

Within the DST approach, cognition and development are modelled as unfolding in real time. They are perceived as processes that are "embodied" (in a physical body) and "embedded" (in context). The complete interconnectedness principle implies that it is not possible to separate mind from body or the individual from the environment, all must be viewed as continuously interacting subsystems of the whole system (see, for example, Thelen, 1995; Smith, 2005). Perception, action and cognition are not separated in a DST approach: they all form part of a single process (Thelen, 1995). This complete interconnectedness of interacting components, or subsystems, and continuous flow through time has led to the rejection of the notion of any kind of static representations or structures anywhere in the system. Instead, cognition and behaviour are thought of as being "assembled from multiple interacting components that can be freely combined from moment to moment on the basis of the context, task, and developmental history of the organism" (Spencer et al., 2012: 403). This is the concept of so-called "soft assembly", introduced by Thelen et al. (see Thelen, 1995; Thelen and Smith, 1994).

In this context, attractors as characterised as being the "preferred" states of a system. These can be more or less stable, depending on the strength of the attractor. Development is thus conceptualised as a series of shifts between periods of greater of lesser stability, with strong attractors characterising the more stable phases, and weak or changing attractors associated with the unstable phases (e.g. Howe and Lewis, 2005; Smith and Thelen, 2003: 344; Thelen, 1995). Selforganisation, that is, the ability of components of the system to organise themselves without any

guiding organising force, has several important implications for the conceptualisation of cognition and development. For instance, self-organisation implies multi-causality, i.e. that no single element has causal priority. (e.g. Smith, 2005; Smith and Thelen, 2003). This means that the brain is not seen as a "control" centre, guiding development. Instead, coherent behaviour emerges as a result of the interaction of the various components and the limitations and opportunities of the environment (Smith and Thelen, 2003). The self-organisation principle also means that there is no need to artificially build pattern into the system: dynamic systems are able to organise themselves around the preferred attractor states, thus forming cohesive pattern (see Spencer et al., 2012; Smith and Thelen, 2003). Finally, within the context of development, the sensitivity to initial conditions property means that even minor changes to one or several components of the system can have a significant effect on the system as a whole, leading to reorganisation and changes in behaviour (e.g. Smith and Thelen, 2003).

DST contrasts sharply with the computational models that have dominated the cognitive sciences since the 1950s, the fundamental assumption of which is that the human brain operates like a digital computer (see Port and Van Gelder, 1995 for a general historical perspective). In the 1980s, an alternative approach called connectionism, or "the modelling of cognitive processes using networks and neural units" (Port and Van Gelder, 1995: 3), was developed and started to gain popularity. However, while connectionism in a modified form is compatible with DST (see, for example, Elman, 2003, 1995; Thelen and Smith, 1994), it is essentially computational. With respect to the cognitive sciences, both connectionism and "traditional" AI are examples of what is called classic cognitivism (de Bruin and Kästner, 2012).

The need for an alternative approach was brought about by the fact that there are certain aspects of human cognition and development that computational models are unable to account for, due to their underlying assumptions. For example, within the computational approach, time is taken to be a discrete sequence of states, whereas cognitive processes unfold continuously in real time (Van Gelder and Port, 1995; Thelen, 1995). Moreover, the computational approach cannot account for the interaction of various timescales involved in development. Thelen and Smith (2003) argue that, since time is unified and coherent for the developing organism, any theory of development must concern itself with how timescales interact. DST, on the other hand, is well suited to modelling development in real time and is able to handle the interaction of nested timescales. In addition, the separation of mind from body and individual from environment, which is implicit in the computational approach, makes it impossible to study cognitive and developmental processes in context, and consider the impact of internal and external environmental factors (see, for example, Howe and Lewis, 2005). DST, by construction, views development as inseparable from the context in which it unfolds. Finally, a computational approach cannot explain how more complex systems emerge from simpler components, and it is also unable to account for the human ability to adapt

and solve problems. It is also unable to explain how development can both have globally similar outcomes, and yet display a high degree of variation at the individual level. This apparent paradox, however, can be explained from a DST perspective as being due to a combination of the self-organisation and sensitivity to initial conditions properties of the system (Smith and Thelen, 2003).

#### 3.3 A DST approach to second language development

Within a DST perspective, language development is not seen as a separate process, but as a subsystem of general cognitive development (de Bot, 2008). Therefore, much of what has been said about human development also applies to language in general, and bilingualism in particular. A key consequence of adopting a DST framework for applied linguistics is the fact that language is seen as an integrated system, and language development a unified process. In other words, language acquisition and attrition are not separate: they are reverse aspects of the same process. Therefore, in line with de Bot et. al. (2013), the term "second language development" (SLD) shall be used in the present thesis to refer to the evolution of the bilingual language system as a whole, including both growth (acquisition) and decline (attrition). While this is already implicit in some definitions of SLA (e.g. Long, 1993), there are other important theoretical implications of this seemingly simple substitution of terms. Crucially, the term SLD not only highlights the "bidirectionality of change", but also emphasizes "a shift from seeing language as a product or thing" (de Bot et. al., 2013: 199). A similar viewpoint is expressed by Larsen-Freeman (2015), who argues for the term SLA to be replaced by SLD.

The first formal attempt to apply the principles of DST to SLD can be found in Diane Larsen-Freeman's pioneering 1997 article entitled "Chaos/Complexity Science and Second Language Acquisition", in which she demonstrates that language, and hence SLD, displays the core properties of a (complex) dynamic system (Larsen-Freeman, 1997). Subsequent key theoretical advances in this area include the publication of Herdina and Jessner's book "A Dynamic Model of Multilingualism: Perspectives of Change in Psycholinguistics" in 2002, and Larsen-Freeman and Cameron's "Complex Systems and Applied Linguistics" (2008), both of which formalise the theoretical implications of the DST approach to SLD and put them in the context of existing theories. While the DST approach cannot as yet be considered mainstream in the field of SLD research, it does appear to be steadily gaining popularity. In particular, the "Five Graces Group", which includes Larsen-Freeman, has adopted a DST view of language and SLD in a position paper ("The Five Graces Group", 2009), and SLD researchers have started to apply the principles in empirical work (see section 3.3.3 below for details). This is in line with the more general trend in SLA research to pay greater heed to social and contextual factors (e.g. Goto Butler, 2013); indeed, the first part of the most recent edition of The Routledge Handbook of Second Language Acquisition (Gass and Mackey, eds., 2012) is entitled "Language in context", and includes a

chapter by Larsen-Freeman on complexity theory in SLA. The study of attrition from a DST perspective is also gaining ground, with de Bot being the first to formally apply the principles to the study of L1 attrition over the lifespan (see de Bot, 2007). Other applications soon followed and, most recently, a special issue of the *International Journal of Bilingualism* has been dedicated to the study of L1 attrition from a DST perspective (2013, Volume 17, Issue 6). More generally, in an attempt to unify the various applications and incarnations of DST in the field of applied linguistics, including those theories that have applied the principles implicitly, there have been calls to adopt DST as a comprehensive framework for SLD research (de Bot et al., 2013). These developments point to the fact that researchers are recognising that, in order to understand the complex processes involved in language development over time, they must be studied in context and viewed as continuously evolving over time.

# 3.3.1 Rationale for adopting a DST framework

The argument for adopting a DST approach to the study of SLD is essentially the same as that for cognition and development. Specifically, that SLD does indeed behave like a (complex) dynamic system, in the sense that it exhibits the core characteristics of such a system as outlined above, which implies that DST is the most appropriate framework for studying SLD. Secondly, in an analogous fashion to the more general field of cognition and development, there is no other theory in applied linguistics that is capable of describing change over time, under the influence of sets of interacting contextual factors. DST is also the only existing theory in SLD that is able to account for both individual variation and the emergence of pattern. So, while the application of DST in the field of applied linguistics is still fairly new, requires further development, and is "not necessarily a convenient research paradigm" (de Bot et al., 2013: 216), to reject DST principles is "to pretend that language *acquisition* is a product of linear development with a clear end state" (*ibid*), despite mounting evidence to the contrary. An overview of this evidence, i.e. that SLD is in line with DST principles, can be found in, for example, Larsen-Freeman (1997), de Bot et al. (2013), de Bot (2008). The main points are summarised below, to illustrate in what sense SLD displays the core properties of a dynamic system

Continuous temporal evolution is clear, and follows directly from the fact that language is a cognitive process, which grows and evolves with time, both synchronically and diachronically (Larsen-Freeman, 1997). It has also been shown that timescales interact and influence each other in the context of language development. In particular, it has been demonstrated that language use at larger time scales impacts language processing at the millisecond level (see de Bot et al., 2013). Language is complex in the sense that it is composed of a large number of different subsystems (e.g. phonology, morphology, lexicon, syntax, semantics, pragmatics etc.), and these subsystems are interdependent (e.g. Larsen-Freeman, 1997). It has been shown that the L1 and L2 are

interconnected and influence each other, as evidenced by phenomena such as interference/transfer, both at the level of speech production and perception (see Chapter 2). Moreover, the bilingual (or multilingual) person can be considered as a subsystem of the wider language community (e.g. Larsen-Freeman and Cameron, 2014), and thus second language development is an embedded process. It has been demonstrated that the bilingual language system is affected by numerous external (e.g. linguistic environment) and internal (e.g. motivation and attitude) factors, both in general and at the subsystems level (Larsen-Freeman, 1997; see also Chapter 2 for details).

Both language acquisition and attrition are subject to a great degree of **individual variation** across different subsystems. For example, Lowie and Verspoor have found the presence of individual variation in the acquisition order of morphemes (2015; see also Verspoor et al., 2008, for a general discussion of variability in SLD). The development of the bilingual language system is a **nonlinear** process, characterised by progress and backsliding (e.g. Larsen-Freeman, 1997). It is also **sensitively dependent on initial conditions**, as evidenced by the impact of the age of acquisition on aspects of L2 proficiency, or the influence of the learner's L1 (initial condition) on L2 acquisition. For instance, the pronunciation of a learner of English whose L1 is Spanish will differ to that of a learner whose L1 is Chinese (Larsen-Freeman, 1997). With respect to attrition, dramatic differences have been observed in the L1 competence of pre- and post-puberty migrants (see, Schmid et al. 2013, for an overview of evidence), i.e. the initial condition for this particular investigation is the age of migration.

Despite variation at the individual level, language learning and forgetting are not random processes. It has been shown that, at the global level, there are stages that learners generally go through (see, for example, Lenzing, 2015). Thus, **pattern emerges in the system around attractor states**. In this sense, the phenomenon of fossilisation is an example of such an attractor state (e.g. Larsen-Freeman, 1997). The system is also able to **self-organise and adapt**. In this regard, Larsen-Freeman gives the example of the acquisition of the past tense in English, in terms of both regular and irregular verbs (Larsen-Freeman, 1997). She notes that, initially, the interlanguage system exhibits chaotic behaviour with respect to the past tense ending, which then subsides. In other words, a restructuring of this aspect of the interlanguage has occurred, i.e. the system has organised itself independently (Larsen-Freeman, 1997).

As previously stated, adopting a DST approach to SLD does not mean adopting a single theory, but rather a set of principles, tools and approaches (de Bot et. al., 2008). Indeed, van Geert points out that "dynamic systems is basically a very general approach to describing and explaining change" (van Geert, 2008: 184). Therefore, if one accepts DST principles, one is also accepting those theories of language development that are compatible with these principles, whether they have been applied directly or implicitly. For example, Larsen-Freeman and Cameron outline the theories and

conceptual tools that are compatible with, and have served to inform, their complexity theory view, which include the DST approach to human development (Thelen and Smith's embodied/embedded theory of development), emergentist approaches to language (see also Ellis and Larsen-Freeman, 2006), connectionism (see Elman, 2003), usage-based and emergent theories of grammar, probabilistic linguistics, Vygotskian sociocultural theory and ecological approaches (Larsen-Freeman and Cameron, 2008). Similarly, de Bot et al. give an almost identical list of SLA theories which are compatible with DST principles (de Bot et al., 2013). In this way, DST can serve as a unifying framework for all such theories, which encompass different levels of granularity and different timescales (de Bot et al., 2013). To illustrate the flexibility of DST, the next section considers different applications of the framework, starting with the theoretical by outlining three key general dynamic models of SLD, before discussing several examples of specific practical applications.

#### 3.3.2 Dynamic models of second language development

The theoretical models surveyed below are distinct and exhibit some terminological differences, however, they share the same fundamental assumption: that language behaves like a complex dynamic system.

# (1) Modelling language development as growth (van Geert)

Van Geert's growth model is a highly influential dynamic model of language development (e.g. 1995). The importance of this model in terms of the DST approach to SLD has been emphasised, in particular, by de Bot et al. (see de Bot et al., 2007; de Bot, 2008) and has served to inform the Dynamic Model of Multilingualism (Herdina and Jessner, 2002), discussed below. Essentially, van Geert models development - both in general and in terms of language development specifically - as growth, taking the iterative growth equation as his starting point (van Geert, 1995; see also de Bot et al., 2007 for an overview). In this context, growth is conceived as a bidirectional process, one that is "concerned with the increase or decrease (negative increase) of one or more properties" (van Geert, 1995). The key principles of the growth model are that (1) there needs to be something that can grow, known as the "minimal structural growth condition" (van Geert, 1995: 314), (2) growth is dependent on resources and (3) these resources are limited and interlinked in a dynamical system (van Geert, 1995). In terms of language learning, van Geert distinguishes between external and internal resources, which are divided further into several general classes, namely: spatial, time, informational, motivational/energetic and material. With regard to internal resources, an example of a spatial resource is available mental capacity. Time in this context is taken to mean how much time a learner must spend on the task, and hence classed as an internal temporal resource, while informational resources relate to knowledge possessed by the learner. Motivational/energetic

resources refer to the individual's effort, interest and motivation to learn. Finally, internal material resources relate to the learner's physical body, for instance, whether the senses are working correctly (van Geert, 1995). The various classes of external resources include: the physical environment (spatial), the learner's linguistic environment (informational), positive or negative reinforcement given by the environment (energetic/motivational), availability of learning materials, or even basic things such as food and shelter (material) (van Geert, 1995). Van Geert terms this interlinked structure of resources the "cognitive ecosystem" (van Geert, 1995: 315). The resources can be compensatory, in the sense that a lack of one resource (e.g. time) can be compensated, to a certain extent, by more of another (e.g. higher motivation) (see de Bot, 2008). The interconnectedness of resources means that they interact and affect each other over time, and specific resources play a different role at the various developmental stages (de Bot, 2008). An important implication that stems from the fact that growth is resource-dependent and resources are limited is that of "carrying capacity", which is the idea that the final attainable growth level is also limited (Van Geert, 1995: 316). Since the growth equation is iterative, the present growth level is determined by the previous growth level, subject to the resources available at that point (see de Bot et al., 2008).

Growth equations are useful for modelling first and second language development because they allow for the simplification of what are otherwise very complex developmental processes. The model can be used in two ways: either to deductively infer precise hypotheses about development, subsequently testing these against available data (see van Geert, 1995 for examples of this kind of application of the growth model). Alternatively, if no such data is available, then the dynamical growth model can serve as a guideline for further inquiry and a source of theoretical questions (van Geert, 1995).

# (2) Complexity theory view of SLD (Larsen-Freeman and Cameron)

This approach is most closely related to Thelen and Smith's embedded/embodied theory of development (see above for discussion, and also Thelen and Smith, 1994) in the sense that no mathematical equations are used, but instead the key tenets of DST/complexity theory are applied in a qualitative way, to conceptualise SLD and thereby inform inquiry and methodology. Larsen-Freeman and Cameron's complexity theory approach is effectively a detailed analysis of the global and methodological implications that stem from viewing SLD as a complex dynamic system. At the heart of the approach is the focus on language *use* in context. Thus, it is argued that change, or processes and factors that result in change, should be the focus of research, as opposed to an attempt to look for static or unchanging entities (Larsen-Freeman and Cameron, 2008). Therefore, the aim is to study how the language system changes with time, what happens before and after relatively steady states, which factors lead to the destabilisation of the system. Consequently, "from

a dynamic systems point of view, what is studied are not single variables, but rather changes in systems" (Larsen-Freeman and Cameron, 2008: 122). These changes are conceptualised in terms of systems stability and attractors of varying strength (Larsen-Freeman and Cameron, 2008).

A further implication of viewing language as a complex system is that, as such, it cannot be understood by studying the various components in isolation. The impact of environmental factors, both internal and external, cannot be ignored, since "context is seen as an intrinsic part of a system, not as a background against which action takes place" (Larsen-Freeman and Cameron, 2008: 16). This interconnectedness applies not only to the subsystems and context-specific variables, but also to timescales. Therefore, Larsen-Freeman and Cameron warn against conflating timescales, but instead advise researchers to "seek linkages across levels and timescales" (Larsen-Freeman and Cameron, 2008: 242). In sum, the aim of the complexity theory approach is to offer a new way of thinking about language development and offer unique conceptual tools for further inquiry by providing a shift in focus and specific methodological recommendations (Larsen-Freeman and Cameron, 2008).

# (3) Dynamic Model of Multilingualism (DMM) (Herdina and Jessner)

Since bilingualism is the simplest form of multilingualism, much of the DMM is directly applicable to the study of bilingualism and SLD. The DMM is concerned with the development of the multilingual language system in terms of what drives development, how far the system will (or can) develop, what leads to stabilisation and perturbations within the system. Within the DMM, development is modelled as a function of interacting finite resources and "efforts". The fundamental assumption of the DMM is that overall resources are limited, and the language system competes for these resources (Herdina and Jessner, 2002). It is postulated that both language acquisition and language maintenance require effort, and that these efforts themselves are functions of the speaker's internal and external environment, i.e. the limited resources available (Herdina and Jessner, 2002). Taking this together, the overall degree of multilingual proficiency achieved by an individual is modelled as a function of general language effort (GLE), which is taken to be the sum of the language acquisition and language maintenance efforts (Herdina and Jessner, 2002). The GLE is divided up between languages of a multilingual speaker and thus the GLE itself is a finite resource (Herdina and Jessner, 2002).

Within the DMM, the driver of development is taken to be the "communicative needs" of the speaker (Herdina and Jessner, 2002: 135). These communicative needs "refer to all situations in which language is used as a means to express oneself or as a tool for thought" (Herdina and Jessner, 2002: 135). A further distinction is made between "effective" and "perceived" communicative needs, with the former defined by the objective "recurrent communicative".

requirements of the speaker" (Herdina and Jessner, 2002: 136), while the latter are subjective and include anticipation of possible other "states" where the (effective) communicative needs may be different (see Herdina and Jessner, 2002). This distinction between effective and perceived communicative needs carries important theoretical implications for the development of a speaker's language system. Specifically, while the effective (objective) communicative needs of the speaker certainly have a bearing on the perceived communicative needs, it is the perceived communicative needs that determine how much GLE a speaker is willing to expend, and thus how far the system will develop and where it will stabilise (Herdina and Jessner, 2002).

The theoretical constructs of GLE, which itself is a function of available resources, and communicative needs are applied in order to model and explain certain multilingual phenomena. For example, the domain specificity principle (see Grosjean, 2010 and also Chapter 2) is explained on the basis of the principle of economy of effort (Zipf's law) and the phenomenon of attractors, which are steady states assumed to require less effort to maintain (Herdina and Jessner, 2002). Since resources are limited and both language learning and maintenance require effort, it follows that the language system and its various subsystems will only be maintained at the minimum level required to meet the individual's perceived communicative needs. The development of domain-specific language skills result in a more stable system, and hence require less maintenance effort than a "fully-fledged all-purpose language system" (Herdina and Jessner, 2002: 105). This notion of stability is applied within the DMM to model other multilingual phenomena, such as, for instance, cross-linguistic interference and transfer. The DMM proposes that interference and transfer can be viewed as functions of the proximity (in terms of relative competence in the languages) and distance respectively between the languages involved. (Herdina and Jessner, 2002).

#### 3.3.3 Examples of DST studies of second language development

The DST-based SLD studies conducted so far demonstrate the broad applications of the DST approach, both in terms of the kinds of multilingual phenomena investigated and the methodologies used. A review of the literature suggests that these studies can be divided into the following broad categories, subject to a certain degree of overlap: (1) studies focusing on individual variation in SLD at different subsystems levels; (2) investigations of state changes in SLD over time, either at a global or subsystems level, and attempts to model aspects of SLD as dynamic processes; (3) studies looking at the interaction of sets of linguistic and extralinguistic factors that are considered relevant for SLD, or attempts to isolate a specific factor and examine its influence.

Examples of (1) include a case study looking at intra-individual variability with respect to changes in L2 systems at a time of rapid development (Verspoor et al., 2008) and an investigation of variation in the acquisition order of English morphemes (Lowie and Verspoor, 2015). In particular,

it was found that, while the acquisition order of morphemes does behave in a stage-like way at a global group level, there is significant degree of variation at the individual learner level. The authors conclude that, such "grand sweep" effects that may be generalizable to large populations of learners may not be meaningful if one is interested in the progress of an individual learner in a specific context (Lowie and Verspoor, 2015).

There are numerous examples of (2), the broadest category of the three as defined here. Studies looking at changes in the states of the system and emergence of pattern at a global level include an investigation of phase transitions in the development of writing fluency (Baba and Nitta, 2014) and an exploration of regularities in L2 development from a DST perspective (Lenzing, 2015). Examples of type (2) studies focusing on modelling specific aspects of SLD as a dynamic process include a study investigating dynamic patterns in the development of complexity and accuracy in the acquisition of Finnish (Spoelman and Verspoor, 2010). The researchers found that the interaction of different complexity measures considered changes over time. In particular, it was observed that as word complexity increases, so do noun phrase and sentence complexity, on the other hand, no such relationship was found between noun phrase and sentence complexity. The results showed no meaningful relationship between accuracy and complexity measures over time and a high degree of individual variation was observed within different subsystems of accuracy and complexity (Spoelman and Verspoor, 2010). Other studies of the second type have focused on modelling L1 attrition as a dynamic process, either in general (Opitz, 2013) or at a subsystems level (de Leeuw et al., 2012 and 2013; Ecke and Hall, 2012). Furthermore, there have been attempts to apply DST principles to the development of grammars and, in particular, examine the compatibility of DST with Universal Grammar (see Plaza Pust, 2008; de Bot et al., 2013 for a discussion on UG within the context of DST).

The most relevant DST studies for the present project are type (3), as they consider the relationships between factors in the context of the development of bilingualism over the lifespan. Of particular interest is a study conducted by Schmid and Dusseldorp, which examines the impact of groups of extralinguistic and sociolinguistic factors on L1 attrition, applying quantitative methods of analysis (Schmid and Dusseldorp, 2010). The specific factor groups considered were exposure to the L1, identification and affiliation with the L1 and attitude towards the L1. The data were collected from two attriting populations with German as their first language, one in a Dutch language context (n = 53) and one in an English linguistic environment (n = 53). The results were compared to those of a control group of Germans in Germany (n = 53). Despite the comparatively large size of the population in the context of SLD research, the predictive power of the factor groups considered was found to be relatively weak (Schmid and Dusseldorp, 2010). Other studies within category (3) have focused on the influence of a single factor in SLD. For example, Cherciov (2012) looked at the impact of attitude on migrants' L1 attrition and L2 proficiency. The

participants were 50 post-puberty Romanian migrants who have spent at least 10 years in Canada. The study used a combination of approaches: attitude was measured by means of questionnaires and interviews, and proficiency through a C-test and verbal fluency task. The results showed that the effect of attitude on L1 attrition is non-linear and prone to fluctuations (see also Cherciov, 2011 for an exploration of the effect of attitude on L1 attrition).

#### 3.4 A DST approach to bilingualism: implications for the present project

The final section of this Chapter aims to apply the DST theory discussed above in order to consolidate the view of individual bilingualism introduced in Chapter 2. In particular, the bilingual profile is discussed as a dynamic concept, as this will form the basis of the study design (see Chapter 5).

## 3.4.1 A dynamic view of bilingualism

Since DST in the field of applied linguistics is concerned with language use in context (Larsen-Freeman and Cameron, 2008), the approach is fully in line with the definition of bilingualism as language use proposed in Chapter 2. Furthermore, the approach supports the unique speaker-hearer view of the bilingual person; indeed, expressly postulates that "the multilingual system is not reducible to multiple monolingualism" (Herdina and Jessner, 2002: 19). This follows from the complete interconnectedness principle, whereby the bilingual's L1 and L2 are viewed as interconnected subsystems of the whole language system, which interact and influence each other. As a consequence, "the acquisition of several language systems results in a qualitative change in the speaker's psycholinguistic system, that is, as the whole psycholinguistic system adapts to meet new psychological and social requirements, it also changes in nature" (Herdina and Jessner, 2002: 92). The findings set out in Chapter 2, pointing to the interconnected nature of the bilingual lexicon and the inability of bilinguals to "switch off" one of their languages can be considered evidence in support of this DST principle. The complete interconnectedness principle also implies that there is no need for any additional mechanism within the bilingual language system that controls language selection (see Chapter 2 for details). The same principle applies to the cultural aspect of bilingualism, meaning that, from a DST perspective, a bicultural individual is a unique product of the two (or more) cultures to which s/he belongs, and not the sum of two monoculturals. In addition, the findings described in Chapter 2, which demonstrate that pragmatic aspects of language use influence linguistic aspects (e.g. sentence parsing, requesting behaviour) emphasise the fact that, within a DST approach, all subsystems are interconnected – linguistic, pragmatic etc.

At the heart of DST is the idea of continuous change over time. Applied to bilingualism, this means that the language system is in constant flux, and thus there can be no permanent states. This

continuous change is conceived as being bidirectional, encompassing both acquisition and attrition, and is influenced by external and internal environmental factors. This implies that there can be no permanent states in the system - states that appear permanent are conceptualised as steady states (attractors) of varying strength. In particular, fossilization is an example of such a steady state (e.g. Larsen-Freeman, 1997; see Chapter 2). In this way, DST does not impose a limit on development. At the same time, DST is able to account for the fact that the language system does not continue developing indefinitely, because language development is conceived as a function of limited available resources and the communicative needs of the individual (Herdina and Jessner, 2002; Van Geert, 1995; see above for details). Furthermore, DST postulates that language development is not forever restricted by the initial conditions (Larsen-Freeman and Cameron, 2008: 97), while being sensitively dependent on these conditions. Taking the example of the critical period for SLA (see Chapter 2 for details), DST is able to account for the seemingly conflicting findings that (1) there is a strong negative correlation between age of SLA and L2 attainment and (2) that some adult learners do appear to achieve native-like proficiency in their L2. These findings are also in line with the DST principle of individual variation, which implies that there can be no set "critical period" for SLA which would hold true for everyone.

Finally, the DST principles imply that bilingualism cannot be "measured" in a way that is absolute, but only in relative terms. It follows that high bilingual proficiency is a concept that is, in effect, arbitrary and depends on the criteria imposed by the researcher. The idea of relative competence in the two languages – and the notions of balance and language dominance – are continuously shifting states and, as postulated by the DMM, can be conceived in terms of the stability of the language system, and how close the two languages are two each other in terms of relative competence (proximity of the languages). We have already seen above that, according to the DMM, the phenomena of interference and transfer (involuntary interference, such as lexical borrowing due to insufficient vocabulary in one of the languages) can be viewed as a function of the proximity or distance (in terms of relative competence) of the languages in question. Specifically, "the closer the levels of both systems are to each other, the greater the degree of interference to be expected, as systems integrity is threatened by the degree of proximity. Transfer on the other hand is defined as a function of the distance between the two (or more) systems" (Herdina and Jessner, 2002: 133). This idea is closely related to the concept of systems stability in terms of looking at "the outcome of perturbing, or pushing, the system away from its stable behaviour" (Larsen Freeman and Cameron, 2008: 238). In other words, this measure of the stability of the language system considers how much input is required to move the system away from its current steady state: the more stable a system is, the greater input or degree of perturbation required to shift the system into a different behaviour (Larsen-Freeman and Cameron, 2008: 238-9). Applied to the above, this suggests that the closer the bilingual's languages are in terms of relative competence, the less stable the system is as a whole, meaning that less input - perhaps, simply reading more in one of the languages - is

required to alter the language dominance balance. It therefore also follows that, the closer a bilingual's languages are in terms of relative competence, the more interference is likely to be experienced by the individual, together with more frequent shifts in language dominance.

#### 3.4.2 The bilingual profile as a dynamic concept

The idea of the bilingual profile introduced in Chapter 2 is important for the present study because it provides a way of describing the bilingualism of individuals or groups on the basis of the main relevant dimensions of bilingualism. In other words, it represents a necessary simplification, which makes it possible to compare groups of bilinguals with reference to these dimensions – sets of internal and external contextual factors. It has already been mentioned in Chapter 2 that the bilingual profile is not a static concept, but rather represents a description at a given moment in time, and the various contextual factors that make up the description interact and influence each other (see Chapter 2, Section 2.2). DST, by construction, provides a useful framework for viewing the bilingual profile as a dynamic concept. This application of the DST principles is similar to van Geert's (1995) growth model of language development, which views language development as unfolding over time, as a function of external and internal resources (see Section 3.3.2 above for details). These resources in effect correspond to the components of the bilingual profile identified in Chapter 2. Therefore, while DST does not alter the theory introduced in the preceding Chapter in terms of the dimensions that are considered relevant for the description of bilingualism, the DST framework represents a conceptual shift and has important methodological implications.

An important consequence of the DST principles applied to the study of bilingualism in the way described above is that the interrelationships between the various components of the bilingual profile are complex and not easily predicted. In particular, we have seen that DST studies looking at relationships between sets of linguistic and extralinguistic factors within the context of the development of bilingualism over time (studies labelled as type (3) in Section 3.3.3 above) have generally not found strong links between factors within the populations investigated. This can be attributed to two things: firstly, the contribution of other factors, which may be highly variable within the populations considered and, secondly, the principle of individual variation. The former suggests that, in order to successfully study the interrelationships between sets of factors in SLD, careful choice of population is needed in so as to isolate the factors of interest as far as possible. This means that the population of bilinguals involved in such a study needs to be fairly homogenous with respect to at least a number of the other significant factors. While there can be no constants from a DST perspective, the aim is to establish which factors behave more or less like constants, and can be taken to be effectively invariable for a given population, at least for a specific time interval (Herdina and Jessner, 2002). Secondly, the principle of individual variation, in turn, suggests that a study aiming to find differences at a group level would need to be fairly large-scale,

otherwise global trends would be obscured by individual differences. These general principles are applied in the design of the present study, the formulation of hypotheses and subsequent interpretation of the results (see Chapters 5, 7 and 8).

# 5. Research Questions and Methodology

This Chapter sets out the research questions and hypotheses formulated on the basis of the review of the literature. The study design is explained and described, as are the participants involved in the study. The Chapter concludes with some further details on the technical aspects of the processing and analysis of the data obtained in the course of the study.

#### 5.1 Research questions and hypotheses

An overview of research at the intersection of bilingualism and interpreting set out in the preceding Chapter identified some important gaps in the literature, which include the following:

- (1) Very little is known about the "exceptional" bilingualism of interpreters and the development of this type of bilingualism; the limited work that has been carried out in this area has tended to focus on conference interpreters. In particular, no studies aiming to describe the bilingualism of interpreters have been carried out with the participation of PSI interpreters.
- (2) The only detailed inquiry seeking to describe the bilingualism of interpreters, including in terms of attitudes and perceptions as well as biographies, is that conducted by Thiéry (1975), which focused on AIIC AA interpreters only. The characteristics identified by Thiéry as pertaining exclusively to AIIC AA interpreters (defined by him as "true bilinguals") have not been investigated further; in particular, the question of whether the characteristics identified are indeed exclusive to AIIC AA interpreters, or are in fact typical of professional conference interpreters in general, remains unanswered. Other questions raised by the study, such as the reasons why some AIIC AA interpreters obtained their double A classification at a later date, and what the properties of this group are, have not subsequently been investigated further.
- (3) The potential of the AIIC register of conference interpreters with corresponding language classifications as a resource for the study of individual bilingualism remains unexplored since Thiéry's pioneering 1975 dissertation. It is argued here that the AIIC A and B language classifications can be viewed as either (i) ratings of native-like competence in a language according to external reviewers, and/or (ii) externally-ascribed native (mother tongue) or non-native speaker identity in a language. Possible avenues for further investigation therefore include both questions pertaining to ultimate attainment in a second language, as well as self-ascribed and externally-ascribed native speaker identity in a language, and the conditions that are associated with this.

### 5.1.1 Research questions

On the basis of the foregoing, the present study aims to describe three populations of professional interpreters as bilinguals, in terms of both biographical and attitudinal factors, in order to further understanding of the nature of this type of bilingualism and its development. The specific populations investigated are AIIC AA and AB conference interpreters, and PSIs on the UK National Register (NRPSI). The research questions addressed as part of this inquiry are set out below. The first question takes a broad view, aiming to identify which properties, if any, are common to all three groups of interpreters surveyed, and thus what characterises professional interpreters as a group of bilinguals. Questions two and three aim to investigate properties of the AIIC AA and AB language classifications and their wider implications for the study of individual bilingualism, exploring in more depth the notions of true bilingualism (Thiéry, 1975) and mother tongue identity in two languages. Question four compares the UK PSI and AIIC AB groups, taken to represent the wider populations of public service and conference interpreters, in order to identify the main ways in which these groups differ as bilinguals. The implications of these differences for the study of individual bilingualism more generally are considered. The specific research questions investigated are as follows:

- 1. Are there any characteristics that broadly define professional interpreters as a group of highly proficient bilinguals, irrespective of language classification (AA/AB) and domain (PSI/conference), in terms of:
  - biographical factors (including age of SLA, childhood linguistic environment, language(s) of education, current language use/exposure);
  - attitudes and perceptions (including language maintenance motivation, metalinguistic awareness (language maintenance effort, fluctuations in language dominance, crosslinguistic interference), emotional attachment towards their languages, self-ratings of proficiency in their languages)?
- 2. (a) (i) Do AIIC AA interpreters, defined as "true bilinguals" by Thiéry (1975), possess special characteristics that set them apart from other comparable populations of highly proficient bilinguals, as argued by Thiéry? Specifically, do AIIC AA interpreters differ from AIIC AB interpreters as bilinguals in terms of:
  - biographical factors (including age of SLA, childhood linguistic environment, language(s)
    of education, current language use/exposure);
  - attitudes and perceptions (including language maintenance motivation, metalinguistic awareness (language maintenance effort, fluctuations in language dominance, crosslinguistic interference), emotional attachment towards their languages, self-ratings of proficiency in their languages)?

- (ii) What do these findings imply for bilingualism more generally, in particular with respect to: ultimate attainment in a second language; the critical or sensitive period for second language acquisition; the predictions of the dynamic systems theory (DST) view of second language development over the lifespan?
- (b) Do AIIC AA interpreters identify themselves as "true bilinguals", and what is the relationship between Thiéry's (1975) definition and: (i) the AIIC AA classification; (ii) mother tongue identity in the two languages; (iii) "true bilingualism" in writing?
- (c) Is there a subgroup of AIIC AA interpreters who were awarded their double A classification at a later date (i.e. who were initially classed as AB upon joining AIIC)? If so, does this group differ from AIIC AA interpreters who obtained their AA classification upon joining in terms of biographical and/or attitudinal factors?
- 3. If the AIIC A and B language classifications are taken to be markers of externally-ascribed native and non-native identity in the respective languages, what does the description of the AIIC AA and AB groups imply in terms of the following questions:
- (a) Does externally-ascribed mother tongue identity in two languages generally correspond to self-ascribed identity?
- (b) Where this is not the case, are there any biographical or attitudinal factors that are associated with such anomalous groups?
- (c) Which factors appear to be the most important in terms of shaping self-ascribed mother tongue identity?
- 4. (a) If PSI and conference interpreters, as represented by the NRPSI and AIIC AB groups, are found to differ with respect to:
  - biographical factors (including age of SLA, childhood linguistic environment, language(s) of education, current language use/exposure);
  - attitudes and perceptions (including language maintenance motivation, metalinguistic awareness (language maintenance effort, fluctuations in language dominance, crosslinguistic interference), emotional attachment towards their languages, self-ratings of proficiency in their languages)?

what are the main differences between these groups as bilinguals?

(b) What are the implications of these findings for bilingualism more generally, in particular, in terms of the relationships between biographical and attitudinal factors?

#### 5.1.2 Hypotheses

The following hypotheses have been formulated on the basis of the literature surveyed in Chapters 2-4 of the dissertation.

# Question 1:

- With respect to biographical factors, it is anticipated that, on average, respondents from all three groups will have acquired both of their strongest working languages before puberty. This is based on the fact that all interpreters, by virtue of their profession, need to have a high level of proficiency in their active working languages, and the strong negative correlation between age of SLA and L2 attainment (see Chapter 2). That having been said, on the basis of the DST principle of individual variation, it is expected that there will be respondents with a higher age of SLA. It is further expected that, in view of the training and qualifications required for entry into the interpreting profession, a substantial proportion of respondents from all groups will have been exposed to both of their strongest languages during higher education, or have been exposed to the active language that is not their first language as part of education.
- As regards attitudes and perceptions, on the basis of the DMM (Herdina & Jessner, 2002) hypothesis that language maintenance requires effort, as well as Thiéry's (1975) finding that his respondents felt that maintaining their working languages required a degree of conscious effort, it is expected that the majority of interpreters from all groups surveyed will demonstrate awareness of a language maintenance effort. On the basis of the literature on bilingualism discussed in Chapter 2, according to which emotional factors and attitudes are strongly influenced by the childhood linguistic environment (in particular, this appears to be the case with regard to perceptions of the emotional force of words; e.g. Dewaele, 2004), it is predicted that attitudinal factors within all three groups will be affected by the linguistic environment in childhood. In particular, it is hypothesised that all respondents will demonstrate greater levels of attachment/emotional response to their first language, or the language most strongly associated with childhood, irrespective of other factors, such as subsequent language use and exposure. Finally, given the link between age and context of SLA and ultimate attainment (see Chapter 2), it is expected that self-ratings of proficiency/confidence in the two strongest languages will again be most strongly influenced by childhood linguistic environment factors. Specifically, it is expected that respondents from all groups will tend to rate their first language, or the dominant language in childhood, as the stronger of their active working languages. In particular, it is expected that this will be the case in terms of active use of the language, based on Paradis' (2004) argument that processing in the first language, acquired by immersion, is more automatic than that in the second language, resulting in more stable performance.

# Question 2:

<u>Part (a)</u> (i) If the AIIC AA group is defined as having two mother tongues in accordance with the AIIC language classifications, while the AB group is defined as having one mother tongue and one foreign language, then the following differences are expected between the AIIC AA and AB groups:

- The AA and AB groups are expected to differ in terms of childhood linguistic environment and average age of SLA. In particular, it is expected that the AA group will tend to have acquired their second language at an earlier age than the AB group, and that both languages will be represented more equally in childhood within the AA group compared with AB group. This hypothesis is based on the literature on bilingualism surveyed in Chapter 2 and Thiéry's (1975) description of the AIIC AA group. A similar pattern is expected in the languages of education for the two groups, with the AA group more likely to have been exposed to both languages than the AB group. At the same time, according to the DST principle of individual variation, individual differences are to be expected, even if the above-described trends are observed at a group level.
- It is expected that the AA interpreters will be characterised by an equal attachment/emotional response to their two strongest languages, while the AB group is expected to demonstrate a greater attachment/emotional response to language A over language B. This is based on the bilingualism literature surveyed in Chapter 2, according to which attitude/emotional response to the bilingual's languages is shaped, in particular, by the childhood linguistic environment and relative proficiency in the languages (e.g. Dewaele, 2004).
- With regard to awareness of language maintenance effort, it is proposed that this will be influenced, to some degree, by the age of acquisition and exposure to the language(s) in childhood. This is based on Paradis' (2004) argument that a language acquired earlier results in more automatic performance and hence a more stable language system. On the basis of the DMM, a more stable system requires less effort to maintain (see Chapter 3). Therefore, if the first hypothesis is correct and the age of acquisition among the AA group is indeed lower than that of the AB group, it can be expected that AA interpreters will tend to perceive language maintenance as a conscious effort to a lesser degree than AB interpreters.
- With respect to motivation to maintain the two languages, it is anticipated that the AA group will exhibit a higher level of non-professional motivation to maintain both languages at their present high standard than the AB group. This expectation is based on the earlier hypothesis that the AA group is likely to exhibit equal attachment towards both languages, and this may in turn influence the group's motivation to maintain both at a high standard, regardless of professional requirements (Chapter 2, Section 2.2 (6)).

- It is expected that a greater degree of awareness of fluctuations in language dominance will be observed among the AA group compared with the AB group, as a result of the closer proximity (in terms of relative competence) of the two languages of the AA group compared with the AB group. With regard to cross-linguistic interference, it is expected that this will be perceived as effectively bi-directional by AA respondents, whereas AB interpreters are more likely to report greater interference in the A→B direction. This prediction is again based on the argument that processing in the first language is more automatic, and therefore more stable/resistant to external influence than that in the second language (Paradis, 2004).
- According to the literature, self-evaluation scores of bilingual proficiency in their two
  languages are a reliable measure of actual relative competence in these languages (e.g.
  Hamers and Blanc, 2000). It is therefore expected that the self-reported proficiency scores
  for the two languages will be closer for the AA group than the AB group, with the latter
  tending to rate their A language as stronger than language B.
- (ii) If differences are observed between the AIIC AA and AB groups with regard to biographical and/or attitudinal factors, then the findings may have implications in terms of the conditions that are necessary and/or sufficient for the attainment of native-like competence in two language, or which factors appear to contribute the most to the attainment of such high levels of competence in a second language. A comparison of the findings for the AA and AB groups may also have implications in terms of a critical/sensitive period for second language acquisition, the duration of this period and whether there are exceptions. With regard to the predictions of the DST approach to individual bilingualism, the findings for the AA and AB groups may reveal, in particular, whether there is evidence of individual variation in the context of trends at a global level; evidence of relationships between external and internal (personal) contextual factors and the interaction of timescales, such as, for example, the influence of linguistic environment in childhood on attitude to the languages in question in later life.

Part (b) On the basis of Thiéry's (1975) findings, it is expected that the majority of AIIC AA interpreters will identify as true bilinguals, and that Thiéry's definition will be taken by most to correspond to the AIIC AA classification. At the same time, as noted in Chapter 4, the true bilingualism definition concerns external perceptions only, and does not address the subjective component of mother tongue identity in two languages. On this basis, it is contended that mother tongue identity in two languages and true bilingualism are not the same concept, in particular, as the term "mother tongue" has additional emotional connotations (see Chapter 2). While, according to Thiéry's (1975) definition, true bilingualism does not necessarily imply true bilingualism in writing, it is nonetheless reasonable to suppose that the latter will tend to accompany the former.

<u>Part (c)</u> On the basis of Thiéry's (1975) findings, it is expected that a subgroup of AA respondents will be identified who obtained their AA classification after the date of joining AIIC. Very little is known about this, however, the following hypotheses are proposed as possible reasons for and consequences of the later acquisition of the AIIC AA classification:

- A later average age of acquisition of the language that was initially classed as a B language, compared with AIIC AA interpreters who obtained the AA classification immediately upon joining AIIC. Also, it is expected that differences will be observed with regard to the relative exposure to the two languages in childhood among this group of respondents compared with those AA respondents who were immediately classed as AA upon joining AIIC. Specifically, it is anticipated that the language that was initially classed as a B will be less prominent in childhood compared with the language that was initially classed as an A among those respondents who obtained their AA classification at a later date. The AA interpreters who were immediately classed as such upon joining AIIC are expected to have had both languages represented more equally in childhood. In addition, it is expected that a lower proportion of respondents who were awarded their AA classification at a later date will have had both A languages as the languages of higher education, compared with the rest of the AA group.
- It is further hypothesised that the subgroup of AA interpreters who were awarded their AA classification at a later date may exhibit differences in terms of attitude towards their A languages, identity and confidence (self-ratings of proficiency relative to a monolingual standard), compared with the rest of the AA interpreters. In particular, it is anticipated that the AA subgroup who were initially classed as AB will report a greater degree of attachment/emotional response to the language that was immediately classed as an A, compared with the language that was initially classed as a B; that this subgroup of AA interpreters will be less likely than the rest of the AA group to identify as having two mother tongues, and less likely to identify as true bilinguals; finally, that they will tend to rate the language initially classed as an A as stronger than the language initially classed as a B. These results are expected to differ from those observed for the rest of the AA group, with the main group of AA interpreters (those immediately classed as AA upon joining), exhibiting more equal attachment to their languages, being more likely to rate their languages as being equal, compared with those AA respondents who were initially classed as AB.

# **Question 3:**

<u>Part (a)</u> On the basis of the literature surveyed in Chapter 2 on native speaker identity, it is expected that externally-ascribed and self-ascribed mother tongue identity in two languages will tend to match for the majority of respondents in both the AA and AB groups. However, as suggested by the literature, this is not always the case, and therefore it is anticipated that there will be subgroups of

respondents within both groups of AIIC interpreters whose self-ascribed identity does not match that which has been ascribed by others. In other words, it is predicted that there will be a minority of AA interpreters who do not feel that both of their A languages are mother tongues, and a minority of AB respondents who, on the contrary, do feel that both of their strongest working languages are mother tongues.

<u>Part (b)</u> The literature surveyed in Chapter 2 suggests that identity and attitude towards the languages are shaped, in particular, by the age of acquisition and childhood linguistic environment. On this basis, the following differences are expected among the subgroups of AA and AB interpreters whose self-ascribed identity does not match that which has been ascribed by others:

- It is expected that the subgroup of AA interpreters who do not feel that they have two mother tongues will be characterised by a higher average age of SLA and less equal exposure to both languages in childhood compared with the rest of the AIIC AA respondents. It is also hypothesised that this subgroup of AA interpreters will report a greater attachment to one of their languages over the other compared with those AA interpreters who do identify as having two mother tongues.
- As regards AB respondents who consider both languages to be mother tongues, it is anticipated that this group will have a lower average age of SLA and more equal exposure to both languages in childhood compared with the rest of the AB group. It is further expected that this subgroup of AB respondents will exhibit more equal attachment to both of their languages compared with the rest of the AB group who, in turn, are anticipated to report a greater attachment to language A over language B.

<u>Part (c)</u> On the basis of the previous hypotheses, it is expected that childhood linguistic environment factors (are of SLA and relative exposure to the languages in childhood) will be revealed as the most important in terms of shaping self-ascribed mother tongue identity in two languages.

#### **Question 4:**

As explained in Chapter 4, both PSI and conference interpreting require the ability use the active languages at a very high level, in a variety of contexts, as well as the ability to mediate across the corresponding cultures. Therefore, there are no grounds to suppose that significant differences will be observed between interpreters as bilinguals on the basis of the setting in which they work. However, it needs to be noted that, due to the nature of the PSI group as represented by interpreters on the UK National Register, it is expected that the dominant linguistic environment of this group will be the L2 (English in this case). No such information is available about the conference interpreters. According to the DST principle of interconnectedness, external contextual factors have a bearing on internal factors in second language development (see Chapter 3). Therefore,

differences in attitudinal factors between the PSI and conference groups may be observed due to the impact of the L2 as the dominant linguistic environment of the PSI group. If this is indeed found to be the case, it is possible that a comparison of the AB and PSI groups may reveal important information about the impact of the factor of dominant linguistic environment (language of the country of residence) on other aspects of individual bilingualism, in particular, on attitudinal factors and perceptions.

# 5.2 Design of the study

This section sets out the details of the study designed to address the research questions outlined above, and explains the theoretical frame work on which it is based. Further information is provided about the groups of interpreters approached to take part in the study, in particular, with respect to their working languages.

#### 5.2.1 Theoretical framework

The aim of the study is to describe three populations of professional interpreters as bilinguals in terms of biographical factors, attitudes and perceptions. The theoretical framework for the study is based on the literature surveyed in Chapters 2 - 4 of the dissertation. Chapter 2 identified the main dimensions of bilingualism, which are relevant for the description of individual bilingualism, and outlined some of the ways in which these dimensions interact and influence each other. The specific dimensions identified were: (1) age and context of language acquisition; (2) competence in the two languages; (3) state of the language system, that is, whether each language is being acquired, maintained or forgotten; (4) domains and context of language use; (5) the bilingual's external environment (linguistic, cultural, social); (6) the bilingual's internal environment (e.g. motivation, attitude towards the languages, identity, education, aptitude, personality) (see Chapter 2, Section 2.2). Chapter 3 consolidated the view of this bilingual profile as a dynamic system, the components of which interact and influence each other over time. The implications of this dynamic view of bilingualism include the idea that second language development is an embedded process, that is, it is viewed as unfolding in context with external environmental factors influencing internal ones and vice versa. Of the dynamic models of second language development described in Chapter 3, it is Van Geert's model of language development as growth (see Section 3.3.2 (1)) that is most relevant in the present case, as it conceptualises the development of bilingualism over time as a function of interacting external and internal resources. As such, the growth model view of bilingualism can be considered as a dynamic bilingual profile. Chapter 4 identified what is known about interpreters as bilinguals and what remains to be discovered, and the specific research questions set out earlier in this Chapter were formulated on this basis.

Taking the findings of the literature together with the specific characteristics of interpreters and the research questions formulated at the beginning of this Chapter, the populations are described as bilinguals in terms of the following sets of interacting factors: (1) mother tongue identity and linguistic situation in childhood<sup>1</sup>, including age of acquisition; (2) language(s) of education; (3) present linguistic environment and language use; (4) level of non-professional motivation to maintain the languages and awareness of language maintenance effort; (5) metalinguistic awareness (fluctuations in language dominance, cross-linguistic interference); (6) attachment and emotional response to the languages; (7) self-ratings of proficiency in the two strongest languages relative to each other; (8) self-ratings of proficiency in the two languages compared to a monolingual standard (this is intended to be a measure of confidence in the languages). Thus components (1) - (3) of the description correspond to the relevant biographical factors, whereas components (4) - (8) concern the respondents' attitudes and perceptions. The next section gives details of the data collection instrument used to obtain this information, as well as additional clarifications about the groups of professional interpreters and their working languages.

#### 5.2.2 Data collection instrument: large-scale questionnaire

In order to investigate the research questions set, the information outlined above needed to be obtained from the three populations of interpreters. This could be done by means of either a questionnaire or interviews with the participants. It was considered that a large-scale questionnaire is the most appropriate data collection instrument in the present case, for the following reasons. First of all, the present study aims to identify trends at group level and compare three populations of interpreters in terms of these trends. As indicated in Chapter 3, the principle of individual variation suggests that, in order to global trends to be observed, a large number of participants is needed. In addition, the populations to be described – NRPSI and AIIC AA and AB interpreters – are themselves large, with over 2,000 members registered with the UK NRPSI and over 3,000 members of AIIC (see Chapter 4). Thus, conducting interviews with participants on this scale is not deemed to be practically feasible, and a questionnaire-based study is more appropriate on these grounds. Furthermore, a questionnaire makes it possible to standardise responses, which can then be more readily compared in terms of general trends, whereas an interview format is more suited to obtaining detailed information from individuals. Given the aims of the project, a questionnaire format is more appropriate in the present context.

Here, childhood is taken to be approximately the period between 0 - 12 years, in other words, before the onset of puberty. This is in line with literature on the critical or sensitive period for language acquisition (see, for example, DeKeyser, 2013).

#### (1) Design and structure of the questionnaire

Three distinct questionnaires were produced for each group of interpreters (see Appendix I). The AIIC AB and PSI questionnaires were identical in terms of content, but differed in the wording of some of the questions, due to the fact the A and B language classifications apply to conference interpreters only. The AIIC AA questionnaire included several additional questions pertaining to the AIIC AA language classification, the true bilingualism definition and the notion of mother tongue competence in childhood in the two languages (Q5-9 in Section 2; Q9-10 in Section 3; Q10 in Section 5). The term "mother tongue" was preserved in all questionnaires, as opposed to the more neutral term "native language", for the following reasons. Firstly, this is the wording used by the AIIC in its definitions of the A and B language classifications (see Chapter 4). Secondly, the additional emotional connotations implicit in the term "mother tongue" (see Chapter 2) are invoked intentionally, in order to explore notions of identity and emotional attachment to the languages in question.

All three questionnaires were divided into the same themed sections: (1) personal details; (2) bilingualism (working languages and mother tongues); (3) linguistic background; (4) present linguistic and professional situation; (5) attitudes and perceptions, to give the questionnaires a logical structure (Dörnyei & Taguchi, 2010). The intended aims of sections 1 and 2 were to establish some general points of reference, such as age and working languages (additionally, to check that only two strongest working languages were entered). Section 2 also aimed to explore the relationship between working languages and mother tongue competence, as well as assist with subsequent data interpretation and coding. Section 3 was designed to obtain information about respondents' linguistic environment in childhood and their languages of education. Section 4 was intended to establish the relative degree of language use and exposure at present, both in a professional context and otherwise. Additionally, section 4 contains questions pertaining to language maintenance effort and respondents' motivation to maintain their two languages (Q10-14/AA, Q11-15/AB/PSI). Section 5 was given the general heading of "attitudes and perceptions", but the questions were divided into several distinct themes within the section. The first two questions addressed awareness of fluctuations in language dominance; Q5.4-6 were intended to investigate attachment/emotional response to the languages; Q5.7-9 concern relative self-ratings of proficiency in the two languages, with Q5.11-13/AA, Q5.10-12/AB/PSI investigating self-ratings of proficiency relative to a monolingual standard, intended to be a measure of confidence in the languages; Q5.14/AA, Q5.13/AB/PSI pertains to awareness of cross-linguistic interference between the two strongest languages.

The questionnaires were designed in such a way as to ensure anonymity: respondents were not asked to give their names or other biographical details that would make identification possible,

unless they chose to do so at the end of the questionnaire. The questionnaires were designed to be sufficiently detailed so as to provide enough information to address the research questions set, but short enough for respondents to complete in their entirety. Thus, there had to be a degree of compromise between detail and brevity; for example, with respect to childhood linguistic environment, it was not possible to enter into the level of detail covered by Thiéry in his survey of AIIC AA interpreters. As pointed out in Chapter 4, Thiéry was able to obtain a high response rate from his participants as they were his colleagues, and it is likely that he would have known at least some of them personally. Thus, in the present study, the childhood linguistic environment is simplified to its key components of languages of the home, street (country) and (primary) school (Thiéry, 1975). Similarly, other sections of the questionnaire were subject to a necessary degree of simplification, so as to cover the information required for the comparison of the populations without compromising response rate.

#### (2) Administration of the questionnaire

The three questionnaires were set up online using the online survey software SurveyMonkey. Contact details of AIIC AA and AB interpreters were obtained from the AIIC website, and the public service interpreters' details were obtained from the NRPSI website. All of this information was freely available to the public at the time. A database of interpreters to contact was created using this information.

An initial pilot version of the questionnaire was tested on a small sample of the NRPSI interpreters (135 pilot questionnaires were sent by email). The response rate for the pilot survey was just over 8% (11 completed questionnaires received). The results were analysed and feedback obtained from several respondents with regard to (1) the clarity of the wording of the questions and (2) the length of the questionnaire. On the basis of the feedback and responses received, the wording of several of the questions was altered to dispel any ambiguity, and some questions were omitted as they were not found to be relevant.

The final version of the online questionnaire was sent to each interpreter in each group individually, by means of embedding the link to the online survey in a covering letter. Three different covering letters were used for the three groups of interpreters, to highlight the most relevant aspects of the research project for that particular group. In total, 3094 questionnaires were sent to the three groups of interpreters (200 AA, 1326 AB, 1568 PSI). Responses were then collected via the SurveyMonkey site.

### 5.2.3 Participants: AIIC AA, AB and UK PSI interpreters

In accordance with the aims and research questions, the populations of interest in the present study are AIIC AA and AB conference interpreters and UK-based public service interpreters registered with the NRPSI. The following additional selection criteria were imposed on the prospective respondents to ensure methodological integrity.

- (1) Of the AIIC conference interpreters, only those with English as an A or B language were considered. This is because the questionnaire and all correspondence was in English, and it was important that all elements of the questionnaire were understood as intended.
- (2) As stated in Chapter 4, interpreters are often multilingual. The present project is concerned with bilingualism as defined in Chapter 2. Thus, while respondents may have additional weaker or passive working languages, there are two distinct languages that can be considered to be the strongest languages. Therefore, in order to minimise the influence of any other languages known by the respondents, only those AB respondents with one B language were selected to take part in the project (of the AA respondents, all had only two A languages). Of the PSI group, only those with two distinct strongest working languages were approached.

Therefore, throughout this and the subsequent chapters of the present thesis, the term "AIIC AB interpreter" (or "AB interpreter") is used to refer to those AIIC AB interpreters who have one "A" and one "B" language only, according to the AIIC language classifications, unless otherwise specified. The term "public service interpreter" (PSI) is used to refer only to those public service interpreters who are (1) registered with the NRPSI and (2) have two languages which they identify as their strongest working languages, unless otherwise specified.

## 5.3 Data processing and analysis

This section gives further information on how the raw data obtained from respondents were subsequently processed and analysed.

## 5.3.1 Initial data processing and software used

Once all responses were received via the SurveyMonkey portal, the online surveys were closed and results collected. The responses for each group were downloaded from SurveyMonkey in Excel format. The spreadsheets with raw data were formatted for ease of interpretation and divided into the corresponding questionnaire sections. An initial data clean-up was conducted by checking individual responses and deleting any incomplete and/or ambiguous responses.

The raw responses received were language-specific, as respondents were asked to state their working languages and, in the case of the AIIC interpreters, indicate the corresponding language classifications. It was only in section 5 of the questionnaire that respondents were asked to give abstract labels to each of their two strongest languages (Q5.3/AA/AB/PSI). The next stage in the initial data processing was therefore to assign labels to respondents' two strongest working languages, and re-code the data where necessary to ensure that these labels were consistent with those in Q5.3. Thus, for the AA group, the two A languages were labelled A1 and A2; A and B for the AB group; L1 and L2 for the PSI group. The remaining questions were then re-interpreted and coded in terms of these language labels. For example, where respondents were asked to state their languages of education (Q3.8/AA/AB/PSI), the responses were entered in terms of the language labels, i.e. for the AA group, for instance, the choices were "A1", "A2", "both A1 and A2", or "other". Similarly for the AB and PSI groups. In the case of respondents' country of residence (Q1.2/AA/AB/PSI) and country of birth (Q1.3/AA/AB/PSI), the responses were coded in terms of the official language(s) of that country and respondents' language labels.

Following the initial data clean-up and re-coding in terms of language labels, all of the responses were prepared for subsequent entry into a statistics software package by assigning numerical values to each response (Brace et al., 2009). For example, questions requiring Yes/No/Unsure answers were coded by means of assigning the numerical values 2/1/0 to each response. Once all responses had been coded numerically, the data was entered into IBM SPSS Statistics (Version 22), a statistical software package widely used in the social sciences for questionnaire analysis among other applications (Brace et al., 2009). SPSS was considered to be the most appropriate statistical software package for the present project, as it combines ease of use with the analytical tools necessary for the present analysis.

Initial checks were then carried out by producing frequency bar charts for each question for each group of interpreters, to ensure that there were no coding errors. Further data clean-up was carried out by removing any empty categories, and grouping together similar categories (Brace et al., 2009; Dörnyei and Taguchi, 2010).

#### 5.3.2 Coding of respondents' working languages

In the AA group the two A languages are labelled A1 and A2. The languages are coded so that A1 is always the first language (with the exception of cases where neither A1 nor A2 is the first language), either acquired on its own or simultaneously with A2, in cases where both languages were learnt from birth. A2 is never the first language on its own: it is either acquired from birth together with A1, or acquired after A1. The AB respondents all have one A language and one B

language only, as per the AIIC A and B language classifications. These two languages are labelled A and B accordingly. Thus, the labels in this case refer only to the current state of the two strongest working languages, and not the age at which these were acquired. The languages for the PSI group are labelled L1 and L2. L1 is always the first language, with the exception of cases where neither L1 nor L2 is the first language. L2 is always the second language, with the exception of cases where both L1 and L2 were acquired simultaneously from birth.

# 5.3.3 Statistical methods used and presentation of results

The majority of the questions give responses in the form of nominal data, and these are reported as frequencies and percentages. Descriptive statistics are used for the most part, on the basis of which the three populations of interest are described and compared in order to investigate the research questions set. The entire populations were sampled, and the responses obtained can therefore be considered to be representative of the three groups of interpreters.

Inferential statistics are used only in the case of means age of SLA, where the independent samples t-test is applied to compare the mean ages of SLA for pairs of groups of respondents. The aim of this is to determine whether the means ages of SLA of two groups are statistically different from each other, taking the variability into account (see, for example, Field, 2013). In all cases, the two-tailed value of p is given. All t-statistics were calculated in IBM SPSS. In addition, the corresponding effect size r was also calculated manually using the formula:  $r = \sqrt{t^2/(t^2 + Df)}$ , where Df is the number of degrees of freedom (see Field, 2013).

The results are presented for groups of interpreters in turn, starting with the additional questions pertaining to the AIIC AA group, followed by a full description and comparison of the AIIC AA and AB groups, concluding with a description of the PSI group and comparison with the AIIC AB group. In the final section, the results for the AB group are repeated so that they can be more readily compared with the findings for the PSI group. For comparison of subgroups of the AA and AB groups with the main populations, the full findings are presented in Appendices II, III and IV, with only the main trends outlined in the text of the dissertation. The discussion of the findings in Chapter 7 reverts to the format of the research questions set out in Section 5.1.1 of the present Chapter.

#### 6. Results

This Chapter reports the responses to the questionnaire obtained from the three groups of interpreters. The first section presents an overview of the questionnaire response rate for the three groups, as well as a general description of the populations (mean age, working languages, mean interpreting experience). The next section deals with questions pertaining to the AIIC AA group, while section three compares the responses given by the AIIC AA and AB groups. The final part of this Chapter sets out the findings for the PSI group and compared this population with the AIIC AB interpreters. The aim of this Chapter is to present and describe the findings; a discussion of their implications in view of the literature and research questions can be found in Chapter 7.

#### 6.1 Data collection results

This section is divided into two parts. The first is an evaluation of the data collection methods with reference to the response rate. The second presents some general information about the three populations of interpreters.

#### 6.1.1 Response rate

As explained in Chapter 5, a pilot version of the questionnaire was initially sent out to a group of UK-based PSI interpreters, with a response rate of 8%. One of the aims was to increase the response rate for the final questionnaire. In order to do this, three separate covering letters were drafted for the individual groups of interpreters, highlighting the aspects of the present project that are of most interest to that particular group. The questionnaires were sent out to respondents individually via email, so as to establish a feeling of personal connection. Table 6.1/1 below shows the overall response rate for each population, as well as the number of the completed questionnaires that were actually used in the final analysis.

Table 6.1/1. Response rate and total responses received for each group of interpreters

	AA	AB	PSI	Total
Questionnaires sent	200	1326	1568	3094
Completed questionnaires	49	232	224	505
Overall response rate	24.5%	17.5%	14.3%	16.3%
Questionnaires used in analysis	48	212	180	440
Adjusted response rate	24.0%	16.0%	11.5%	14.2%

The overall response rate was 16.3%, a substantial improvement compared with the response rate for the pilot questionnaire (even within the PSI group only, which was the group used for the pilot, the response rate increased; see Table 6.1/1 above). Note that the number of actual questionnaires used in the subsequent analysis is lower than the total number of completed questionnaires received. This is due to the fact that some of the questionnaires received were either incomplete or ambiguous. The adjusted response rate, which corresponds to the number of useable questionnaires received, is therefore 14.2% (see Table 6.1/1). Even with the adjustment for usability, the number of responses received is sufficient for the purposes of the present study. Therefore, the data collection methods used can be considered successful.

It is noted that a particularly high response rate was observed among the AA group (24%; see Table 6.1/1). This may be indicative of this group's particular interest in bilingualism research, in view of their special status as bilinguals with two native languages, according to the AIIC classification. The responses took around 8 months to collect, as they were sent individually and the researcher personally responded to any questions the interpreters had about the project. The researcher also individually thanked respondents who had completed the questionnaire.

## 6.1.2 General description of the populations

As shown in Table 6.1/1 above, the final population sizes are as follows: AA – 48; AB – 212; PSI – 180. A general comparison of the groups is set out below, with respect to (1) working languages; (2) age and number of years of interpreting experience.

#### (1) Working languages

Table 6.1/2 below gives the main categories of working languages for the three groups, split into the two strongest working languages (see Chapter 5 for details on the coding and labelling of respondents' working languages). The categories with more than 10% are emphasised in bold.

Table 6.1/2. Main categories of working languages for the three groups of interpreters

Language	AA Group		AB Group		PSI Group	
category	A1	A2	A	В	L1	L2
English	64.6%	25.0%	25.5%	69.8%	16.7%	83.3%
French	14.6%	14.6%	22.2%	15.6%	4.7%	3.6%
Spanish	6.3%	31.3%	9.0%	2.4%	7.3%	3.1%
German	2.1%	4.2%	14.2%	7.5%	1.0%	1.0%
East European	8.3%	4.2%	9.0%	1.4%	43.2%	2.6%
Scandinavian	2.1%	0%	3.3%	0.5%	0%	0%
Other European	0%	14.6%	9.0%	1.9%	7.3%	3.1%
Non-European	2.1%	6.3%	8.0%	0.9%	19.8%	3.1%

Similar language distribution patterns are observed for the AA and AB groups, with English and French together dominating across all working languages. English was found to dominate as the main language category for language A1 among the AA group (64.6%) and B among the AB group (69.8%). Other European languages also feature prominently, with Spanish particularly prevalent in within the A2 language category among the AA group (31.3%), and German in the A language category among the AB group (14.2%). In general, non-European languages do not form a significant proportion of the working languages for either group of conference interpreters (see Table 6.1/2 above).

The working language distribution pattern is rather different for the PSI group, with East European languages dominating in the L1 language category (43.2%), followed by non-European languages (19.8%) and English (16.7%), as shown in Table 6.1/2 above. English dominates as the L2 language for the majority of PSI respondents (83.3%), with no other language category making a significant contribution (see Table 6.1/2). The working languages of the PSI group are coded so that language L1 is always the first language spoken (unless L1 and L2 are acquired simultaneously at birth; see Chapter 5 for details). Therefore the L1 can be thought of as the first language, and L2 the second. The results therefore suggest that, based on this sample of UK-based (NRPSI) public service interpreters, the majority of professional public service interpreters working in the UK have English as their second language. This finding is in line with the expectations set out in Chapter 4.

### (2) Age and interpreting experience

In addition to the spread of working language for the three groups, the mean age and interpreting experience (in years) were compared for each population. The average (mean) age in years for each group of interpreters, together with the range, is presented in Table 6.1/3 below.

Table 6.1/3. Average (mean) age in years and age range for each group of interpreters

	Group of interpreters		
	AA	AB	PSI
Mean age (years)	56.5	53.9	47.8
Min. age (years)	29.0	27.0	25.0
Max. age (years)	81.0	84.0	78.0
Range (years)	52.0	57.0	53.0

We see that the average age in years of the PSI population is lower than that of the two groups of conference interpreters. The AA group has the highest mean age and largest age range (see Table 6.1/3).

The average length of interpreting experience (in years), together with the range for each group, is presented in Table 6.1/4 below.

Table 6.1/4. Average (mean) interpreting experience in years and range for each group of interpreters

	Group of interpreters		
	AA	AB	PSI
Mean interpreting experience (years)	29.1	25.1	12.6
Min. (years)	7.0	4.0	1.0
Max. (years)	50.0	55.0	50.0
Range (years)	43.0	51.0	49.0

The results presented in Table 6.1/4 show that the AA group has the longest interpreting experience on average (29 years), followed by the AB group (25 years). The PSI group has the shortest average interpreting experience of just over 12 and a half years, which is less than half of that of the AA and AB groups. We also observe that the minimum interpreting experience is highest among the AA group, followed by AB and then PSI. However, all three groups exhibit a large range between the minimum and maximum number of years of experience.

## 6.2 Questions pertaining to the AIIC AA group

This section sets out the responses to the additional questions included in the AIIC AA questionnaire. The presentation of the findings is divided into three main sections: questions on the notion of two mother tongue and the AIIC AA classification, questions concerning perceptions of the true bilingualism definition, and further investigation of the characteristics of anomalous groups of respondents.

#### 6.2.1 Two mother tongues and the AIIC AA classification

Interpreters from the AIIC AA group were asked whether they consider both of their A languages to be mother tongues. The responses to this question are presented in Table 6.2/1 below.

Table 6.2/1. AIIC AA interpreters' responses to the question of whether they consider both A languages to be mother tongues

Would you agree that both of your A languages are mother tongues?		
esponse Frequency and % of total respond		
Yes	41 (85.4%)	
No	2 (4.2%)	
Unsure	5 (10.4%)	

It was found that the overwhelming majority of AA respondents agreed that they do have two mother tongues (85.4%; see Table 6.2/1). This indicates that, on the whole, respondents' own perceptions of mother tongue competence/identity in their A languages are in line with external perceptions, as represented by the AIIC language classifications. However, as can be seen from Table 6.2/1 above, 14.6% of respondents did not answer Yes to this question, suggesting that self-ascribed mother tongue identity does not necessarily correspond neatly to the linguistic identity ascribed by others. If the AIIC AA classification is assumed to correspond to mother tongue competence and/or identity in two languages as perceived by others, then the above findings suggest that internal perceptions of this are stronger than external ones. The characteristics of the subgroup of AA respondents who do not feel that both of their A languages are mother tongues (i.e. those who answered No or Unsure to Q2/6/AA) are investigated further in Section 6.2.3 (1) below.

Respondents from the AA group were also asked whether they felt that both of their A languages were mother tongues in childhood. The responses are presented in Table 6.2/2 below.

Table 6.2/2. AIIC AA interpreters' responses to the question of whether both of their A languages were mother tongues in childhood

Would you say that both of your A languages were mother tongues in childhood?		
Response	Frequency and % of total respondents	
Yes	33 (68.8%)	
No	15 (31.3%)	
Unsure	0	

It was found that a little over two thirds (68.8%) of AA respondents believe that both of their A languages were mother tongues in childhood, while the remaining respondents (31.3%) did not consider this to be the case (see Table 6.2/2 above). Those respondents who did not agree that their two languages were mother tongues in childhood were asked to specify at what age they attained this level of competence in both languages and what led to it. Table 6.2/3 below shows the mean age and range at which mother tongue competence was achieved in both A languages if not in childhood.

Table 6.2/3. Average (mean), min. and max. age (years) at which AA interpreters achieved mother tongue competence in both A languages, if not in childhood

	Age (years)
Mean age mother tongue competence achieved in both A languages	19.7
Min. age mother tongue competence achieved in both A languages	12
Max. age mother tongue competence achieved in both A languages	30

It was found that, for those respondents who did not feel that both languages were mother tongues in childhood, the mean age at which this level of competence was attained in both languages is 19.7 years, with a range from 12 to 30 years (see Table 6.2/3 above). The reasons that led to mother tongue competence in both languages were reported as either change in language of education (secondary school or university) or change in dominant linguistic environment (moving country), or both. Only one participant gave a different reason, namely "extensive reading, the practice of translation". It is notable that no respondents cited interpreting as the reason for attaining mother tongue competence in both A languages.

Another factor considered was the date the AA classification was attained relative to date of joining AIIC. Table 6.2/4 below gives the breakdown of respondents according to whether the AA classification was attained upon joining AIIC or at a later date.

Table 6.2/4. Breakdown of AIIC AA respondents according to whether they obtained their AA classification upon joining AIIC or at a later date

	No. of respondents and % of total
AA classification awarded upon joining	41 (85.4%)
AA classification awarded later	7 (14.6%)

It was found that 14.6% of AA respondents attained their AA classification at a later date, i.e. they were initially classed as AB upon joining AIIC, and subsequently "upgraded" to AA status (see Table 6.2/4 above). This means that, for this particular subgroup of respondents, their A languages were not both considered to be mother tongues by external reviewers (peers). For those AA respondents who were not classed as AA upon joining AIIC, the time taken to achieve double A classification was calculated. The results are presented in Table 6.2/5 below.

Table 6.2/5. Average (mean), min. and max. time taken (in years) to achieve the AIIC AA classification for those AA respondents who did not obtain this classification upon joining AIIC

	Time (years)
Average (mean) time taken to achieve AA classification	5.6
Min. time taken to achieve AA classification	2
Max. time taken to achieve AA classification	10

It was found that the time taken to achieve the AA classification if it was not awarded upon joining AIIC ranged from 2 to 10 years, with just over 5 and a half years being the average time taken (see Table 6.2/5). The characteristics of this subgroup of the AA population are examined in more detail in Section 6.2.3 (2) below.

#### 6.2.2 Perceptions of the true bilingualism definition

The AA respondents' perceptions of the true bilingualism definition proposed by Thiéry (1975) were investigated, as well as the relationship between true bilingualism and true bilingualism in writing (i.e. being indistinguishable in writing from a monolingual in both A languages). Respondents were first asked whether they consider themselves true bilinguals according to Thiéry's definition. The responses to this question are given in Table 6.2/6 below.

Table 6.2/6. AIIC AA interpreters' responses to the question of whether they consider themselves to be true bilinguals in accordance with the definition

Do you feel that you are a true bilingual as defined by Thiéry (1975)?		
Response	Frequency and % of total respondent	
Yes	44 (91.7%)	
No	1 (2.1%)	
Unsure	3 (6.3%)	

It was found that an overwhelming majority of the AA respondents (91.7%) consider themselves to be true bilinguals as defined by Thiéry (see Table 6.2/6 above). Three of the four respondents who did not answer Yes to this question gave the following additional clarifications: (1) "I have greater difficulty as concerns speed and concordance in one of the two languages", (2) "I have a very slightly detectable accent [in one of my A languages]" and (3) "Both [members of the A1 linguistic community] and [members of the A2 linguistic community] invariably assume I am a compatriot, an impression strengthened not weakened by non-job related conversation. The same is true re friends and family from both countries. Usually, people tend to doubt my abilities in "the other" language and cultural environment, the "foreign" one". Note that comment (1) essentially refers to an imbalance between the two languages although, as pointed out in Chapter 4, true bilingualism does not imply bilingual balance. Thus, the author of comment (1) may in fact be a true bilingual. Comment (3) is particularly interesting, and it also does not prevent its author from being considered a true bilingual under Thiéry's definition, as s/he is certainly taken to be "one of them" by members of both the A1 and A2 linguistic community. The fact that members of each community doubt his/her belonging to the "other" linguistic community may in fact be taken as a sign of the strength of this acceptance. It is therefore only comment (2) that may prevent its author from being considered a true bilingual within the meaning of Thiéry's definition, if the presence of a slight accent prevents him/her from being accepted as one of them by the members of the corresponding linguistic community. Therefore, given that the fourth respondent who did not consider him or herself to be a true bilingual did not provide additional clarifications, and comments (1) and (3) do not in themselves prevent their authors from being defined as true bilinguals, it can be considered that 46 out of the 48 respondents are (or would define themselves to be) true bilinguals, amounting to 95.8% of the total.

Next, respondents were asked whether they believe that the AIIC AA classification is equivalent to the true bilingualism definition, as intended by Thiéry (1975). The table below gives the responses to this question.

Table 6.2/7. AIIC AA interpreters' responses to the question of whether they believe that the true bilingualism definition is equivalent to the AIIC AA classification

Do you think that the true b	ilingualism definition corresponds to the AIIC AA classification?
Response	Frequency and % of total respondents
Yes	34 (70.8%)
No	6 (12.5%)
Unsure	7 (14.6%)

Table 6.2/7 above shows that almost three quarters of AA respondents (72.3%) agreed that the true bilingualism definition corresponds to the AIIC AA classification. One respondent who did not agree that this was the case clarified in the space for additional comments provided at the end of the questionnaire that s/he believes that the AA classification is stronger than the true bilingualism definition, suggesting that the AA classification refers to "someone who is taken to be one of them by the members of two different linguistic communities, *at exactly the same* social and cultural level". This is interesting because, since all of the respondents in this group are in possession of the AIIC AA classification, yet not all would describe themselves as true bilinguals with certainty, the results imply the opposite: that true bilingualism is a stronger concept than the AA classification.

Finally, it was explored whether true bilingualism as originally defined by Thiéry as pertaining to spoken language only, is accompanied by true bilingualism in both languages in writing. Respondents were asked whether they consider themselves to be indistinguishable from a monolingual in writing in both of their A languages (which is equivalent to the idea of being a true bilingual in writing). The results of this inquiry are presented in Table 6.2/8 below.

Table 6.2/8. AIIC AA interpreters' responses to the question of whether they consider themselves to be true bilinguals in writing

Are you a true bilingual in writing?		
Response Frequency and % of total responde		
Yes	35 (72.9%)	
No	6 (12.5%)	
Unsure	7 (14.6%)	

It was found that almost three quarters of AA respondents consider themselves to be true bilinguals in writing, in other words, they believe that they are indistinguishable from a monolingual in writing in each of their A languages (see Table 6.2/8 above). The relationship between the concepts of "true bilingualism" and "true bilingualism in writing" was explored further by considering whether AA respondents who believed themselves to be true bilinguals also felt that they are true bilinguals in writing. It was found that the majority of AA respondents (77.3%) who agreed that they are "true bilinguals" according to Thiéry's definition also considered themselves to be "true bilinguals in writing". Thus, while being indistinguishable from a monolingual in both languages in writing is not a requirement for true bilingualism, the findings suggest that it tends to accompany true bilingualism, at least according to respondents' own perceptions.

#### 6.2.3 Further investigation of results: a closer look at anomalous groups

Two important anomalous groups have been identified during the examination of the findings, namely: (1) respondents who do not feel that both of their A languages are mother tongues (i.e. those who answered No or Unsure when asked whether they consider both A languages to be mother tongues) and (2) respondents who obtained their AA classification at a later date, and not immediately upon joining AIIC. The specific properties of these groups in terms of biographical factors, attitudes and perceptions are investigated below, and compared with the rest of the AA group in each case.

#### (1) Respondents who do not consider both of their A languages to be mother tongues

This investigation concerns the subgroup of AA respondents who answered either No or Unsure to the questions of whether they feel that both of their A languages are mother tongues (7 respondents in total, 14.6% of total group; see Section 6.2.1 above). For convenience, this group shall be referred to as the "Not Two MTs" group. The responses of this group are compared with the rest of the AA group, i.e. those who answered Yes to the question of whether both of their A languages are mother tongues ("Two MTs" group). The full comparison of responses for the two groups is presented in Appendix II, with the main trends outlined below.

It was first investigated whether this subgroup of respondents felt that both of their A languages were mother tongues in childhood, as well as whether they consider themselves to be true bilinguals. The responses to these questions are presented in the table below.

Table 6.2/9. Responses of the Not Two MTs and Two MTs groups to the questions of whether both A languages were mother tongues in childhood and whether respondents consider themselves to be true bilinguals

Group	Questions and responses (freq and %)				
	Were both of your A languages mother tongues in childhood?				
	Yes	No	Unsure		
Two MTs	31 (75.6%)	10 (24.4%)	0		
Not Two MTs	2 (28.6%)	5 (71.4%)	0		
	Are you a true bilingual in accordance with the definition?				
	Yes	No	Unsure		
Two MTs	41 (100.0%)	0	0		
Not Two MTs	3 (42.9%)	1 (14.3%)	3 (42.9%)		

We see from Table 6.2/9 above that, while the majority of respondents (71.4%) who do not consider both A languages to be mother tongues currently also did not feel that both of these languages were mother tongues in childhood, two respondents (28.6%) from the did "Not Two MTs" group stated that their A languages were mother tongues in childhood, but have since ceased to be. This suggests several things, depending on the viewpoint taken on what the idea of "mother tongue" stands for - competence, or identity. If the findings are interpreted as native competence in the two languages, then the implication is that: (1) native competence in two languages in childhood does not necessarily result in this level of competence in adulthood (as perceived by the individual), despite intensive use of both languages as demanded by the interpreting profession and (2) respondents in the AA group are aware of shifts in language competence over the lifespan. If the identity angle is taken, then the findings imply that identifying as having two mother tongues in childhood does not mean that this identity will be preserved in later life, thus perceptions of identity also shift over the lifespan.

With regard to the question on true bilingualism, it was found that respondents in the "Not Two MTs" group were split equally between those who agreed that they are true bilinguals and those who were unsure (42.9% in each case; see Table 6.2/9), with one respondent indicating that s/he is not a true bilingual. On the other hand, 100% of the rest of the AA group ("Two MTs") agreed that they are true bilinguals (see Table 6.2/9). These findings suggest, first of all, that the concept of having two mother tongues is stronger than that of true bilingualism, as the results indicate that it is possible for a respondent to consider him/herself a true bilingual without necessarily feeling that both languages are mother tongues. At the same time, we see that the percentage of AA respondents in the "Not Two MTs" who consider themselves true bilinguals is lower than the same figure among the rest of the AA respondents, which implies that there is a relationship between the two concepts, i.e. those who identify as having two mother tongues are more likely to identify as true bilinguals, compared with those who do not think that they have two mother tongues.

Next, the general characteristics of the "Not two MTs" group were compared with the rest of the AA group in terms of biographical and attitudinal factors (see Appendix II for full results tables). With respect to differences in biographical factors between the two groups, some important observations can be made about the differences in childhood linguistic environment for the two groups. First of all, it was found that the mean age of SLA was higher among the "Not two MTs" group compared with the rest of the AA respondents, with 8 years and 3.7 years as the mean ages of SLA respectively (see Appendix II (1)). While these results are not statistically significant, they do represent a fairly large effect (see Appendix II (1) for full results and independent t-statistics). The range of SLA for the "Not two MTs group was found to be 0-18 years. These findings suggest

that those AA respondents who identify as having two mother tongues also tended to acquire both languages at an earlier age, compared with those who do not feel that they have two mother tongues. At the same time, given the range of the age of SLA for the "Not two MTs" group, mother tongue identity in two languages in later life is not an immediate consequence of the acquisition of both languages at a very early age.

With regard to other components that are taken to make up the respondents' childhood linguistic environment it was also found that, in terms of relative exposure to the two A languages in childhood, the "Not two MTs" group behaved as expected on the basis of the literature (see Appendix II (3) for full comparison of responses). As a general trend, it was found that language A1 dominated as the language that the respondents from the "Not two MTs" group were exposed to across almost all childhood linguistic environment categories considered, including "home", "country" and "school", whereas A1 was less dominant among the AA respondents who did feel that they have two mother tongues in the same categories. This latter group can be characterised instead by a greater exposure to both languages A1 and A2 simultaneously across the various childhood linguistic environment categories considered, compared with the "Not two MTs" group. Specifically, it can be seen that the majority of respondents from the "Not two MTs" group indicated A1 as the first language spoken (71.4%), with the remaining respondents selecting the "other language" category (28.6%) (see Appendix II (3), Table II.3). No respondents from this group had A2 or both languages as the first language spoken. Comparing this with the responses given by the rest of the AA group, it was observed that a lower number indicated A1 as the first language spoken (65.9%), with just under a third (31.7%) giving the answer of both A1 and A2 to this question (see Appendix II (3), Table II.3). It was also observed that all of the respondents within the "Not two MTs" group can be characterised by the fact that none of them had A2 or both languages (A1 and A2) as the language(s) mostly spoken at home during childhood (see Appendix II (3), Table II.3). The language spoken at home during childhood for this group was overwhelmingly the A1 (85.7%, 6 out of 7 respondents), with the remaining respondent indicating that another language (not A1 or A2) was mostly spoken at home during childhood (see Appendix II (3), Table II.3). These results differ from the pattern observed among the rest of the AA respondents, whose responses to this question were mostly split unevenly between A1 and both languages (58.5% and 34.1% respectively, with no respondents indicating that some other language was the language mostly spoken at home during childhood; see Appendix II (3), Table II.3). The same kind of patterns as those identified above can be observed for primary education and, to a lesser extent, secondary education, for the two groups (see Appendix II (3) for full results). Finally, with regard to the childhood environment, there is a marked difference between the groups in terms of the number of respondents who relocated to a different country (at least once) in childhood:

70.7% of the AA respondents with two mother moved to a different country in childhood, but only 28.6% of the "Not two MTs" group did so (see Appendix II (3), Table II.4).

As regards language use and linguistic environment in later life, no major differences were observed between the two groups in the language(s) of higher education and dominant linguistic environment at present categories (see Appendix II (4) and (5) respectively). In terms of selfreported daily language use across various categories, the AA respondents who feel they have two mother tongues exhibited a greater tendency to use both languages in the categories of reading, writing, TV and, to a lesser extent, radio compared with the "Not Two MTs" group, who favoured one language or the other – mostly the A1, with the exception of the writing category, where A2 dominated (see Appendix II (5), Table II.7). In the categories of socialising and internet, both groups exhibited very similar language use patterns, with the majority of both groups indicating that they use both of their A languages for these activities (see Appendix II (5), Table II.7). These results are difficult to interpret in terms of their relationship to mother tongue identity/competence in two languages because the responses are, to different degrees, influenced by external circumstances (e.g. language(s) of the country, social circle, work etc.) and do not necessarily imply deliberate personal choice to use a particular language. Furthermore, it is not clear which direction the influence goes in – whether feeling that both languages are mother tongues results in using both of these more equally on a daily basis, or vice versa.

With regard to language use in a professional context (interpreting and translation), there are several interesting observations that can be made (see Appendix II (6) for full results). First of all, we see that the respondents from the "Not Two MTs" group tend to interpret mostly into either the A2 or both languages (42.9% in each response category), with only one respondent interpreting mostly into the A1 (see Appendix II (6), Table II.8). However, this intensive active use of the A2 or both languages to carry out a cognitively demanding activity (simultaneous interpreting) does not result in both languages being considered mother tongues by the individual. In addition, it is also observed that, while most of the AA respondents who feel that they have two mother tongues have no preference when it comes to direction of interpreting (63.9%), less than a third of the "Not Two MTs" group (28.6%) chose this response category (see Appendix II (6), Table II.8). However, these findings may or may not be significant in terms of the respondents' bilingualism, because preferred interpreting direction may be influenced by a number of other factors that are specific to the practice of interpreting (e.g. conciseness of a language, or the specificities of a particular language combination). Finally, there was a difference in terms of how the respondents from the two groups felt about the relative importance of linguistic vs interpreting skill in interpreting. All of the respondents from the "Not Two MTs" group felt that both interpreting and linguistic skill are

equally important, whereas 50% of the rest of the AA group chose this response – 40% of this group chose interpreting skill with 10% considering linguistic skill to be more important (see Appendix II (6), Table II.8).

Turning our attention to the respondents' attitudes and perceptions, some further distinctions between the two subgroups of the AA group were observed. The "Not Two MTs" group appear to be aware to a greater extent of the conscious effort required to maintain their two A languages at the present high standard compared with the rest of the AA group (85.7% and 65.9% respectively; see Appendix II (7), Table II.11). We also see that, while the majority of the "Not Two MTs" group consider that it would not be possible to maintain both A languages at their present high standard if not working as an interpreter (57.1%), only a minority of the rest of the AA group chose this response (19.5%) (see Appendix II (7), Table II.11). It may also tentatively be concluded that the "Not Two MTs" group exhibit a slightly lower level of non-professional motivation to maintain both languages at their current high standard compared with the rest of the AA group. This is based on the findings that fewer respondents from the "Not Two MTs" group stated that they would make the effort to maintain both languages at this standard if not working as an interpreter compared with those who do have two mother tongues (42.9% and 62.5% respectively), and that more respondents from the "Not Two MTs" group indicated that their motivation for filling gaps in linguistic knowledge is purely professional, compared with the rest of the AA group (42.9% and 10.0% respectively; see Appendix II (7), Table II.11).

Respondents from the "Not Two MTs" group were also compared with the rest of the AA group in terms of their perceptions of language dominance shifts and cross-linguistic interference between their two A languages (see Appendix II (8) for full results). The response patterns for cross-linguistic interference and the direction it occurs in were almost identical for the two groups, with almost a third of respondents from each group indicating that cross-linguistic interference does not occur in either language (28.6% in the "Not Two MTs" group, 29.3% in the rest of the group) or occurs in both languages equally (28.6% in the "Not Two MTs" group, 31.7% in the rest of the group; see Appendix II (8), Table II.13). In terms of awareness of fluctuations in languages dominance, however, the groups exhibited differing response patterns. A greater proportion of the respondents from the "Not Two MTs" group felt that language dominance fluctuates between their two A languages compared with the rest of the AA group (71.4% and 46.3% respectively; see Appendix II (8), Table II.12). If identifying both languages as mother tongues is taken to be a measure of proximity of the two languages (in terms of competence), whereas not having two mother tongues is taken to imply that one of the languages is stronger than the other, then this finding goes against the prediction that the closer the two languages, the less stable the system and

hence more likely to be prone to fluctuations. However, in this case, all respondents have the same external proficiency ratings in both languages (the AA classification), and therefore identifying as having two mother tongues may imply an additional emotional component related to exposure to the languages in childhood, and is not necessarily to be taken as an objective marker of relative competence in the languages. Indeed, the responses to the question concerning attachment to the languages show that a greater percentage of the "Not Two MTs" group felt a special attachment towards their A1 compared with the rest of the AA group (57.1% and 29.3% respectively; see Appendix II (9), Table II.14). In addition, just over half of AA respondents who do consider both A languages to be mother tongues reported that their do not feel a special attachment towards one language over the other (51.2%), whereas this figure was slightly lower among the "Not Two MTs" group (see Appendix II (9), Table II.14). These findings are in line with expectations, and may be interpreted as confirmation of the hypothesis that the concept of the mother tongues carries an additional emotional attachment component. Interestingly, with respect to perceptions of word weight/emotional force of words (love/hate, swear words) in the two languages, the responses of the two groups appear to directly contradict expectations. An overwhelming majority of the "Not Two MTs" group reported that words carry equal weight in both of their A languages (85.7%), whereas this figure was lower among the rest of the AA group (56.1%; see Appendix II (9), Table II.14).

Finally, AA respondents who do not feel that they have two mother tongues were compared to the rest of the AA group in terms of self-ratings of proficiency in the two languages relative to each other, as well as self-ratings in each language compared to an average monolingual standards (taken to be a marker of language confidence), across different categories (see Appendix II (10) for full results). On the whole, it was found that the groups behaved as expected in terms of relative competence ratings in their A languages across the three categories of command of registers, gaps in vocabulary and writing, with the "Not Two MTs" group tending to rate the A1 as stronger, with the rest of the AA respondents tended to feel that both languages are equal. Specifically, 70.7% of AA respondents who feel that they have two mother tongues rated their A languages as equal in terms of command of registers (the corresponding figure for the "Not Two MTs" group is 28.6%), while 42.2% of the "Not Two MTs" group favoured the A1 (compared with 12.2% of the rest of the AA respondents; see Appendix II (10), Table II.15). In the relative number of gaps in vocabulary category, 28.6% of the "Not Two MTs" group rated both languages as equal (compared with 56.1% of the rest of the AA group), while 57.1% indicated that they have fewer gaps in the A1 (compared with 14.6% of the rest of the AA respondents; Appendix II (10), Table II.15). The difference was even more marked in the category of writing, with no respondents from the "Not Two MTs" group stating that they write equally well/easily in both of their A languages (while 70.7% of the rest of

the AA group indicated that this is the case), but 71.4% chose the A1 as the stronger language in writing (compared with 12.2% of the rest of the AA group) (see Appendix II (10), Table II.15). In terms of true bilingualism in writing – being indistinguishable from a monolingual in writing in both languages – it was found that the number of respondents from the "Not Two MTs" group who felt this was the case was lower at 42.9% compared with the same figure for the rest of the AA group (78.0%) (see Appendix II (10), Table II.15). These observed differences suggest that those AA respondents who do not consider both A languages to be mother tongues are also less likely to rate both languages as being equal in terms of competence, compared with the rest of the AA group.

With regard to self-ratings in each language relative to a monolingual standard across the categories of richness of vocabulary, instinctive/native comprehension and pragmatic/cultural competence, there are some slight differences between the two groups. As an overall trend, it can be observed that, across the first two categories (richness of vocabulary, native comprehension), slightly more respondents from the "Not Two MTs" group tend to rate both languages as being equal to the monolingual standard compared with the rest of the AA group, while more of the AA respondents who do feel they have two mother tongues tend to rate both languages as stronger than the monolingual standard compared with the "Not Two MTs" group (see Appendix II (10), Table II.16). However, it must be noted that this difference is small, especially in the case of native comprehension in the A2.. Where the difference is greatest is in the category of richness of vocabulary in the A2, with 71.4% of respondents from the "Not Two MTs" group rating this as being equal to the monolingual standard (compared with 48.8% of the rest of the AA group), while only 28.6% of the "Not Two MTs" group felt that their A2 is stronger in this category (compared with 48.8% of the rest of the AA group) (see Appendix II (10), Table II.16). The difference in responses is smallest with respect to native comprehension in the A2, with the majority of both groups rating this as equal to the monolingual standard (71.4% for the "Not Two MTs" group, 68.3% for the rest of the AA respondents; see Appendix II (10), Table II.16). Where the trend alters is in the third category - pragmatic/cultural competence - in both languages. Here, it was observed that, while the "Not Two MTs" group was more or less equally split between the "same" and "stronger" responses for both languages (42.9% and 57.1% respectively for A1; 57.1% and 42.9% respectively for A2), the rest of the AA respondents overwhelmingly rated both languages as equal to the monolingual standard (80.0% for A1, 75.6% for A2; see Appendix II (10), Table II.16). While the small differences and size of the "Not Two MTs" group make it impossible to extrapolate too much from this, it may be possible to speculate that these findings suggest that those AA respondents who feel that both languages are mother tongues have slightly higher confidence in both languages when it comes to active use (richness of vocabulary) compared with the "Not Two

MTs" group, whereas confidence in passive language use is around the same in both groups. Interestingly, the trend is reversed with respect to pragmatic/cultural competence in the two languages, with the "Not Two MTs" group exhibiting more confidence in both languages compared with the rest of the AA group.

### (2) Respondents who obtained their AIIC AA classification at a later date

This section takes a closer look at the group of AA respondents who obtained their AA classification at a later date, and not immediately upon joining AIIC (referred to as the "AA Later" group) in order to identify whether there are any biographical or attitudinal characteristics that set this group apart from the rest of the AA respondents (referred to as the "AA Same Date" group). The "AA Later" group is made up of 7 respondents, 14.6% of the total AA population surveyed. The full comparison of responses for the "AA Later" group and the rest of the AA group is presented in Appendix III, with the main observations outlined below.

It was first considered whether the AA interpreters who obtained their AA classification at a later date had different perceptions of mother tongue competence/identity in the two languages and of true bilingualism compared with the rest of the AA group. The comparison of responses is presented in the table below.

Table 6.2/10. Responses of the AA Later and AA Same Date groups to the questions pertaining to mother tongue competence and true bilingualism

Group	Questions and responses (freq and %)				
	Were both of your A languages mother tongues in childhood?				
	Yes	No	Unsure		
AA Same Date	27 (65.9%)	14 (34.1%)	0		
AA Later	6 (85.7%)	1 (14.3%)	0		
	Do you consider both of your A languages to be mother tongues?				
	Yes	No	Unsure		
AA Same Date	35 (85.4%)	1 (2.4%)	5 (12.2%)		
AA Later	6 (85.7%)	1 (14.3%)	0		
	Are you a true bilingual in accordance with the definition?				
	Yes	No	Unsure		
AA Same Date	37 (90.2%)	1 (2.4%)	3 (7.3%)		
AA Later	7 (100.0%)	0	0		

As we see from the Table 6.2/10 above, whether the AA classification was awarded at a later date or not appears to have no bearing on whether the languages are considered by respondents to be mother tongues, both currently and in childhood, or on respondents' tendency to class themselves

as true bilinguals. Indeed, contrary to expectations, slightly more respondents from the "AA Later" group reported that both of their A languages were mother tongues in childhood, compared with the rest of the AA group (85.7% and 65.9% respectively; see Table 6.2/10). Also, all of the "AA Later" group felt that they are true bilinguals, compared with 90.2% of the rest of the AA group (see Table 6.2/10). These findings suggest that achieving the AA classification at a later date, i.e. the fact that one of the languages was initially classes as a non-native language by external reviewers, is not connected to one's own perceptions of mother tongue identity or competence. Furthermore, it does not appear to be connected to whether both languages were mother tongues in childhood or not.

The latter observation finds further confirmation in the comparison of childhood linguistic environment categories for the two groups. Overall, no differences were observed between the two groups that would suggest that attaining the AA classification at a later date is related to reduced exposure to one of the languages in childhood (see Appendix III (1), (3) for full results). First of all, a comparison of the mean age of SLA for the "AA Later" group with the rest of the AA respondents revealed this to be virtually the same (4.4 years and 4.3 years respectively; see Appendix III (1), Table III.1). The only observable difference here is the standard deviation, which is higher among the "AA Later" group compared with the rest of the AA respondents (8.25 and 3.78 respectively; see Appendix III (1), Table III.1), which is indicative of a larger age range. As regards relative exposure to the two A languages across the childhood linguistic environment components of "home", "country" and "primary school", both groups exhibited very similar patterns (see Appendix III (3)). In particular, contrary to predictions, the "AA Later" group did not demonstrate increased exposure to the A1 compared with the rest of the AA respondents. The only category where this was found to be the case to a small degree is the first language spoken, with slightly more of the "AA Later" respondents indicating A1 as their first language compared with the rest of the AA group (71.4% and 65.9% respectively), and a smaller percentage of "AA Later" respondents stating that they acquired both languages simultaneously compared with the rest of the group (14.3% and 29.3% respectively; see Appendix III (3), Table III.3). No major differences were observed in terms of the number of respondents within each group who relocated to a different country during childhood, with 57.1% of the "AA Later" group and 65.9% of the rest of the AA group indicating that this was the case (see Appendix III (3), Table III.4).

As regards language use and exposure in later life, again, very similar patterns emerged for both groups, with the majority of both groups indicating that they are generally exposed to and use both of their A languages in everyday life (see Appendix III (5) for full results). The only marked difference observed was in the category of language(s) of higher education, but this difference was found to go against expectations that the "AA Later" group are less likely to have had both

languages as the languages of higher education. Instead, the opposite was found to be true, with 71.4% of respondents from the "AA Later" group indicating that both A languages were their languages of higher education, compared with 33.3% of the rest of the AA group see Appendix III (4). Table III.5). Some slight differences were found between the two groups in terms of language use in a professional context (interpreting and translation). In particular, fewer respondents from the "AA Later" group reported that they interpret into both languages equally, compared with the rest of the AA group (14.3% and 43.9% respectively; see Appendix III (6), Table III.8). The "AA Later" respondents tended to be split equally between those who mostly interpret into the A1 or the A2 (42.9% in each category; see Appendix III (6), Table III.8). Thus, obtaining the AA classification at a later date does not correspond to being required to interpret predominantly into one language. While it was found that fewer of the "AA Later" respondents expressed to preference for interpreting direction compared with the rest of the group (20.0% and 63.2% respectively; see Appendix III (6), Table III.8), this finding may be explained by practice or habit, in light of the previous observation that fewer "AA Later" respondents interpret into both languages equally. There was a difference between the two groups' perceptions of the relative importance of interpreting vs linguistic skill. Specifically, a higher proportion of "AA Later" respondents felt that both were equally important, compared with the rest of the group (85.7% and 52.5% respectively; see Appendix III (6), Table III.8). Only one respondent (14.3%) from the "AA Later" group considered interpreting skill to be the most important (compared with 37.5% of respondents from the rest of the group), while no "AA Later" respondents considered linguistic skill to be the most important (while a small minority, 10.0%, of the rest of the AA group felt that this was the case; see Appendix III (6), Table III.8).

Next, the two groups were compared in terms of their attitudes and perceptions. With regard to the questions concerning effort and motivation it was observed, first of all, that respondents from the "AA Later" group were less likely to think that maintaining both languages at their present high standard requires conscious effort, compared with the rest of the AA group (57.1% and 70.7% respectively; see Appendix III (7), Table III.11). With respect to levels of non-professional motivation to maintain the languages, the results are ambiguous. On the one hand, 71.4% of the "AA Later" group stated that they would make the effort to maintain both A languages at their current standard if not working as a professional interpreter, compared with 57.5% of the rest of the AA group (see Appendix III (7), Table III.11). However, the percentage of "AA Later" respondents who indicated that their motivation for filling gaps in vocabulary in their languages is not purely professional is slightly lower compared with the rest of the AA group (71.4% and 82.5% respectively; see Appendix III (7), Table III.11). Some differences were observed between the groups in terms of awareness of fluctuations in language dominance. In particular, a much larger

percentage of the "AA Later" respondents felt that language dominance does not fluctuate between their A languages, compared with the rest of the AA group (71.4% and 34.1% respectively; see Appendix III (8), Table III.12). The response patterns for perceptions of cross-linguistic interference were similar for the two groups (see Appendix III (8), Table III.13). An examination of the response patterns for the two groups to the questions concerning emotional attachment to the languages revealed results that are contrary to expectations. In all three categories intended to measure attachment/emotional response to the languages - language preference when tired/upset, perception of weight of words, attachment towards one language over the other – a greater proportion of the "AA Later" respondents tended to rate both languages as being equal compared with the rest of the AA group (see Appendix III (9), Table III.14). Specifically, 71.4% of the "AA Later" group indicated that they would not revert to one of their A languages if possible when tired/upset, compared with 36.6% of the rest of the group; 85.7% of the "AA Later" respondents felt that words carry equal weight in both of their A languages, compared with 56.1% of the rest of the group; 71.4% of the "AA Later" group stated that they do not have a special attachment to one of their A languages over the other, compared with 46.3% of the rest of the group (see Appendix III (9), Table III.14).

Finally, the two groups were compared in terms of self-ratings of proficiency in their A languages. With respect to self-ratings of proficiency in the two languages relative two each other in the categories of command of registers, number of gaps in vocabulary and writing, the response patterns for command of registers were very similar, with the majority of both groups tending to rate both A languages as being equal (57.1% of the "AA Later" group and 65.9% of the rest of the group; see Appendix III (10), Table III.15). However, in the latter two categories, the "AA Later" group tended to favour the A1, whereas the rest of the AA respondents tended to rate both languages as being equal. Specifically, 28.6% of "AA Later" respondents stated that their A languages are equal in terms of gaps in vocabulary, while 57.1% felt that they had fewer gaps in the A1 (the corresponding figures for the rest of the AA group are 56.1% and 14.6% respectively); 28.6% of the "AA Later" respondents felt that they write equally well/easily in both languages, while 42.9% rated the A1 as stronger in this category (the figures for the rest of the AA group are 65.9% and 17.1% respectively; see Appendix III (10), Table III.15). As regards true bilingualism in writing, the majority of both groups - 71.4% of the "AA Later" group and 73.2% of the rest of the AA group – felt that they are indistinguishable from monolinguals in writing in both A languages (see Appendix III (10), Table III.15). Taken together, these findings suggest that, with respect to certain language use categories at least, respondents who achieved their AA classification at a later date tend to rate the A1 as being slightly stronger than the A2, whereas the rest of the AA respondents are more likely to consider both languages as equal.

Differences were observed between the two groups in terms of self-ratings of proficiency in each language compared to a monolingual standard. In particular, with respect to richness of vocabulary and native comprehension in both languages, it was found that the "AA Later" respondents were more likely to rate both languages as corresponding to the monolingual standard than the rest of the AA group, and less likely to rate the languages as being above this standard (see Appendix III (10), Table III.16). Specifically, 85.7% of the "AA Later" respondents rated their richness of vocabulary in the A1 as being equal to the monolingual standard, with 14.3% rating it as stronger (compared with 41.5% and 51.2% respectively for the rest of the AA group); the figures for richness of vocabulary in the A2 were the same among the "AA Later" respondents (the corresponding figures for A2 for the rest of the AA group are 46.3% and 51.2% for same and stronger respectively; see Appendix III (10), Table III.16). As regards native comprehension, all respondents from the "AA Later" group rated both languages as being equal to the monolingual standard, whereas the percentage of the rest of the group who rated the A1 as equal to the monolingual standard in this category is 61.0%, and 63.4% for the A2; just over a third of the rest of the AA group rated both languages as being stronger in this category (34.1% for A1 and 36.6% for A2; see Appendix III (10), Table III.16). The response patterns for pragmatic/cultural competence in each language were similar for both groups, in particular in the case of the A2, with the majority of respondents in the two groups rating the languages as being equal to the monolingual standard, and around a quarter of respondents from both groups rating the languages as stronger (see Appendix III (10), Table III.16). These findings suggest that, if the ratings relative to an average monolingual standard are interpreted as markers of language confidence then, with respect to certain parameters at least, the "AA Later" group display slightly lower confidence levels compared with the rest of the AA respondents. However, the overwhelming majority of respondents from both groups tend to consider both languages as being at least equal to the monolingual standard

## 6.3 Description and comparison of the AIIC AA and AB groups

This Section presents the results for the AA and AB groups, comparing the responses of the two groups in terms of biographical and attitudinal factors. The final part of this Section focuses on the characteristics of the subgroup of AB respondents who consider both of their languages to be mother tongues – in other words, the internal perceptions of this group do not match external perceptions of mother tongue identity/competence (as represented by the AIIC AB language classification).

#### 6.3.1 Mother tongue identity, biographical factors and language use

It was first considered whether AB respondents consider one of their languages to be the mother tongue, as expected on the basis of the AIIC language classification. The responses to this question are presented in the table below.

Table 6.3/1. Responses of the AB group to the question of which languages they consider to be mother tongues

Group	Mother tongue			
	A	В	Both	Other
AB	179 (84.4%)	0	28 (13.2%)	5 (2.4%)

As we can see from Table 6.3/1 above, the overwhelming majority (84.4%) of AB respondents consider the A language to be the mother tongue, with only 13.2% (28 respondents in total) indicating that both A and B are mother tongues. Compare this figure with 85.4% of AA respondents who feel that they have two mother tongues (see Section 6.2.1). These findings suggest that internal perceptions of mother tongue competence/identity tend to match external perceptions, as represented by the AIIC language classifications. The characteristics of the subgroup of AB respondents who do consider both languages to be mother tongues are explored in more detail in Section 6.3.3 below and compared with the rest of the AB group.

Next, the AA and AB groups were compared in terms of age of SLA and childhood linguistic environment, including languages of primary and secondary education. The first notable difference observed between the groups was the mean and of SLA, which is given in the table below.

Table 6.3/2. Comparison of mean age of SLA in years for the AA and AB groups

Group	No. in group	Mean age SLA (years)	Std. Deviation
AA	47	4.3	4.841
AB	205	8.6	4.889

It was found that the average (mean) age of SLA for the AA group was 4.3 years, and that the AB group tended to acquire their second language 4 years later on average than the AA group, at age the age of 8.6 years (see Table 6.3/2 above). An independent samples t-test showed that the difference in mean age of SLA for the AA and AB groups is significant and represents a medium sized effect: t(250) = -5.414, p = .000; t = .324. It can therefore be inferred that the AIIC AA group acquired both of their strongest working languages at an earlier age than the AB group. We note also that the mean age of SLA for the AB group is close to the mean age of SLA for the subgroup of AA respondents who do not feel that they have two mother tongues (see Section 6.2.3).

In addition to calculating the mean age of SLA for the AB and AA groups, the data were examined more closely. In particular, it was observed that not only is the average age of SLA lower among the AA group compared with the AB group, the maximum age of SLA is also lower among the AA respondents (18 years for the AA group, 30 years for AB). It was also found that, while the SLA range for the AA group is 0 - 18 years, the majority of respondents fall within the 0 - 8 year range. Within the AB group, the actual SLA range is 0 - 30 years, but the majority of respondents fall into the 5 - 11 year range. Thus, while the overall ranges for both groups are indicative of a high degree of individual variation, there is a general tendency to cluster within a smaller set of values, which is lower among the AA population.

The AA and AB groups were next compared in terms of relative exposure to their two strongest languages in childhood. The full comparison of responses across all childhood linguistic environment categories considered in the questionnaire is presented in Table 6.3/3 below.

Table 6.3/3. Comparison of childhood linguistic environment (by category) and language(s) of primary and secondary education for the AA and AB groups

		Childhood linguistic environment/education category and responses (free and %)				
First language(s) spoken						
A1/A	A2/B	Both	Other			
32 (66.7%)	0	13 (27.1%)	3 (6.3%)			
183 (86.7%)	14 (6.6%)	4 (1.9%)	10 (4.7%)			
	Mothe	er's language(s)				
A1/A	A2/B	Both	Other			
27 (56.3%)	11 (22.9%)	6 (12.5%)	4 (8.3%)			
171 (80.7%)	20 (9.4%)	4 (1.9%)	17 (8.0%)			
Father's language(s)						
A1/A	A2/B	Both	Other			
31 (64.6%)	9 (18.8%)	7 (14.6%)	1 (2.1%)			
179 (85.2%)	11 (5.2%)	8 (3.8%)	12 (5.7%)			
Language(s) mostly spoken in the home during childhood						
A1/A	A2/B	Both	Other			
30 (62.5%)	3 (6.3%)	14 (29.2%)	1 (2.1%)			
185 (87.3%)	7 (3.3%)	13 (6.1%)	7 (3.3%)			
	Childhood linguis	stic environment (co	ountry)			
A1/A	A2/B	Both	Other/Unclear			
13 (27.1%)	12 (25.0%)	18 (37.5%)	5 (10.4%)			
152 (72.0%)	4 (1.9%)	14 (6.6%)	41 (19.4%)			
Language(s) of primary education						
A1/A	A2/B	Both	Other			
20 (43.5%)	12 (26.1%)	14 (30.4%)	0			
164 (78.5%)	13 (6.2%)	27 (12.9%)	5 (2.4%)			
	Language(s)	of secondary educati	ion			
A1/A	A2/B	Both	Other			
11 (23.9%)	15 (32.6%)	20 (43.5%)	0			
139 (66.5%)	26 (12.4%)	42 (20.1%)	2 (1.0%)			
	32 (66.7%) 183 (86.7%)  A1/A 27 (56.3%) 171 (80.7%)  A1/A 31 (64.6%) 179 (85.2%)  Lang  A1/A 30 (62.5%) 185 (87.3%)  A1/A 13 (27.1%) 152 (72.0%)  A1/A 20 (43.5%) 164 (78.5%)  A1/A 11 (23.9%)	32 (66.7%) 0  183 (86.7%) 14 (6.6%)  Mother  A1/A A2/B  27 (56.3%) 11 (22.9%)  171 (80.7%) 20 (9.4%)  Fathe  A1/A A2/B  31 (64.6%) 9 (18.8%)  179 (85.2%) 11 (5.2%)  Language(s) mostly spole  A1/A A2/B  30 (62.5%) 3 (6.3%)  185 (87.3%) 7 (3.3%)  Childhood linguis  A1/A A2/B  13 (27.1%) 12 (25.0%)  152 (72.0%) 4 (1.9%)  Language(s)  A1/A A2/B  20 (43.5%) 12 (26.1%)  164 (78.5%) 13 (6.2%)  Language(s)  A1/A A2/B  11 (23.9%) 15 (32.6%)	32 (66.7%) 0 13 (27.1%) 183 (86.7%) 14 (6.6%) 4 (1.9%)  Mother's language(s)  A1/A A2/B Both  27 (56.3%) 11 (22.9%) 6 (12.5%)  171 (80.7%) 20 (9.4%) 4 (1.9%)  Father's language(s)  A1/A A2/B Both  31 (64.6%) 9 (18.8%) 7 (14.6%)  179 (85.2%) 11 (5.2%) 8 (3.8%)  Language(s) mostly spoken in the home dur  A1/A A2/B Both  30 (62.5%) 3 (6.3%) 14 (29.2%)  185 (87.3%) 7 (3.3%) 13 (6.1%)  Childhood linguistic environment (condition of the condition of			

We see from Table 6.3/3 above that the response patterns for the two groups are in line with expectations: the proportion of AB respondents who were exposed to one of their languages (A) is greater than the proportion of AA respondents who were exposed to language A1 only across all categories considered, and a greater percentage of AA respondents were exposed to both of their languages simultaneously, compared with the AB group. Specifically, it was found that while one

language dominates as the first language spoken within both the AA and AB groups (A1 with 66.7% for AA; A with 86.7% for AB), the proportion of AA respondents who acquired both languages simultaneously from birth is much higher than the same proportion of AB respondents (27.1% for AA compared with 1.9% for AB; see Table 6.3/3). A similar pattern is observed for the parents' languages of the two groups, with one language dominating as the language of the mother and father in both cases, but to a lesser extent within the AA group (A1 with 56.3% and 64.6% respectively for AA; A with 80.7% and 85.2% for AB; see Table 6.3/3). As regards the languages mostly spoken at home during childhood ("home"), language A1 was the dominant category within the AA group, and language A within the AB group (62.5% and 87.3% respectively; see Table 6.3/3). That said, the category of "both languages together" was again found to be represented to a higher degree within the AA group compared with the AB group for the variable "home" (29.2% and 6.1% respectively; see Table 6.3/3).

With regard to the dominant linguistic environment of the country of residence in childhood, it was found that a substantially larger proportion of AA respondents had both languages present in the linguistic environment of the country of residence in childhood compared with the AB group (37.5% and 6.6% respectively; see Table 6.3/3). This overall statistic was calculated by taking into account the official language(s) of the country of birth, together with the official language(s) of the country of relocation in childhood for those respondents who relocated. In addition, for those AA respondents who did not have both A languages present in the linguistic environment of the country of residence in childhood, there was a much more even split between the two strongest working languages (27.1% A1, 25.0% A2) compared with the AB group, where language A dominated (72.0%, see Table 6.3/3). Similar patterns emerged in the category of language(s) of primary education and, to a lesser extent, secondary education. With regard to the language(s) of primary education for the AA and AB respondents, it was again found that both of the strongest working languages together were represented to a much higher degree within the AA group compared to the AB group (30.4% and 12.9% respectively; see Table 6.3/3). Furthermore, while language A was found to be the dominant language of primary education for the AB group, and language A1 the dominant category within the AA group, this dominance was more pronounced within the AB group compared with the AA group (see Table 6.3/3). As regards secondary education, while language A was again the dominant language in this category for the AB group with 66.5%, and both together languages were again represented to a greater degree among AA respondents than AB respondents (43.5% and 20.1% respectively), in the case of secondary education, A2 was more widespread as the language of education among AA respondents that A1 for the first time (with 32.6% and 23.9% respectively; see Table 6.3/3).

While this question is not language-specific, the percentage of AA and AB respondents who relocated to a different country during childhood was also taken into consideration. The responses to this question are presented below.

Table 6.3/4. Comparison of responses of the AA and AB groups to the question of whether they relocated to a different country in childhood

Group	Did you relocate to a different country during childhood? (freq and %)		
	Yes	No	
AA	31 (64.6%)	17 (35.4%)	
AB	56 (26.4%)	156 (73.6%)	

It was observed that over twice as many AA respondents relocated to a different country in childhood compared with AB respondents (64.6% and 26.4% respectively; see Table 6.3/4). Although this relocation does not necessarily correspond to a change in linguistic environment in every case, it can be taken to indicate at least a change in cultural environment. The average age of relocation was compared for those AA and AB respondents who did relocate to a different country in childhood, and it was found to be around 5 and a half years for both groups. It is also worth noting that, with respect to the general trends in terms of childhood linguistic environment, the subgroup of AA respondents who do not feel that they have two mother tongues described in Section 6.2.3 (1) lies somewhere between the AA and AB groups.

The AA and AB groups were next compared in terms of languages use and exposure in later life, including in a professional context. First of all, with respect to language(s) of higher education, the two groups exhibited different patterns, summarised in Table 6.3/5 below.

Table 6.3/5. Comparison of language(s) of higher education for the AA and AB groups

Group	Language(s) of higher education (freq and %)			
	A1/A	A2/B	Both	Other
AA	7 (15.2%)	19 (41.3%)	18 (39.1%)	2 (4.3%)
AB	64 (31.2%)	34 (16.6%)	104 (50.7%)	3 (1.5%)

The AA respondents were split almost equally between A2 and both languages in the higher education category (41.3% and 39.1% respectively), with only a small proportion of the group indicating that A1 was their language of higher education (15.2%; see Table 6.3/5 above). On the other hand, half of AB respondents reported that both of their languages were languages of higher education (50.7%), followed by language A with 31.2%, and only 16.6% of AB respondents had language B was their language of higher education (see Table 6.3/5).

The two groups were next compared in terms of their dominant linguistic environment and relative exposure to each language at the time of completion of the questionnaire. The results are presented in Table 6.3/6 below.

Table 6.3/6. Comparison of present linguistic environment for the AA and AB groups

Group	Present linguistic environment (freq and %)					
	Dominant language(s) of country of residence					
	A1/A	A2/B	Both	Other		
AA	10 (22.2%)	27 (60.0%)	3 (6.7%)	5 (11.1%)		
AB	97 (49.2%)	50 (25.4%)	21 (10.7%)	29 (14.7%)		
	Language(s) exposed to the most on a daily basis					
	A1/A	A1/A A2/B Both Other				
AA	12 (25.0%)	21 (43.8%)	14 (29.2%)	1 (2.1%)		
AB	98 (46.2%)	41 (19.3%)	66 (31.1%)	7 (3.3%)		

The responses showed that the majority of AA respondents live in an A2-speaking country (60%), followed by those who live in an A1-speaking country (22.2%), whereas the dominant language of the country of residence for AB interpreters was found to be language A (49.2%), followed by B (25.4%) (see Table 6.3/6). Only a small minority of each group reported living in a country where some other language is mostly spoken (11.1% of AA respondents, 14.7% of AB respondents). Looking next at the language(s) respondents perceive to be exposed to the most on a daily basis, we see that the response patterns reflect the influence of the dominant language of the country of residence, with A2 being the most popular choice among AA respondents (43.8%) and A1 among AB respondents (with 46.2%; see Table 6.3/6). However, within both groups this dominance was less pronounced compared with the figures observed for the language(s) of the country of residence – instead, a higher percentage of respondents within both groups indicated that they are exposed to both languages equally on a daily basis (29.2% among the AA group and 31.1% among the AB group; see Table 6.3/6). This suggests that, while the linguistic environment is a significant contributing factor to perceived degree of language exposure, other factors (e.g. work, leisure, family) can, in part, redress the balance between languages.

Respondents' daily language use habits were also considered across several different categories, namely reading, writing, TV, radio, socialising and the internet. The results are summarised in Table 6.3/7 below.

Table 6.3/7. Comparison of self-reported daily language use (by category) for the AA and AB groups

Group	La	nguage use categor	y and responses (fro	eq and %)			
			Reading				
	A1/A	A2/B	Both	Neither			
AA	10 (20.8%)	2 (4.2%)	35 (72.9%)	1 (2.1%)			
AB	17 (8.0%)	18 (8.5%)	175 (82.5%)	2 (0.9%)			
		·	Writing				
	A1/A	A2/B	Both	Neither			
AA	2 (4.2%)	3 (6.3%)	42 (87.5%)	1 (2.1%)			
AB	40 (18.9%)	9 (4.2%)	160 (75.5%)	3 (1.4%)			
		Watching TV					
	A1/A	A2/B	Both	Neither			
AA	8 (16.7%)	7 (14.6%)	23 (47.9%)	10 (20.8%)			
AB	32 (15.1%)	34 (16.0%)	122 (57.5%)	24 (11.3%)			
		Listening to the radio					
	A1/A	A2/B	Both	Neither			
AA	8 (16.7%)	18 (37.5%)	16 (33.3%)	6 (12.5%)			
AB	38 (17.9%)	60 (28.3%)	28 (13.2%)	38 (17.9%)			
		Socialising					
	A1/A	A2/B	Both	Neither			
AA	5 (10.4%)	8 (16.7%)	35 (72.9%)	0			
AB	47 (22.2%)	20 (9.4%)	136 (64.2%)	9 (4.2%)			
		Using the internet					
	A1/A	A2/B	Both	Neither			
AA	7 (14.6%)	3 (6.3%)	35 (72.9%)	3 (6.3%)			
AB	16 (7.5%)	28 (13.2%)	155 (73.1%)	13 (6.1%)			

It was found that the majority of respondents from both groups use both languages for reading, writing, socialising and viewing internet pages/using social media, with the responses spread more evenly across different response options in the TV and radio categories (see Table 6.3/7). Taken together with the previous results on the dominant linguistic environment of the two groups on interpreters, these findings suggest that members of both groups use both of their strongest languages on a daily basis for a range of activities, irrespective of the main language of the country of residence. This finding is in line with expectations, however, it is not clear whether daily use of both languages in a variety of contexts is a consequence of advanced bilingualism, or a necessary condition for maintaining both languages at this standard. What was not found was a greater use of both languages by the AA group compared with the AB group across the different categories. For

example, in the case of reading, a higher percentage of AB respondents reported using both languages compared with the same figure for AA respondents (82.5% and 72.9% respectively; see Table 6.3/7 above). However, in the case of writing and socialising, this was found to be the case, with more AA respondents reporting that the use both languages on a daily basis for these activities compared with AB respondents (87.5% and 72.9% respectively for AA; 75.5% and 64.2% respectively for AB; see Table 6.3/7 above).

The final part of the comparison of the AIIC AA and AB respondents' current language use concerns language use in a professional context (interpreting and translation). The results for questions pertaining to interpreting are presented in the table below.

Table 6.3/8. Comparison of responses for the AA and AB groups to questions pertaining to

interpreting

Group	Responses to qu	Responses to questions concerning interpreting (freq and %)				
	Lang	Language(s) you mostly interpret into:				
	A1/A	A2/B	Both			
AA	15 (31.3%)	14 (29.2%)	19 (39.6%)			
AB	127 (59.9%)	56 (26.4%)	29 (13.7%)			
	Preferred interpreting direction					
	A1/A	A2/B	No preference			
AA	9 (20.9%)	9 (20.9%)	25 (58.1%)			
AB	127 (62.3%)	32 (15.7%)	45 (22.1%)			
	What is more in	What is more important: interpreting skill or linguistic skill?				
	Interpreting skill	Linguistic skill	Both equally important			
AA	16 (34.0%)	4 (8.5%)	27 (57.4%)			
AB	90 (43.1%)	24 (11.5%)	95 (45.5%)			

The results showed a fairly even spread between the three response categories (A1, A2, both) within the AA group in terms of the languages respondents interpret into the most, with the "both A1 and A2" category receiving the highest proportion of responses (39.6%; see Table 6.3/8). Within the AB group, language A was found to be the target language used most frequently in interpreting for the majority of AB respondents (59.9%; see Table 6.3/8). In terms of preferred interpreting direction, a greater proportion of AA respondents reported no preference (58.1%) compared with the same figure for AB respondents (22.1%; see Table 6.3/8). However, these findings are not necessarily indicative of greater ease or confidence in one of the languages, and could be related to interpreting-specific factors (language pair, practice etc.) as opposed to bilingualism. Respondents were also asked whether they consider linguistic skill or interpreting

skill to be most important in interpreting, and similar response patterns were observed for both groups. In particular, the most popular response category within both the AA and AB groups was that both are equally important (selected by 57.4% of AA and 45.5% of AB respondents respectively), with interpreting skill coming next (34.0% for AA, 43.1% for AB) and linguistic skill being the least popular option (8.5% for AA, 11.5% for AB; see Table 6.3/8).

The proportion of respondents from both groups who carry out translation work was also considered, as well as the languages they translate into. Table 6.3/9 below gives the number of respondents from each group who engage in translation work in addition to interpreting.

Table 6.3/9. Number of respondents from the AA and AB groups who also carry out translation work

Group	Do you do any translation work? (freq and %)		
	Yes	No	
AA	23 (47.9%)	25 (52.1%)	
AB	149 (70.6%)	62 (29.4%)	

It was found that a higher proportion of AB interpreters also work as translators, compared with AA interpreters (70.6% and 47.9% respectively; see Table 6.3/9 above). For those AA and AB interpreters who also carry out translation work, the languages respondents translate into the most are given below.

Table 6.3/10. Languages respondents from the AA and AB groups mostly translate into

Group	Language(s) you mostly translate into (freq and %):			
	A1/A	A2/B	Both	
AA	6 (26.1%)	3 (13.0%)	14 (60.9%)	
AB	69 (46.6%)	4 (2.7%)	75 (50.7%)	

Of those respondents who do carry out translation work, the majority of the AA respondents indicated that they translate into both of their A languages (60.9%; see Table 6.3/10 above). Of the AB respondents who also work as translators, almost half translate into both languages A and B, with slightly fewer translating into language A only (50.7% and 46.6% respectively; see Table 6.3/10 above). This last finding is shows that a large number of respondents translate into a language that is not considered to be their native language.

# 6.3.2 Attitudes, perceptions and self-ratings

The two groups were next compared in terms of attitudinal factors and self-ratings, with respect to the categories outlined in Chapter 5. First of all, respondents' perception of language maintenance effort and their motivation to maintain their two strongest working languages at their current high standard was investigated and, in particular, the extent to which respondents' motivation to maintain their two strongest languages is connected to professional requirements. The responses to these questions are presented in Table 6.3/11.

Table 6.3/11. Comparison of responses for the AA and AB groups to questions pertaining to language maintenance effort and motivation

Group	Responses to questions on language maintenance effort and motivation (freq and %)  Do you feel that maintaining your two strongest languages at their present high standard requires a degree of conscious effort?		
	AA	33 (68.8%)	13 (27.1%)
AB	154 (73.7%)	42 (20.1%)	13 (6.2%)
	Would you make the effort to maintain both of your strongest languages at their present high standard if not working as a professional interpreter?		
	Yes	No	Unsure
AA	28 (59.6%)	7 (14.9%)	12 (25.5%)
AB	105 (50.5%)	57 (27.4%)	46 (22.1%)
	Would it be possible to maintain both of your strongest languages at their present high standard if not working as a professional interpreter?		
	Yes	No	Unsure
AA	22 (45.8%)	12 (25.0%)	14 (29.2%)
AB	65 (31.1%)	96 (45.9%)	48 (23.0%)
	Do you seek to fill gaps in your linguistic knowledge when you notice them?		
	Yes	No	Unsure
AA	46 (95.8%)	1 (2.1%)	1 (2.1%)
AB	206 (98.1%)	1 (0.5%)	3 (1.4%)
	Is your motivation for filling gaps in linguistic knowledge purely professional?		
	Yes	No	Unsure
AA	7 (14.9%)	38 (80.9%)	2 (4.3%)
AB	38 (18.1%)	148 (70.5%)	24 (11.4%)

Respondents were asked whether they feel that maintaining their two strongest working languages at their present high level requires a degree of conscious effort (Q4.10/AA, 4.11/AB), and also whether they believe it would be possible to maintain both languages at this standard without working as professional interpreters (Q4.12/AA, 4.13/AB). The results showed that the majority of respondents in both groups feel that language maintenance requires a degree of conscious effort, but this proportion was slightly higher within the AB group compared with the AA group (73.7% and 68.8% respectively; see Table 6.3/11). Interestingly, different response patterns were observed within the two groups with regard to whether language maintenance would be possible without the practice of interpreting. Within the AA group, the largest proportion of respondents (45.8%) felt that it would be possible to maintain their two strongest languages at their current high standard without working as professional interpreters, however, within the AB group, the largest proportion of respondents (45.9%) felt that this would not be possible (see Table 6.3/11). This suggests that the practice of interpreting is perceived to be a significant factor in language maintenance by AIIC AB respondents, whereas there appears to be no such connection among the AA group, who tend to feel that they would be able to maintain both of their A languages at the present standard without working as interpreters.

Further, respondents from both groups were asked to indicate whether they believe that they would make the effort to maintain both of their strongest languages at the current high standard if they were not working as professional interpreters (Q4.11/AA, Q4.12/AB), in an attempt to elicit whether maintaining the two languages is important to them regardless of the demands of their profession. Additionally, and for the same purpose, AA and AB respondents were asked to indicate whether they seek to fill gaps in their linguistic knowledge when they notice them, and whether their motivation for this is purely professional (Q4.13/AA, Q4.14/AB and Q4.14/AA, Q4.15/AB respectively). It was found that more than half of respondents within both groups believe that they would strive to maintain both of their strongest working languages at the present high standard even if they were not working as professional interpreters, but this figure was slightly higher within the AA group compared with the AB group (59.6% and 50.5% respectively; see Table 6.3/11 above). It was also observed that almost all respondents within both groups seek to fill gaps in their linguistic knowledge when they notice them (95.8% AA, 98.1% AB; see Table 6.3/11). The motivation for this was not purely professional for the majority of respondents within both groups, but this figure was slightly higher within the AA group compared with the AB group (80.9% and 70.5% respectively; see Table 6.3/11).

The two groups were compared in terms of metalinguistic awareness, specifically, awareness of fluctuations in language dominance and perceptions of cross-linguistic interference between their two strongest languages. The responses to the first question are set out in Table 6.3/12 below.

Table 6.3/12. Responses of the AA and AB groups to the question of whether language dominance is perceived to fluctuate between their strongest languages

Group	Do you feel that language dominance fluctuates between your strongest languages? (freq and %)		
	Yes	No	Unsure
AA	24 (50.0%)	19 (39.6%)	5 (10.4%)
AB	79 (37.3%)	120 (56.6%)	13 (6.1%)

It was found that the largest proportion of AA respondents (50%) agreed that language dominance does fluctuate between their two strongest working languages, while this general trend was reversed within the AB group, where over half of respondents indicated that language dominance does not fluctuate between their two strongest working languages (56.6%; see Table 6.3/12). This finding is in line with expectations (see Chapter 7). Respondents were also asked to indicate the reasons they feel are responsible for such shifts in language dominance. It was found that both groups cited similar reasons, with the most popular categories among both the AA and AB groups being travel/change of country or work (interpreting), followed by reading or interacting socially in one of the languages.

Respondents were also asked to indicate whether they were aware of making unnatural utterances in either of their strongest working languages, due to the influence of the other strongest language (Q14/AA, 13/AB). This is taken to be a marker of the relative level of perceived linguistic interference in each language. The responses to this question are presented in Table 6.3/13 below.

Table 6.3/13. Comparison of responses for the AA and AB groups to the question pertaining to perceptions of non-linguistic interference between their strongest languages

Group		Do you notice yourself making unnatural utterances in either of your two strongest languages due to the influence of the other language? (freq and %)					
	In neither	Mostly in A1/A	Mostly in A2/B	In both equally	Unsure		
AA	14 (29.2%)	7 (14.6%)	5 (10.4%)	15 (31.3%)	7 (14.6%)		
AB	51 (24.3%)	33 (15.7%)	63 (30.0%)	43 (20.5%)	20 (9.5%)		

We see that the majority of AA respondents reported either making unnatural utterances in both of their strongest languages equally (31.3%) or in neither language (29.2%; see Table 6.3/13). On the other hand, the largest proportion of AB respondents reported either making unnatural utterances in language B only (30.0%) or in neither language (24.3%), followed closely by the "both languages equally" response category (20.5%; see Table 6.3/13). The findings suggest the following. First of all, taking the "neither" and "both" categories together and comparing the responses for the two groups shows that 60.5% of AA respondents chose one of these two response categories, compared with 44.8% of AB respondents. This implies that AA respondents are more likely to perceive both languages as being equal in terms of cross-linguistic interference (either equally susceptible or equally immune to it) compared with AB respondents. Comparing the responses for each of the languages separately, we see that languages A1 and A2 within the AA group are closer in terms of perceived levels of interference than languages A and B within the B group (14.6% for A1, 10.4% for A2; 15.7% for A, 30.0% for B; see Table 6.3/13).

Attention was next turned to the respondents' attitudes towards their two strongest languages and, in particular, whether there is any special attachment to one of the languages over the other among the AA and AB groups. Respondents were asked to indicate whether they have a preference for speaking in one of their two strongest languages when tired/emotionally upset (Q5.4/AA/AB), whether they feel that words carry the same "weight" in both of their strongest languages (Q5.5/AA/AB), and if they feel an attachment to one of the languages over the other (Q5.6/AA/AB). The responses to these questions are summarised in the Table 6.3/14 on the next page.

Table 6.3/14. Comparison of responses for the AA and AB groups to questions pertaining to emotional response and attachment to the languages

Group	Responses to questions on emotional response and attachment to the languages (freq and %)						
	If possible, w	If possible, would you revert to one of your two strongest languages when tired/upset?					
	No	A1/A	A2/B	Unsure			
AA	20 (41.7%)	16 (33.3%)	5 (10.4%)	7 (14.6%)			
AB	65 (31.1%)	106 (50.7%)	18 (8.6%)	20 (9.6%)			
			-	as "love" and			
AA	29 (60.4%)	9 (18.8%)	3 (6.3%)	7 (14.6%)			
AB	77 (36.8%)	76 (36.4%)	18 (8.6%)	38 (18.2%)			
	Do you feel a greater attachment towards one of your strongest languages over the other?						
	·	0	e other?				
	No	0	e other? To A2/B	Unsure			
AA		over the					

With respect to all three components of attachment considered here ("language preference when tired/upset", "relative word weight in the two languages" and "attachment to one language over the other"), the largest proportion of AA respondents consistently selected the category that corresponds to both languages being equal (41.7%, 60.4% and 50.0% respectively; see Table 6.3/14 above). Within the AB group, language A was the preferred language for the largest proportion of respondents when tired/upset (50.7%; see Table 6.3/14). However, with regard to the other two components of attachment ("word weight" and "special attachment"), the AB respondents were split fairly evenly between the "no preference" and "preference for language A" categories (36.8% and 36.4% respectively for "word weight", 37.3% and 40.2% for "special attachment"; see Table 6.3/14). These observations are in line with the hypothesis that AA interpreters are more likely to view both of their languages as being equal in terms of emotional attachment or the emotional response they illicit, compared with AB interpreters. Looking at the percentages for each of the languages separately, we see that language A1 scored consistently higher than A2 in all three categories in the AA group, and language A scored higher in all three categories than language B in the AB group (see Table 6.3/14 above). If we look back at the childhood linguistic environment results for the two groups (see Section 6.3.1 above), we see that language A1 and language A within the AA and AB groups respectively are associated more strongly with the languages the respondents were exposed to the most in childhood.

Finally, the AA and AB groups were compared in terms of perceived proficiency in each of their strongest languages, both relative to each other and relative to an average monolingual standard. In the first case, respondents were asked to rate their proficiency in their two strongest languages relative to each other in the following three categories: command of registers (Q5.7/AA/AB), relative number of gaps in vocabulary (Q5.8/AA/AB) and writing skills (Q5.9/AA/AB). The results are presented in the Table 6.3/15 below.

Table 6.3/15. Self-ratings of proficiency in the strongest languages relative to each other (by category) of the AA and AB groups

Group	Self-ratings in the two languages relative to each other across difference categories (freq and %)						
	-	at your command of re nguages, or better in o	_	ooth of your strongest he other?			
	Equal	Better in A1/A	Better in A2/B	Unsure			
AA	31 (64.6%)	8 (16.7%)	3 (6.3%)	6 (12.5%)			
AB	58 (27.6%)	140 (66.7%)	2 (1.0%)	10 (4.8%)			
	Do you have	Do you have fewer gaps in vocabulary in one of your strongest languages compared with the other?					
	Equal	Fewer in A1/A	Fewer in A2/B	Unsure			
AA	25 (52.1%)	10 (20.8%)	5 (10.4%)	8 (16.7%)			
AB	58 (27.8%)	126 (60.3%)	11 (5.3%)	14 (6.7%)			
	Do you write equally well/just as easily in both of your strongest languages?						
	Equal	Better in A1/A	Better in A2/B	Unsure			
AA	29 (60.4%)	10 (20.8%)	6 (12.5%)	3 (6.3%)			
AB	73 (35.1%)	123 (59.1%)	5 (2.4%)	7 (3.4%)			

The results showed consistent response patterns across all three proficiency categories considered within the AA and AB groups. The majority of AA respondents felt that their proficiency in both of their strongest working languages is equal with respect to command of registers, gaps in vocabulary and writing skills (64.6%, 52.1% and 60.4% respectively; see Table 6.3/15). On the other hand, the majority of AB respondents indicated that language A is their strongest language in all three categories (66.7%, 60.3% and 59.1% respectively; see Table 6.3/15). This general observation is in line with the expectation that AA respondents would be more likely to rate both languages as being equal compared with AB respondents, whereas AB respondents would tend to rate language A as being the stronger of the two languages. Considering the results for languages A1 and A2 within the AA group, we see that respondents tended to rate language A1 as stronger than A2 in all three categories (16.7% for A1, 6.3% for A2 in command of registers; 20.8% for A2, 10.4% for A2 in fewer gaps in vocabulary; 20.8% for A1 and 12.5% for A2 in writing; see Table

6.3/15). Given that language A1 within the AA group is characterised by earlier acquisition (first language spoken tended to be the A1) and greater exposure in childhood (see Section 6.3.1 above), this finding is in line with the literature on age of SLA and subsequent proficiency.

Respondents were also asked to rate their proficiency in their two strongest languages against a monolingual standard in the following three categories: "richness of vocabulary" (Q5.11/AA, Q5.10/AB), "instinctive native understanding" (Q5.12/AA, Q5.11/AB) and "pragmatic competence/cultural awareness" (Q5.13/AA, Q5.12/AB), by indicating whether they believe that they are stronger, weaker or about the same as a monolingual of roughly the same educational/cultural level in each of their strongest languages. These self-ratings relative to a monolingual standard are taken to be an indicator of language confidence. The results for the three categories set out above for the groups' two languages are presented in the Table 6.3/16 on the next page.

An examination of the results reveals several important trends. First of all, we notice that the largest portion of the AA respondents consistently choose the "same" response category for both languages, with the next most popular response category being "stronger" (although this difference is very slight in the richness of vocabulary category for both languages, where the responses are split almost equally between "same" and "stronger"; see Table 6.3/16). On the other hand, the majority of AB respondents tend to rate language A as being above the monolingual standard across all three categories (71.2% for richness of vocabulary; 63.2% for native comprehension; 64.4% for pragmatic/cultural competence; see Table 6.3/16). As regards language B, only around a third of AB respondents rate this language as being above the monolingual standard in all three categories (32.8%, 27.2% and 36.9% respectively), with the highest proportion of AB respondents indicating that the languages is equal to the monolingual standard in all three categories (51.5%, 55.9% and 44.4% respectively; see Table 6.3/16). In addition, in all three categories, the percentage of AB respondents who rate their B language as being weaker than the monolingual standard is much higher compared with the same figures for language A (see Table 6.3/16). Taken together the above-listed findings imply the following. Firstly, that AA interpreters tend to not only perceive both languages as being about the same, but both are generally regarded as being equal to the monolingual standard. Among the AB group, on the other hand, there is a tendency to rate language A as being stronger than language B, with language A regarded as being above the monolingual standard by most respondents, and language B as being about the same. Comparing the results for the AA and AB groups, it can be concluded that the two populations are characterised by two languages of about the same proficiency (which is perceived to be more or less equal to the monolingual standard) in the case of AA interpreters, and two languages with differing levels of

proficiency – one (language A) very strong language (above the monolingual standard) and one weaker language (B) – in the case of AB interpreters. If these self-ratings are viewed as markers of language confidence, the results imply that AA respondents exhibit about the same level of confidence in both languages, whereas AB respondents display greater confidence in language A compared with language B.

Table 6.3/16. Self-ratings of proficiency relative to an average monolingual standard in each language (by category) of the AA and AB groups

Group		each language relative ( cross different categories	to a monolingual standard s (freq and %)		
		Richness of vocabular	ry in A1/A		
	Weaker	Same	Stronger		
AA	3 (6.3%)	23 (47.9%)	22 (45.8%)		
AB	3 (1.4%)	57 (27.4%)	148 (71.2%)		
		Richness of vocabular	ry in A2/B		
	Weaker	Same	Stronger		
AA	1 (2.1%)	25 (52.1%)	22 (45.8%)		
AB	31 (15.7%)	102 (51.5%)	65 (32.8%)		
	Ins	Instinctive/native comprehension in A1/A			
	Weaker	Same	Stronger		
AA	2 (4.2%)	32 (66.7%)	14 (29.2%)		
AB	1 (0.5%)	74 (36.3%)	129 (63.2%)		
	Ins	Instinctive/native comprehension in A2/B			
	Weaker	Same	Stronger		
AA	0	33 (68.8%)	15 (31.3%)		
AB	33 (16.9%)	109 (55.9%)	53 (27.2%)		
	P	ragmatic/cultural compe	tence in A1/A		
	Weaker	Same	Stronger		
AA	2 (4.3%)	35 (74.5%)	10 (21.3%)		
AB	3 (1.4%)	71 (34.1%)	134 (64.4%)		
	P	ragmatic/cultural compe	etence in A2/B		
	Weaker	Same	Stronger		
AA	1 (2.1%)	35 (72.9%)	12 (25.0%)		
AB	37 (18.7%)	88 (44.4%)	73 (36.9%)		

# 6.3.3 Self-ascribed mother tongue identity vs. identity ascribed by others among the AB group

It was observed in Section 6.3.1 that 13.2% (28 respondents) of AB interpreters consider both of their strongest languages to be mother tongues, although this does not correspond to external perceptions (as represented by the AIIC AB classification). This subgroup of AB respondents was compared to the rest of the AB group in order to determine whether there is a difference between the two populations in terms of biographical and attitudinal factors. The full comparison is presented in Appendix IV, with the main trends summarised below. For convenience, the subgroup of AB respondents who consider that they have two mother tongues shall be referred to as the "AB Two MTs" group. The findings for this subgroup of AB respondents are contrasted with those observed for the rest of the AB group (referred to as the "AB Not Two MTs" group).

First of all, the "AB Two MTs" group was compared to the rest of the AB group with respect to the age of SLA. It was found that the mean age of SLA among the "AB Two MTs" group was lower than that of the rest of the AB respondents, with 4.8 and 9.2 years as the mean age of SLA for the two groups respectively (see Appendix IV (1), Table IV.1). This difference in mean ages of SLA was found to be significant, representing a medium sized effect (see Appendix IV (1) for t-statistics). Therefore, AB interpreters who self-identify as having two mother tongues have generally acquired their second language at an earlier age than those AB interpreters who feel that language A is their only mother tongue.

Turning out attention to other aspects of language acquisition and the linguistic environment in childhood, it was observed that the "AB Two MTs" group behaved as expected in comparison with the rest of the AB group. Namely, it was found that, while language A tended to dominate among both groups, this dominance was less pronounced among the "AB Two MTs" group across all childhood linguistic environment categories considered (see Appendix IV (2)). In particular, in the categories of "home", "street" and "(primary) school" identified as the major components of the childhood linguistic environment by Thiéry (see Chapter 4), language A was the main language for 50.0%, 42.9% and 57.1% of the "AB Two MTs" respondents respectively, and for 95.0%, 77.0% and 81.8% of the rest of the AB respondents in the respective categories (see Appendix IV (2), Table IV.2). In addition, a particularly large difference was observed between the number of respondents in each group who reported that both languages were spoken in the home during childhood: 28.6% of the "AB Two MTs" group indicated that this was the case, compared with just 2.8% of the rest of the AB group (see Appendix IV (2), Table IV.2). This figure for the "AB Two MTs" group is comparable to that observed among the AA interpreters (see Section 6.3.1 above). It was also noted that a higher percentage of respondents from the "AB Two MTs" group relocated to

a different country in childhood compared with the rest of the AB group (53.6% and 22.3% respectively; see Appendix IV (2), Table IV.3).

The groups were next compared in terms of language use and exposure in later life. With regard to higher education, the findings did not suggest that exposure to both languages in this context is connect with the two languages being considered mother tongues. In fact, a higher proportion of AB respondents who do not consider both languages to be mother tongues were found to have had both A and B as their languages of higher education, compared with the same figure for the "AB Two MTs" group (54.0% and 26.9% respectively; see Appendix IV (3), Table IV.4). A comparison of overall daily language use/exposure revealed that a higher proportion of respondents from the "AB Two MTs" group reported being exposed to both languages equally on a daily basis, compared with the same figure among the rest of the AB group (42.9% and 28.5% respectively; see Appendix IV (4), Table IV.5). However, this finding does not necessarily imply a greater effort on the part of the "AB Two MTs" to use both languages in their daily lives, as proposed in Chapter 5, and could be related to other, practical factors (country of residence, work etc.) Indeed, a comparison of selfreported daily language use across various categories (reading, writing, TV, radio, socialising, internet) appears to contradict this hypothesis, since it was found that in all cases except for socialising and radio a lower percentage of the "AB Two MTs" group used both of their languages equally compared with the rest of the AB group (see Appendix IV (4), Table IV.6).

With respect to language use in a professional context, the response patterns were very similar for the two groups in the case of interpreting, although respondents from the "AB Two MTs" group were more evenly split between those who mostly interpreter into language A and those who mostly interpret into language B compared with the same figures for the rest of the AB group (46.4% for A, 35.7% for B in the "AB Two MTs" group; 61.5% for A, 25.1% for B in the rest of the group; see Appendix IV (5), Table IV.7). The results for preferred interpreting direction were very similar for both groups, with around 60% of both expressing a preference for language A, and the rest spread more or less evenly between the "language B" or "no preference" response categories, although it was observed that slightly more AB respondents who do not feel that they have two mother tongues chose this latter category compared with the "AB Two MTs" group (23.3% and 18.5% respectively; see Appendix IV (5), Table IV.7). However, as previously noted, interpreting preferences are not necessarily directly related to bilingualism. The two groups differed in terms of their perceptions of the relative importance of interpreting vs linguistic skill, with a lower proportion of "AB Two MTs" respondents choosing interpreting skill compared with the rest of the AB group (28.6% and 44.9% respectively), and a higher proportion of "AB Two MTs" respondents choosing linguistic skill compared with the rest of the group (21.4% and 9.1%

respectively; see Appendix IV (5), Table IV.7). The number of respondents who also carry out translation work was similar within both groups, although the percentage of respondents who translate into both languages was higher within the "AB Two Mts" group than in the rest of the AB group (68.2% and 48.4% respectively; see Appendix IV (5), Table IV.8).

In terms of attitudes and perceptions, the two groups were first compared with respect to perceptions of language maintenance effort and motivation. With regard to perceptions of language maintenance effort, the majority of respondents from both groups felt that maintaining their strongest languages requires a degree of conscious effort, although this figure was slightly lower among the "AB Two MTs" respondents (67.9% for the "AB Two MTs" group, 74.4% for the rest of the AB group; see Appendix IV (6), Table IV.10). As regards non-professional motivation to maintain the two languages, it was found that the largest proportion of respondents from both groups stated that they would continue to make the effort to maintain both languages at their current high standard if not working as professional interpreters (44.4% of the "AB Two MTs" group, 51.7% of the rest of the group), and that their motivation for filling gaps in linguistic knowledge is not purely professional (78.6% of the "AB Two MTs" group, 69.5% of the rest of the group see Appendix IV (6), Table IV.10). However, we note that the percentage was lower among the "AB Two MTs" group compared with the rest of the group in the first instance, but higher in the second case. The two groups differed most substantially in their responses to the question of whether they believe that it would be possible to maintain both languages at their present high standard without interpreting, with the greatest proportion of respondents from the "AB Two MTs" group feeling that it would (42.9%), while the greatest proportion of the rest of the AB group felt that it would not be possible (48.3%) (see Appendix IV (6), Table IV.10).

Looking at the perceptions of fluctuations in language dominance for the two groups, response patterns were similar, with just over half of respondents within both groups reporting that language dominance does not fluctuate between their two languages (see Appendix IV (7), Table IV.11). However, the percentage of respondents who did feel that language dominance fluctuates was higher among the "AB Two MTs" group compared with the rest of the AB group (42.9% and 36.9% respectively; see Appendix IV (7), Table IV.11). With regard to perceptions of cross-linguistic interference, the two groups exhibited fairly similar response patterns, although respondents from the "AB Two MTs" group were more likely to rate both languages as being equal with respect to interference compared with the rest of the AB group (28.6% of "AB Two MTs" and 23.7% of the rest of the group reported that interference does not occur in either language, while 28.6% of "AB Two MTs" and 19.8% of the rest of the group felt that interference occurs in both languages equally; see Appendix IV (7), Table IV.12). Comparing interference levels for languages

A and B separately it was found that both groups perceived interference levels as being greater in language B compared with language A (see Appendix IV (7), Table IV.12).

Comparing the responses for the two groups with regard to the questions concerning emotional factors and attachment to the two languages, some differences were observed. It was found that, while just under a third of respondents from both groups indicated that they would not revert to one of their languages when tired/upset if given the choice, a considerably larger percentage of AB respondents who do not feel that they have two mother tongues would revert to language A as opposed to language B (52.8% and 7.4% respectively), whereas the figures for the two languages were more evenly matched within the "AB Two MTs" group (35.7% and 17.9% respectively; see Appendix IV (8), Table IV.13). In terms of perceptions of word weight in each language, a greater percentage of "AB Two MTs" respondents considered this to be equal in both languages, compared with the same figure for the rest of the AB group (53.6% and 34.1% respectively), while a greater proportion of AB respondents who do not feel that they have two mother tongues considered words to have more weight in the A languages, compared with the "AB Two MTs" group (40.3% and 14.3% respectively; see Appendix IV (8), Table IV.13). Finally, with respect to emotional attachment, the largest proportion of "AB Two MTs" respondents (44.4%) reported that they do not feel a special attachment towards one of their languages over the other (the corresponding figure for the rest of the group is 36.7%), while the largest proportion of AB respondents who do not have two mother tongues (44.1%) stated that they feel a special attachment towards language A (the figure for the "AB Two MTs" group is 18.5%; see Appendix IV (8), Table IV.13).

Respondents were also compared in terms of self-ratings of proficiency in each language, relative to each other and relative to a monolingual standard. Similar overall trends were observed for the two groups in their responses to the questions concerning relative proficiency in each language. In all three categories considered – command of registers, least amount of gaps in vocabulary, writing – the largest percentage of respondents from both groups felt that language A is the stronger language (see Appendix IV (9), Table IV.14). However, the percentage of respondents who considered both languages to be equal was higher in all three categories within the "AB Two MTs" group compared with the rest of the AB group (35.7% and 27.1% respectively for command of registers; 39.3% and 25.6% for gaps in vocabulary; 39.3% and 34.3% for writing; see Appendix IV (9), Table IV.14). This suggests that, while both groups tend to feel that language A is their strongest language (compared with language B), those AB respondents who feel that they have two mother tongues are slightly more likely to regard both languages as being equal, compared with those AB interpreters who have one mother tongue. In terms of self-ratings of proficiency in each language relative to a monolingual standard, two main differences between the groups can be

observed. Firstly, the findings imply that AB interpreters who have one mother tongue feel a greater confidence in language A compared with the "AB Two MTs" group in the categories of richness of vocabulary and native comprehension. This is suggested by the fact that a greater proportion of AB respondents with one mother tongue tended to rate language A as being above the monolingual standard in these categories, compared with respondents from the "AB Two MTs" group (see Appendix IV (9), Table IV.15). As regards language B, the response patterns for the two groups in both categories were very similar, with the largest proportion of both rating this language as being equal to the monolingual standard (see Appendix IV (9), Table IV.15). However, slightly more respondents in the "AB Two MTs" group rated language B as being above the monolingual standard in both of these categories compared with the rest of the AB group, suggesting that confidence in language B is slightly higher among the "AB Two MTs" group. With regard to pragmatic/cultural competence, a larger percentage of "AB Two MTs" respondents rated both languages as being above the monolingual standard, compared with the same figure among the rest of the group (see Appendix IV (9), Table IV.15). This implies that, with respect to pragmatic/cultural competence, "AB Two MTs" respondents feel more confidence in both languages overall compared with the rest of the AB group. However, confidence in language A was higher than in language B within both groups in this category.

# 6.4 Description of the PSI group and comparison with the AIIC AB group

This final section of the present Chapter presents the findings for the PSI groups, comparing and contrasting the responses of this population of interpreters with those given by the AIIC AB group. The format followed is the same as in the case of the AA/AB comparison presented earlier. While the findings for the AB group have already been presented in Section 6.3, they are reproduced here for convenience so that they can be more readily compared with the responses of the PSI group.

# 6.4.1 Mother tongue identity, biographical factors and language use

It was first explored whether PSI respondents consider both of their strongest working languages (L1, L2) to be mother tongues, or if they feel that they have one mother tongue only. The responses to this question are presented in the Table 6.4/1 below, together with those for the AB group.

Table 6.4/1. Responses of the PSI and AB groups to the question of which languages they consider to be mother tongues

Group	Mother tongue				
	L1/A	L2/B	Both	Other	
PSI	156 (86.7%)	0	23 (12.8%)	1 (0.6%)	
AB	179 (84.4%)	0	28 (13.2%)	5 (2.4%)	

We see that the overwhelming majority consider L1 to be their mother tongue (86.7% of respondents), with only 12.8% identifying as having two mother tongues (see Table 6.4/1). These findings are almost identical to the results observed for the AB group (see Table 6.4/1). Therefore, the PSI group can be characterised, in general, as having one distinct mother tongue, the L1, and a second foreign language (L2). Looking ahead to Table 6.4/3 below, we also see that L1 was the first language spoken for 95.0% of PSI respondents. Thus, similarly for the A language within the AB group, the L1 in the PSI group can be taken to be the first language spoken and the mother tongue.

The average (mean) age of SLA for the PSI group was next considered, and the findings are compared with those for the AB group in the Table 6.4/2 below.

Table 6.4/2. Comparison of mean age of SLA in years for the PSI and AB groups

Group	No. in group	Mean age SLA (years)	Std. Deviation
PSI	178	11.0	6.134
AB	205	8.6	4.889

We see that the mean age of SLA for the PSI group is slightly higher at 11 years than that of the AB group (8.6 years). The range of age of SLA was the same for both groups, from 0 to 30 years. An independent samples t-test showed that this difference in mean ages of SLA is significant, but represents a fairly small effect: t(337.111) = -4.310, p = .000; r = .229. It can therefore be concluded that PSI interpreters tend to have acquired their second language at a slightly higher age than conference interpreters. However, we note that both groups generally acquired their second language before puberty.

Looking at the childhood linguistic environment of the PSI group more generally, Table 6.4/3 on the next page shows the relative exposure to each of the languages in childhood across the difference components of childhood linguistic environment considered in the questionnaire, for both the PSI and AB groups. The results revealed strikingly similar response patterns for the PSI and AB groups, with language L1/A dominating as the language each population of interpreters was exposed to the most in all components of childhood linguistic environment, although this dominance was slightly less pronounced in the AB group, with the exception of primary and secondary education, where this was reversed (see Table 6.4/3). In particular, for the main components of childhood linguistic environment of "home", "street" and "(primary) school", L1 was the language mostly spoken at home during childhood for 93.9% of PSI respondents (compared with 87.3% for language A within the AB group), L1 was the dominant language of the country of residence in childhood for 91.1% of PSI respondents (compared with 72.0% for language A within the AB group), and L1 was the language of primary education for 78.3% of PSI respondents (compared with 78.5% for language A within the AB group; see Table 6.4/3). The response patterns for languages of secondary education were found to be nearly identical for the two groups, with around two thirds of each indicating that L1/A was the dominant language (60.6% for PSI, 66.5% for AB), around 20% of each group reported that both languages featured as the languages of secondary education, followed by language B, which was selected by a little over 10% of respondents within each group (see Table 6.4/3)

Table 6.4/3. Comparison of childhood linguistic environment (by category) and language(s) of primary and secondary education for the PSI and AB groups

Group	Childhood lin	Childhood linguistic environment/education category and responses (freq and %)			
	First language(s) spoken				
	L1/A	L2/B	Both	Other	
PSI	171 (95.0%)	0	3 (1.7%)	5 (2.8%)	
AB	183 (86.7%)	14 (6.6%)	4 (1.9%)	10 (4.7%)	
		Mothe	er's language(s)		
	L1/A	L2/B	Both	Other	
PSI	168 (93.3%)	3 (1.7%)	1 (0.6%)	8 (4.4%)	
AB	171 (80.7%)	20 (9.4%)	4 (1.9%)	17 (8.0%)	
		Fathe	r's language(s)		
	L1/A	L2/B	Both	Other	
PSI	169 (93.9%)	3 (1.7%)	0	8 (4.4%)	
AB	179 (85.2%)	11 (5.2%)	8 (3.8%)	12 (5.7%)	
	Lang	uage(s) mostly spol	ken in the home dur	ing childhood	
	L1/A	L2/B	Both	Other	
PSI	169 (93.9%)	2 (1.1%)	3 (1.7%)	6 (3.3%)	
AB	185 (87.3%)	7 (3.3%)	13 (6.1%)	7 (3.3%)	
		Childhood linguis	stic environment (co	ountry)	
	L1/A	L2/B	Both	Other/Unclear	
PSI	164 (91.1%)	7 (3.9%)	7 (3.9%)	2 (1.1%)	
AB	152 (72.0%)	4 (1.9%)	14 (6.6%)	41 (19.4%)	
		Language(s)	of primary education	on	
	L1/A	L2/B	Both	Other	
PSI	141 (78.3%)	13 (7.2%)	14 (7.8%)	4 (2.2%)	
AB	164 (78.5%)	13 (6.2%)	27 (12.9%)	5 (2.4%)	
		Language(s)	of secondary educat	ion	
	L1/A	L2/B	Both	Other	
PSI	109 (60.6%)	21 (11.7%)	35 (19.4%)	5 (2.8%)	
AB	139 (66.5%)	26 (12.4%)	42 (20.1%)	2 (1.0%)	

Respondents were also asked to indicate whether they relocated to a different country in childhood at any point. We see from Table 6.4/4 below that the overall trend is similar for both the PSI and AB groups, with the majority of both indicating that they did not relocate to a different country in childhood (87.2% among PSI respondents, 73.6% among AB respondents), although the number of respondents who did relocate in childhood is higher within the AB group compared with the PSI group.

Table 6.4/4. Comparison of responses of the PSI and AB groups to the question of whether they relocated to a different country in childhood

Group	Did you relocate to a different country during childhood? (freq and %)		
	Yes	No	
PSI	21 (11.7%)	157 (87.2%)	
AB	56 (26.4%)	156 (73.6%)	

It can therefore be concluded that the childhood linguistic environment of both PSI and AB interpreters is comparable in terms of the relative exposure to each of their strongest languages in the components of childhood linguistic environment considered in the questionnaire. The main differences identified between the groups include a slightly higher age of SLA among the PSI respondents, a slightly less pronounced dominance of the first language (A/L1) within the AB group, and a higher proportion of respondents who relocated to a different country in childhood within the AB group compared with the PSI group.

Considering next the languages of higher education of the PSI group, it was found that respondents were split almost equally between L1, L2 and both languages (33.9%, 27.8% and 31.7% respectively; see Table 6.4/5 below). This pattern differs from that observed for the AB group, just over half of which reported that both languages were their languages of higher education, with a third selecting language A and a little over 15% language B (see Table 6.4/5 below).

Table 6.4/5. Comparison of language(s) of higher education for the PSI and AB groups

Group		Language(s) of higher education (freq and %)				
	L1/A	L2/B	Both	Other		
PSI	61 (33.9%)	50 (27.8%)	57 (31.7%)	4 (2.2%)		
AB	64 (31.2%)	34 (16.6%)	104 (50.7%)	3 (1.5%)		

The dominant linguistic environment of the country of residence, as well as the language(s) respondents feel that they are exposed to or use the most on a daily basis, was also considered, and these results are presented in Table 6.4/6 below.

Table 6.4/6. Comparison of present linguistic environment for the PSI and AB groups

Group		and %)			
		Dominant language	e(s) of country of re	esidence	
	L1/A	L2/B	Both	Other	
PSI	23 (12.8%)	153 (85.0%)	4 (2.2%)	0	
AB	97 (49.2%)	50 (25.4%)	21 (10.7%)	29 (14.7%)	
	Language(s) exposed to the most on a daily basis				
	L1/A	L2/B	Both	Other	
PSI	42 (23.3%)	110 (61.1%)	26 (14.4%)	1 (0.6%)	
AB	98 (46.2%)	41 (19.3%)	66 (31.1%)	7 (3.3%)	

As expected on the basis of what is known about PSIs, the overwhelming majority of respondents from this group indicated that the L2 (i.e. English) is currently the dominant language of their country of residence (85.0%; see Table 6.4/6). This is reflected in the next figure of perceived degree of exposure to each language, with the majority of PSI respondents (61.1%) reporting that they are exposed predominantly to the L2 on a daily basis (see Table 6.4/6). The responses to these questions for the AB group, as we saw earlier in Section 6.3, are more spread out, and thus the AB group cannot be characterised as neatly as the PSI group in terms of a dominant linguistic environment of the country of residence.

The PSI respondents' daily language use across difference categories was also considered, and the results are presented in Table 6.4/7 on the next page, together with those for the AB group for comparison.

We see that the majority of PSI respondents report using both languages equally on a daily basis in the categories of reading (71.7%), writing (60.0%), socialising (71.1%) and the internet (68.3%) (see Table 6.4/7). These trends are in line with the findings for the AB group (see Table 6.4/7). Language L2 was the most popular choice in the categories of TV and radio (43.3% and 47.8% respectively) among the PSI respondents, which is likely a reflection of the dominant linguistic environment of the country of residence. Therefore, it can be concluded that, while the language of the country of residence certainly affects language use and exposure, PSI interpreters tend to use both languages on a daily basis for a variety of activities.

Table 6.4/7. Comparison of self-reported daily language use (by category) for the PSI and AB groups

Group	La	nguage use categor	y and responses (fro	eq and %)		
			Reading			
	L1/A	L2/B	Both	Neither		
PSI	14 (7.8%)	34 (18.9%)	129 (71.7%)	0		
AB	17 (8.0%)	18 (8.5%)	175 (82.5%)	2 (0.9%)		
			Writing			
	L1/A	L2/B	Both	Neither		
PSI	19 (10.6%)	49 (27.2%)	108 (60.0%)	1 (0.6%)		
AB	40 (18.9%)	9 (4.2%)	160 (75.5%)	3 (1.4%)		
		Wa	ntching TV			
	L1/A	L2/B	Both	Neither		
PSI	22 (12.2%)	78 (43.3%)	66 (36.7%)	11 (6.2%)		
AB	32 (15.1%)	34 (16.0%)	122 (57.5%)	24 (11.3%)		
	Listening to the radio					
	L1/A	L2/B	Both	Neither		
PSI	24 (13.3%)	86 (47.8%)	54 (30.0%)	13 (7.2%)		
AB	38 (17.9%)	60 (28.3%)	28 (13.2%)	38 (17.9%)		
		Socialising				
	L1/A	L2/B	Both	Neither		
PSI	20 (11.1%)	25 (13.9%)	128 (71.1%)	4 (2.3%)		
AB	47 (22.2%)	20 (9.4%)	136 (64.2%)	9 (4.2%)		
		Using	g the internet			
	L1/A	L2/B	Both	Neither		
PSI	10 (5.6%)	41 (22.8%)	123 (68.3%)	3 (1.7%)		
AB	16 (7.5%)	28 (13.2%)	155 (73.1%)	13 (6.1%)		

Language use in the context of interpreting and translation among the PSI group was also considered. The responses to the questions pertaining to interpreting are summarised in Table 6.4/8 below.

Table 6.4/8. Comparison of responses for the PSI and AB groups to questions pertaining to interpreting

Group	Responses to questions concerning interpreting (freq and %)				
	Language(s) you mostly interpret into:				
	L1/A	L2/B	Both		
PSI	76 (42.2%)	43 (23.9%)	55 (30.6%)		
AB	127 (59.9%)	56 (26.4%)	29 (13.7%)		
	Preferred interpreting direction				
	L1/A	L2/B	No preference		
PSI	56 (31.1%)	37 (20.6%)	76 (42.2%)		
AB	127 (62.3%)	32 (15.7%)	45 (22.1%)		
	What is more important: interpreting skill or linguistic skill?				
	Interpreting skill	Linguistic skill	Both equally important		
PSI	69 (38.3%)	35 (19.4%)	73 (40.6%)		
AB	90 (43.1%)	24 (11.5%)	95 (45.5%)		

While the largest proportion of PSI respondents reported interpreting into the L1 the most (42.2%), the number who indicated that that interpret into both languages equally was not far behind (30.6%)(see Table 6.4/8. In particular, the percentage of PSI respondents who interpret into both languages is greater than that for the AB group (13.7%; see Table 6.4/8). In terms of preferred interpreting direction, the largest proportion of PSI respondents reported that they have no preference (42.2%), with just under a third expressing a preference for the L1 (31.1%; see Table 6.4/8). Compared with the AB group, we see that the percentage of PSI respondents who do not have a preference when it comes to interpreting direction is greater (the corresponding figure for AB is 22.1%; see Table 6.4/8). With regard to perceptions of the relative importance of interpreting and linguistic skill in interpreting, the responses of the PSI group followed the same trends as those observed within the AB group. Specifically, the largest proportion of respondents felt that both were equally important (40.6%), followed closely by interpreting skill (38.3%), with linguistic skill the least popular choice (19.4%; see Table 6.4/8). However, the percentage of PSI interpreters who felt that linguistic skill is most important is almost twice as high as the same figure within the AB group (11.5%; see Table 6.4/8).

The practice of translation was next considered and, as can be seen from Table 6.4/9 below, almost all PSI interpreters also work as translators.

Table 6.4/9. Number of respondents from the PSI and AB groups who also carry out translation work

Group	Do you do any translation work? (freq and %)		
	Yes	No	
PSI	165 (91.7%)	15 (8.3%)	
AB	149 (70.6%)	62 (29.4%)	

The breakdown of the languages that respondents mostly translate into is presented in the Table 6.4/10 below.

Table 6.4/10. Languages respondents from the PSI and AB groups mostly translate into

Group	Language(s) you mostly translate into (freq and %):			
	L1/A	L2/B	Both	
PSI	59 (32.8%)	11 (6.1%)	95 (52.8%)	
AB	69 (46.6%)	4 (2.7%)	75 (50.7%)	

The responses for the PSI group were found to be similar to those observed for the AB group, with just over half of respondents reporting that they translate into both languages (52.8%), followed by L1 only (32.8%), with only a small proportion translating exclusively into the L2 (6.1%; see Table 6.4/10). Again, these findings are perhaps surprising, as they appear to go against the widely-held belief in the field of translation that one is expected to translate into the native language only.

# 6.4.2 Attitudes, perceptions and self-ratings

The responses of the PSI group to the questions concerning awareness of language maintenance effort and motivation for maintaining both languages were compared with the responses given by the AB group. As can be seen from Table 6.4/11 on the next page, the two groups' responses to four of the five questions were strikingly similar.

Table 6.4/11. Comparison of responses for the PSI and AB groups to questions pertaining to language maintenance effort and motivation

Group	Responses to ques	Responses to questions on language maintenance effort and motivation (freq and $\%)$				
	Do you feel that maintaining your two strongest languages at their present high standard requires a degree of conscious effort?					
	Yes	No	Unsure			
PSI	134 (74.4%)	39 (21.7%)	7 (3.9%)			
AB	154 (73.7%)	42 (20.1%)	13 (6.2%)			
		Would you make the effort to maintain both of your strongest languages at their present high standard if not working as a professional interpreter?				
	Yes	No	Unsure			
PSI	99 (55.0%)	47 (26.1%)	33 (18.3%)			
AB	105 (50.5%)	57 (27.4%)	46 (22.1%)			
		Would it be possible to maintain both of your strongest languages at their present high standard if not working as a professional interpreter?				
	Yes	No	Unsure			
PSI	62 (34.4%)	85 (47.2%)	33 (18.3%)			
AB	65 (31.1%)	96 (45.9%)	48 (23.0%)			
	Do you seek to fill	Do you seek to fill gaps in your linguistic knowledge when you notice them?				
	Yes	No	Unsure			
PSI	175 (97.2%)	3 (1.7%)	2 (1.1%)			
AB	206 (98.1%)	1 (0.5%)	3 (1.4%)			
	Is your motivation for filling gaps in linguistic knowledge purely professional?					
	Yes	No	Unsure			
PSI	55 (30.6%)	111 (61.7%)	11 (6.1%)			
AB	38 (18.1%)	148 (70.5%)	24 (11.4%)			

It was found that perceptions of language maintenance efforts among PSI respondents mirrored the responses of the AB group, with the majority of both groups indicating that language maintenance requires a degree of conscious effort (74.4% for PSI, 73.7% for AB; see Table 6.4/11). Similar proportions of respondents within the two groups felt that it would be possible to maintain both languages at their current high standard without the practice of interpreting (34.4% for PSI, 31.1% for AB; see Table 6.4/11) and, similarly to the AB group, almost all PSI respondents reported that they seek to fill gaps in their linguistic knowledge when they notice them (97.2%). The greatest differences between the two groups were observed with regard to the nature of their motivation to maintain their languages (i.e. professional or otherwise). While the responses of the two groups to the question of whether they would continue to make the effort to maintain both languages at their present high standard if not working as interpreters were similar, with just over half of both

indicating that they would (55.0% for PSI, 50.5% for AB; see Table 6.4/11) and a little over a quarter that they would not (26.1% for PSI, 27.4% for AB; see Table 6.4/11), we note that the percentage of PSI respondents who would make this effort is slightly higher than that observed among the AB group. On the other hand, a substantially larger proportion of PSI respondents than AB respondents stated that their motivation for filling gaps in linguistic knowledge was purely professional (30.6% and 18.1% respectively; see Table 6.4/11), and a lower percentage of PSI than AB respondents indicated that it was not (61.7% and 70.5% respectively; see Table 6.4/11). It is therefore difficult to reach a conclusion on the basis of these results as to the relative levels of non-professional language maintenance motivation within the two groups.

Next, the PSI group's responses to the questions concerning awareness of dominance shifts between their two strongest languages are set out in Table 6.4/12 below, together with the findings for the AB group.

Table 6.4/12. Responses of the PSI and AB groups to the question of whether language dominance is perceived to fluctuate between their strongest languages

Group	Do you feel that language dominance fluctuates between your two strongest languages? (freq and %)			
	Yes	No	Unsure	
PSI	82 (45.6%)	76 (42.2%)	22 (12.2%)	
AB	79 (37.3%)	120 (56.6%)	13 (6.1%)	

It was found that the PSI respondents were split almost equally between those who felt that language dominance fluctuates between their two strongest languages (45.6%) and those who feel that it does not (42.2%; see Table 6.4/12). Comparing the results for the PSI group with those observed for the AB group, we see that the percentage of PSI respondents who feel that language dominance does fluctuate is slightly higher compared with the same figure within the AB group (37.3% for AB; see Table 6.4/12). Given that the childhood linguistic environment conditions for the two groups are very similar (see Section 6.4.1 above), this difference may be due to the influence of the dominant linguistic environment, which is overwhelmingly the L2 in the case of the PSI group (see Chapter 7, Section 7.4 for details).

Differences were also observed between the PSI and AB groups with respect to perceptions of cross-linguistic interference between their two strongest languages. The responses for both groups are summarised in Table 6.4/13 below.

Table 6.4/13. Comparison of responses for the PSI and AB groups to the question pertaining to perceptions of non-linguistic interference between their strongest languages

Group		Do you notice yourself making unnatural utterances in either of your two strongest languages due to the influence of the other language? (freq and %)				
	In neither	Mostly in L1/A	Mostly in L2/B	In both equally	Unsure	
PSI	52 (28.9%)	55 (30.6%)	28 (15.6%)	30 (16.7%)	14 (7.8%)	
AB	51 (24.3%)	33 (15.7%)	63 (30.0%)	43 (20.5%)	20 (9.5%)	

In particular, while comparable proportions of respondents within each group felt that interference occurs in neither language (28.9% for PSI, 24.3% for AB) or both languages equally (16.7% for PSI, 20.5% for AB), the trend was reversed for perceptions of interference in each of the languages separately (see Table 6.4/13). While 30.6% of PSI respondents felt interference occurs mostly in the L1, and only 15.6% reported that it occurs mostly in the L2, within the AB group the opposite trend was observed: 15.7% of respondents reported greater interference in language A, while 30.0% felt that it occurred mostly in language B (see Table 6.4/13). Once again, this observation could be a manifestation of the influence of the dominant linguistic environment on the bilingualism of the PSI group (see Chapter 7, Section 7.4).

The responses of the PSI group to the questions pertaining to emotional attachment/attitudes towards their languages were considered next, and the results are summarised in Table 6.4/14 on the next page, together with those for the AB group. First of all, it was observed that the PSI group exhibited very similar response patterns to the AB group in all the three questions in this category. In particular, it was found that the largest proportion (45.6%) of PSI respondents would revert to L1 when tired/emotionally upset, with just under a third (29.4%) indicating that they have no preference (the corresponding figures for AB are 50.7% and 31.1%; see Table 6.4/14). In terms of perceptions of word weight in the two languages, the majority of PSI respondents were split more or less evenly between those who felt that words in the L1 carry more weight (38.3%) and those who perceived this as being equal in both languages (35.0%), with the corresponding figures for the AB group 36.4% and 36.8% respectively (see Table 6.4/14). With regard to attachment to the languages, the largest percentage of PSI respondents reported feeling a special attachment towards the L1 (44.4%), with the next largest group indicating that they do not feel an attachment to one of their languages over the other (36.7%); the corresponding figures for the AB group are 40.2% and 37.3% (see Table 6.4/14). The significance of these findings is discussed in the context of the literature in Chapter 7, Section 7.4.2.

Table 6.4/14. Comparison of responses for the PSI and AB groups to questions pertaining to emotional response and attachment to the languages

Group	Responses to questions on emotional response and attachment to the languages (freq and %)  If possible, would you revert to one of your two strongest languages when tired/upset?				
	No	L1/A	L2/B	Unsure	
PSI	53 (29.4%)	82 (45.6%)	25 (13.9%)	19 (10.6%)	
AB	65 (31.1%)	106 (50.7%)	18 (8.6%)	20 (9.6%)	
	Do you feel that words carry the same weight in your two strongest languages (in particular, swear words and words such as "love" and "hate")?				
	Equal	L1/A more weight	L2/B more weight	Unsure	
PSI	63 (35.0%)	69 (38.3%)	21 (11.7%)	27 (15.0%)	
AB	77 (36.8%)	76 (36.4%)	18 (8.6%)	38 (18.2%)	
	Do you feel a greater attachment towards one of your strongest languages over the other?				
	No	To L1/A	To L2/B	Unsure	
PSI	66 (36.7%)	80 (44.4%)	19 (10.6%)	14 (7.8%)	
AB	78 (37.3%)	84 (40.2%)	26 (12.4%)	21 (10.0%)	

Finally, the responses of the PSI group to the questions concerning self-ratings of proficiency in their languages were considered, and compared to the responses given by the AB group. Table 6.4/15 on the next page sets out the results for proficiency self-ratings in each language relative to each other, in the categories of command of registers, gaps in vocabulary and writing.

Table 6.4/15. Self-ratings of proficiency in the strongest languages relative to each other (by category) of the PSI and AB groups

Group	Self-ratings in the two languages relative to each other across difference categories (freq and %)  Do you feel that your command of registers is equal in both of your strongest languages, or better in one language than the other?					
	Equal	Better in L1/A	Better in L2/B	Unsure		
PSI	69 (38.3%)	74 (41.1%)	24 (13.3%)	13 (7.2%)		
AB	58 (27.6%)	140 (66.7%)	2 (1.0%)	10 (4.8%)		
	Do you have fewer gaps in vocabulary in one of your strongest languages compared with the other?					
	Equal	Fewer in L1/A	Fewer in L2/B	Unsure		
PSI	35 (19.4%)	88 (48.9%)	35 (19.4%)	18 (10.0%)		
AB	58 (27.8%)	58 (27.8%) 126 (60.3%) 11 (5.3%) 14 (6.7%)				
	Do you write equally well/just as easily in both of your strongest languages?					
	Equal	Better in L1/A	Better in L2/B	Unsure		
PSI	75 (41.7%)	73 (40.6%)	25 (13.9%)	6 (3.3%)		
AB	73 (35.1%)	123 (59.1%)	5 (2.4%)	7 (3.4%)		

With regard to command of registers, we see that PSI respondents were split almost evenly between those who felt that the L1 is stronger (41.1%) and those who felt that both languages are equal (38.3%), with a much smaller percentage considering the L2 to be the strongest language (13.3%; see Table 6.4/15). In terms of gaps in vocabulary, PSI respondents generally felt that that L1 is stronger in this regard (48.4%), with an equal number of respondents (19.4%) indicating that either both languages are equal or that they have fewer gaps in language L2 (see Table 6.4/15). In the writing category, the response patterns were very similar for those observed for command of registers, with the majority of PSI respondents split almost equally between those who felt that the L1 is stronger (40.6%) and those who considered both languages to be equal (41.7%), with a minority choosing L2 as their stronger language in writing (13.9%; see Table 6.4/15). Overall, these findings suggest that PSI respondents generally feel that the L1 is stronger than the L2, however, a substantial proportion of the group consider both languages to be equal, in particular in terms of command of registers and writing. Comparing the responses of the PSI group with those of the AB group, we observe that there is a greater distance in relative proficiency scores for languages A and B in the AB group compared with L1 and L2 in the PSI group (see Table 6.4/15). Specifically, in all three categories considered, more respondents within the AB group felt that language A was their strongest languages, compared with the figures for L1 within the PSI group; fewer respondents within the AB group gave preference to language B compared with the same figures for the L2 within the PSI group (see Table 6.4/15). These findings suggest that the PSI

respondents perceive their two strongest languages to be closer in terms of proficiency compared with the AB group, with respect to the categories considered. This finding may be a manifestation of the influence of the L2 as the dominant linguistic environment of the PSI group (see Chapter 7, Section 7.4.2).

Table 6.4/16 below presents the results for the PSI and AB groups to the questions concerning self-ratings in each language relative to a monolingual standard.

Table 6.4/16. Self-ratings of proficiency relative to an average monolingual standard in each language (by category) of the PSI and AB groups

Group	Self-ratings in each language relative to a monolingual standard across different categories (freq and %)						
	Richness of vocabulary in L1/A						
	Weaker	Same	Stronger				
PSI	12 (6.7%)	68 (37.8%)	93 (51.7%)				
AB	3 (1.4%)	57 (27.4%)	148 (71.2%)				
		Richness of vocabular	ry in L2/B				
	Weaker	Same	Stronger				
PSI	34 (18.9%)	78 (43.3%)	52 (28.9%)				
AB	31 (15.7%)	102 (51.5%)	65 (32.8%)				
	In	Instinctive/native comprehension in L1/A					
	Weaker	Same	Stronger				
PSI	6 (3.3%)	73 (40.6%)	96 (53.5%)				
AB	1 (0.5%)	74 (36.3%)	129 (63.2%)				
	In	Instinctive/native comprehension in L2/B					
	Weaker	Same	Stronger				
PSI	37 (20.6%)	90 (50.0%)	38 (21.1%)				
AB	33 (16.9%)	109 (55.9%)	53 (27.2%)				
	Pragmatic/cultural competence in L1/A						
	Weaker	Same	Stronger				
PSI	12 (6.7%)	70 (38.9%)	94 (52.2%)				
AB	3 (1.4%)	71 (34.1%)	134 (64.4%)				
	Pragmatic/cultural competence in L2B						
	Weaker	Same	Stronger				
PSI	28 (15.6%)	77 (42.8%)	57 (31.7%)				
AB	37 (18.7%)	88 (44.4%)	73 (36.9%)				

The following general observations can be made about the PSI group on the basis of the above responses. The findings for the L1 in all three categories - richness of vocabulary, native comprehension, pragmatic/cultural competence - are very similar, with just over half of PSI respondents rating the L1 as stronger than the monolingual standard (51.7%, 53.5% and 52.2% in the three categories respectively), and around 40% rating this language as being equal to the monolingual standard in each category (37.8%, 40.6% and 38.9% respectively; see Table 6.4/16). Only a small minority of PSI respondents (a maximum of 6.7%) felt that the L1 was weaker than the monolingual standard in any of the categories (see Table 6.4/16). The response patterns for the L2 were similar to each other in all three categories considered, but differed to those observed for the L1. In this case, the greatest proportion of respondents felt that the L2 is equal to the monolingual standard (51.5%, 55.9% and 42.8% in the three categories respectively), with under a third of respondents rating the language as being stronger than the monolingual standard (28.9%, 21.1% and 31.7% in the three categories respectively), with 15-20% of respondents rating the language as being weaker (18.9%, 20.6% and 15.6% respectively; see Table 6.4/16). These findings suggest that the PSI group are generally more confident in the L1 compared with the L2. This seems to be the case in particular with respect to instinctive/native comprehension, where the least number of PSI respondents rated the language as being stronger, and the greatest felt that it was weaker than the monolingual standard (see Table 6.4/16). On the whole, however, both languages are generally considered to be at least equal to the monolingual standard in all three categories by the PSI respondents.

Comparing the findings for the PSI group with those for the AB group, we see that, in the case of languages L1 and A, a higher percentage of AB respondents tended to rate A as being above the monolingual standard, compared with the same figure for the L1 among PSI respondents, and a lower percentage of the AB group rated A as being below the monolingual standard, compared with the same figure for the L1 in the PSI group (see Table 6.4/16). This was observed in all three categories, suggesting that, on the whole, AB respondents display a greater degree of confidence in their strongest language than PSI respondents. The same pattern was observed for languages B and L2 in the categories of richness of vocabulary and native comprehension (see Table 6.4/16). However, with respect to pragmatic/cultural competence, while the responses for the two groups followed the same pattern in the "same" and "stronger" response categories, a slightly higher percentage of AB respondents rated language B as being below the monolingual standard, compared with the same figure for the L2 within the PSI group (18.7% and 15.6% respectively; see Table 6.4/16). On the whole, however, the responses concerning the L2 for the two groups suggest that the AB group exhibit a greater degree of confidence in their second language compared with the PSI group. Possible explanations for these findings include the slightly higher age of SLA

observed within the PSI group compared with the AB group, as well as the dominant linguistic environment (L2) of the PSI group. This, as well as the other trends observed, are explained in more detail and discussed in light of the literature in Chapter 7, Section 7.4.

#### 7. Discussion

In this Chapter, the findings presented in Chapter 6 are discussed in light of the literature on bilingualism and interpreting surveyed in the first part of the dissertation, in order to address the research questions set and determine whether the results support the hypotheses put forward. The structure of the ensuing discussion is determined by the four main research questions outlined in Chapter 5, which are addressed in order. Therefore, the discussion is divided into four main sections as follows: a general overview of the characteristics of interpreters as bilinguals; an investigation of the findings pertaining to the AIIC AA and AB groups, and the implications of these results within the wider context of bilingualism; an inquiry into self-ascribed mother tongue identity among AA and AB respondents, and whether this corresponds to externally-ascribed identity as represented by the AIIC A and B classifications; an investigation of the differences between PSI and conference interpreters as bilinguals, as represented by the NRPSI and AIIC AB groups, and the implications of these findings for bilingualism more generally.

# 7.1 Interpreters as a group of highly proficient bilinguals

A look at the results for the three populations of interpreters presented in Chapter 6 reveals that, while the responses of the AIIC AB and PSI groups exhibit similar patterns with respect to many of the biographical and attitudinal factors considered, the AIIC AA group is set apart from these two populations in terms of both linguistic biographies and attitudes. The nature of and reasons for these observed differences are discussed in Section 7.2. However, the findings reveal that there are nonetheless some characteristics that are common to all groups, and which can be taken to describe professional interpreters as bilinguals at a global level. These main properties are outlined below, divided into the categories of biographies and attitudes/perceptions.

# 7.1.1 Biographical factors and language use

It was first of all observed that all three groups can be characterised by the fact that one of their current strongest languages was acquired before the other, and this language also tended to dominate as the language mostly spoken at home during childhood for all three groups (A1 within the AA group; A within the AB group; L1 within the PSI group), although this was the case to a lesser degree within the AA group (see Section 7.2). As hypothesised, the findings revealed that, on average, respondents from all three groups started learning their second language before puberty, with the highest average age of SLA of 11 years observed among the PSI respondents. This corresponds to the predictions of the literature on the incidence of a sensitive period for second language acquisition, and the strong negative correlation observed between age of SLA and degree of attainment in a second language (see Chapter 2). At the same time, respondents from all three

groups exhibited individual variation with respect to age of SLA, with a range from 0 - 18 years observed for the AA group, and 0-30 years for both the AB and PSI groups. This finding is as expected and is in line with the predictions of the DST view of second language development, which postulates that variation is to be expected at the individual level. With respect to language use in later life, two important characteristics were identified that pertain to all three groups. Firstly, as hypothesised, only a minority of respondents from all three groups were exposed exclusively to their first language as the language of higher education (a maximum of a third of respondents, observed within the PSI and AB groups; this figure was lower among the AA group). With respect to attainment in a second language, this finding suggests that exposure to the second language in the context of higher education plays an important role in the attainment of the level of proficiency in this language required of interpreters, including command of different registers and cultural knowledge. Secondly, it was found that all three groups were characterised by the tendency to use both languages more or less equally on a daily basis for a number of non-professional activities, regardless of the dominant language of their country of residence. The specific language use categories in which the majority of respondents from all groups reported using both of their strongest languages equally were: reading, writing, socialising and using the internet. What is not known is whether this extensive use of both languages in everyday life is a conscious choice or a natural consequence of this level of bilingualism, and whether it is a necessary condition for the maintenance of both languages at the high standard required of interpreters.

Several interpretations of the above finding can be proposed on the basis of the literature. On the one hand, it is reasonable to suppose that, in addition to objective circumstances (e.g. work requirements, country of residence, family circle) language use among bilinguals in later life is influenced by attitude towards the languages in question. In particular, Chapter 2 discussed examples of bilinguals choosing not to use certain languages due to negative attitudes towards them, and vice versa. Furthermore, the studies looking at emotional response towards their languages among bilinguals conducted by Dewaele (2004, 2008; see Chapter 2) found a link between L2 proficiency and perception of the emotional force of words in the L2. It is possible, therefore, that bilinguals with such high levels of proficiency in both languages as that required of interpreters feel a strong emotional attachment towards both languages, with in turn induces them to use both languages in daily life (consciously or unconsciously). This idea is explored further in Section 7.2, which looks at the findings for the AA and AB groups. Another possible interpretation is that interpreters make a conscious effort to use both languages in domains outside of work on a daily basis solely for professional reasons, i.e. to maintain the languages at the level required to succeed as interpreters. However, such an interpretation does not appear to be in line with the findings on motivation discussed below, as it was found that interpreters' motivation to maintain their languages is not purely professional in the majority of cases (see Section 7.1.2). It is also possible that this daily use of both languages is simply a consequence of the interpreters' high

bilingual proficiency in their two languages, and occurs naturally without conscious effort. More information is needed from the respondents about their daily language use habits, as well as their motivations for this, in order to interpret the findings with more confidence.

#### 7.1.2 Attitudes and perceptions

As hypothesised, the majority of respondents from all three groups reported awareness of a language maintenance effort, required to keep both of their languages at their present high standard. This is in line with Thiéry's (1975) findings for the AIIC AA group, as well as the prediction of the DMM (Herdina and Jessner, 2002) that language maintenance, as well as acquisition, requires effort (see Chapter 3). It appears also that interpreters consciously make this effort, as suggested by the fact that almost all respondents from the three groups indicated that they seek to fill gaps in their linguistic knowledge when they notice them. This corresponds to expectations, as it was proposed in Chapter 4 that interpreters are characterised by a strong motivation to maintain their working languages at a high standard, at least for professional reasons. However, further investigation of respondents' motivation to maintain their languages revealed that this motivation is not entirely professional, with the majority of all three groups indicating (1) that they would continue to make the effort to maintain both languages at their present high standard even if not working as professional interpreters, and (2) that their motivation for filling gaps in linguistic knowledge is not purely professional. While the questions pertaining to the nature of language maintenance motivation among interpreters are limited, the responses obtained suggest that interpreters can be characterised by instrumental as well as integrative motivation (see Chapter 2, Section 2.2 (6)) to maintain and/or continue to improve their languages.

It was further observed, although this was not one of the hypotheses originally put forward, that the majority of interpreters from all groups surveyed demonstrated awareness of cross-linguistic interference between their strongest languages, with a minority of respondents (less than a third) in each group reporting that they do not notice interference in either language. This finding could be explained by the fact that, as professional linguists who are required to function at a high level in monolingual mode (Grosjean, 2008; see Chapter 2), keeping their languages free from interference as far as possible, interpreters are therefore sensitively aware of cross-linguistic interference and its direction. The next hypotheses concerns the behaviour of the populations with respect to relative attachment/emotional response to their two strongest languages. Indeed, comparing the responses for each of the groups' strongest languages, the findings seem to confirm the hypothesis that respondents are likely to demonstrate a greater attachment/emotional response to their first language (the language most associated with childhood), compared with their second language. It was found that all three groups favoured their first language (A1 within the AA group; A within the AB group; L1 within the PSI group) over their second language (A2, B and L2 respectively) in all

three categories of language preference when tired/emotionally upset, relative weight of words and special attachment. Given that subsequent factors affecting language use and/or experiences in the language(s) (e.g. country of residence; social/family circle etc.) were not controlled for and differed among respondents, these findings suggest that, as predicted by the literature, emotional attachment to a bilingual's languages is influenced most strongly by the childhood linguistic environment. That having been said, it was also observed that a substantial proportion of respondents from all three groups rated both languages as being equal in the three categories outlined above, and this figure was particularly high within the AA group. Possible conditions that contribute to equal attachment to the two languages are analysed in more detail in the subsequent discussion of the findings for the AA and AB groups.

Finally, it was proposed that interpreters would tend to rate their first language as being stronger than their second language, in particular in terms of speech production, regardless of subsequent language use/exposure. This was based on Paradis' (2004) argument that processing in the first language is more automatic than that in the second language, resulting in more stable performance. As a general trend, this was indeed found to be the case, with greater numbers of respondents within each group rating their first language as stronger than their second language in the categories of command of registers, gaps in vocabulary and writing. At the same time, while this overall trend was observed among all three groups, there were nonetheless important differences between the three populations, with the AA group in particular exhibiting response patterns that set it apart from the other two populations (see Section 7.2 below for details). Finally, with respect to respondents' self-ratings relative to an average monolingual standard, it was found that the majority of interpreters from all three groups tended to rate both languages as being at least equal to this standard in terms of richness of vocabulary, instinctive/native comprehension and pragmatic/cultural competence. These results could be explained by the fact that interpreters use their languages in a variety of domains and contexts and must, as indicated in Chapter 4, be able to communicate across cultures, grasping the meaning or idea of an utterance in order to convey the speaker's intention (in accordance with Seleskovitch's theorie du sens; see Chapter 4). It is therefore reasonable that interpreters as a population of bilinguals are characterised by confidence in both of their languages, tending to rate both as being at least equal to an average native monolingual standard, at least with respect to certain skills.

The foregoing focused on the similarities between the three groups of interpreters surveyed, identifying some general characteristics that pertain to all three. The broad properties outlined below can, on the basis of the findings obtained, be taken to characterise interpreters with two active working languages as a group of bilinguals in the most general sense, regardless of language classification or habitual interpreting setting. However, the responses uncovered important differences between groups, and it is these differences and their context, discussed below, that are

believed to be the most revealing in terms of both the bilingualism of interpreters, and individual bilingualism more generally.

## 7.2 True bilingualism and the AIIC AA and AB classifications

This section of the discussion examines the findings for the AA and AB groups in light of the literature, the implications of these findings for bilingualism, as well as the exploring in more detail the true bilingualism definition itself. The subsections set out below correspond to the three parts of research question 2, with sections (i) and (ii) of part (a) addressed at the same time (see Chapter 5, Section 5.1.1).

#### 7.2.1 Comparison of AIIC AA and AB groups and implications for bilingualism

The results revealed differences between the AA and AB groups, both in terms of biographical factors and attitudes and perception. Therefore, on the whole, the findings appear to support Thiéry's (1975) observations that true bilinguals – AIIC AA interpreters – are indeed a population of bilinguals that are set apart from other highly proficient bilinguals by a number of characteristics that are unique to this group. In particular, childhood linguistic factors apparently play an important role in the development of this kind of bilingualism, which in turn have a bearing on the group's attitudes and perceptions. The specific distinctions between the AA and AB groups are discussed below.

It was first of all noted that the majority of AA respondents consider both of their languages to be mother tongues (85.4%), while this was the case for only a minority of AB respondents (13.2%), suggesting that external perceptions of native and non-native proficiency and/or identity tend to correspond to the bilinguals' own perceptions. The characteristics of the subgroups of respondents from both groups where this was not the case are analysed in detail in Section 7.3 below. Turning to the first hypothesis concerning expected differences between the AA and AB groups with respect of age of SLA and other childhood linguistic environment factors, the findings appear to be fully in line with expectations and the predictions of the literature. In particular, a lower mean age of SLA was observed among the AA group, compared with the AB group (4.3 and 8.6 years respectively). What is more, this difference was found to be statistically significant, representing a medium sized effect. This finding is in line with the literature on the role played by age of acquisition in the degree of proficiency attained in a second language, suggesting that age of SLA is an important predictor of ultimate proficiency attained. However, in accordance with DST principles, individual variation was observed within both groups in terms of age of SLA: the range for SLA was found to be 0-18 years among the AA group, and 0-30 years among the AB group. If the AIIC AA language classification is taken to correspond to native-like proficiency in two languages, then the

above findings suggest that it is possible to attain this level of proficiency in two languages even if one of the languages is acquired after puberty. This is in line with the findings of Bongaerts et al. (2000) that there is no critical period for SLA, even in the case of accent (see Chapter 2 for details). The findings also correspond to the Larsen-Freeman's claim that, while the language system is sensitively dependent on its initial conditions (in this case, the age of acquisition of the languages), it is not forever bound by these conditions (see Larsen-Freeman, 1997; Larsen-Freeman and Cameron, 2008; Chapter 3). However, this does not contradict DeKeyser's (2013) contention that native-like attainment in a second language is less likely after puberty, since it was observed that, with the actual range of SLA for the AA group ranged from 0-18 years, the majority of respondents fell within the 0-8 year range. It was also found that, while the majority of both the AA and AB groups reported that they acquired one of their languages before the other (A1 within the AA group, A within the AB group), the proportion of respondents who acquired both languages simultaneously from birth was substantially larger among the AA group compared with the AB group (27.1% and 1.9% respectively).

In addition to differences in terms of age of SLA, the AA and AB groups were also found to differ with respect to relative exposure to their two strongest languages in childhood, in all categories considered. As hypothesised, the results revealed that the two strongest languages within the AA group tended to be more evenly represented in childhood compared with the AB group, who were predominantly exposed to language A. These group-level differences in relative exposure to the two strongest languages in childhood were observed in all three categories of "home", "street" and "school", identified as key by Thiery (1975) (see Chapter 4). Another important difference observed between the two groups was that a much larger proportion of AA respondents relocated to a different country during childhood compared with AB respondents (64.6% and 26.4% respectively). While this relocation does not necessarily imply a change in linguistic environment, it is reasonable to suppose that it did in most cases, given the higher incidence of both languages as the languages of school education and country of residence in childhood among the AA group compared with the AB group. This finding is in line with Thiery's (1975) conclusion that, in children, true bilingualism is the result of chance or exceptional circumstances. These findings taken together support the predictions of the literature on the importance of the role of age and context of language acquisition among bilinguals in determining subsequent proficiency levels attained in the languages. At the same time, while the trends observed at group level were as expected, there were nonetheless individual differences. This gives rise to the potential further research question of why, if there are AB interpreters with childhood linguistic environments similar to those of the AA group in terms of relative exposure to the two languages and age of acquisition, did they not obtain an AA classification in both languages? Are the differences more subtle distinctions in childhood linguistic environment that are not captured by the present questionnaire, is it a matter of other factors (e.g. motivation, attitude, subsequent language use), or

does the difference pertain specifically to interpreting rather than bilingualism? Conversely, what is different about those AA interpreters whose childhood linguistic environment was more typical of the AB group, and yet they are still classified as having two native languages? Further information is needed to investigate these questions, and suggestions for future inquiry along these lines are given in Chapter 8.

The AA and AB groups were next compared in terms of language use and exposure in later life. In the category of higher education, perhaps surprisingly, it was found that a greater number of AB respondents were exposed to both of their languages in this context, compared with AA respondents (50.7% and 39.1%). This suggests that, as postulated by Thiéry (1975), with respect to true bilingualism, in the majority of cases, everything is determined in childhood. In other words, even if one subsequently uses both languages at a high level, such as in an academic setting, without this initial foundation in childhood, one may not be considered a true bilingual in accordance with the definition. With respect to the dominant linguistic environment of respondents' country of residence, this was found to be mixed within both groups, and it can be supposed that the country of residence is determined by work commitments or other circumstances.

Respondents were also asked to indicate the language(s) they are exposed to or use the most on a daily basis. It was observed that, while the answer to this question appeared to be influenced by the dominant language(s) of the country of residence among both groups, the answers were not identical: instead, a substantially higher percentage of respondents within both the AA and AB groups indicated that they are exposed to both languages equally on a daily basis. These figures were almost identical for the AA and AB groups (almost a third within each) suggesting that, at least with regard to subjective perceptions, the AA respondents are not exposed to/do not use their two languages more on a daily basis compared with AB respondents. This finding was echoed by the results of the comparison of daily language use across different categories, namely: reading, writing, TV, radio, socialising and the internet. It was found that the majority of respondents from both groups use both languages for reading, writing, socialising and viewing internet pages/using social media, with the responses spread more evenly across different response options in the TV and radio categories, but a greater incidence of use of both languages was not generally observed among the AA respondents. This was found to be the case only in the categories of writing and socialising, with more AA respondents reporting that the use both languages on a daily basis for these activities compared with AB respondents. These two language use categories are associated with more active use of the languages, and it is possible that this increased active use of both languages on a daily basis is an important distinguishing characteristic of the AA group. In other words, while the results for daily language use for the two groups do not shed any light on the necessary/sufficient conditions for the maintenance of two languages at the native-like level, as both AA and AB groups tend to use their two languages extensively in everyday life, the findings

suggest that the type of language use (active/passive) may play a greater role than extent. In any event, questions arise as to whether native-like competence in two languages requires a different kind of language use (e.g. more extensive; active as opposed to passive) to maintain than simply a high level of competence in two languages, or if any such observed differences are a consequence of this degree of bilingual proficiency.

With regard to language use in the context of interpreting, the responses were in line with reasonable expectations on the basis of what is known about conference interpreting and directionality (see Chapter 4). It was found that a greater percentage of AA respondents interpret into both of their languages equally compared with the AB group, who tend to interpret mostly into language A. A greater proportion of AA respondents also reported no preference in terms of interpreting direction, whereas AB respondents tended to favour interpreting into their native language. With regard to perceptions of the relative importance of interpreting and linguistic skill, no major differences were observed between the two groups, with respondents from both groups tending to consider both to be equally important, closely followed by interpreting skill, and only a minority from each group choosing linguistic skill. Interestingly, the latter figure was slightly lower among the AA group compared with AB group (8.5% and 11.5% respectively). Thus, it appears that the special status of the double A classification does not influence interpreters' perceptions in this regard. Comparing the number of respondents in each group who also translate, it was found that this figure was substantially higher within the AB group than among the AA group (70.6% and 47.9% respectively). This could be explained by the fact that, as indicated in Chapter 4, conference interpreters usually work into their A language, AA interpreters therefore have more interpreting opportunities than their AB colleagues (at least in international institutions, as opposed to the freelance market where the retour is more common), and thus do not need to carry out translation work. It was further found that, of those AB interpreters who also translate, half (50.7%) indicated that they translate into both their A and B languages. This is interesting because it is generally expected that translators work into their native language(s) only. Depending on the type of translation carried out, and the standards set, it may be that these AB interpreters are indistinguishable from monolinguals in writing in both of their languages (i.e. they can be considered true bilinguals in writing). If this is the case, the finding supports Thiéry's (1975) claim that true bilingualism in writing does not necessarily correspond to true bilingualism.

The AA and AB groups were next compared in terms of their attitudes, perceptions and self-ratings of proficiency in their two languages, and several hypotheses were put forward in this regard on the basis of the literature. In particular, it was postulated that a lower percentage of the AA group than the AB group would report that language maintenance requires conscious effort, that AA respondents would tend to feel an equal attachment to both of their languages, whereas the AB group would be characterised by a stronger attachment to language A, and that AA respondents

would demonstrate higher levels of non-professional (integrative) motivation to maintain their two languages. The responses obtained from the two groups appear to support these hypotheses, although the observed differences were slight in some cases. First of all, it was indeed found that slightly more AB respondents than AA respondents felt that maintaining both languages at the present high standard requires a degree of conscious effort (73.7% and 68.8% respectively), and this trend was reversed among those who felt that it does not. Furthermore, a greater proportion of AA respondents than AB respondents believed that maintaining both languages at this standard would be possible without working as a professional interpreter (45.8% and 31.1% respectively). Taken together, these findings imply that, compared with the AB group, the AA respondents are less likely to consider that the maintenance of both languages requires conscious work to maintain. As proposed in Chapter 5, one explanation for this could be the earlier average age of SLA, greater exposure to both languages in childhood and tendency to have acquired both by immersion observed within the AA group, resulting in more automatic processing in the two languages (Paradis, 2004). In other words, this foundation acquired in childhood means that both languages within the AA group remain more resistant to change in linguistic environment, levels of use/disuse of the two in later life. On the other hand, it is also possible that a consequence of having two native languages (the AA classification) is increased active use of both in everyday life, without the application of conscious effort (as observed above, the AA respondents tended to use both languages in the active use categories of socialising and writing to a greater extent than AB respondents). Therefore, it may be that, for AA respondents, language maintenance is something that is perceived as occurring naturally, and would happen regardless of professional activity, whereas AB interpreters are more likely to feel that they are making a conscious effort to keep both languages at their present high standard.

With respect to the nature of the two groups' motivation to maintain their languages, the findings are also in line with expectations. It was observed that a greater proportion of AA than AB respondents believe that they would still make the effort to maintain both languages at their present high standard even if not working as interpreters (59.6% and 50.5% respectively). Also, a higher proportion of the AA group than the AB group reported that their motivation for filling gaps in linguistic knowledge is not purely professional (80.9% and 70.5% respectively). Taken together, these responses suggest that the AA group does tend to be characterised by higher levels of non-professional motivation to maintain both languages than the AB group. As explained in Chapter 5, this hypothesis was based on the expectation that the AA group would tend to demonstrate a more equal attachment to both languages compared with the AB group, and the connection between attitude and motivation (for SLA and language maintenance) documented in the literature (see Chapter 2). The expectations for attitude/attachment appear to have been confirmed by the responses obtained. It was indeed found that a greater percentage of AA than AB respondents reported that they do not feel a greater attachment to one language over the other (50.0% and

37.3% respectively), and that words carry the same weight in both languages (60.4% and 36.8% respectively). Within the AB group, on the other hand, language A dominated in both cases. Additionally, a greater proportion of AA than AB respondents indicated that they do not have a language preference when tired or upset (41.7% and 31.1% respectively), whereas the AB group again favoured language A. However, while this last question was initially intended to be a measure of emotional attachment, it is possible that this is not in fact the case, as some respondents clarified in the additional comments that language preference in this context is influenced by the interlocutor or topic. Nonetheless, on the basis of the responses to the questions concerning attachment to the languages and perception of the weight (emotional force) of words in the two languages, it can be concluded that the AA group is characterised by a more equal attachment to the two languages compared with the AB group. Therefore, the findings for attitude/attachment to the two languages provide a possible explanation for the findings concerning language maintenance motivation in both groups. In other words, as a consequence of the more equal attachment to the languages observed within the AA group compared with the AB group, it is more important to AA respondents to maintain both languages at a high standard for personal reasons than it is for the AB group.

Hypotheses were also advanced with respect to awareness of language dominance shifts and perceptions of the direction of cross-linguistic interference within the two groups. The findings for the AA and AB groups in both cases appear to be in line with expectations. First of all, it was found that a greater proportion of AA than AB respondents feel that language dominance fluctuates between their two strongest languages (50.0% and 37.3% respectively). This overall trend was observed without controlling for external factors, such as dominant linguistic environment or language(s) respondents work into the most. Therefore, it can be concluded that, regardless of changes in external linguistic environment, AA respondents are characterised by a greater awareness of shifts in dominance between their two languages compared with AB respondents. This prediction was made in Chapter 5 on the basis of the closer proximity (with respect to relative competence) of the two languages within the AA group compared with the AB group. In accordance with the predictions of the DST approach to second language development, this closer proximity of the languages in terms of relative competence within the AA group means that less external input is needed to shift the language system to a different state, such as to a state where the relative strength of the languages changes (see Chapter 3; Larsen-Freeman and Cameron, 2008). On this basis, it is reasonable to expect that shifts in language dominance would occur more frequently within the AA group than within the AB group, as more subtle changes in the external linguistic environment (e.g. simply reading or socialising more in one of the languages) are required to bring about such shifts within the AA group. Additionally, respondents from both groups demonstrated similar levels of awareness of the circumstances that lead to these fluctuations in language dominance. In other words, they were aware of the influence of external environmental

factors on internal ones (relative competence in the languages). This is in line with the DST view of bilingualism, that the language system is never in a static state, but rather it continually evolves and changes due to the influence of contextual factors (see Chapter 3 for details).

With regard to the direction of cross-linguistic interference (in terms of speech production) as perceived by respondents, the findings demonstrated that, as anticipated, the AA group were more likely to view interference as being bi-directional, whereas the AB group tended to feel that language A exerted a greater degree of influence on language B. Again, these results were observed without controlling for other factors that have a bearing on cross-linguistic interference and its direction, such as language use and exposure. The implication for bilingualism more generally, therefore, is that relative competence in the two languages and/or the age and context of SLA contribute the most to perceptions of dominance shifts and direction of interference, since differences were observed between the AA and AB groups without controlling for any other factors. Additionally, the finding that the majority of AA respondents reported that they do make unnatural utterances in one or both languages due to the influence of the other language appears to support Grosjean's (1989) claim that the bilingual language system is different in nature to the monolingual system (see Chapter 2), and hence is not reducible to "multiple monolingualism", regardless of the level of competence attained in each language (Herdina & Jessner, 2002; see Chapter 3).

Finally the two groups were compared in terms of self-ratings of proficiency in their strongest languages, relative to each other and relative to an average monolingual standard. This latter category was intended to measure confidence in the languages in question. On the basis of the literature on the correspondence between external ratings of bilingual proficiency and self-ratings (Hamers and Blanc, 2000; see Chapter 2), it was hypothesised that the AA group would tend to view both languages as being more or less equal, whereas the AB groups would be more likely to regard language A as being stronger than language B. This was indeed found to be the case, with a greater proportion of AA respondents than AB respondents rating both languages as being equal in the three categories of command of registers, number of gaps in vocabulary and writing (64.6% and 27.6 percent respectively; 52.1% and 27.8% respectively; 60.4% and 35.1% respectively). While both groups tended to rate their first language as being stronger than their second language in all three categories (i.e. language A1 in the AA group, language A in the AB group), the difference between languages was less pronounced within the AA group. It is notable that these results were observed without controlling for other variables, such as language(s) of higher education and current dominant linguistic environment. This suggests again that, while it has been demonstrated that other contextual factors contribute to proficiency levels in the languages (see Chapter 2), age and context of SLA play the biggest role, at least with respect to self-ratings.

In terms of respondents' self-ratings relative to an average monolingual standard in the categories of richness of vocabulary, instinctive/native comprehension and pragmatic/cultural competence, similar patterns to the above were observed within the AA and AB groups. That is, AA respondents tended to give similar ratings for both of their languages, whereas AB interpreters consistently gave higher scores to language A than to language B in all three categories. An additional observation, which was not anticipated, was that AB respondents tended to rate their first language (A) as stronger than AA respondents rated either of their languages (A1, A2). Language B, on the other hand was rated by AB respondents as weaker relative to the ratings given by AA respondents to each of their languages. The first finding is perhaps surprising, given that languages A1, A2 in the AA group and language A in the AB group all have the same AIIC language classification (i.e. they are all considered to be native languages). At the same time, it was noted in Chapter 2 that there is a high degree of variation in terms of language competence even within monolingual populations, and therefore defining a language as "native" does not imply that some set standard has been attained in this language.

The finding that, at least in terms of subjective perceptions, the AA group can be characterised by two languages of about the same proficiency (which is perceived to be more or less equal to the monolingual standard), and the AB group by two languages with differing levels of proficiency one very strong language (above the monolingual standard) and one weaker language (B) may be explained in terms of the DST view of bilingualism. The explanation is similar to that proposed by the DMM (Herdina and Jessner, 2002) for Grosjean's domain specificity principle (see Chapter 3). According to the DMM, domain specificity emerges as a consequence of the principle of economy of effort (Zipf's law; see also Larsen-Freeman and Cameron, 2008), as such a system is more stable and requires less effort to maintain than an all-purpose fully-fledged language system (Herdina and Jessner, 2002). It is possible that such a compensatory mechanism also applies to relative competence in two languages. Consequently, languages A1, A2 in the AA group are generally ranked below language A (AB group), since maintaining two languages at the higher standard attributed to language A in the AB group would require too much effort. If the communicative needs (DMM, see Chapter 3) of the individual can be met by maintaining both languages at a slightly lower standard, there is no reason to expend additional effort to maintain the languages at a higher level. Language A in the AB group, however, can be maintained at a higher relative standard due to the lower relative ranking of language B. On the other hand, if the above-described findings are viewed in terms of language confidence rather than actual proficiency, it can be concluded that having one clear native language results in greater confidence in that language, compared to cases where two native languages are present.

## 7.2.2 The true bilingualism definition and the AIIC AA classification

This section explores the questions pertaining specifically to the true bilingualism definition (Thiéry, 1975) and AIIC AA respondents' attitudes towards it, as well as the relationship between true bilingualism and other notions, such as mother tongue identity in two languages. Several hypotheses were proposed in this regard. First of all, it was anticipated that the majority of AIIC AA respondents would identify themselves as true bilinguals, as both the double A classification and Thiéry's definition concern external perceptions of mother tongue competence (identity) in two languages. This was indeed found to be the case, with 91.7% of AA respondents agreeing that they are true bilinguals. Indeed, as explained in Chapter 6, Section 6.2.2, following an examination of the reasons given for their answer by respondents who did not consider themselves to be true bilinguals, two of the reasons given do not prevent their authors from being classed as true bilinguals according to the definition. Consequently, the actual proportion of AA respondents who can be taken to be true bilinguals is 95.8%. Interestingly, one respondent commented that "....usually, people tend to doubt my abilities in "the other" language and cultural environment, the "foreign" one" (see Section 6.2.2 for full details). This corresponds to the point raised by Grosjean (2015) on biculturalism and others' perceptions of this phenomenon: in particular, he noted that people often find it hard to accept that biculturals are members of both cultures - invariably, they are seen as being one or the other, but not both (see Chapter 2).

Respondents were also asked to indicate whether they believe that the AIIC AA classification is equivalent to the true bilingualism definition proposed by Thiéry. While the majority of the AA group considered this to be the case (72.9%), it is notable that this figure is lower than the percentage of respondents who would class themselves as true bilinguals. The questionnaire did not expressly ask respondents who did not feel that the true bilingualism definition corresponds to the AIIC AA classification to give reasons for this. However, one respondent did provide clarifications in the additional comments at the end of the questionnaire, suggesting that the AA classification is a stronger concept that true bilingualism, and refers to "someone who is taken to be one of them by the members of two different linguistic communities, at exactly the same social and cultural level". Such a position appears to contradict the findings, i.e. that all respondents are in possession of the AIIC AA classification, yet not all would describe themselves as true bilinguals, which implies that true bilingualism is the more demanding definition of the two. In light of this, a possible explanation for the finding that not all respondents believe that the true bilingualism definition corresponds to the AIIC AA classification is that some respondents may have interpreted Thiéry's definition as implying bilingual balance. This is supported by a comment made by one of the respondents, who indicated that s/he does consider her/himself to be a true bilingual because "I have greater difficulty as concerns speed and concordance in one of the two languages". As pointed

out by Thiéry, however, true bilingualism does suggest that there must be some kind of balance between the two languages, or that such a state is even possible (see Chapter 4).

It was next hypothesised that the true bilingualism definition is not necessarily equivalent to the idea of having two mother tongues because, as admitted by Thiéry himself (1975; see Chapter 4), his definition only addresses the objective component of bilingualism (others' perceptions), and does not take into consideration subjective ideas about proficiency and/or identity in the two languages. The notion of a mother tongue, on the other hand, has certain emotional connotations (see Chapter 2), and includes the subjective component. The findings appear to support this hypothesis: while the majority of respondents agreed that both of their A languages are mother tongues (85.4%), there was nonetheless a subgroup who did not necessarily agree that this is the case. This suggests that the idea of having two mother tongues is stronger than the AIIC AA classification. Furthermore, it was observed that, of those respondents who did not indicate that both of their A languages are mother tongues, a number (42.9%; 3 respondents) nonetheless considered themselves to be true bilinguals (see Chapter 6, Section 6.2.3 (1) for details). This implies that the idea of having two mother tongues is a stronger notion that true bilingualism. This suggests that, in view of the emotional connotation of the term "mother tongue", for a language to be considered by an individual as a mother tongue, there needs to be some kind of emotional connection and/or childhood associations with the language in question, in addition to high proficiency in the two languages.

With regard to mother tongue identity in two languages, AA respondents were additionally asked whether they feel that both A languages were mother tongues in childhood. While the majority of respondents indicated that this was the case, the number who felt that it was not was found to be equal to almost a third of respondents (31.3%). It is notable that no respondents selected the Unsure response category when asked about the situation in childhood, suggesting that perceptions of mother tongue competence/identity in childhood are more clear-cut compared to adulthood (10.4% of respondents were unsure whether their A languages are currently both mother tongues). Respondents who did not feel that both of their A languages were mother tongues in childhood were asked to indicate the age at which they achieved this level of competence in both languages and what led to it. The average age at which this level of competence was attained in both languages is 19.7 years, with a range from 12 to 30 years. The reasons given for this were either a change in language(s) of education or linguistic environment of the country of residence, with only one participant giving a difference reason, namely "extensive reading, the practice of translation". No respondents cited interpreting as the reason for the shift in competence. Taken together, these findings suggest that respondents in the AA group are aware of shifts in competence in their A languages over the lifespan, and the circumstances that led to these changes. The finding that almost a third of AA respondents did not feel that both of their A languages were mother tongues in

childhood, as well as the fairly high mean age at which mother tongue competence was attained (i.e. several years after puberty) indicates that the AIIC AA classification is not necessarily accompanied by mother tongue competence in childhood, at least with respect to subjective perceptions. This is in line with the DST view of bilingualism as a system that undergoes constant change under the influence of contextual factors.

Finally, the relationship between true bilingualism as defined by Thiéry and true bilingualism in writing (i.e. being indistinguishable from a native monolingual speaker in both languages in writing) was explored further. As expected, the majority of AA respondents considered themselves to be true bilinguals in writing (72.9%). This figure was found to be even higher among those AA respondents who also identified themselves as true bilinguals (77.3%). It can therefore be concluded that, while Thiéry's definition refers to oral competence in the languages only, true bilingualism does tend to be accompanied by the same proficiency in writing, at least with respect to respondents' perceptions. However, one does not necessarily imply the other. In particular, we recall that it was observed in Section 7.2.1 above that a substantial proportion of AB interpreters reported that they translate into both their A and B languages, suggesting that they are indistinguishable in these languages in writing, but not in speech (at least according to the AIIC classification).

## 7.2.3 AIIC AA interpreters who obtained their AA classification at a later date

As anticipated on the basis of Thiéry's (1975) original findings, a subgroup of AA respondents (14.6%; 7 respondents in total) was identified who were awarded their AA classification after joining AIIC. The proportion of AA respondents with this experience is similar to that observed by Thiéry (see Chapter 4). A number of hypotheses were proposed regarding the characteristics of this subgroup and how it is likely to differ from the main group of AA respondents. However, as explained below, these hypotheses were generally not confirmed by the findings; indeed, the results were at times the opposite of what was expected. It should be noted that, given the small size of the population of AA interpreters who were awarded their double A classification at a later date, it is not necessarily possible to generalise on the basis of these findings. At the same time, the present investigation of the characteristics of this particularly interesting group has made it possible to test some preliminary hypotheses and suggest further directions for research.

It was first of all observed that obtaining the AA classification at a later date had no bearing on perceptions of mother tongue identity/competence in the two A languages, either at the time of completion of the questionnaire or in childhood. Indeed, a higher proportion of the subgroup of AA respondents who obtained their classification at a later date (hereinafter referred to as the "AA Later" group) indicated that both of their A languages were mother tongues in childhood, compared

with the rest of the AA group. Furthermore, whether the AA classification was awarded upon joining AIIC or subsequently appeared to have no impact on respondents' tendency to identify themselves as true bilinguals. It was found that 100% of the "AA Later" group consider themselves to be true bilinguals. The hypotheses that the "AA Later" group would be characterised by a higher age of SLA than the rest of the AA group, and a less equal exposure to both languages in childhood did not find confirmation in the results obtained. The mean ages of SLA were found to be virtually identical for both groups (4.4 years for "AA Later" and 4.3 years for the rest of the AA group), and the two groups exhibited similar patterns of relative exposure to the two languages in childhood (see Chapter 6, Section 6.2.3 (2) for details). Therefore, these findings contradict the hypothesis that a possible reason to the later acquisition of the AA classification is down to a later age of acquisition of the language originally classed as a B, and/or less exposure to this language in childhood in general. The findings do, however, appear to support Thiéry's (1975) argument that, in the case of true bilingualism (the AA classification), everything is determined in childhood. In other words, a language may become less active due to disuse in later life, however, if the foundation of sufficient exposure to that language in childhood is in place, it may subsequently be reactivated with intensive use, such as interpreting. On the other hand, while the present questionnaire failed to reveal differences in the childhood linguistic environment of the "AA Later" group compared with the main group, it is possible that the information obtained as part of the present study is too general, and a more detailed investigation is required in order to uncover subtle differences.

It was further proposed that the "AA Later" group may differ from the rest of the AA group with respect to the languages of higher education, in particular, that the former would be less likely to have been exposed to both languages in this setting. However, this was not found to be the case. In fact, the results showed that a greater proportion of "AA Later" respondents were exposed to both A languages in the context of higher education, compared with the rest of the AA group. The two groups also exhibited very similar response patterns for language use and exposure at the time of completion of the questionnaire (see Chapter 6, Section 6.2.3 (2) for details). Therefore, the data obtained as part of the present survey does not make it possible to identify any differences in language use/exposure in later life that could account for the AA classification being awarded at a later date. If any such differences exist between the groups, a more fine-grained analysis is needed to reveal them.

A comparison of the responses of the "AA Later" group with those of the main AA group with respect to attitudes and perceptions revealed some mixed results as regards the proposed hypotheses. The hypothesis that "AA Later" respondents would tend to exhibit a greater attachment/emotional response to one of their languages over the other, whereas the rest of the AA group would be more likely to rate both languages as being equal in the corresponding questions, was not supported by the findings. Indeed, the opposite was found to be true, with a greater

proportion of "AA Later" respondents rating both A languages as being equal in all three questions intended to measure attachment/emotional response (language preference when tired/upset, weight of words, special attachment) compared with the main AA group (see Chapter 6, Section 6.2.3 (2) for details). While this finding goes against the expectations outlined in Chapter 5, it is not surprising with the earlier observations regarding the similar ages of SLA and childhood linguistic environments of the two groups. However, the hypothesis that the two groups are likely to differ with respect to self-ratings of proficiency in the two languages, as well as language confidence (as represented by the self-ratings in each language relative to a monolingual standard), was found to be in line with some of the trends observed. Specifically, it was found that, with respect to certain language use categories at least, respondents within the "AA Later" group tended to rate the A1 as being slightly stronger than the A2, whereas the rest of the AA respondents were more likely to consider both languages as equal. This finding is in line with expectations. Furthermore, it was observed that, with regard to some of the aspects of language use considered, the "AA Later" group tended to display slightly lower confidence levels compared with the rest of the AA respondents (see Chapter 6, Section 6.2.3 (2) for details). This result was also in line with expectations. Taken together, the findings concerning differences in self-ratings of proficiency between the two groups could be interpreted as the presence of a link between external and subjective perceptions of proficiency. However, it is not clear is whether the fact that, according to external perceptions (AIIC classifications), one language was initially rated below the other affected "AA Later" respondents' perceptions of relative proficiency and/or confidence in their languages, giving rise to the observed differences.

In addition, although this did not form part of the hypotheses formulated in Chapter 5, the responses given by the two groups to some of the other questions were also compared, in particular with respect to awareness of language maintenance effort, motivation, perceptions of fluctuations in language dominance and cross-linguistic interference. Differences between the groups were observed in terms of perceptions of language maintenance effort, with a smaller proportion of "AA Later" respondents considering that language maintenance requires conscious effort, compared with the rest of the AA group (57.1% and 70.7% respectively). Secondly, it was found that a substantially larger proportion of "AA Later" respondents felt that language dominance does not fluctuate between their two A languages, compared with the main group (71.4% and 34.1% respectively). In the remaining categories of questions, the two groups either exhibited very similar response patterns, or the findings were inconclusive (see Chapter 6, Section 6.2.3 (2)). The finding regarding awareness of language maintenance effort appears to contradict what could reasonably be expected, i.e. that the "AA Later" group would tend to exhibit greater awareness of the effort required to maintain both languages, given that some degree of effort/increased language use was likely to have been required to bring the two languages to AA standard for this group of respondents. The result concerning awareness of shifts in language dominance, on the other hand,

could be interpreted as being in line with reasonable expectations, if it is supposed that the languages within the "AA Later" group are not as close (in terms of relative competence) as within the rest of the AA group, However, as previously stated, given the small sample size, it is difficult to infer too much from these findings. What can be concluded on the basis of the fact that some of the AA interpreters were awarded their AA classification at a later date is that the language system does appear behave in accordance with DST principles. In the case of the "AA Later" group, it appears as though their language system had stabilised at AB level (which could be considered an attractor or steady state), however, as a result of increased use of the B language in a professional interpreting context, the language system entered a new state.

## 7.3 Self-ascribed and externally-ascribed mother tongue identity in two languages

This section takes a closer look at the subgroups of AA and AB respondents whose own perceptions of mother tongue identity in their languages do not correspond to external perceptions, as represented by the AIIC AA and AB classifications. In the first part, the characteristics of the subgroup of AA respondents who do not feel that they have two mother tongues are investigated and compared with the main AA group, while the second part investigates the subgroup of AB respondents who do consider both of their languages to be mother tongues. The aim of the ensuing discussion is to understand which factors appear to have the most impact on shaping self-ascribed mother tongue identity and the relationships, if at any, between such subjective perceptions of identity and bilinguals' attitudes towards their languages.

## 7.3.1 AIIC AA interpreters who do not consider that they have two mother tongues

As anticipated, while the majority of AA respondents did agree that both of their A languages are mother tongues, indicating that self-ascribed native speaker identity tends to correspond to that which has been ascribed by others, a small subgroup of AA interpreters who do not necessarily feel that this is the case was identified (14.6%, 7 respondents; see Chapter 6, Section 6.2.3 (1) for details). While the small size of the population means that generalisations cannot necessarily be made on the basis of these results, it is nonetheless informative to examine the characteristics of this group in light of the predictions of the literature, in order to explore preliminary hypotheses. It was first considered whether those AA interpreters who did not state that both of their A languages are mother tongues (referred to as the "Not Two MTs" group) were more or less likely to report that both languages were mother tongues in childhood than the rest of the AA group. It was found that a lower percentage of "Not Two MTs" respondents indicated that both A languages were mother tongues in childhood compared with the main group (28.6% and 75.6% respectively). This overall trend is in line with expectations, as it was anticipated that childhood environmental factors would have a bearing on perceptions of identity/competence in adulthood. At the same time, the finding

that there were respondents within the "Not Two MTs" group who felt that both A languages were mother tongues in childhood suggests that the presence of two mother tongues in childhood does not necessarily translate into the same situation in adulthood (at least with regard to subjective perceptions). This is in line with Thiéry's (1975) argument that true bilingualism in childhood is the result of external circumstances, whereas in adulthood this state requires a conscious effort to maintain. On this basis, Thiéry argues that there are numerous cases of childhood true bilingualism, but very few adult true bilinguals (*ibid*). Furthermore, the findings indicate that perceptions of native and non-native identity shift over the lifespan.

On the basis of the literature on bilingualism surveyed in Chapter 2, it was hypothesised that, compared with the rest of the AA group, the "Not Two MTs" group would be characterised by a higher average age of SLA and less equal exposure to both languages in childhood. The findings appear to confirm these hypothesis, with some substantial differences observed between the two groups. First of all, the mean age of SLA among the "Not Two MTs" group was found to be 8 years, and 3.7 years for the main AA group (note that that mean age of SLA for the "Not Two MTs" group is closer to that observed for the AB group than it is to the main AA group). While this difference in means was not found to be statistically significant, it did represent a fairly large effect (see Appendix II). Although, as noted above, the size of the "Not Two MTs" group is small (n = 7), it is striking that, even at this small scale the mean age of SLA is more than twice that of respondents who do feel that they have two mother tongues. This finding in in line with the literature both on the impact of childhood linguistic factors on identity in adulthood. At the same time, it was observed that the age range of SLA within the "Not Two MTs" group was 0-18 years. This suggests that, while an earlier age of SLA appears to be associated with subsequent perceptions of native/non-native identity, native speaker identity in two languages is not an immediate consequence of the acquisition of both languages at a very early age.

With respect to other childhood linguistic environment factors considered in the questionnaire, the response patterns exhibited by the two groups were in line with expectations (see Chapter 6, Section 6.2.3 (1) for details). In particular, the "Not Two MTs" group was characterised by the fact that the A1 was the language of the home for 6 out of the 7 respondents, with the remaining respondent indicating that another language (not the A1 or A2) was the language mostly spoken at home during childhood. While the differences in relative exposure to the languages in the categories of school and country of residence also followed the expected pattern, these differences were not as striking as those observed for languages of the home. These observations suggest that, in terms of shaping mother tongue identity, it is the home that plays the biggest role. This finding is in line with the additional emotional connotations of the term "mother tongue", and the literature suggesting that context of acquisition/exposure influences emotional response to the language in question (see Chapter 2 for details). In addition, a marked difference was observed between the two

groups in terms of the proportion of respondents who relocated to a different country in childhood. This was the case for 28.6% of the "Not Two MTs" group, and for 70.7% of the main AA group. This finding mirrors the differences observed between the AA and AB groups suggesting that, while more information is needed on the changes in linguistic/cultural environment that accompany this relocation, such major life events in childhood are significant not only in terms of shaping linguistic proficiency (as perceived by others), but also the bilinguals' own perceptions of identity.

While, as previously noted, the size of the population of AA respondents who do not feel that both of their A languages are mother tongues is small, the degree to which the patterns of language exposure in childhood and age of SLA compared to the rest of the AA group follow the predictions of the literature is striking. What is more, these differences are observed between two subgroups of a population who are, by definition, considered by others to have two mother tongues. These findings imply two main things. Firstly, that perceptions of mother tongue identity or competence depending on how one views the idea of having two mother tongues - are strongly influenced by the childhood linguistic environment and the age of SLA. A lower age of SLA and more equal representation of both languages in childhood appears to be linked with the tendency to consider both languages mother tongues. Secondly, it seems that these differences in terms of exposure to the two A languages in childhood do not necessarily have an impact on external perceptions of mother tongue competence in both languages, as all respondents have the AA classification. This gives rise to the question of why, despite the differences in childhood linguistic environment observed between the two groups, which mirror the differences observed between the AA and AB groups described earlier, are the "Not Two MTs" group considered by others to have two native languages (i.e. they were awarded the AA classification), while AB respondents are not?

The next hypothesis proposed in Chapter 5 concerns differences in emotional attachment to the two languages between the groups. On the basis of the literature, it was suggested that those AA respondents who consider both languages to be mother tongues would report more equal attachment to their languages, compared with the "Not Two MTs" group. The findings for the two groups were in line with expectations, and it was indeed observed that a greater proportion of the "Not Two MTs" group felt a special attachment towards the A1 compared with the rest of the AA group (57.1% and 29.3% respectively). However, the findings concerning the relative weight of words in the two languages appeared to directly contradict expectations: it was observed that the majority of the "Not Two MTs" group reported that words carry equal weight in both of their A languages (85.7%), whereas this figure was lower among the rest of the AA group (56.1%). While this latter finding seems to contradict the predictions of the literature that perceptions of the emotional force of words in a given language are influenced by the childhood linguistic environment, it must also be noted that other contextual factors are also at play. In particular, Dewaele (2004, 2008; see Chapter 2 for details) found that factors such as the language of the

country of residence, length of residence and relative proficiency in the language(s) also influences the perception of the emotional force of words in that language. It is therefore possible that the findings reflect the contribution of other factors, and are not necessarily indicative of the absence of a link between childhood linguistic factors/identity and perception of the emotional force of words (in particular, in view of the small size of the "Not Two MTs" group).

While no hypotheses had been put forward in this regard, the two groups were additionally compared in terms of other aspects covered by the questionnaire. The most notable differences observed between the groups and their possible implications are outlined below. With regard to language use in the context of interpreting, it was found that respondents from the "Not Two MTs" group tend to interpret mostly into either the A2 (language least associated with childhood) or both languages (42.9% in each response category), with only one respondent interpreting mostly into the A1. This result suggests that intensive active use of the A2 or both languages does not necessarily result in both languages being identified as mother tongues by the bilingual, which may point to the presence of an additional, emotional, component inherent in the notion of a mother tongue. In terms of attitudes and perceptions, a difference was observed between the two groups with regard to awareness of language maintenance effort, and the results mirrored the differences between the AA and AB groups described above. Specifically, the "Not Two MTs" group appear to be aware to a greater extent of the conscious effort required to maintain their two A languages at the present high standard compared with the rest of the AA group (85.7% and 65.9% respectively). These results appear to be in line with the predictions of the literature, in view of the earlier age of SLA and more equal exposure to both languages in childhood of the main AA group compared with the "Not Two MTs" group (see Section 7.2.1 above for analogous argument in the AA/AB comparison). It was further observed that respondents from the "Not Two MTs" group appear to exhibit a slightly lower level of non-professional motivation to maintain both languages at their current high standard compared with the rest of the AA group (see Chapter 6, Section 6.2.3 (1) for details). This observation is in line with what can be expected on the basis of the literature review, that seeing a language as a mother tongue may increase the individual's motivation to maintain that language for personal, rather than purely practical, reasons.

Finally, differences were found between the groups in terms of self-ratings of proficiency in each language. A comparison of the responses to the questions concerning self-rating in the languages relative to each other suggests that respondents from the "Not Two MTs" group are less likely to rate both languages as being equal in terms of competence, compared with the rest of the AA group (see Chapter 6, Section 6.2.3 (1)). With respect to self-ratings in the two languages relative to a monolingual standard (taken to be a marker of language confidence), findings suggest that those AA respondents who feel that both languages are mother tongues have slightly higher confidence in both languages when it comes to active use (richness of vocabulary) compared with the "Not Two

MTs" group, whereas confidence in passive language use is around the same in both groups (see Chapter 6, Section 6.2.3 (1)). Both of the above findings are in line with what can reasonably be expected, in view of the differences in mean age of SLA and relative exposure to the two languages in childhood between the groups.

## 7.3.2 AIIC AB interpreters who consider that they do have two mother tongues

A subgroup of AB respondents was identified in Chapter 6 who consider both of their strongest working languages to be mother tongues (13.2%, 28 respondents; see Chapter 6, Section 6.3.3 for details). This finding is in line with expectations, as the majority of AB respondents feel that they have one mother tongue (language A), suggesting that self-ascribed native speaker identity generally corresponds to that which has been ascribed by others (as represented by the AB classification). At the same time, as predicted by the literature, this is not always the case. On the basis of the literature reviewed in Chapter 2 on the influence of age and context of SLA on bilinguals' identity and attitudes towards their languages, it was hypothesised that the subgroup of AB respondents who feel that both languages are mother tongues (referred to as the "AB Two MTs" group) are likely to be characterised by a lower mean age of SLA than the rest of the AB group, as well as more equal exposure to both languages in childhood. A comparison of the responses for the two groups appears to confirm both of these hypotheses. In particular, a lower mean age of SLA of 4.8 years was observed within the "AB Two MTs" group, compared with 9.2 years for the rest of the AB group. This difference was found to be statistically significant, representing a medium sized effect (see Appendix IV for details). We note also that the mean age of SLA for the "AB Two MTs" group is very close to that of the AA group (see Chapter 6 for details).

Furthermore, a comparison of the relative exposure to the two languages in childhood for the two groups was in line with expectations. Specifically, while both groups reported being predominantly exposed to language A in childhood across the main childhood linguistic environment categories of "home", "street" and "school", this dominance was less pronounced among the "AB Two MTs" group (see Chapter 6, Section 6.3.3). A particularly large difference was observed between the number of respondents in each group who reported that both languages were spoken in the home during childhood: 28.6% of the "AB Two MTs" group indicated that this was the case, compared with just 2.8% of the rest of the AB group. Taken together, these findings suggest that the most important factors in shaping self-ascribed identity are the age of SLA and the language(s) associated with the home during childhood. This is in line with the literature on the influence the context of language acquisition on subsequent attitudes/emotional response towards that language (see Chapter 2). In particular, a key consideration appears to be whether a language was acquired/used in an emotionally neutral setting (e.g. school), or a setting that, on the contrary, is associated with a greater emotional response (e.g. the home).

With regard to emotional attachment to the languages, for the same reasons as those set out in Section 7.3.1 above, it was hypothesised that the "AB Two MTs" group would tend to exhibit a more equal attachment towards their two languages compared with the rest of the AB group. The findings for the two groups were in line with expectations, and it was indeed observed that a higher proportion of "AB Two MTs" respondents reported an equal attachment to both languages, while the AB group was characterised by a greater attachment towards language A. The same pattern was observed for perception of the emotional force of words (word weight) in the two languages (see Chapter 6, Section 6.3.3). These results correspond to the general trends observed for the subgroup of AA respondents who do not feel that they have two mother tongues, described earlier. Taken together, the findings are indicative of a connection between childhood linguistic factors and emotional response/attachment towards the language(s), as well as the additional emotional component inherent in identifying a language as a mother tongue.

While hypotheses had not been put forward with regard to any other potential differences between the "AB Two MTs" group and the main AB group, the populations were additionally compared in terms of other attitudinal and biographical factors. Some of the main, potentially significant, differences are discussed below. First of all, it was observed that a slightly lower proportion of the "AB Two MTs" group felt that language maintenance requires conscious effort, compared with the rest of the AB group (67.9% and 74.4% respectively). In view of the lower age of SLA and more equal exposure to both languages in childhood observed among the "AB Two MTs" group compared with main AB group, it may be interpreted as being in favour of the earlier hypothesis that perceptions of language maintenance effort are influenced by the childhood linguistic environment (see comparison of AA and AB groups for a similar argument; Section 7.2.1). The findings comparing levels of non-professional motivation of the two groups were ambiguous, with different trend observed in the two questions intended to measure this factor, i.e. whether respondents would continue to make the effort to maintain both languages if not working as interpreters, and whether their motivation for filling gaps in linguistic knowledge is purely professional (see Chapter 6, Section 6.3.3 for details). It is worth noting that, according to the findings set out in Section 6.3.2 of Chapter 6, the AA group is set apart by the highest levels of non-professional motivation to maintain both languages in both cases.

Finally, some differences were observed between the "AB Two MTs" group and the main AB group in terms of self-ratings of proficiency in the two languages, both relative to each other and relative to a monolingual standard. In the first case, while it was found that both groups tended to rate language A as the stronger of the two languages in all three categories considered, respondents from the "AB Two MTs" group were slightly more likely to regard both languages as being equal, compared with those AB interpreters who feel that they have one mother tongue (see Chapter 6,

Section 6.3.3 for details). In the second case, if self-ratings relative to a monolingual standard are taken to be a marker of language confidence, a comparison of the two groups' responses suggests that AB interpreters who have one mother tongue feel a greater confidence in language A compared with the "AB Two MTs" group in the categories of richness of vocabulary and native comprehension. However, confidence in language B was found to be slightly higher among the "AB Two MTs" group compared with the main AB group in the same categories. As regards pragmatic/cultural competence, while it was observed that both groups tended to exhibit more confidence in language A compared with language B, confidence in both languages overall in this category was higher among the "AB Two MTs" group compared with the rest of the AB group (see Chapter 6, Section 6.3.3 for details). Overall, the findings described above are in line with what can reasonably be expected on the basis of the literature. The observation that "AB Two MTs" respondents are slightly more likely to regard both languages as being equal in terms of relative competence compared with the main AB group can be anticipated on the basis of the age of SLA and childhood linguistic environment differences between the groups. Similarly, the trends observed with respect to self-ratings relative to a monolingual standard mirror the results for the AA and AB groups (Section 7.2.1).

## 7.4 Comparison of PSI and AB conference interpreters as bilinguals

This Section addresses the final research question formulated in Chapter 5, concerning the differences between groups of PSIs and conference interpreters as bilinguals. The conference interpreters are represented by the AIIC AB group; the AIIC AA group is not included in this comparison since this population appears to be set apart by a number of characteristics, and is therefore not considered to be representative of conference interpreters in general. As we have seen in Chapter 6, the size of the PSI and AIIC AB groups is comparable (180 and 212 respondents respectively; see Section 6.1.2).

## 7.4.1 Biographical factors and language use

As explained in Chapter 5, it was not expected that the PSI and AB groups would differ greatly as bilinguals with respect to biographical factors. However, on the basis of what is known about UK-based PSIs (see Chapter 4), it was anticipated that this group would tend to have the L2 (English) as the dominant linguistic environment of the country of residence, whereas no such trend has been observed for the AB group (see AA/AB comparison, Chapter 6, Section 6.3). On this basis, it was proposed that some differences may be observed between the PSI and AB groups in terms of attitudinal factors and self-ratings due to the influence of this factor. A comparison of the responses for the two groups appears to be in line with these general hypotheses.

It was first of all noted that a similar proportion of PSI respondents as AB respondents consider one of their languages (L1) to be their mother tongue (86.7% and 84.4% respectively). Therefore, like the AB group, the PSIs can be characterised in general by the presence of one distinct mother tongue, which is also the first language spoken in the majority of cases. Comparing the mean ages of SLA for the two groups, it was found that this was slightly higher at 11 years among the PSI group, compared with 8.6 years among the AB group. While this difference in means was found to be statistically significant, it represents a fairly small effect (see Section 6.4.1). Therefore, although it was observed that the PSI respondents tended to have acquired their second language slightly later than the AB respondents, this difference was not as great as that observed between the AB and AA groups (see Chapter 6, Section 6.3). The range of age of SLA was found to be the same for both the PSI and AB groups (0 – 30 years). With respect to relative exposure to the two languages in childhood, the PSI and AB groups exhibited very similar patterns, with the overwhelming majority of both groups exposed predominantly to the first language (L1, A respectively) in all components of childhood linguistic environment considered, although this dominance was slightly less pronounced among the AB group (see Section 6.4.1 for details). As regards the number of respondents who relocated to a different country during childhood, the overall trend within both groups was the same, with the majority reporting that this was not the case, although the percentage of respondents who did relocate was higher among the AB group compared with the PSI group (26.4% and 11.7% respectively). On the whole, it can be concluded that both groups of interpreters have similar childhood linguistic environments in terms of the relative exposure to their two strongest languages, as well as comparable ages of SLA.

As regards language use and exposure in later life, as anticipated, the biggest difference between the two groups was in terms of the dominant linguistic environment of the country of residence. In the case of the overwhelming majority of PSI respondents, this was found to be the L2 (85.0%), while the AB group was more equally split between the various response categories (see Section 6.4.1 for details). With respect to self-reported daily language use for various activities, however, similar patterns were observed within both groups in most of the categories considered. Specifically, the majority of respondents from both the PSI and AB groups reported using both of their languages equally for reading, writing, socialising and the internet. Language L2 was the most popular choice among PSIs in the TV and radio categories, which is likely to be a consequence of the language of the country of residence (see Section 6.4.1 for details). Therefore, it can be concluded that, while the language of the country of residence affects language use and exposure, both PSIs and AB interpreters tend to use both languages on a daily basis for a variety of activities, which are not necessarily related to professional requirements. However, it is not known whether this is a conscious effort on the part of the interpreters in order to maintain both languages, or a consequence of their bilingualism.

Some differences were observed between the two groups in terms of language use in a professional context (translation and interpreters), which are likely to be a consequence of the specificities of conference and public service interpreting. In particular, it was found that a higher percentage of PSIs than AB interpreters reported that they interpret into both of their languages equally, and also that they have no preference as regards interpreting direction (see Section 6.4.1). These findings are in line with reasonable expectations, as the nature of PSI interpreting is generally bi-directional, while conference interpreting tends to be monologic (see Chapter 4). In terms of perceptions of the relative importance of interpreting and linguistic skill in the practice of interpreting, both groups exhibited similar response patterns, with the largest proportion of each indicating that both are equally important, closely followed by interpreting skill. However, the proportion of PSIs who rated linguistic skill as being more important was almost double that observed within the AB group (19.4% and 11.5% respectively). This finding could, again, be down to the fact that PSI mostly takes the form of short consecutive interpreting, whereas conference interpreting is mostly in the simultaneous mode, or long consecutive. The latter two types of interpreting arguably require more interpreting-specific skills, such as the development of special cognitive and other skills (working memory, note-taking, split attention etc.; see Chapter 4 for details). As regards translation work, it was found that over 90% of PSIs also work as translators - the highest percentage out of all of the populations of interpreters considered here (see Section 6.4.1). This is not surprising, given the nature of PSI work: for example, as part of an assignment PSIs may be required to translate documents, such as witness statements or other information. Indeed, the DPSI exam includes two written translation components. It was also observed that, similarly to the AB group, just over half of PSIs indicated that they translate into both languages (see Section 6.4.1).

## 7.4.2 Attitudes and perceptions

A comparison of the PSI and AB groups' responses to the questions concerning language maintenance effort and motivation revealed that the groups were generally very similar in terms of their views (see Section 6.4.2). In particular, perceptions of language maintenance effort were virtually the same among both groups of respondents. This is in line with reasonable expectations, given the very similar childhood linguistic environments of the two populations. In view of the fact that the main difference between the PSI and AB groups in terms of biographical factors is the dominant linguistic environment of the country of residence, which is mixed in the AB group, but skewed towards the L2 in the PSI group, the finding of almost identical perceptions of language maintenance effort among the groups suggests that this factor is influenced mostly by childhood linguistic factors (age and context of acquisition), and to a lesser degree by subsequent language exposure. This appears to be in line with the findings for the AA and AB groups and the hypothesis that the effort required to maintain the languages is influenced by the age of acquisition and degree of exposure in childhood (see Chapter 5, Section 5.1.2, Question 2 for details). The findings

concerning the nature of the PSI and AB groups' motivation to maintain their languages, however, appear to be ambiguous. Specifically, while it was found that a slightly higher proportion of PSI than AB respondents indicated that they would make the effort to maintain both languages at their present high standard if not working as interpreters, a considerably higher percentage of PSIs than AB respondents stated that their motivation for filling gaps in linguistic knowledge was purely professional (see Chapter 6, Section 6.4.2 for details). It is therefore difficult to reach any kind of preliminary conclusions regarding the nature of language maintenance motivation within the two groups, and further information is needed. The reasons for the observed differences in response to the latter question are also unclear.

Differences were observed between the PSI and AB groups in terms of their perceptions of fluctuations in language dominance and direction of cross-linguistic interference between their two strongest languages. It is argued below that these differences are due to the influence of the L2 as the dominant linguistic environment of the PSI group, and as such may be considered evidence of the effect of the external contextual factor of linguistic environment on bilinguals' attitudes and perceptions. Firstly, it was observed that a slightly higher proportion of PSI than AB respondents felt that language dominance fluctuates between their two languages (45.6% and 37.3% respectively; see Section 6.4.2 for details). Initially, this result may seem surprising, in view of the findings for the AA group (recall that 50.0% of AA respondents were aware of language dominance shifts between their languages; see Chapter 6, Section 6.3 for details). Indeed, with regard to the AA/AB comparison, it was hypothesised that AA interpreters are likely to demonstrate greater awareness of language dominance shifts than AB interpreters, due to the increased proximity (in terms of relative competence) of their two languages (see Chapter 5, Section 5.1.2). As we saw in Chapter 2, the literature on bilingualism points to a strong negative correlation between the level of proficiency attained in a language and age of SLA. Thus, given the slightly lower mean age of SLA of the AB group compared with the PSI group, it could reasonably be expected that the languages of the AB group would be closer (in terms of relative competence) than those of the PSIs. As a consequence, if differences were to be observed between these groups with respect to awareness of fluctuations in language dominance, it would be reasonable to suppose that this awareness would be slightly higher among AB respondents than the PSIs. However, a possible explanation for the apparently contradictory finding is that, as a result of extensive (see Chapter 4) exposure to the L2, language L1 within the PSI group has undergone attrition, whereas L2 proficiency has improved, bringing the two languages closer together (in terms of relative competence) at a group level, compared with the AB population. It may therefore be concluded this greater proximity of the two languages (with respect to relative competence) has given rise to the observed differences in terms of awareness of language dominance shifts between the PSI and AB groups.

Secondly, differences between the AB and PSI groups were found with respect to the perceived direction of cross-linguistic interference (unnatural utterances made in a language due to the influence of the other language). In particular, while the proportions of respondents from the two groups who felt that interference occurs in neither language or in both equally were comparable (see Section 6.4.2 for details), the two groups were characterised by opposite trends in terms of interference in each of their languages separately. Specifically, while more PSI respondents reported interference in the L1 than in the L2 (30.6% and 15.6% respectively; Section 6.4.2), a greater proportion of AB respondents felt that interference occurs in language A rather than in language B (15.7% and 30.0% respectively; see Section 6.4.2). Once again, given that languages L1 and A are the first languages spoken by PSI and AB respondents, as well as the similar childhood linguistic environments and mean ages of SLA of the two groups, this observed difference in the groups' perceptions of the direction of cross-linguistic interference is likely to be due to the influence of the L2 as the dominant linguistic environment of the PSI group. Thus, the AB group behaves as anticipated, reporting that the influence of their second language on their first language is stronger than in the other direction. The PSIs, on the other hand, as a consequence of increased and prolonged exposure to their second language, notice the L2 exerting a greater influence on the L1. With regard to more general implications for bilingualism, the above two findings could be taken as evidence of the influence of external contextual factors on perceptions. In other words, with respect to subjective perceptions at least, the findings are in line with the DST ("embedded") view of second language development (see Chapter 3 for details).

As regards the questions concerning emotional response/attachment towards the two languages, the PSI and AB groups again exhibited very similar response patterns. However, the differences that were observed between the groups point to the influence of external contextual factors (age and context of acquisition, dominant language of the country of residence) on bilinguals' attitudes. First of all, while it was found that the largest proportion of both the PSI and AB groups indicated that they would revert to their first language (L1, A) if possible when tired/upset, with a minority of both groups choosing their second language (L2, B), some subtle differences were observed between the groups. Namely, the proportion of respondents who chose their first language was lower among the PSIs than the AB group (45.6% and 50.7% respectively; see Chapter 6, Section 6.4.2), the proportion who chose their second language was higher among PSIs than AB interpreters (13.9% and 8.6% respectively; see Chapter 6, Section 6.4.2). In view of the main biographical differences observed between the two groups (i.e. a slightly lower mean age of SLA among the AB group compared with the PSIs, and the increased exposure of the PSI group to the L2 as the current dominant linguistic environment), it is likely that the question regarding language preference when tired/upset does not actually measure emotional attachment as originally intended (see Section 7.2.1 above, where this was also noted), but possibly ease of expression in a language, particularly when under stress. This is based on the literature surveyed in Chapter 2, according to

which emotional attachment to a language is most greatly influenced of the age and context of SLA. Consequently, in light of the biographical differences observed between the PSI and AB groups, had this question measured attachment to the languages, a higher proportion of PSIs than AB interpreters would have chosen the first language as the language preferred when tired/upset, and the trend would have been reversed for the second language. The fact that this was not the case, and that the main characteristic that distinguishes the PSIs from the AB group is the increased, prolonged exposure to the L2 as their linguistic environment, suggests that this factor influenced the PSI group's responses. That is, as a result of this increased exposure to the L2, ease of expression in the L1 may have decreased (e.g. in terms of lexical availability; see Paradis, 2004), while it may have improved in the L2. Thus, respondents' language choice when tired/upset appears to be influenced not by a special emotional attachment to the language, but by how easily they are able to express themselves in that language. Such an interpretation explains both the general observation that the first language was the most popular choice among both groups, as well as the slight differences in language preference observed between the two groups.

Secondly, it was observed that, while both the PSI and AB groups gave similar responses to the question concerning perceptions of the relative weight of words in their two languages, with both groups split more or less evenly between those who felt that words in the L1 carry more weight (38.3% for PSI, 36.4% for AB; see Chapter 6, Section 6.4.2) and those who perceived this as being equal in both languages (35.0% for PSI, 36.8% for AB; see Chapter 6, Section 6.4.2), some subtle differences were found between the groups. We note from the foregoing that the percentage of PSI respondents who felt that words carry more weight in the first language is higher among the PSIs than the AB group, but the same trend was observed for the second language also – a slightly higher percentage of PSIs indicated that words carry more weight in the second language than AB respondents (11.7% and 8.6% respectively; see Chapter 6, Section 6.4.2). These results seem to be in line with the literature surveyed in Chapter 2, in particular Dewaele's (2004; 2008) findings that bilinguals' perceptions of the emotional force of words in their languages are influenced both by the age and context of SLA, as well as, in the case of the second language, the extent and duration of exposure to this language. Finally, with respect to a greater attachment to one language over the other, while both groups again exhibited similar trends (see Chapter 6, Section 6.4.2), the PSI group was characterised by a higher level of attachment to the first language than the AB group (44.4% and 40.2% respectively; see Chapter 6, Section 6.4.2), and a slightly lower level of attachment to the second language compared with AB respondents (10.6% and 12.4% respectively). On the basis of the literature, there are two possible (not necessarily mutually exclusive) explanations for the observed differences between the groups. The first is the influence of age of SLA and childhood linguistic environment. That is, the slightly higher age of SLA among the PSIs, as well as slightly greater exposure to the first language in childhood among this group (compared with the AB respondents) has resulted in a greater attachment towards the L1, and a

slightly lower attachment towards the L2 (see Chapter 2). Another possibility is the impact of prolonged exposure to the L2 (residence in an L2-speaking country) on the attitudes of the PSI group towards their languages. In particular, in one of the studies discussed in Chapter 2, Prescher (2007) found that, as the duration of immigration increased, the more the migrants tried to return to their original identity and language. In other words, the L1 becomes more important, and attachment to this language increases with time spent in the L2 environment. Both explanations are plausible, and warrant further investigation, for example, by controlling for country of residence and length of residence among the AB population and seeing whether they exhibit the same patterns as the PSIs with respect to language attachment.

A comparison of the PSI and AB groups' self-ratings of proficiency in their languages, both relative to each other and relative to a monolingual standard (language confidence) also revealed differences between the groups that could be attributed to the influence of the L2 as the dominant linguistic environment of the PSI group. In terms of self-rating of relative proficiency in the two languages, it was found that PSIs tend to perceive their two strongest languages as being closer in terms of proficiency compared with the AB group, in the three categories of command of registers, number of gaps in vocabulary and writing (see Chapter 6, Section 6.4.2 for full details). Specifically, the first language was selected by a lower percentage of the PSI respondents compared with the AB group, while the second language was selected by a greater percentage of PSI than AB respondents (see Chapter 6, Section 6.4.2 for figures). These findings suggest that, as a consequence of prolonged exposure to the L2 as the dominant language of the country of residence, PSI respondents' perceptions of proficiency in their first language were affected negatively, but positively in their second language. The AB group, on the other hand, was not characterised by a single dominant linguistic environment, and therefore their self-ratings of proficiency in their languages were influenced, at a group level, by childhood linguistic factors and age of SLA. Again, this hypothesis could be tested further by controlling for language of the country of residence and length of residence within the AB group, and seeing if those AB respondents who have been predominantly exposed to their second language for a prolonged period of time tend to exhibit the same response patterns in the three self-ratings categories as the PSIs.

As regards self-ratings relative to a monolingual standard in the categories of richness of vocabulary, instinctive/native comprehension and pragmatic cultural competence, which are taken to be a marker of language confidence, it was found that, on the whole, PSI respondents tended to feel more confident in the L1 compared with the L2 (see Chapter 6, Section 6.4.2 for full details). However, both languages tended to be perceived as being at least equal to the monolingual standards in the three categories considered. Comparing the findings for those of the AB group, it was observed that, overall, the PSIs exhibited a lower level of confidence in their first language than AB respondents, as well as a lower level of confidence in their second language than AB

interpreters (see Chapter 6, Section 6.4.2). At first glance, these findings may appear to be counterintuitive. While the lower confidence of the PSIs in their first language than that of the AB group could reasonably be explained as a consequence of first language attrition due to prolonged exposure to the second among the PSIs, the lower confidence in the second language observed in the PSI group runs counter to expectations. In other words, it is reasonable to expect that bilinguals who have spent a long time living in their L2 environment would feel increased confidence in this language. However, it is also possible that, at a group level, living for a prolonged period of time in the L2-speaking community may have the opposite effect on language confidence. That is, being surrounded by (monolingual) native speakers of the L2 may raise the bilinguals' own standards with respect to their L2 performance, and possibly make them more aware of the fact that their L2 use differs from that of monolingual natives. Consequently, bilinguals' confidence in their L2 may be negatively impacted (note that this does not necessarily correspond to actual L2 proficiency, which is likely to increase with prolonged exposure to the L2). This hypothesis may warrant further investigation, by controlling for country of residence and length of residence among the AB and PSI respondents and examining the impact of these factors on language confidence. This and other questions for further research identified as a result of the present study are discussed in the next Chapter.

#### 8. Conclusion

The final Chapter of the dissertation summarises what are considered to be the main findings of the study and outlines potential avenues for further inquiry, pertaining both to interpreting and bilingualism. The strengths and limitations of the study are also discussed.

## 8.1 Summary of key findings

The main findings obtained as part of the study are set out below, themed according to the research questions identified in Chapter 5 (Section 5.1.1).

## 8.1.1 Interpreters as highly proficient bilinguals

Several general observations can be made on the basis of a comparison of all three populations of interpreters surveyed. First of all, it is noted that the group that stands out from the three, in terms of both biographical and attitudinal factors, is the population of AIIC AA interpreters. Meanwhile, the AIIC AB and PSI groups exhibited similar response patterns. This suggests that, as argued by Thiéry (1975), AIIC AA interpreters, or "true bilinguals", are indeed a group of bilinguals that are set apart from other highly proficient bilinguals by a number of properties that appear to be unique to them (the specific characteristics of the AA group and their implications are discussed in Section 8.1.2 below). On the other hand, controlling for interpreting setting (PSI or conference; as represented by the AIIC AB and UK PSI groups) does not reveal major differences between groups as bilinguals.

Nevertheless, it can be concluded that there are some characteristics that appear to be common to all interpreters with two strongest working languages, irrespective of setting or AIIC AA/AB language classification. These include: a tendency to have acquired both languages before puberty; a tendency to use both languages on a daily basis for a range of activities that are not necessarily related to professional activity, socialising in particular, irrespective of the dominant language of the country of residence; awareness of language maintenance as a conscious effort; awareness of cross-linguistic interference between the two strongest languages; both professional and non-professional (instrumental and integrative; see Chapter 2) motivation to maintain the languages at a high standard; a degree of confidence in both languages (i.e. both languages are considered to be at least equal to the native monolingual standard). These findings are in line with the literature on bilingualism and the factors that influence bilingual proficiency (see Chapter 7, Section 7.1 for details). Furthermore, it was observed that the behaviour of all three groups with respect to relationships between external contextual factors (biographies) and attitudes and perceptions was in line with the predictions of the literature. In particular, the findings suggest that attachment towards

the languages, as well as the perception of the emotional force of words, are largely influenced by childhood linguistic factors (age of acquisition, relative exposure to the languages in childhood). Similarly, the childhood linguistic environment appears to exert the greatest influence on respondents' self-ratings of proficiency in their languages, as well as confidence in the languages (see Chapter 7, Section 7.1 for details). Despite these trends at a group level, individual differences were observed between respondents in all three groups (e.g. with respect to age of SLA), which is in line with DST principles (see Chapter 3).

## 8.1.2 Characteristics of AIIC AA interpreters and their implications

As noted above, differences were observed between the AIIC AA and AB groups, in terms of both biographical factors and attitudes and perceptions, suggesting that AA interpreters are, as argued by Thiéry (1975), a group of bilinguals with a unique set of properties. Before the specific differences observed between the AA and AB respondents are addressed, it is worth noting that the fact that the groups were found to differ as bilinguals is important in itself, in view of how an A language is distinguished from a B language. Recall that, as explained in Chapter 4, an AIIC member's language is classified as an A if listeners (peers) have the impression that it is the speaker's native language; otherwise, the language can only be classed as a B. Note that a B language in the context of professional conference interpreting is also of a very high standard, in view of the demands of the profession (see Chapter 4). In this context, the decision whether a language is native (A) or not (B) is taken intuitively; indeed, as pointed out by Valdés and Angelelli (2003), in the context of interpreting, expectations of interpreters' bilingual competence are often based in assumptions, as opposed to more rigorous definitions or categorisations. Therefore, the finding that AA and AB interpreters differ as bilinguals can be taken as indicative of the human ability to intuitively grasp the difference between a native and non-native language. It is this quality that "distinguishes a native speaker from someone who speaks the language beautifully for a foreigner" Thiéry's (1975), a quality which seemingly evades definition or measurement (see Chapter 2). We note also that the difference between an A and a B language is not as simple as a case of the presence or absence of a foreign accent in the language. This is based on the following findings of the present study and Thiéry's (1975) investigation of true bilingualism: (i) a small number of AA respondents did not consider themselves to be true bilinguals due to the presence of a slight accent in one of the languages, and (ii) a number of AA respondents were initially classed as AB upon joining AIIC, and only achieved the AA classification at a later date. Finding (i) implies that the presence of a slight foreign accent does not necessarily dispel the impression that the language in question is the speaker's native language. Finding (ii) may imply the same as (i), or may be indicative of the fact that the absence of an accent is not a sufficient condition for an A classification.

It was found that the AA respondents are set apart from the AB respondents as a group of bilinguals by the following characteristics: (1) a much greater tendency to identify as having two mother tongues; (2) a significantly lower mean age of SLA, and a smaller age range for SLA; (3) more equal representation of the two languages in childhood, and a greater tendency to have changed country of residence in childhood; (4) a slightly greater tendency to use both languages actively (for socialising, writing) in everyday life; (5) a slightly lower tendency to perceive language maintenance as requiring conscious effort; (6) a slightly higher level of non-professional (integrative) motivation to maintain both languages at a high standard; (7) more equal attachment to the two strongest languages; (8) a greater awareness of dominance shifts between the two languages; (9) a tendency to perceive cross-linguistic interference as being bi-directional, in contrast with the AB interpreters, who tend to view this as occurring mostly from A to B; (10) a greater tendency to regard both languages as being equal in terms of relative proficiency; (11) in terms of language confidence, as measured by self-ratings of proficiency relative to a monolingual standard, the AA group is characterised by two languages in which they have about the same confidence (both are perceived to be more or less equal to the monolingual standard), while the AB group exhibits a very high level of confidence in the A language (taken to be above the monolingual standard) and a lower level of confidence in the B language (see Chapter 7, Section 7.2 for details).

If the AA classification is taken to correspond to native-like competence in two languages, the above-listed differences between the AA and AB respondents have important implications for the study of bilingualism more generally. Finding (1) suggests that, as predicted by the literature, selfascribed native speaker identity tends to correspond with that which has been ascribed by others (although, as discussed in Section 8.1.3 below, this is not always the case). Findings (2) and (3) can be interpreted as the conditions that are favourable for the development of native-like competence in two languages. In particular, the findings confirm the predictions of the literature regarding the important role played by age and context of language acquisition in determining subsequent proficiency in the languages among bilinguals (see Chapter 2). At the same time, individual variation was observed within the AA group with respect to both age of SLA, which ranged from 0 to 18 years, as well as childhood linguistic environment. This is in line with DST principles (see Chapter 3), and also DeKeyser's (2013) assertion that, while there is a strong negative correlation between the age of L2 acquisition and many aspects of grammar and pronunciation, there some individuals do pass for native speakers, despite learning the language in later life (see Chapter 2). The finding that some of the AA interpreters acquired their second A language after puberty can be taken as evidence against the presence of a critical period for SLA, although it does appear that the likelihood of attaining native-like competence in a second language decreases with age. What is more, individual variation with respect to age of SLA and childhood linguistic environment was also observed among the AB interpreters, with some respondents from this group exhibiting similar

characteristics to those of the AA group (see Chapter 7, Section 7.2 for details). This finding implies that, while certain conditions in childhood appear to favour the development of native-like competence in two languages, there are no factors that can be considered necessary and/or sufficient for this to occur. Examining the findings from a DST perspective, it can be said that, while the bilingual language system is sensitively dependent on its initial conditions, it is not forever determined by these conditions, and there will always be variation at the individual level (see Chapter 3).

Finding (4) represents a preliminary conclusion, based on the limited assessment of daily language use among respondents. However, as pointed out in Chapter 7 (Section 7.2.1), the observation that AA interpreters appear to use both languages actively to a greater extent than AB interpreters in their daily life raises important questions as to the kind of language use associated with maintaining native-like competence in two languages, and whether there are any necessary and/or sufficient conditions for this to be possible. As indicated in the preceding Chapter (Section 7.2.1), finding (5) may be explained on the basis of Paradis' (2004) argument that processing in the first language (i.e. a language acquired at an early age and by immersion) is more automatic than that in the second language (typically acquired at a later age and by formal instruction). This greater automaticity results in performance that is more stable and resilient to change. Given that AA respondents typically acquired both languages before the age of 5 and by immersion (see Chapter 6), the finding that this group perceives language maintenance as requiring conscious effort to a lesser degree than AB respondents makes sense in this context. At the same time, it is possible that having two native languages is naturally accompanied by more extensive use of both languages in daily life (for example, in a social setting, as observed here), and this extensive use is not perceived as conscious effort (see Section 7.2.1). In any event, the foregoing points to the presence of group-level differences in perceptions of language maintenance effort between populations of bilinguals characterised by differing relative proficiency levels in their languages.

As argued in Chapter 7 (Section 7.2.1), in view of the connection between attitude and motivation in SLD (see Chapter 2 for details), it can be supposed on the basis of the literature that findings (6) and (7) are related. In other words, it is proposed that the finding of more equal attachment to both languages among the AA group has influenced this group's integrative motivation to maintain both languages at their present high standard. Taken separately, the finding of a more equal attachment to both languages among the AA group compared with the AB group, as well as AA respondents' greater tendency to perceive the emotional force of words (weight of words) as being equal in both languages, is in line with the literature and suggests that these two attitudinal factors are most strongly influenced by the childhood linguistic environment (see Chapter 2; Chapter 7, Section 7.2.1). Finding (8) was observed without controlling for the influence of other contextual factors, such as the main language of the country of residence or language use, and as such suggests that

AA respondents experience more shifts in language dominance between their languages compared with AB respondents, regardless of their external environment. This finding was anticipated on the basis of the DST literature surveyed in Chapter 3, and explained as being a consequence of the greater proximity of the languages in terms of relative competence within the AA group compared with the AB group (see Section 7.2.1 for details). Finding (9) was again observed without controlling for other factors that have a bearing on cross-linguistic interference and its direction, such as language use and exposure, which suggests that these trends are typical of the AA and AB groups, irrespective of their specific context at a given time. The fact that the majority of AA respondents reported making unnatural utterances in one or both of their languages due to the influence of the other language appears to support Grosjean's (1989) claim that bilinguals are not two monolinguals in one person (see Chapter 2), regardless of the level of competence attained in the languages. It is also worth noting that the presence of this cross-linguistic interference among the AA group does not appear to influence external perceptions of native competence; if the converse were true, these interpreters would not have been awarded the AA classification.

Findings (10) and (11) concern respondents' self-ratings of proficiency in their two strongest languages, relative to each other and relative to an average monolingual standard respectively. The latter is taken to be a marker of confidence in each language. The observation that AA respondents are more likely to view both languages as being equal in terms of proficiency, compared with AB respondents who favour language A, suggests, first of all, that self-ratings of bilingual proficiency tend to correspond to external ratings (see Chapter 2; Chapter 7, Section 7.2.1). Furthermore, these results were observed without controlling for other factors that influence relative proficiency levels in each language (e.g. language(s) of higher education, dominant linguistic environment). Given that subsequent language use/exposure varied among the AA and AB respondents, while strong trends were observed with respect to childhood linguistic factors and age of SLA, the results suggest that age and context of SLA have the biggest impact on bilinguals' self-ratings of proficiency in their languages, as well as language confidence. Finding (11) also appears to imply that having one distinct native language results in greater confidence in that language, compared with those bilinguals who have two native languages, or where the distinction between native and foreign is less clear-cut. If finding (11) is interpreted not in terms of language confidence, but instead is taken to signify that, at least in terms of subjective perceptions, the AA group can be characterised by two languages of about the same proficiency, while the AB group is characterised by one very strong language and one weaker language, this result can be viewed as corresponding to the DST principle of economy of effort (Herdina and Jessner, 2002; Larsen-Freeman and Cameron, 2008; see Chapter 7, Section 7.2.1 for details and explanation).

More generally, the comparison of the AA and AB groups and the main trends observed point to the existence of relationships between external and internal contextual factors within the bilingual

language system. This can be taken as evidence in support of the DST view of bilingualism, according to which second language development can only be understood as unfolding in context ("embedded" approach), where all subsystems are interconnected (see Chapter 3). Other findings pertaining to the AA group also appear to support the view that bilingualism behaves like a dynamic system, in particular, with respect to the principle of continuous temporal evolution. Evidence in support of this was observed both in terms of subjective and objective (external) perceptions. With regard to the former, it was found that perceptions of mother tongue competence/identity in childhood did not necessarily correspond to those in later life, which is indicative of shifting perceptions of identity and/or competence over the life span (see Chapter 7, Section 7.2.2 for details). With respect to the latter, it was observed, in line with Thiéry's (1975) findings, that a subgroup of AA respondents were awarded their AA classification at a later date, and not immediately upon joining AIIC (i.e. they were originally classed as AB). This finding is in line with the DST view of the bilingual language system as undergoing continuous change under the influence of contextual factors. In other words, in the case of the subgroup of AA interpreters who obtained their double A classification at a later date, it appears as though their language system had stabilised at AB level (an attractor or steady state; see Chapter 3), however, as a result of increased use of the B language in a professional context, the system entered a new state, in which both languages are considered native by others.

The above-mentioned subgroup of AA interpreters who obtained their AA classification at a later date is considered to be a particularly interesting group, the study of which may provide important insights into individual bilingualism. For example, insights may be gained into such aspects as why certain individuals are able to achieve native-like proficiency into two languages, whereas others are not, and what the most important factors are in determining ultimate attainment in a second language (and maintenance of the first). As part of the present study, the characteristics of the subgroup of AA respondents who obtained their AA classification later were investigated further, and compared with the rest of the AA group. Given the small size of the population (n = 7), it is not possible to generalise on the basis of this inquiry; at the same time, the comparison made it possible to test some preliminary hypotheses. It was found that the majority of the hypotheses advanced in Chapter 5 were not confirmed by the findings (see Chapter 7, Section 7.2.3 for details). In particular, whether the AA classification was obtained at a later date or not appeared to have no relationship to mother tongue identity in two languages, at present or in childhood, or to whether respondents tended to consider themselves true bilinguals as defined by Thiéry (1975). Furthermore, no differences were observed in terms of age of SLA and relative exposure to the two languages in childhood between the groups. This observation could be interpreted as being in support of Thiéry's claim that, in the case of the AIIC AA classification/true bilingualism, everything is determined in childhood, and it is this foundation that plays the most important role in whether the bilingual is considered to have two native languages in later life. At the same time, it is

noted that a more fine-grained analysis is needed to capture differences in exposure to the two languages in childhood, which was not possible on the basis of the data obtained as part of the present study (see Section 8.2 below for suggestions of alternative methodology). As regards attitudes and perceptions of the subgroup of AA respondents who were awarded their AA classification later, the small sample size made it difficult to reach reliable conclusions (see Section 7.2.3). It was tentatively concluded that the observed slight differences in language confidence/self-ratings of proficiency between the groups (see Section 7.2.3 for details) may be indicative of a link between external and subjective perceptions of proficiency, although further information is needed to determine whether such differences in perceptions are due to the influence of external proficiency ratings (see Section 8.2 below).

Finally, some observations can be made regarding the implications of Thiéry's (1975) true bilingualism definition and AA respondents' perceptions of it. While almost all AA respondents indicated that they consider themselves to be true bilinguals, a lower proportion of the group felt that the AA classification and Thiéry's definition were equivalent (91.7% and 72.9% respectively; see Section 7.2.2 for details). On the basis of additional comments received from respondents, as well as Thiéry's original findings, a possible explanation for this was the suggestion that some respondents interpreted the true bilingualism definition as implying bilingual balance (see Section 7.2.2 for details). The relationship between mother tongue identity in two languages (at present and in childhood) and true bilingualism was also considered. It was found that, with respect to subjective perceptions at least, true bilingualism is not always accompanied by the presence of two mother tongues in childhood. The findings also suggest that identifying as having two mother tongues is a stronger concept than true bilingualism (see Section 7.2.2). This observation is in line with reasonable expectations, given that true bilingualism addresses the external/objective component of bilingual identity only, whereas the idea of the mother tongue has certain emotional connotations, and implies a degree of emotional attachment to the language(s) (see Chapter 2; Section 7.2.2). It was additionally found that, while true bilingualism does tend to be accompanied by true bilingualism in writing, at least according to subjective perceptions, this is not always the case, and one does not necessarily imply the other (see Chapter 7, Sections 7.2.2, 7.2.1).

## 8.1.3 Factors associated with self-ascribed mother tongue identity in two languages

An examination of the responses of the AIIC AA and AB groups revealed the presence of a subgroup of respondents within each group whose own perceptions of mother tongue identity in their languages were not in line with external perceptions. Specifically, 14.6% (7 respondents) of the AA group did not feel confident that both of their A languages are mother tongues, while 13.2% (28 respondents) of the AB group indicated that they consider both languages to be mother tongues (see Section 7.3). These findings imply that, while self-ascribed native speaker identity does tend to

correspond to that which has been ascribed by others (as represented by the AIIC A and B classifications), this is not always the case. The characteristics of the above anomalous subgroups of AA and AB respondents were compared with the main group in each case, in order to determine which factors appear to be the most important in shaping self-ascribed mother tongue identity, and whether the groups exhibit differences with respect to attitudinal factors. It is noted that, in view of the small size of the population of AA respondents who do not consider both languages to be mother tongues, it is not necessarily possible to generalise on the basis of these findings, however, the comparison did provide the opportunity to test some preliminary hypotheses. Overall, in the case of both the AA and AB groups, the investigation of the responses of the subgroups whose self-ascribed identity did not correspond to identity ascribed by others were in line with expectations based on the literature and the proposed hypotheses (see Section 7.3; Chapter 5).

The first key finding is that differences in childhood linguistic environment, including mean age of SLA, were observed between the anomalous subgroups and the main groups in both the AA and AB cases. Specifically, AA respondents who did not necessarily feel that both languages are mother tongues had a higher mean age of SLA than the main AA group (8 years and 3.7 years respectively), and less equal exposure to both languages in childhood than the main group, in particular in the setting of the home, where the first language dominated in 6 out of 7 cases (see Section 7.3.1). The opposite trend was observed for the subgroup of AB respondents who considered both languages to be mother tongues, i.e. this subgroup had a lower mean age of SLA than the main AB group (4.8 years and 9.2 years respectively) and was characterised by more equal exposure to the two languages in childhood compared with the main AB group, with the greatest difference observed in the setting of the home. Taken together, the findings correspond to the predictions of the literature, that age and context of acquisition are highly influential in terms of shaping identity/emotional response to the languages (if the concept of the mother tongue is assumed to carry emotional connotations). The findings for both groups suggest that a key consideration in this respect appears to be whether a language was acquired/used in an emotionally neutral setting (e.g. school), or a setting that, on the contrary, is associated with a greater emotional response (e.g. the home), which is in line with the literature surveyed in Chapter 2. The conclusion regarding emotional attachment is also supported by the findings comparing relative attachment to the languages between the anomalous subgroups and the main groups. Specifically, it was found that AA respondents who do not necessarily consider both languages to be mother tongues had a greater attachment to A1 (the first language) compared with the main group (see Section 7.3.1). The opposite trend was observed among AB respondents who do consider both languages to be mother tongues, i.e. this subgroup was characterised by a more equal attachment to both languages compared with the main AB group, which in turn exhibited a greater attachment to language A (see Section 7.3.2 for details).

Differences were observed between the anomalous subgroups and the main groups in both the AA and AB cases, which largely reflected the influence of the difference in age of SLA and childhood linguistic environment as predicted by the literature (see Section 7.3 for full details). In particular, it was observed that the response patterns of the subgroup of AB respondents who consider both languages to be mother tongues tended to mirror those of the full AA group (see Section 7.3.2). In view of these similarities, the question arises of what it is that makes the subgroup of AB respondents who feel that both languages are mother tongues different from the AA group, i.e. why is it that their languages are not considered native by external reviewers, at least in the context of interpreting? Conversely, it could be asked why is it that AA respondents who do not necessarily consider both languages to be mother tongues are considered to have two native languages by others, in spite of the observed differences in the key areas of age of SLA and childhood linguistic environment with the main AA group? These questions, among others, are discussed in more detail in Section 8.2. Finally, the finding of differences in terms of both biographical and attitudinal factors between groups of bilinguals whose self-ascribed identity in both languages does not correspond to that ascribed by others highlights the complex nature of bilingualism, the numerous factors at play and the non-linear relationships between proficiency, identity, subjective and external perceptions.

## 8.1.4 Implications of the comparison of the AIIC AB and PSI groups for bilingualism

The comparison of the AB and PSI groups as bilinguals revealed the groups to be largely similar in terms of biographical factors, attitudes and perceptions (see Chapter 7, Section 7.4 for details). However, an important difference was observed between the groups with respect to the dominant linguistic environment of the country of residence, which was mixed for the AB respondents, but predominantly the L2 among the PSI group (see Section 7.4.1). In view of the largely similar childhood linguistic conditions and mean ages of SLA of the two groups, it was argued in Chapter 7 that the differences in attitudes/perceptions that were observed between the groups were indicative of the influence of the external factor of linguistic environment on bilinguals' subjective perceptions. In other words, the comparison of the AB and PSI groups as bilinguals made it possible to isolate the factor of dominant linguistic environment and examine its impact on internal environmental factors, and to advance hypotheses on the basis of the literature. Before turning to the specific trends observed, it is worth noting that the finding of possible relationships between the external contextual factor of language of the country of residence and internal I factors (attitudes and perceptions) is in itself important, as it supports the DST approach to SLD, emphasising the importance of contextual factors in the study of bilingualism.

The main differences in attitudes and perceptions observed between the PSI and AB groups were in the following areas: (1) perceptions of fluctuations in language dominance; (2) perceptions of the

direction of cross-linguist interference; (3) aspects of emotional response/attachment to the languages; (4) self-ratings of proficiency in the languages, both relative to each other and relative to a monolingual standard (language confidence). As regards finding (1), it was observed that a slightly higher proportion of PSI than AB respondents felt that language dominance fluctuates between their two languages (see Section 7.4.2). As previously argued in the context of the AA/AB comparison (see Chapter 7, Section 7.2.1), a greater awareness of language dominance shifts is taken to be indicative of greater proximity (in terms of relative competence) of the two languages. In view of the slightly lower mean age of SLA observed among the AB group compared with the PSI group, it was proposed that the observed difference between the PSI and AB groups with respect to awareness of language dominance fluctuations is a consequence of the PSI group's increased exposure to the L2. Specifically, it was hypothesised that language L1 within the PSI group has undergone attrition, whereas L2 proficiency has improved, bringing the two languages closer together (in terms of relative competence) at a group level, compared with the AB interpreters. This, in turn, resulted in the increased awareness of fluctuations in language dominance observed among the PSI group.

Finding (2) is particularly striking, as it seems to be directly indicative of the influence of the language of the country of residence on bilinguals' perceptions of cross-linguistic interference. It was found that, while the AB group reported that the influence of their second language on their first language is stronger than in the other direction, the PSI group exhibited the reverse trend, tending to view interference as occurring in the L2-L1 direction (see Section 7.4.2 for details). As regards finding (3), the differences observed between the groups in these categories were more subtle, but potentially indicative of important relationships between factors. The results pertaining to the perception of the emotional force of words (words weight) were in line with the literature (e.g. Dewaele, 2004; 2008), suggesting that bilinguals' perceptions in this regard are influenced both by the age and context of SLA, as well as, in the case of the second language, the extent and duration of exposure to this language (see Section 7.4.2 for details). A comparison of the groups' responses to the question on language attachment showed that the PSI group tended to exhibit a greater attachment to their first language compared with the AB group, and a lower level of attachment to their second language than the AB interpreters (see Section 7.4.2 for figures). Two possible, not necessarily mutually exclusive, explanations for these findings were proposed. Namely, that the observed trends are due to the slight differences in age of SLA/childhood linguistic environment among the PSI and AB groups, and/or the impact of prolonged exposure to the L2 among the PSIs. In the first case, the PSIs are characterised by a slightly higher mean age of SLA, and slightly greater exposure to the first language in childhood, compared with the AB group which, on the basis of the literature, may have resulted in the observed differences in language attachment. In the second case, it has been found by Prescher (2007) that among L1-speaking migrants living in their L2 environment, as length of residence in the L2 country increased, so did

the migrants' desire to return to their original L1 identity and languages. Therefore, prolonged residence in the L2 linguistic community may also have contributed to the differences in language attachment observed between the groups (see Section 7.4.2)

Finding (4) gave rise to two main observations. Firstly, with regard to self-ratings of relative proficiency in the two languages, it was observed that PSIs tend to perceive their two strongest languages as being closer in terms of proficiency compared with the AB group, in the three categories of command of registers, number of gaps in vocabulary and writing. Secondly, it was observed that, overall, the PSIs exhibited a lower level of confidence in their first language than AB interpreters, as well as a lower level of confidence in their second language than AB interpreters (see Section 7.4.2 for details). The first observation can be taken as a pointing to the greater proximity in terms of relative competence of the two languages within the PSI group, as argued in the context of finding (1) above. The second result may initially appear counter-intuitive. That is, while a decrease in L1 confidence among the PSIs may be expected due to prolonged exposure to the L2 linguistic environment, a decrease in L2 confidence seems to go against expectations. However, as argued in Chapter 7 (Section 7.4.2), it is possible that being surrounded by (monolingual) native speakers of the L2 may raise the bilinguals' own standards with respect to their L2 performance, and possibly make them more aware of the fact that their L2 use differs from that of monolingual natives, negatively impacting L2 confidence. It must be noted these self-ratings do not necessarily correspond to actual L2 proficiency, which is likely to increase with prolonged exposure to the L2. Overall, the comparison of the AB and PSI groups has given rise to several new hypotheses, which were not anticipated at the start of the study. Suggestions for how these preliminary explanations could be tested further are given in the next section.

#### 8.2 Directions for future research

The present study and analysis of findings has raised some additional questions, helping to identify potential areas for further investigation, some of which have already been mentioned briefly in this and the preceding Chapter. These are divided below into the broad categories of questions pertaining to the AIIC AA and AB groups, and further questions raised by the AB/PSI comparison.

## 8.2.1 Questions pertaining to the AIIC AA and AB groups

The following list of further questions is by no means exhaustive, and it is proposed that the AIIC register of interpreters with associated language classifications – effectively, a list of highly proficient bilinguals with relative proficiency ratings in each language – is a valuable resource for the study of individual bilingualism (and multilingualism). The main directions set out below are questions that have arisen directly as a consequence of the results obtained in the present study.

## (1) The necessary and/or sufficient conditions for native-like attainment in two languages

The comparison of the AA and AB groups conducted as part of the present study revealed differences with respect to the age and context of language acquisition between the two groups. This general observation emphasises the importance of the role played by exposure to the languages in childhood and the age of acquisition in determining subsequent bilingual proficiency. However, the information collected as part of the present study simplifies the childhood linguistic environment into several main components (see Chapter 5), and therefore potentially important differences between the groups may have been missed. In particular, it was found that some AB respondents had similar childhood linguistic conditions as the AA group (e.g. the subgroup of AB respondents who consider both languages to be mother tongues), and vice versa. This raises the question of why are some bilinguals considered to have two native languages whereas others or not, in spite of similar childhood linguistic conditions? Conversely, why do some AA interpreters with childhood linguistic environments that are more typical of the AB group still obtain the AA classification?

Several explanations are possible, such as more subtle differences in childhood linguistic environment, which were not picked up by the questionnaire; differences in subsequent language use/exposure, which were not picked up by the questionnaire; differences in motivation to acquire/maintain two languages at native-like standard; a result of individual variation. On the other hand, it is possible that the AA and AB classifications are connected to conference interpreting rather than bilingualism (something alluded to by Thiéry (1975) in his justification for including interpreters with major international languages only in his study; see Chapter 4). These questions could be investigated further by means of more detailed questionnaires, making a more fine-grained analysis of the differences between groups possible, and/or by conducting in-depth interviews with anomalous groups of respondents (i.e. those AA and AB interpreters whose ages of SLA and childhood linguistic environments are not typical of the main group).

# (2) The necessarily and/or sufficient conditions for the maintenance of two languages at native-like standard

The comparison of the AA and AB groups revealed some differences in self-reported daily language use in a non-professional setting, in particular, it was observed that AA respondents tend to use both languages actively (for socialising, writing) on a daily basis to a greater extent than AB interpreters. However, similarly to (1) above, the data on language use collected as part of the present study represents a simplification. Consequently, it was not possible to reach more definite conclusions as to the presence or nature of language use differences between the groups. This question could be investigated further by collecting more detailed information about AA and AB

interpreters' use of both languages in everyday life. A related question is, if differences in language use outside the professional setting are indeed observed between AA and AB interpreters, the extent to which these differences are the result of a conscious effort (e.g. the decision to use both languages actively daily in order to maintain them at a native-like level) on the part of the interpreters. To this end, interviews could be conducted with interpreters from each group, in order to gain further insight into their perceptions of language maintenance effort and the conscious/unconscious decisions behind language use.

## (3) A closer look at differences in motivation and attitude between the AA and AB groups

The findings of the present study revealed differences between the AA and AB groups in terms of levels of non-professional (integrative) motivation to maintain both languages, as well as emotional attachment to their languages. However, for practical reasons (see Chapter 5), the number of questions that addressed these aspects of the bilinguals' internal environment were limited. Therefore, the observed differences in motivation and attitude/attachment to the languages between the groups warrant further investigation. This could be done by means of detailed questionnaires designed for this purpose, such as the survey used by Cherciov (2011) in her exploration of the role of attitude in first language attrition. Additionally, a more thorough investigation of differences in emotional attachment to the languages could be accompanied by a more detailed analysis of respondents' childhood linguistic environment, in order to determine which factors, if any, appear to be particularly influential in shaping subsequent emotional attachment to the languages.

# (4) A closer look at AA respondents who were awarded their double A classification at a later date

The results of the study revealed that there is a subgroup of AA interpreters who were initially classed as AB upon joining AIIC, and obtained their AA classification at a later date. The responses of this subgroup were analysed in more detail and compared with those of the main AA group. However, the majority of the hypotheses advanced were not confirmed, and differences between the groups were generally not observed. At the same time, it is proposed that a more detailed study of this subgroup of AA interpreters may provide insights into such aspects of bilingualism as the factors that influence ultimate L2 attainment, the relationships between external and subjective perceptions of bilingual proficiency/identity, and the conditions/effort required to bring a language originally considered foreign to a native-like standard. Additionally, it could be explored whether the shift in external perceptions of the two languages in the context of interpreting (i.e. "upgrade" from AB to AA classification) tends to be accompanied by a similar change outside this setting. In other words, whether the respondents' two languages were considered native by members (non-interpreters) of both linguistic communities even when they were classed as AB in the sphere of

conference interpreting, of whether external perceptions outside interpreting underwent a similar shift. Such a study could be conducted by means of in-depth semi-structured interviews with the participants.

## (5) Self-ratings of proficiency among AA and AB interpreters: perceptions vs. reality

Differences in self-ratings of bilingual proficiency were observed among the AA and AB groups, both in the languages relative to each other and relative to a monolingual standard ("language confidence"). Some experimental data could be collected in order to investigate the extent to which such subjective perceptions of proficiency in each language are in line with objective findings. In particular, with respect to richness of vocabulary, the ratings attributed by AB respondents to their A language were higher than those attributed by AA respondents to either of their languages (A1, A2), whereas language B in the AB group seemed to be ranked slightly below the AA groups' languages (see Chapter 6, Section 6.3 for details). Thus, an experiment could be conducted to measure lexical richness of the AA and AB groups in each language. An example of a study looking at lexical richness among highly proficient bilinguals is described by Treffers-Daller (2011), and such an approach could be adopted in the case of the AA and AB groups. The experimental findings would then be compared with respondents' own perceptions of lexical richness in their languages. Such a comparison would not only provide insights into the degree to which the interpreters' own perceptions correspond with objective data, but also into what the AIIC A and B language classifications correspond to in practice, and the language dominance patterns among AA and AB interpreters. Furthermore, the hypothesis advanced in Chapter 7 (Section 7.2.1) that, in accordance with the DST principle of economy of effort, the AA groups' languages are maintained at a similar standard, while language A in the AB group is maintained at a slightly higher standard than this, with language B in the AB group at a slightly lower standard, could be tested using the empirical data obtained.

## 8.2.2 Questions raised by the comparison of the PSI and AIIC AB groups

The comparison of the AB and PSI groups gave rise to some important questions, in particular, with respect to the influence of the factor of dominant linguistic environment on bilinguals' attitudes and perceptions. Hypotheses were proposed with respect to the impact of the language of the country of residence on the following: perceptions of fluctuations in language dominance; perceptions of the direction of cross-linguistic interference; emotional attachment to the languages; self-ratings of proficiency/confidence in the languages (see Section 8.1.4 above for details). However, these hypotheses are based on a general comparison of the AB and PSI groups, without controlling for important contextual factors such as length of residence in the present linguistic community. Therefore, in order to investigate the above hypotheses further, more information is

needed from the participants with regard to language(s) of the country of residence and length of residence, as well as childhood linguistic environment factors. Controlling more carefully for these conditions among the participants would make it possible to examine in more detail the impact of the isolated factor of dominant linguistic environment (language of the country of residence), as well as length of residence, on attitudes and perceptions. For example, such an investigation could look into the effect of the length of residence in an L2-speaking country on attachment to the languages among L1-speaking migrants, and test the hypotheses that attachment to the L1 increases, while attachment to the L2 decreases, as length of residence increases (see Section 8.1.4 above, and also Prescher (2007), discussed in Chapter 2). The advantage of using interpreters in such a study is that it is then possible to control for at least some of the other important factors, as some things are already known about this group of bilinguals (e.g. the presence of strong motivation to maintain the languages), and these can be taken as being more or less constant within the group (see Chapter 3, Section 3.4.2; Chapter 4, Section 4.3).

In light of the foregoing, it is proposed that the PSI group is a valuable population for the study of first language attrition among L1-speaking migrants living in their L2 environment. As argued in Chapter 3, given the large number of interacting factors in second language development, in studies looking to isolate and examine the influence of a specific factor (e.g. attitude, motivation) on an aspect of bilingualism, it is important to choose the population carefully so as to reduce the contribution of other factors. That is, select a population of bilinguals in which some of the other variables can be taken as being more or less constant at a group level (see Chapter 3, Section 3.4.2 for details). An example of such a study could be an investigation that is methodologically similar to Cherciov's (2011) inquiry into the role of attitude on first language attrition among L1-speaking migrants living in their L2 linguistic community, but involving PSIs (with additional selection criteria to control for factors such as length of residence, age of migration etc.) An investigation of this kind may make it possible to better isolate and study the effect of the factor of attitude on L1 attrition among highly proficient bilinguals. In addition, as noted in Chapter 2, it has been observed by some researchers (e.g. Isurin, 2007; Laufer, 2003) that the effect of living in the L2 environment for a prolonged period of time is such, that L1 attrition occurs in spite of strong motivation to avoid it, and despite regular use of the L1. This finding could be tested further with PSIs as the population of bilinguals, as this group is characterised by a high level of motivation to maintain the L1 in its "pure" form, due to professional requirements, as well as intensive regular use of the language in a variety of contexts. Furthermore, if any participants are found to be particularly successful at counteracting the effects of extensive exposure to the L2 on the L1, this group could be studied further, for example, through in-depth interviews, in order to better understand which conditions are the most important in terms of preserving the integrity of the L1 among migrants living in their L2 environment.

### 8.3 Strengths and limitations of the study

On the whole, the present study can be considered successful, as the information gathered from respondents was sufficient to enable the investigation of the research questions set. In particular, it is noted that a large volume of data has been obtained from all of the populations of interest, suggesting that the research topic and questionnaire managed to draw respondents' attention and arouse their interest. In addition, the questionnaire was able to cover a wide range of different factors, while maintaining a balance between questionnaire length and response rate. Finally, an analysis of the findings obtained as part of the study has made it possible to not only explore the original research questions and hypotheses, but has revealed potential avenues for further investigation, and yielded new hypotheses.

That said, the study has a number of shortcomings, some of which are regarded as necessary compromises, whereas others would need to be remedied if the study were to be repeated. In the first case, as has already been noted at various points in Chapters 7 and 8 of the dissertation, much of the data obtained represents a simplification. For example, respondents' childhood linguistic environment has been taken to consist of the three main components of the home, country of residence and primary school; relative daily language use was measured using only a limited number of categories; factors such as motivation or emotional attachment to the languages were measured using a limited number of questions. However, as pointed out in Chapter 5, a compromise had to be reached between the level of detail included in the questionnaire and its overall length, to ensure both sufficient information and an adequate response rate. In the second case, there were some shortcomings in the formulation of the questions, which were not picked up on during the pilot run of the questionnaire. For example, the questions concerning respondents' countries of birth and current residence (Q1.2-3/AA/AB/PSI) were actually intended to establish the dominant linguistic environment in terms of respondents' two strongest languages (i.e. L1, L2, both or neither). However, at the time of writing the questionnaire, countries with two official languages were not taken into account (Canada, Belgium etc.) Therefore, valuable information was lost in these cases, as the linguistic environment had to be coded as "missing" or "unclear". Similarly, the formulation of the questions concerning relocation to a different country in childhood made it difficult to track the change in linguistic environment, and consequently the responses were presented as non-language specific. Meanwhile, certain questions included in the questionnaire did not yield any useful information, and were subsequently discarded from the final analysis (e.g. Os 4.4-6/AA/AB/PSI on interpreting; the question pertaining to respondents' nationality).

Despite the limitations of the questionnaire, it is believed that the present study has made a contribution to the literature at the intersection of interpreting and bilingualism, making steps towards answering some of the questions raised by Thiéry's (1975) inquiry into true bilingualism,

as well as those identified by Valdés and Angelelli (2003). In addition, it is hoped that the present study has served to bring attention to the potential of the AIIC register of interpreters with associated classifications as a valuable resource for bilingualism research, in particular in view of the difficulties involved in measuring bilingual proficiency. Finally, it is also hoped that the study has managed to highlight the fact that research on interpreting and bilingualism need not be focused on conference interpreters only; the inclusion in such studies of interpreters who specialise in other settings, such as the public services, would serve to enrich and broaden the scope of the research.

#### References

AIIC website. Available at: http://aiic.net/. [Last accessed 1 March 2017].

Aneja, G.A. 2016. "Rethinking Nativeness: Towards a Dynamic Paradigm of (Non)Native Speakering." *Critical Inquiry in Language Studies* 13(4):351-379.

Baba, K. and Nitta, R. 2014. "Phase Transitions in Development of Writing Fluency From a Complex Dynamic Systems Perspective." *Language learning* 64(1):1-35.

Babcock, L. and Vallesi, A. 2017. "Are simultaneous interpreters expert bilinguals, unique bilinguals, or both?" *Bilingualism: Language and Cognition*, 20(2): 403-417.

Baetens Beardsmore, H. 1986. *Bilingualism: basic principles*. Clevedon: Multilingual Matters Ltd 2<sup>nd</sup> ed.

Baigorri-Jalón, J. 2015. The history of the interpreting profession. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 11-28.

Bancroft, M.A. 2015. Community Interpreting: A profession rooted in social justice. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 217-235.

Bardovi-Harlig, K. 2012. Pragmatics in second language acquisition. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 147-162.

Ben-Rafael, M. and Schmid, M.A. 2007. Language attrition and ideology: Two groups of immigrants in Israel. In: B. Köpke, M.S. Schmid, M. Keijzer and S. Dostert, eds. Language Attrition: Theoretical Perpectives. Amsterdam/Philadelphina: John Benjamins, pp. 205-226.

Birdsong, D. 2005. "Nativelikeness and Non-Nativelikeness in L2A research." *International Review of Applied Linguistics in Language Teaching (IRAL)* 43(4):319-28.

Birdsong, D. 2006. "Age and Second Language Acquisition and Processing: A Selective Overview." *Language Learning* 56(1):9-49.

Birdsong, D. 2014. "Dominance and Age in Bilingualism." Applied Linguistics 35(4):374-392.

Bloomfield, L. 1969. Language. London: Allen and Unwin.

Bongaerts, T., Mennen, S. and van der Silk, F. 2000. "Authenticity of pronunciation in naturalistic second language acquisition: The case of very advanced late learners of Dutch as a second language." *Studia Linguistica* 54(2):298-308.

Bontempo, K. 2015. Signed language interpreting. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 112-128. Brace, N., Kemp, R. and Snelgar, R. 2009. *SPSS for psychologists 4<sup>th</sup> ed.* Basingstoke: Palgrave Macmillan.

Braun, S. 2015. Remote interpreting. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 352-367.

Brutt-Griffler, J. and Samimy, K.K. 2001. "Transcending the Nativeness Paradigm." *World Englishes* 20(1):99-106.

Brysbaert, M. and Duyck, W. 2010. "Is it time to leave behind the Revised Hierarchical Model of bilingual language processing after fifteen years of service?" *Bilingualism* 13(3):359-371.

Cherciov, M. 2011. Between Attrition and Acquisition: the Dynamics between Two Languages in Adult Migrants. Thesis (Ph.D.), University of Toronto.

Cherciov, M. 2013. "Investigating the impact of attitude on first language attrition and second language acquisition from a Dynamic Systems Theory perspective." *The International Journal of Bilingualism* 17(6):716-733.

Cheung, K.F.A. 2013. "Non-native accents and simultaneous interpreting quality perceptions." *Interpreting* 15(1): 25-47.

Choi, L.J. 2015. "Revisiting the issue of native speakerism: 'I don't want to speak like a native speaker of English'." *Language and Education* 30(1):72-85.

Chomsky, N. 1969. Aspects of the theory of syntax. Cambridge, Mass.: M.I.T. Press.

Christoffels, I.K. and de Groot, A.M.B. 2005. Simultaneous Interpreting: A Cognitive Perspective. In: J.F. Kroll and A.M.B. de Groot, eds. Handbook of Bilingualism: Psycholinguistic approaches Oxford: Oxford University Press, pp. 454-479.

Christoffels, I.K., de Groot, A.M.B. and Waldorp, L.J. 2003. "Basic skills in a complex task: A graphical model relating memory and lexical retrieval to simultaneous interpreting." *Bilingualism: Language and Cognition*, 6(3):201-211.

Collins English Dictionary, online resource. Definition of bilingualism available at: http://www.collinsdictionary.com/dictionary/english/bilingualism. [Last accessed 1 March 2017].

Cook, V.J. 2003. The effects of the second language on the first. Clevedon: Multilingual Matters.

Cook, V. 2013. Linguistic Contributions to Bilingualism. In: J. Altarriba and R.R. Heredia, eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 245-264.

Corsellis, A. 2008. Public Service Interpreting: The First Steps. UK: Palgrave Macmillan.

Daller, M.H. 2011. "The measurement of bilingual proficiency: introduction." *International Journal of Bilingualism*, 15(2):123-127.

Davies, A. 2003. The Native Speaker: Myth and Reality. Clevedon; Multilingual Matters 2<sup>nd</sup> ed.

de Bot, K. 2007. Dynamic systems theory, lifespan development and language attrition. In: B. Köpke, M.S. Schmid, M. Keijzer and S. Dostert, eds. Language Attrition: Theoretical Perpectives. Amsterdam/Philadelphina: John Benjamins, pp. 53-68.

de Bot, K. 2008. "Introduction: Second Language Development as a Dynamic Process." *The Modern language journal* 92(2):166-178.

de Bot, K., Lowie, W., and Verspoor, M. 2007. "A Dynamic Systems Theory approach to second language acquisition." *Bilingualism* 10(1):7-21.

de Bot, K., Lowie, W., Thorne, S.L. and Verspoor, M. 2013. "Dynamic Systems Theory as a Theory of Second Language Development." In M. Mayo, M. Gutierrez-Mangado and M. Adrián, eds. Contemporary Approaches to Second Language Acquisition. Amsterdam: John Benjamins, pp. 199-220.

de Bruin, L.C. and Kastner, L. 2012. "Dynamic Embodied Cognition." *Phenomenology and the Cognitive Sciences* 11(4):541-563.

de Leeuw, E., Mennen, I. and Scobbie, J.M. 2012. "Singing a different tune in your native language: first language attrition of prosody." *The International Journal of Bilingualism* 16(1):101-116.

de Leeuw, E., Mennen, I. and Scobbie, J.M. 2013. "Dynamic systems, maturational constraints and L1 phonetic attrition." *The International Journal of Bilingualism* 17(6):683-700.

Degani, T., Prior, A. and Tokowicz, N. 2011. "Bidirectional transfer: The effect of sharing a translation." *Journal of Cognitive Psychology* 23(1):18-28.

DeKeyser, R. 2013. "Age Effects in Second Language Learning: Stepping Stones Toward Better Understanding." *Language learning* 63(1):52-67.

Dewaele, J.-M. 2004. "The Emotional Force of Swearwords and Taboo Words in the Speech of Multilinguals." *Journal of Multilingual and Multicultural Development* 25(2-3):204-222.

Dewaele, J.-M. 2008. "The emotional weight of *I love you* in multilinguals' languages." *Journal of Pragmatics* 40(10):1753-1780.

Dewaele, J.-M. 2010. Emotions in Multiple Languages. UK: Palgrave Macmillan.

Dewaele, J.-M. and van Oudenhoven, J.P. 2009. "The Effect of Multilingualism/Multiculturalism on Personality: No Gain without Pain for Third Culture Kids?" *International Journal of Multilingualism* 6(4):443-459.

Dijkstra, T. 2002. "The Architecture of the Bilingual Word Recognition System: From Identification to Decision." *Bilingualism* 5(3):175-197.

Diriker, E. 2015. Conference interpreting. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 171-185.

Dong, Y. and Xie, Z. 2014. "Contributions of second language proficiency and interpreting experience to cognitive control differences among young adult bilinguals." *Journal of Cognitive Psychology* 26(5):506-519.

Dörnyei, Z. and Taguchi, T. 2010. *Questionnaires in second language research : construction, administration, and processing.* New York; London : Routledge 2<sup>nd</sup> ed.

Dussias, P.E. 2001. Sentence Parsing in Fluent Spanish-English Bilinguals. In: J. Nicol, ed. One mind, two languages: Bilingual language processing. Malden, MA: Blackwell, pp. 159-176.

Dussias, P.E. and Sagarra, N. 2007. "The effect of exposure on syntactic parsing in Spanish–English bilinguals." *Bilingualism* 10(1):101-116.

Ecke, P. and Hall, C.J. 2013. "Tracking tip-of-the-tongue states in a multilingual speaker: Evidence of attrition or instability in lexical systems?" *The International Journal of Bilingualism* 17(6):734-751.

Edwards, J. 2013. Bilingualism and Multilingualism: Some Central Concepts. In: T.K. Bhatia and W.C. Ritchie, eds. The Handbook of Bilingualism and Multilingualism. UK: Wiley-Blackwell 2<sup>nd</sup> ed., pp. 5-25.

Ellis, N.C and Larsen-Freeman, D. 2006. "Language Emergence: Implications for Applied Linguistics - Introduction to the Special Issue." *Applied Linguistics* 27(4):558.

Elman, J.L. 1995. Language as a Dynamical System. In: R.F. Port and T. Van Gelder, eds. Mind as Motion: Cambridge, Mass: MIT Press, pp. 195-225.

Elman, J. 2003. "Development: it's about time." Developmental science 6(4):430-433.

Faez, F. 2011. "Are you a native speaker of English? Moving beyond a simplistic dichotomy." *Critical Inquiry in Language Studies* 8(4):378-399.

Field, A. 2013 *Discovering statistics using IBM SPSS statistics*, 4<sup>th</sup> ed. Los Angeles ; London : SAGE

Foote, R. 2007. "Mathematics and Complex Systems." Science 318(5849):410-412.

Garcia, E.E. 2013. Bilingual Education in the United States. In: J. Altarriba and R.R. Heredia, eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 321-343.

Gass, S. and Glew, M. 2013. Second Language Acquisition and Bilingualism. In: J. Altarriba and R.R. Heredia, eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 265-294.

Gianico, J.L. and Altarriba, J. 2013. The Psycholinguistics of Bilingualism. In: Altarriba, J. and Heredia, R.R. eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 71-104.

Giunti, M. 1995. Dynamical Models of Cognition. In: R.F. Port and T. Van Gelder, eds. Mind as Motion: Cambridge. Mass: MIT Press, pp.549-571.

Goto Butler, Y. 2013. Bilingualism/Multilingualism and Second-Language Acquisition. In: T.K. Bhatia and W.C. Ritchie, eds. The Handbook of Bilingualism and Multilingualism. UK: Wiley-Blackwell 2<sup>nd</sup> ed., pp. 109-136.

Grosjean, F. 1982. *Life with two languages : an introduction to bilingualism*. Cambridge, Mass. ; London: Harvard University Press.

Grosjean, F. 1989. "Neurolinguists, beware! The bilingual is not two monolinguals in one person." *Brain and language* 36(1):3-15.

Grosjean, F. 2008. Studying Bilinguals. Oxford, GBR: Oxford University Press.

Grosjean, F. 2010. Bilingual: Life and Reality. Cambridge, MA, USA: Harvard University Press.

Grosjean, F. 2012. "An attempt to isolate, and then differentiate, transfer and interference." *The International Journal of Bilingualism* 16(1):11-21.

Grosjean, F. 2015. "Bicultural bilinguals." International Journal of Bilingualism 19(5):572-586.

Grosjean, F. and Py, B. 1991. "La restructuration d'une première langue: l'intégration de variantes de contact dans la compétence de migrants bilingues." *La Linguistique* 27:35-60.

Gruber, C. 2012. "What Is The Most Creative Thing You Have Ever Seen? "Useful" is the New "Valuable" and Dynamic Systems are the Key!" *Integrative Psychological and Behavioral Science* 46(2):243-248.

Hale, S. 2014. "Interpreting culture. Dealing with cross-cultural issues in court interpreting." In *Perspectives: Studies in Translatology* 22(3): 321-331.

Hamers, J.F. and Blanc, M.H.A. 2000. *Bilinguality and Bilingualism*. Cambridge, England; New York, NY: Cambridge University Press 2nd ed.

Han, Z. 2012. Fossilization – A classic concern of SLA research. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 476-490.

Han, Z. and Singleton, D.M. 2004. *Fossilization in adult second language acquisition*. Clevedon: Multilingual Matters.

Herdina, P. and Jessner, U. 2002. *A dynamic model of multilingualism : perspectives of change in psycholinguistics*. Clevedon: Multilingual Matters.

Hiltunen, S. and Vik, G.-V. 2017. "Interpreters – experts in careful listening and efficient encoding? Findings of a prose recall test." *International Journal of Bilingualism*, 21(2):194-212.

Hoffmann, C. 1991. An Introduction to Bilingualism. Harlow: Longman.

Howe, M. and Lewis, M. 2005. "The importance of dynamic systems approaches for understanding development" *Developmental review* 25(3-4):247-251.

Hummel, K.M. (2014), Introducing Second Language Acquisition. UK: Wiley-Blackwell.

Isurin, L. 2007. "Teachers' Language: L1 Attrition in Russian-English Bilinguals." *The Modern Language Journal*, 91(3):357-371.

Javier, R.A. 2007. The Bilingual Mind: Thinking, Feeling and Speaking in Two Languages. New York, Springer.

Kantola, L. 2011. "Between- and within-language priming is the same: Evidence for shared bilingual syntactic representations." *Memory & Cognition* 39(2):276-290.

Köpke, B. 2007. Language attrition at the crossroads of brain, mind and society. In: B. Köpke, M.S. Schmid, M. Keijzer and S. Dostert, eds. Language Attrition: Theoretical Perpectives. Amsterdam/Philadelphina: John Benjamins, pp. 9-38.

Köpke, B. and Signorelli, T.M. 2012. "Methodological aspects of working memory assessment in in simultaneous interpreters." *International Journal of Bilingualism*, 16(2):183-197.

Kroll, J.F., van Hell, J.G., Tokowicz, N. and Green, D. 2010. "The Revised Hierarchical Model: A critical review and assessment." *Bilingualism* 13(3):373-381.

Larsen-Freeman, D. 1997. "Chaos/Complexity Science and Second Language Acquisition." *Applied Linguistics* 18(2):141-165.

Larsen-Freeman, D. 2007. "On the complementarity of Chaos/Complexity Theory and Dynamic Systems Theory in understanding the second language acquisition process." *Bilingualism* 10(1):35-37.

Larsen-Freeman, D. 2012. Complexity theory. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 73-87.

Larsen-Freeman, D. 2015. "Saying what we mean: Making a case for 'language acquisition' to become 'language development'." *Language Teaching* 48(4):491-505.

Larsen-Freeman, D. and Cameron, L. 2008. *Complex systems and applied linguistics*. Oxford: Oxford University Press.

Laufer, B. 2003. The influence of L2 on L1 collocational knowledge and on L1 lexical diversity in free written expression. In: V. Cook, ed. Effects of the Second Language on the First. Clevedon, UK: Multilingual Matters, pp. 19-32

Lenzing, A. 2015. "Exploring Regularities and Dynamic Systems in L2 Development." *Language learning* 65(1):89-122.

Lewis, M. 2000. "The Promise of Dynamic Systems Approaches for an Integrated Account of Human Development." *Child development* 71(1):36-43.

Lewis, M. 2005. "Bridging emotion theory and neurobiology through dynamic systems modelling." *Behavioral and Brain Sciences* 28(2):169-194.

Long, M. 1993. "Assessment Strategies for Second Language Acquisition Theories." *Applied Linguistics* 14(3):225-249.

Lorenz, E. 1993. The essence of chaos. London: UCL Press

Lowie, W., and Verspoor, M. 2015. "Variability and Variation in Second Language Acquisition Orders: A Dynamic Reevaluation." *Language learning* 65(1):63-88.

Mackey, W.F. 2000. The description of bilingualism. In: L. Wei, ed. The Bilingualism Reader. London: Routledge, pp.22-51.

Mackey, A., Abbuhl, R. and Gass, S.M. 2012. Interactionist approach. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 7-23.

MacNamara, J. 1967. "The Bilingual's Linguistic Performance - A Psychological Overview." *Journal of social issues* 23(2):58-77.

Marian, V. 2013. Bilingual Research Methods. In: J. Altarriba. And R.R. Heredia, eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 13-38.

Martinovic, I. and Altarriba, J. 2013. Bilingualism and Emotion: Implications for Mental Health. In: T.K. Bhatia W.C. and Ritchie, eds. The Handbook of Bilingualism and Multilingualism. UK: Wiley-Blackwell 2<sup>nd</sup> ed., pp. 292-320.

Merriam Webster Dictionary, online resource. Definition of bilingualism available at: http://www.merriam-webster.com/dictionary/bilingualism. [Last accessed 1 March 2017].

Mikkelson, H. 2012. Community interpreting. In: C. Millàn and F. Bartrina, eds. The Routledge Handbook of Translation Studies. UK: Routledge, pp. 389-401.

Mikkelson, H. and Jourdenais, R. 2015. Introduction. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 1-7.

Molnar, M., Ibáñez Molina, A. and Carreiras, M. 2015. "Interlocutor identity affects language activation in bilinguals." *Journal of memory and language* 81:91-104.

Morales, J., Padilla, F., Gómez-Ariza, C.J. and Bajo, T.M. 2015. "Simultaneous interpretation selectively influences working memory and attentional networks." *Acta Psychologica*, 155: 82-91.

Norton, A. 1995. Dynamics: An Introduction. In: R.F. Port and T. Van Gelder, eds. Mind as Motion: Cambridge, Mass: MIT Press, pp.45-68.

NRPSI website. Available at: http://www.nrpsi.org.uk/. [Last accessed 1 March 2017].

Obler, L.K. 2012. "Conference interpreting as extreme language use." *The International Journal of Bilingualism* 16(2):177-182.

Opitz, C. 2013. "A dynamic perspective on late bilinguals' linguistic development in an L2 environment." *The International Journal of Bilingualism* 17(6):701-715.

Ortega, L. 2013. Understanding Second Language Acquisition. London and New York: Routledge.

Ottino, J.M. 2003. "Complex systems." AIChE journal 49(2):292-299.

Oxford English Dictionary, online resource. Definition of bilingualism available at: https://en.oxforddictionaries.com/definition/bilingualism. [Last accessed 1 March 2017].

Paradis, M. 2004. *Neurolinguistic Theory of Bilingualism*. Amsterdam, NLD: John Benjamins Publishing Company.

Paradis, M. 2007. L1 attrition features predicted by a neurolinguistic theory of bilingualism. In: B. Köpke, M.S. Schmid, M. Keijzer and S. Dostert, eds. Language Attrition: Theoretical Perpectives. Amsterdam/Philadelphina: John Benjamins, pp. 121-134.

Peréa, F.C. and García Coll, C. 2013. The Social and Cultural Contexts of Bilingualism. In: J. Altarriba and R.R. Heredia, eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 199-241

Pfenninger, S.E. and Singleton, D. 2016. "Affect Trumps Age: A Person-in-Context Relational View of Age and Motivation in SLA." In: *Second Language Research* 32(3):311-345.

Plaza Pust, C. 2008. "Dynamic Systems Theory and Universal Grammar: Holding up a Turbulent Mirror to Development in Grammars." *The Modern language journal* 92(2):250-269.

Pöchhacker, F. 2004. Introducing Interpreting Studies. London: Routledge.

Pöchhacker, F. and Shlesinger, M. 2002. Introduction. In: F. Pöchhacker and M. Shlesinger, eds. The Interpreting Studies Reader. London: Routledge, pp. 1-12.

Pollock, D.C. and Van Reken, R.E. 2009. *Third culture kids: Growing up among worlds*. Boston, MA: Nicholas Brealey.

Port, R.F. and Van Gelder, T. 1995. It's About Time: An Overview of the Dynamical Approach to Cognition. In: R.F. Port and T. Van Gelder, eds. Mind as Motion: Cambridge, Mass: MIT Press, pp.1-43.

Prescher, P. 2007. Identity, immigration and first language attrition. In: B. Köpke, M.S. Schmid, M. Keijzer and S. Dostert, eds. Language Attrition: Theoretical Perpectives. Amsterdam/Philadelphina: John Benjamins, pp. 189-204.

Rampton, M.B.H. 1990. "Displacing the 'native speaker': Expertise, affiliation, and inheritance." *ELT Journal* 44(2):97-101.

Runnqvist, E., Fitzpatrick, I., Strijkers, K. and Costa, A. 2013. An Appraisal of the Bilingual Language Production System: Quantitatively or Qualitatively Different from Monolinguals? In: T.K. Bhatia W.C. and Ritchie, eds. The Handbook of Bilingualism and Multilingualism. UK: Wiley-Blackwell 2<sup>nd</sup> ed., pp. 244-265.

Russell, D. and Takeda, K. 2015. Consecutive interpreting. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 96-111.

Schmid, M.S. 2007. The role of L1 use for L1 attrition. In: B. Köpke, M.S. Schmid, M. Keijzer and S. Dostert, eds. Language Attrition: Theoretical Perpectives. Amsterdam/Philadelphina: John Benjamins, pp. 135-154.

Schmid, M.S. and Dusseldorp, E. 2010. "Quantitative analyses in a multivariate study of language attrition: the impact of extralinguistic factors." *Second Language Research* 26(1):125-160.

Schmid, M.S., Kopke, B. 2004. Language Attrition: The next phase. In: S. Schmid, B. Kopke, M. Keijer and L. Weilemar, eds. First Language Attrition: Interdisciplinary perspectives on methodological issues. Amsterdam: John Benjamins, pp. 1-43.

Schmid, M.S., Kopke, B. 2009. L1 Attrition and the Mental Lexicon. In: A. Pavlenko, ed. The bilingual mental lexicon: interdisciplinary approaches. Bristol: Multilingual Matters pp. 209-238.

Schmid, M.S., Kopke, B. and de Bot, K. 2013. "Language attrition as a complex, non-linear development." *The International Journal of Bilingualism* 17(6):675-682.

Schrauf, R.W. 2013. Bilingualism and Aging. In: J. Altarriba and R.R. Heredia, eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 105-128.

Seeber, K.G. 2015. Simultaneous interpreting. In: H. Mikkelson and R. Jourdenais, eds. The Routledge Handbook of Interpreting. New York and London: Routledge, pp. 79-95.

Seeber, K.G. and Kerzel, D. 2012. "Cognitive load in simultaneous interpreting: Model meets data." *International Journal of Bilingualism* 16(2):228-242.

Selinker, L. 1972. "Interlanguage." *International Review of Applied Linguistics in Language Teaching (IRAL)* 10(3):209-231.

Sieloff Magnan, S. and Lafford, B.A. 2012. Learning through immersion during study abroad. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 525-540.

Signorelli, T.M., Haarmann, H.J. and Obler, L.K. 2012. "Working memory in simultaneous interpreters: Effects of task and age." *International Journal of Bilingualism*, 16(2):198-212.

Skehan, P. 2012. Language aptitude. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 381-395.

Smith, L.B. 2005. "Cognition as a dynamic system: Principles from embodiment." *Developmental review* 25(3-4):278-298.

Smith, L.B. and Thelen, E. 2003. "Development as a dynamic system." *Trends in cognitive sciences* 7(8):343-348.

Spencer, J., Austin, A. and Schutte, A. 2012. "Contributions of dynamic systems theory to cognitive development." *Cognitive development* 27(4):401-418.

Spencer, J., Perone, S. and Buss, A. 2011. "Twenty Years and Going Strong: A Dynamic Systems Revolution in Motor and Cognitive Development." *Child development perspectives* 5(4):260-266.

Spoelman, M. and Verspoor, M. 2010. "Dynamic Patterns in Development of Accuracy and Complexity: A Longitudinal Case Study in the Acquisition of Finnish." *Applied Linguistics* 31(4):532-553.

Su, I. 2010. "Transfer of Pragmatic Competences: A Bi-Directional Perspective." *The Modern language journal* 94(1):87-102.

Sunderman, G. 2007. "How languages are learned." The Modern language journal 91(4):702-703.

Sunderman, G. and Kroll, J. 2006. "First Language Activation During Second Language Lexical Processing: An Investigation of Lexical Form, Meaning, and Grammatical Class." *Studies in Second Language Acquisition* 28(3):387-422.

Tannenbaum, M. and Tseng, J. 2015. "Which one is Ithaca? Multilingualism and sense of identity among Third Culture Kids." *International Journal of Multilingualism* 12(3):276-297.

The Five Graces Group. 2009. "Language Is a Complex Adaptive System: Position Paper." *Language Learning* 59(1):1-26.

Thelen, E. 1995. Time-Scale Dynamics and the Development of an Embodied Cognition. In: R.F. Port and T. Van Gelder, eds. Mind as Motion: Cambridge, Mass: MIT Press, pp.69-100.

Thelen, E. and Smith, L.B. 1994. *A Dynamic systems approach to the development of cognition and action*. Cambridge, Mass; London: MIT Press.

Thiéry, C. 1975. *Le bilinguisme vrai chez les interprètes de conférence professionels*. Thesis (Ph.D), Université de la Sorbonne Nouvelle, Paris.

Thiéry, C. 1978. True Bilingualism and Second-language Learning. In: D. Gerver and H. Sinaiko, eds. Language Interpretation and Communication. New York and London: Plenum Press, pp.145-153.

Thiéry, C. 1982. "Le bilinguisme vrai." Multilingua 1(4):203-212.

Townsley, B. 2007. "Interpreting in the UK Community: Some Reflections on Public Service Interpreting in the UK." *Language and Intercultural Communication* 7(2):163-170.

Treffers-Daller, B. 2011. "Operationalizing and measuring language dominance." *International Journal of Bilingualism* 15(2):147-163.

Tzou, Y.-Z., Eslami, Z.R., Chen, H.-C., Vaid, J. 2012. "Effect of language proficiency and degree of formal training in simultaneous interpreting on working memory and interpreting performance: Evidence from Mandarin-English speakers." *International Journal of Bilingualism* 16(2):213-227.

Ushioda, E. and Dörnyei, Z. 2012. Motivation. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 396-409.

Valdés, G. and Angelelli, C. 2003. "Interpreting, Interpreters and the Study of Bilingualism." *Annual Review of Applied Linguistics* 23:58-78.

van Geert, P. 1995. Growth Dynamics in Development. In: R.F. Port and T. Van Gelder, eds. Mind as Motion: Cambridge, Mass: MIT Press, pp.313-337.

van Geert, P. 2008. "The Dynamic Systems Approach in the Study of L1 and L2 Acquisition: An Introduction." *The Modern language journal* 92(2):179-199.

van Heuven, W. 2008. "Language Conflict in the Bilingual Brain." *Cerebral cortex* 18(11):2706-2716.

Vega, L.A. 2013. Social Psychological Approaches to Bilingualism. In: J. Altarriba and R.R. Heredia, eds. An Introduction to Bilingualism: Principles and Processes. New York and London: Psychology Press, pp. 185-198.

Verspoor, M., Lowie, W. and Van Dijk, M. 2008. "Variability in Second Language Development From a Dynamic Systems Perspective." *The Modern language journal* 92(2):214-231.

Williams, J. 2012. Classroom research. In: S.M. Gass and A. Mackey, eds. The Routledge Handbook of Second Language Acquisition. New York and London: Routledge, pp. 541-554.

Woumans, E., Ceuleers, E., Van der Linden, L., Szmalec, A., Duyck, W. 2015. "Verbal and Nonverbal Cognitive Control in Bilinguals and Interpreters." *Journal of Experimental Psychology: Learning, Memory and Cognition*, 41(5):1579-1586.

### Appendix I

Questionnaires: AIIC AA, AIIC AB and PSI groups

ection 1: Personal Details	
Age:	
Country(ies) of residence:	
Country of birth:	
Nationality(ies):	
Nationality(ies) at birth:	

### Questionnaire: AIIC Interpreters (A-A)

### Section 2: AIIC classification

1. Language comb	oination (with AIIC classification):
A Languages:	
B Languages:	
C Languages:	
2. Interpreting exp	perience (years):
3. Date of joining	AIIC:
4. Original classif	ication upon joining:
A Languages:	
B Languages:	
C Languages:	
5. Date A-A classif	fication obtained (if different from above):
	to the mother tongue.
7. The concept of	"true bilingualism" has been defined as follows (Thiéry, 1975):
	s someone who is taken to be one of them by the members of two communities, at roughly the same social and cultural level".
Do you feel that you have had not been seen as a seen as	ou are a "true" bilingual in the sense of Thiéry's definition?
Unsure	
8. If you answered	d "no" to the above, why would you say this is?

Questionnaire: AIIC Interpreters (A-A)		
9. Do you feel that Thiéry's definition correspo	onds to the AIIC A-A clas	ssification?
Yes		
○ No		
Unsure		

Questionnaire: AIIC Interpreters (A-A)
Section 3: Linguistic background
1. First language spoken:
2. Language spoken by mother:
3. Language spoken by father:
4. At what age did you start learning your second A language?
5. Language mostly spoken at home during childhood (here, "childhood" is taken to be approximately the period from 0-12 years):
6. Did you relocate to a different country during childhood?  Yes  No  No  7. If you answered "yes" to the above, please specify the country and your age when
you relocated:
8. Please specify your language(s) of education:
Primary
Secondary
Higher
9. Would you say that both of your A languages were "mother tongues" in childhood?  Yes  No  Unsure
10. If not, at approximately what age would you say you attained this level of
competence in both languages and what led to it?

Section 4: Present linguistic and professional sit	uation
1. Which of your A languages do you mostly interpret i	nto?
2. Which direction do you prefer to interpret into (if you	have a preference)?
3. If you have a preferred direction, is this because (fee answer):	I free to tick more than one
Performance in this direction is more stable / less prone to error	
I get less tired / can interpret for longer periods in this direction	
Comprehension is easier in this direction	
Lexical availability is better in this direction	
I feel more confident interpreting in this direction	
Other (please specify)	
-	
Quality is mostly down to interpreting skill/experience  Both are important, but interpreting skill is more so  Both are equally important  Both are important, but linguistic skill is more so	
Quality is mostly down to linguistic skill	
Other (please specify)	
5. Do you do any translation work?	
Yes	
○ No	
6. If yes, which language(s) do you translate into?	

Questionnaire: AIIC Interpreters (A-A)
7. If you live in a country where one of your A languages (A1) is spoken, do you
regularly spend time in a country where your other A language (A2) is spoken?
Yes - live between A1 and A2-speaking countries
Yes - regularly visit A2-speaking country
No - mostly stay in A1-speaking country
N/A - do not live in A1 or A2-speaking country
2. Places indicate which of your Alenguages you do the following in on a doily basis
8. Please indicate which of your A languages you do the following in on a daily basis (please put "both" if both languages are used and "neither" if neither is one used):
(please put both in both languages are used and heither in heither is one used).
Read
Write
Watch TV
Listen to the radio
Socialise
View internet pages/social networking sites
9. Which language do you generally use more, or are more exposed to, on a daily
basis?
10. Would you say that maintaining two A languages requires a degree of conscious
effort?
Yes
○ No
Unsure
44.16
11. If you were not working as a professional interpreter, do you believe that you would
still make an effort to maintain your two current A languages at such a high level?
Yes
○ No
Unsure
12. If you were not working as a professional interpreter, do you believe that it would be
possible to maintain both of your A languages at such a high level without the
professional motivation/level of exposure to the languages that stems from the
profession?
Yes
○ No
Unsure

Questionnaire: AIIC Interpreters (A-A)	
13. Do you consciously seek to fill any gaps in your linguistic knowledge w	hen you
notice them?	
Yes	
○ No	
Unsure	
14. If yes, is your motivation for this purely professional?	
Yes	
○ No	
Unsure	

### Questionnaire: AIIC Interpreters (A-A) Section 5: Attitudes and perceptions 1. Do you feel that language dominance fluctuates between your two A languages? ) Unsure 2. If you feel that language dominance does fluctuate, when does this occur (e.g. if you move country, work into one of the languages for a long time, or simply read more in one of the languages)? 3. For the purpose of the remaining questions, please label your two current A languages as "Language A1" and "Language A2" by entering them in the boxes below: Language A1 Language A2 4. When you feel tired or emotionally upset, is there one language you would naturally revert to if possible? Yes, Language A1 Yes, Language A2 No Unsure 5. Do you feel that words carry the same "weight" in both of your A languages (in particular, swear words, words such as "hate" and "love")? No, Language A1 words carry more weight No, Language A2 words carry more weight Unsure 6. Do you feel any special attachment to one of the languages over the other? Yes, Language A1 Yes, Language A2 No Unsure

Questionnaire: AII0	C Interpreters	(A-A)	
7. Would you say that	your command o	of registers in your two A la	anguages is:
☐ Equal			
Better in Language A1			
Better in Language A2			
Unsure			
8. Would you say that	you have more ဇ္	gaps in vocabulary in one o	of your A languages?
Yes, in Language A1			
Yes, in Language A2			
No			
Unsure			
9. Do you feel that you	u write equally w	ell/easily in both of your A	anguages?
Yes			
No, better in Language A1			
No, better in Language A2			
Unsure			
10. Do you believe tha	at in writing you a	re indistinguishable from a	a monolingual in your
A languages?			
Yes, in both languages			
In Language A1 only			
In Language A2 only			
In neither language			
Unsure			
11. Compared to a mo	nolingual of roug	hly the same cultural level	educational level.
		h of your A languages) in te	
vocabulary:			
1-2	Stronger overall	About the same	Weaker overall
Language A1			
-		hly the same cultural level	
		n of your A languages) in te	rms of instinctive
"native" understandii	ng: Stronger overall	About the same	Weaker overall
Language A1	Stronger overall	About the same	Trearer overall
Language A2			

Questionnaire. Ai	IIC Interpreters (	A-A)	
13. Compared to a n	nonolingual of roughl	y the same cultural lev	el/educational level,
how do you think yo	ou compare (in each	of your A languages) in	terms of pragmatic
competence/cultura	ıl awareness:		
	Stronger overall	About the same	Weaker overall
Language A1			
Language A2			
14. Do you ever noti	ce yourself making u	tterances in your curre	ent strongest working
languages which so	ound unnatural due to	being influenced by t	he other language?
No, in neither language			
Yes, in language A1 only			
Yes, in language A2 only			
Yes, in both languages, b	ut more frequently in Language A	11	
Yes, in both languages, b	ut more frequently in Language A	λ2	
Yes, in both languages, w	ith about the same frequency		
Unsure			
45 51	to add any additional		

Questionnaire: AIIC Interpreters (A-A)
Please enter your email if you would like to receive an electronic copy of the research project once it is complete:
2. If you have entered you email above, would you object to possibly being contacted by the researcher regarding another aspect of the project?
Yes
○ No

Questionnaire: AIIC Interpreters	
Section 1: Personal Details	
1. Age: 2. Country(ies) of residence:	
3. Country of birth:	
4. Nationality(ies):	
5. Nationality(ies) at birth:	

# Questionnaire: AIIC Interpreters Section 2: Bilingualism

	inination (with AIIC	classification):		
A Languages:				
B Languages:				
C Languages:				
2. Which of the a languages?	bove would you sa	y are your two cu	rrent strongest	working
	ge would you class applies to more th			
No Unsure				

## Questionnaire: AIIC Interpreters **Section 3: Linguistic background** 1. First language spoken: 2. Language spoken by mother: 3. Language spoken by father: 4. At what age did you start learning the foreign language which is currently your strongest foreign language? 5. Language mostly spoken at home during childhood (here, "childhood" is taken to be approximately the period from 0-12 years): 6. Did you relocate to a different country during childhood? 7. If you answered "yes" to the above, please specify the country and your age when you relocated: 8. Please specify your language(s) of education: Primary Secondary Higher

## Questionnaire: AIIC Interpreters **Section 4: Present linguistic and professional situation** 1. Interpreting experience (years/months): 2. Which language combinations are you required to use the most during interpreting assignments? 3. Which language would you say you interpret into the most? 4. Which direction do you prefer to interpret into (if you have a preference)? 5. If you have a preferred direction, is this because (feel free to tick more than one answer): Performance in this direction is more stable / less prone to error I get less tired / can interpret for longer periods in this direction Comprehension is easier in this direction Lexical availability is better in this direction I feel more confident interpreting in this direction Other (please specify) 6. In your opinion, to what extent is performance quality affected by interpreting skill or experience, and to what extent does it depend on linguistic skill? Quality is mostly down to interpreting skill/experience Both are important, but interpreting skill is more so Both are equally important Both are important, but linguistic skill is more so Quality is mostly down to linguistic skill Other (please specify)

Questionnaire: AIIC Interpreters
7. Do you do any translation work?
Yes
○ No
8. If yes, which language(s) do you translate into?
9. Please indicate which of your two strongest working languages you do the following
in on a daily basis (please put "both" if both languages are used and "neither" if neither
is one used):
Read
Write
Watch TV
Listen to the radio
Socialise
View internet pages/social networking sites
10. Which language do you generally use more, or are more exposed to, on a daily
basis?
11. Would you say that maintaining your two strongest languages at such a high level
requires a degree of conscious effort?
Yes
○ No
Unsure
42 16
12. If you were not working as a professional interpreter, do you believe that you would
still make an effort to maintain your two current strongest working languages at such a high level?
Yes
○ No
Unsure
Unsure

Questionnaire: AIIC Interpreters
13. If you were not working as a professional interpreter, do you believe that it would be possible to maintain both of your current strongest working languages at such a high level without the professional motivation/level of exposure to the languages that stems from the profession?    Yes  No  Unsure
14. Do you consciously seek to fill any gaps in your linguistic knowledge when you
notice them?
○ Yes ○ No
Unsure
15. If yes, is your motivation for this purely professional?
Yes
○ No
Unsure

## Questionnaire: AIIC Interpreters

### Section 5: Attitudes and perceptions

1. Do you feel that language dominance fluctuates between your two current strongest
working languages?
Yes
○ No
Unsure
2. If you feel that language dominance does fluctuate, when does this occur (e.g. if you move country, work into one of the languages for a long time, or simply read more in one of the languages)?
3. For the purpose of the remaining questions, please label your two current strongest working languages as "Language 1" and "Language 2" by entering them in the boxes below:  Language 1
Language 2
4. When you feel tired or emotionally upset, is there one language you would naturally revert to if possible?  Yes, Language 1  Yes, Language 2  No  Unsure
5. Do you feel that words carry the same "weight" in both of your current strongest
working languages (in particular, swear words, words such as "hate" and "love")?
No, Language 1 words carry more weight
No, Language 2 words carry more weight
Unsure
6. Do you feel any special attachment to one of the languages over the other?
Yes, Language 1
Yes, Language 2
No No
Unsure

7. Would you say that	your command	of registers in your curren	t strongest working
languages is:			
Equal			
Better in Language 1			
Better in Language 2			
Unsure			
8. Would you say that strongest working la		gaps in vocabulary in one	of your current
Yes, in Language 1			
Yes, in Language 2			
○ No			
Unsure			
_	u write equally w	ell/easily in both of your cu	irrent strongest
working languages?			
Yes			
No, better in Language 1			
No, better in Language 2			
Unsure			
10. Compared to a mo	nolingual of roug	hly the same cultural leve	/educational level,
how do you think you	compare (in eac	h of your current stronges	t working languages)
in terms of richness o	f vocabulary:		
	Stronger overall	About the same	Weaker overall
Language 1			
Language 2			
		hly the same cultural leve	
how do you think you	compare (in eacl	h of your current stronges	
	compare (in each "native" underst	h of your current stronges anding:	t working languages)
how do you think you	compare (in eacl	h of your current stronges	
how do you think you in terms of instinctive	compare (in each "native" underst	h of your current stronges anding:	t working languages)
how do you think you in terms of instinctive  Language 1  Language 2	compare (in each	h of your current stronges anding:  About the same	Weaker overall
how do you think you in terms of instinctive  Language 1  Language 2  12. Compared to a mo	compare (in each "native" underst Stronger overall  onolingual of rough	h of your current stronges anding:  About the same  D  hly the same cultural level	Weaker overall  Weaker overall  Weaker overall
how do you think you in terms of instinctive  Language 1  Language 2  12. Compared to a mohow do you think you	compare (in each "native" underst  Stronger overall  onolingual of rough compare (in each	h of your current stronges anding:  About the same  hly the same cultural level h of your current stronges	Weaker overall  Weaker overall  Weaker overall
how do you think you in terms of instinctive  Language 1  Language 2  12. Compared to a mo	compare (in each "native" underst  Stronger overall  onolingual of rough compare (in each	h of your current stronges anding:  About the same  hly the same cultural level h of your current stronges	Weaker overall  Weaker overall  Weaker overall
how do you think you in terms of instinctive  Language 1  Language 2  12. Compared to a mohow do you think you	compare (in each "native" underst  Stronger overall  onolingual of rough compare (in each competence/cult	h of your current stronges anding:  About the same  hly the same cultural level h of your current stronges ural awareness:	Weaker overall  Weaker overall  /educational level, t working languages)
how do you think you in terms of instinctive  Language 1  Language 2  12. Compared to a mohow do you think you in terms of pragmatic	compare (in each "native" underst  Stronger overall  onolingual of rough compare (in each competence/cult	h of your current stronges anding:  About the same  hly the same cultural level h of your current stronges ural awareness:	Weaker overall  Weaker overall  /educational level, t working languages)

Questionnaire: AIIC Interpreters
13. Do you ever notice yourself making utterances in your current strongest working
languages which sound unnatural due to being influenced by the other language?
No, in neither language
Yes, in Language 1 only
Yes, in Language 2 only
Yes, in both languages, but more frequently in Language 1
Yes, in both languages, but more frequently in Language 2
Yes, in both languages, with about the same frequency
Unsure
14. Please feel free to add any additional comments.

Questionnaire: AIIC Interpreters
1. Please enter your email if you would like to receive an electronic copy of the research project once it is complete:
2. If you have entered you email above, would you object to possibly being contacted by the researcher regarding another aspect of the project?
○ No

Age:  Country(ies) of residence:  Country of birth:  Nationality(ies):  Nationality(ies) at birth:	
Country(ies) of residence:  Country of birth:  Nationality(ies):	
Country of birth:  Nationality(ies):	
Country of birth:  Nationality(ies):	
Nationality(ies):	
Nationality(ies) at birth:	

Questionnaire: Public Service Interpreters	
Section 2: Bilingualism	
1. What are your working languages?	
2. Which of the above are your two strongest languages?	
3. Which language would you class as your "mother tongue" or "native language"? If you feel that this applies to both languages, please put "both" (and "neither" for neither language)	r
4. Is the language indicated above also your current strongest language?  Yes No Unsure	

# Questionnaire: Public Service Interpreters Section 3: Linguistic background 1. First language spoken: 2. Language spoken by mother: 3. Language spoken by father: 4. At what age did you start learning the foreign language which is currently your strongest foreign language? 5. Language mostly spoken at home during childhood (here, "childhood" is taken to be approximately the period from 0-12 years): 6. Did you relocate to a different country during childhood? No 7. If you answered "yes" to the above, please specify the country and your age when you relocated: 8. Please specify your language(s) of education: Primary Secondary Higher

## Questionnaire: Public Service Interpreters Section 4: Present linguistic and professional situation 1. Interpreting experience (years/months): 2. Which language combinations are you required to use the most during interpreting assignments? 3. Which language would you say you interpret into the most? 4. Which direction do you prefer to interpret into (if you have a preference)? 5. If you have a preferred direction, is this because (feel free to tick more than one answer): Performance in this direction is more stable / less prone to error I get less tired / can interpret for longer periods in this direction Comprehension is easier in this direction Lexical availability is better in this direction I feel more confident interpreting in this direction Other (please specify) 6. In your opinion, to what extent is performance quality affected by interpreting skill or experience, and to what extent does it depend on linguistic skill? Quality is mostly down to interpreting skill/experience Both are important, but interpreting skill is more so Both are equally important Both are important, but linguistic skill is more so Quality is mostly down to linguistic skill Other (please specify)

Questionnaire: P	ublic Service Interpreters
7. Do you do any tra	anslation work?
Yes	
No	
a Maria subjektora	
8. If yes, which lang	guage(s) do you translate into?
	which of your two strongest working languages you do the following
	please put "both" if both languages are used and "neither" if neither
is one used):	
Read	
Write	
Watch TV	
Listen to the radio	
Socialise	
View internet pages/social networking sites	
10. Which language	e do you generally use more, or are more exposed to, on a daily
basis?	
11. Would you say t	hat maintaining your two strongest languages at such a high level
requires a degree o	f conscious effort?
Yes	
No	
Unsure	
	working as a professional interpreter, do you believe that you would to maintain your two current strongest working languages at such a
Yes	
○ No	
Unsure	

Questionnaire: Public Service Interpreters
13. If you were not working as a professional interpreter, do you believe that it would be possible to maintain both of your current strongest working languages at such a high level without the professional motivation/level of exposure to the languages that stems from the profession?
○ No ○ Unsure
14. Do you consciously seek to fill any gaps in your linguistic knowledge when you notice them?
<ul><li>✓ Yes</li><li>✓ No</li><li>✓ Unsure</li></ul>
15. If yes, is your motivation for this purely professional?  Yes
○ No ○ Unsure

## Questionnaire: Public Service Interpreters

### Section 5: Attitudes and perceptions

1. Do you feel that language dominance fluctuates between your two current strongest working languages?
Yes
○ No
Unsure
O Shadre
2. If you feel that language dominance does fluctuate, when does this occur (e.g. if you
move country, work into one of the languages for a long time, or simply read more in
one of the languages)?
3. For the purpose of the remaining questions, please label your two current strongest
working languages as "Language 1" and "Language 2" by entering them in the boxes
below:
Language 1
Language 2
4. When you feel tired or emotionally upset, is there one language you would naturally
revert to if possible?
Yes, Language 1
Yes, Language 2
○ No
Unsure
5. Do you feel that words carry the same "weight" in both of your current strongest
working languages (in particular, swear words, words such as "hate" and "love")?
Yes
No, Language 1 words carry more weight
No, Language 2 words carry more weight
Unsure
6. Do you feel any special attachment to one of the languages over the other?
Yes, Language 1
Yes, Language 2
○ No
Unsure

= Wanted way and that your command of registers in your current change	
7. Would you say that your command of registers in your current stronge	st working
languages is:	
Equal	
Better in Language 1	
Better in Language 2	
Unsure	
8. Would you say that you have more gaps in vocabulary in one of your c	urrent
strongest working languages?	
Yes, in Language 1	
Yes, in Language 2	
○ No	
Unsure	
9. Do you feel that you write equally well/easily in both of your current str	ongest
working languages?	
Yes	
No, better in Language 1	
No, better in Language 2	
Unsure	
10. Compared to a monolingual of roughly the same cultural level/educati	onal level.
how do you think you compare (in each of your current strongest working	*
in terms of richness of vocabulary:	<b>3</b> **** <b>3</b> *** <b>7</b>
Stronger overall About the same We	eaker overall
Language 1	
Language 2	
11. Compared to a monolingual of roughly the same cultural level/education	onal level,
how do you think you compare (in each of your current strongest working	
in terms of instinctive "native" understanding:	
Stronger overall About the same We	eaker overall
Language 1	
Language 2	
12. Compared to a monolingual of roughly the same cultural level/education	onal level,
how do you think you compare (in each of your current strongest working	g languages)
in terms of pragmatic competence/cultural awareness:	
Stronger overall About the same We	eaker overall
Language 1  Language 2	

aira: Dublia Sanziaa Interpretare

Questionnaire: Public Service Interpreters
13. Do you ever notice yourself making utterances in your current strongest working
languages which sound unnatural due to being influenced by the other language?
No, in neither language
Yes, in Language 1 only
Yes, in Language 2 only
Yes, in both languages, but more frequently in Language 1
Yes, in both languages, but more frequently in Language 2
Yes, in both languages, with about the same frequency
Unsure
14. Please feel free to add any additional comments.
- Please reer free to add any additional comments.

Questionnaire: Public Service Interpreters
1. Please enter your email if you would like to receive an electronic copy of the research project once it is complete:  2. If you have entered you email above, would you object to possibly being contacted
by the researcher regarding another aspect of the project?
Yes No

#### Appendix II

Comparison of responses given by the subgroup of AIIC AA interpreters who do not consider both A languages to be mother tongues ("Not Two MTs" group) with those of the rest of the AA group ("Two MTs" group).

#### (1) Age of SLA:

Table II.1. Comparison of mean age of SLA in years for the Not Two MTs and Two MTs groups

Group	No. in group	Mean age SLA (years)	Std. Deviation
Two MTs	40	3.65	3.779
Not Two MTs	7	8.00	8.246

The range for age of SLA for the Not Two MTs group was found to be 0-18 years.

An independent samples t-test showed that the difference in mean ages of SLA between the two groups is not significant, however, it does represent a fairly large effect:

$$t(6.448) = -1.371$$
,  $p = .216$  (SE = 3.174);  $r = 0.475$ .

#### (2) Mother tongue competence in childhood and true bilingualism:

Table II.2. Responses of the Not Two MTs and Two MTs groups to the questions of whether both A languages were mother tongues in childhood and whether respondents consider themselves to be true bilinguals

Group	Questions and responses (freq and %) Were both of your A languages mother tongues in childhood?				
	Yes	No	Unsure		
Two MTs	31 (75.6%)	10 (24.4%)	0		
Not Two MTs	2 (28.6%)	5 (71.4%)	0		
	Are you a true bilingual in accordance with the definition?				
	Yes	No	Unsure		
Two MTs	41 (100.0%)	0	0		
Not Two MTs	3 (42.9%)	1 (14.3%)	3 (42.9%)		

<sup>1</sup> Those AIIC AA respondents who answered No or Unsure to Q2/6/AA.

(3) Childhood linguistic environment and languages of primary and secondary education:

Table II.3. Comparison of childhood linguistic environment (by category) and language(s) of primary and secondary education for the Not Two MTs and Two MTs groups

Group	Childhood linguistic environment/education category and responses (freq and %)				
	First language(s) spoken				
	A1	A2	Both	Other	
Two MTs	27 (65.9%)	0	13 (31.7%)	1 (2.4%)	
Not Two MTs	5 (71.4%)	0	0	2 (28.6%)	
		Moth	er's language(s)		
	A1	A2	Both	Other	
Two MTs	22 (53.7%)	11 (26.8%)	5 (12.2%)	3 (7.3%)	
Not Two MTs	5 (71.4%)	0	1 (14.3%)	1 (14.3%)	
	Father's language(s)				
	A1	A2	Both	Other	
Two MTs	28 (68.3%)	6 (14.6%)	7 (17.1%)	0	
Not Two MTs	3 (42.9%)	3 (42.9%)	0	1 (14.3%)	
	Language(s) mostly spoken in the home during childhood				
	A1	A2	Both	Other	
Two MTs	24 (58.5%)	3 (7.3%)	14 (34.1%)	0	
Not Two MTs	6 (85.7%)	0	0	1 (14.3%)	
		Childhood lingui	stic environment (cour	ntry)	
	A1	A2	Both	Other/Unclear	
Two MTs	8 (19.5%)	11 (26.8%)	17 (41.5%)	5 (12.2%)	
Not Two MTs	5 (71.4%)	1 (14.3%)	1 (14.3%)	0	
	Language(s) of primary education				
	A1	A2	Both	Other	
Two MTs	14 (35.9%)	11 (28.2%)	14 (35.9%)	0	
Not Two MTs	6 (85.7%)	1 (14.3%)	0	0	
		Language(s)	of secondary education	1	
	A1	A2	Both	Other	
Two MTs	7 (17.9%)	14 (35.9%)	18 (46.2%)	0	
Not Two MTs	4 (57.1%)	1 (14.3%)	2 (28.6%)	0	

Table II.4. Comparison of responses of the Not Two MTs and Two MTs groups to the question of whether they relocated to a different country in childhood

Group	Did you relocate to a different country during childhood? (freq and %)		
	Yes	No	
Two MTs	29 (70.7%)	12 (29.3%)	
Not Two MTs	2 (28.6%)	5 (71.4%)	

#### (4) Language(s) of higher education:

Table II.5. Comparison of language(s) of higher education for the Not Two MTs and Two MTs groups

Group	Language(s) of higher education (freq and %)			
	A1	A2	Both	Other
Two MTs	5 (12.8%)	15 (38.5%)	17 (43.6%)	2 (5.1%)
Not Two MTs	2 (28.6%)	4 (57.1%)	1 (14.3%)	0

#### (5) Present linguistic environment and language use:

Table II.6. Comparison of present linguistic environment for the Not Two MTs and Two MTs groups

Group		Present linguistic	environment (freq a	nd %)	
	Dominant language(s) of country of residence				
	A1	A2	Both	Other	
Two MTs	8 (21.1%)	23 (60.5%)	2 (5.3%)	5 (13.2%)	
Not Two MTs	2 (28.6%)	4 (57.1%)	1 (14.3%)	0	
	Language(s) exposed to the most on a daily basis				
	A1	A2	Both	Other	
Two MTs	12 (29.3%)	19 (46.3%)	9 (22.0%)	1 (2.4%)	
Not Two MTs	2 (28.6%)	3 (42.9%)	2 (28.6%)	0	

Table II.7. Comparison of self-reported daily language use (by category) for the Not Two MTs and Two MTs groups

Group		Language use catego	ory and responses (fre	q and %)		
	Reading					
	A1	A2	Both	Neither		
Two MTs	8 (19.5%)	1 (2.4%)	32 (78.0%)	0		
Not Two MTs	2 (28.6%)	1 (14.3%)	3 (42.9%)	1 (14.3%)		
	Writing					
	A1	A2	Both	Neither		
Two MTs	2 (4.9%)	1 (2.4%)	37 (90.2%)	1 (2.4%)		
Not Two MTs	0	2 (28.6%)	5 (71.4%)	0		
	Watching TV					
	A1	A2	Both	Neither		
Two MTs	6 (14.6%)	6 (14.6%)	21 (51.2%)	8 (19.5%)		
Not Two MTs	2 (28.6%)	1 (14.3%)	2 (28.6%)	2 (28.6%)		
	Listening to the radio					
	A1	A2	Both	Neither		
Two MTs	6 (14.6%)	16 (39.0%)	15 (36.6%)	4 (9.8%)		
Not Two MTs	2 (28.6%)	2 (28.6%)	1 (14.3%)	2 (28.6%)		

	Socialising				
	A:1	A2	Both	Neither	
Two MTs	4 (9.8%)	7 (17.1%)	30 (73.2%)	0	
Not Two MTs	1 (14.3%)	1 (14.3%)	5 (71.4%)	0	
	Using the internet				
	A1	A2	Both	Neither	
Two MTs	6 (14.6%)	3 (7.3%)	30 (73.2%)	2 (4.9%)	
Not Two MTs	1 (14.3%)	0	5 (71.4%)	1 (14.3%)	

#### (6) Interpreting and translation:

Table II.8. Comparison of responses for the Not Two MTs and Two MTs groups to questions pertaining to interpreting

Group	Responses to questions concerning interpreting (freq and %)  Language(s) you mostly interpret into:					
	A1	A2	Both			
Two MTs	14 (34.1%)	11 (26.8%)	16 (39.0%)			
Not Two MTs	1 (14.3%)	3 (42.9%)	3 (42.9%)			
	Preferred interpreting direction					
	A1	A2	No preference			
Two MTs	6 (16.7%)	7 (19.4%)	23 (63.9%)			
Not Two MTs	3 (42.9%)	2 (28.6%)	2 (28.6%)			
	What is more important: interpreting skill or linguistic skill?					
	Interpreting skill	Linguistic skill	Both equally important			
Two MTs	16 (40.0%)	4 (10.0%)	20 (50.0%)			
Not Two MTs	0	0	7 (100.0%)			

Table II.9. Number of respondents from the Not Two MTs and Two MTs groups who also carry out translation work

Group	Do you do any translation work? (freq and %)		
	Yes	No	
Two MTs	19 (46.3%)	22 (53.7%)	
Not Two MTs	4 (57.1%)	3 (42.9%)	

Table II.10. Languages respondents from the Not Two MTs and Two MTs groups mostly translate into

Group	Language(s) you mostly translate into (freq and %):				
	A1	A2	Both		
Two MTs	4 (21.1%)	2 (10.5%)	13 (68.4%)		
Not Two MTs	2 (50.0%)	1 (25.0%)	1 (25.0%)		

#### (7) Language maintenance effort and motivation:

Table II.11. Comparison of responses for the Not Two MTs and Two MTs groups to questions pertaining to language maintenance effort and motivation

Group	Responses to ques	tions on language maintenanc	e effort and motivation (freq and %)
	Do you feel that	maintaining your two A langu requires a degree of cons	ages at their present high standard scious effort?
	Yes	No	Unsure
Two MTs	27 (65.9%)	12 (29.3%)	2 (4.9%)
Not Two MTs	6 (85.7%)	1 (14.3%)	0
		he effort to maintain both of y ndard if not working as a pro	our A languages at their present high fessional interpreter?
	Yes	No	Unsure
Two MTs	25 (62.5%)	5 (12.5%)	10 (25.0%)
Not Two MTs	3 (42.9%)	2 (28.6%)	2 (28.6%)
	_	sible to maintain both of your ndard if not working as a pro	A languages at their present high fessional interpreter?
	Yes	No	Unsure
Two MTs	19 (46.3%)	8 (19.5%)	14 (34.1%)
Not Two MTs	3 (42.9%)	4 (57.1%)	0
	Do you seek to	o fill gaps in your linguistic kn	owledge when you notice them?
	Yes	No	Unsure
Two MTs	39 (95.1%)	1 (2.4%)	1 (2.4%)
Not Two MTs	7 (100%)	0	0
	Is your motivat	ion for filling gaps in linguisti	c knowledge purely professional?
	Yes	No	Unsure
Two MTs	4 (10.0%)	34 (85.0%)	2 (5.0%)
Not Two MTs	3 (42.9%)	4 (57.1%)	0

<sup>(8)</sup> Awareness of fluctuations in language dominance and cross-linguistic interference:

Table II.12. Responses of the Not Two MTs and Two MTs groups to the question of whether language dominance is perceived to fluctuate between their A languages

Group	Do you feel that language dominance fluctuates between y languages? (freq and %)			
	Yes	No	Unsure	
Two MTs	19 (46.3%)	17 (41.5%)	5 (12.2%)	
Not Two MTs	5 (71.4%)	2 (28.6%)	0	

Table II.13. Comparison of responses for the Not Two MTs and Two MTs groups to the question pertaining to perceptions of non-linguistic interference between the A languages

Group	Do you notice	ou notice yourself making unnatural utterances in either of your two A langu- due to the influence of the other language? (freq and %)					
	In neither	Mostly in A1	Mostly in A2	In both equally	Unsure		
Two MTs	12 (29.3%)	6 (14.6%)	4 (9.8%)	13 (31.7%)	6 (14.6%)		
Not Two MTs	2 (28.6%)	1 (14.3%)	1 (14.3%)	2 (28.6%)	1 (14.3%)		

#### (9) Emotional response and attachment to the languages:

Table II.14. Comparison of responses for the Not Two MTs and Two MTs groups to questions pertaining to emotional response and attachment to the languages

Group	Responses to questions on emotional response and attachment to the languages (freq and %)  If possible, would you revert to one of your two A languages when tired/upset?					
Two MTs	18 (43.9%)	13 (31.7%)	5 (12.2%)	5 (12.2%)		
Not Two MTs	2 (28.6%)	3 (42.9%)	0	2 (28.6%)		
	Do you feel that words carry the same weight in your two A languages (in particular, swear words and words such as "love" and "hate")?					
	Equal	A1 more weight	A2 more weight	Unsure		
Two MTs	23 (56.1%)	8 (19.5%)	3 (7.3%)	7 (17.1%)		
Not Two MTs	6 (85.7%)	1 (14.3%)	0	0		
	Do you feel a greater attachment towards one of your A languages over the other					
	No	To A1	To A2	Unsure		
Two MTs	21 (51.2%)	12 (29.3%)	2 (4.9%)	6 (14.6%)		
Not Two MTs	3 (42.9%)	4 (57.1%)	0	0		

(10) Self-ratings of proficiency in the two strongest languages:

Table II.15. Self-ratings of proficiency in the A languages relative to each other (by category) of the Not Two MTs and Two MTs groups

Group	Self-ratings in th	Self-ratings in the two languages relative to each other across difference categories (freq and %)					
	Do you feel that your command of registers is equal in both of your A languages, or better in one language than the other?						
	Equal	Better in A1	Better in A2	Unsure			
Two MTs	29 (70.7%)	5 (12.2%)	2 (4.9%)	5 (12.2%)			
Not Two MTs	2 (28.6%)	3 (42.9%)	1 (14.3%)	1 (14.3%)			
	Do you have fe	wer gaps in vocabular	y in one of your A lang other?	uages compared with the			
	Equal	Fewer in A1	Fewer in A2	Unsure			
Two MTs	23 (56.1%)	6 (14.6%)	5 (12.2%)	7 (17.1%)			
Not Two MTs	2 (28.6%)	4 (57.1%)	0	1 (14.3%)			
	Do you	write equally well/jus	t as easily in both of yo	our A languages?			
	Equal	Better in A1	Better in A2	Unsure			
Two MTs	29 (70.7%)	5 (12.2%)	5 (12.2%)	2 (4.9%)			
Not Two MTs	0	5 (71.4%)	1 (14.3%)	1 (14.3%)			
	Are you indistinguishable from a monolingual in writing in both of your A languages?						
	Yes – in both	In A1 only	In A2 only	Unsure			
Two MTs	32 (78.0%)	3 (7.3%)	0	6 (14.6%)			
Not Two MTs	3 (42.9%)	2 (28.6%)	1 (14.3%)	1 (14.3%)			

Table II.16. Self-ratings of proficiency relative to an average monolingual standard in each language (by category) of the Not Two MTs and Two MTs groups

Group	Self-ratings in	each language relative to a different categories (fi	monolingual standard across req and %)			
	Richness of vocabulary in A1					
	Weaker	Same	Stronger			
Two MTs	2 (4.9%)	19 (46.3%)	20 (48.8%)			
Not Two MTs	1 (14.3%)	4 (57.1%)	2 (28.6%)			
		Richness of vocabula	ary in A2			
	Weaker	Same	Stronger			
Two MTs	1 (2.4%)	20 (48.8%)	20 (48.8%)			
Not Two MTs	0	5 (71.4%)	2 (28.6%)			
	Instinctive/native comprehension in A1					
	Weaker	Same	Stronger			
Two MTs	2 (4.9%)	27 (65.9%)	12 (29.3%)			
Not Two MTs	0	5 (71.4%)	2 (28.6%)			
	Instinctive/native comprehension in A2					
	Weaker	Same	Stronger			
Two MTs	0	28 (68.3%)	13 (31.7%)			
Not Two MTs	0	5 (71.4%)	2 (28.6%)			
	Pragmatic/cultural competence in A1					
	Weaker	Same	Stronger			
Two MTs	2 (5.0%)	32 (80.0%)	6 (15.0%)			
Not Two MTs	0	3 (42.9%)	4 (57.1%)			
		Pragmatic/cultural comp	petence in A2			
	Weaker	Same	Stronger			
Two MTs	1 (2.4%)	31 (75.6%)	9 (22.0%)			
Not Two MTs	0	4 (57.1%)	3 (42.9%)			

#### Appendix III

Comparison of responses given by the subgroup of AIIC AA interpreters who were awarded their AA classification after joining AIIC ("AA Later" group) with those of the rest of the AA group ("AA Same Date" group).

#### (1) Age of SLA:

Table III.1. Comparison of mean age of SLA in years for the AA Later and AA Same Date groups

Group	No. in group	Mean age SLA (years)	Std. Deviation
AA Same Date	40	4.28	3.779
AA Later	7	4.43	8.246

An independent samples t-test showed that the difference in mean ages of SLA between the two groups is not significant and represents a negligible effect: t(45) = -.077, p = .939 (SE = 2.005); r = .011.

#### (2) Mother tongue competence in childhood and true bilingualism:

Table III.2. Responses of the AA Later and AA Same Date groups to the questions pertaining to mother tongue competence and true bilingualism

Group	Questions and responses (freq and %)				
	Were both of your A languages mother tongues in childhood?				
	Yes	No	Unsure		
AA Same Date	27 (65.9%)	14 (34.1%)	0		
AA Later	6 (85.7%)	1 (14.3%)	0		
	Do you consider both of your A languages to be mother tongues?				
	Yes	No	Unsure		
AA Same Date	35 (85.4%)	1 (2.4%)	5 (12.2%)		
AA Later	6 (85.7%)	1 (14.3%)	0		
	Are you a true bilingual in accordance with the definition?				
_	Yes	No	Unsure		
AA Same Date	37 (90.2%)	1 (2.4%)	3 (7.3%)		
AA Later	7 (100.0%)	0	0		

(3) Childhood linguistic environment and languages of primary and secondary education:

Table III.3. Comparison of childhood linguistic environment (by category) and language(s) of primary and secondary education for the AA Later and AA Same Date groups

Group	Childhood lii	nguistic environment/e	education category and	d responses (freq and %		
		First l	anguage(s) spoken			
	A1	A2	Both	Other		
AA Same Date	27 (65.9%)	0	12 (29.3%)	2 (4.9%)		
AA Later	5 (71.4%)	0	1 (14.3%)	1 (14.3%)		
		Motl	her's language(s)			
	A1	A2	Both	Other		
AA Same Date	25 (61.0%)	8 (19.5%)	5 (12.2%)	3 (7.3%)		
AA Later	2 (28.6%)	3 (42.9%)	1 (14.3%)	1 (14.3%)		
	Father's language(s)					
	A1	A2	Both	Other		
AA Same Date	26 (63.4%)	8 (19.5%)	6 (14.6%)	1 (2.4%)		
AA Later	5 (71.4%)	1 (14.3%)	1 (14.3%)	0		
	Language(s) mostly spoken in the home during childhood					
	A1	A2	Both	Other		
AA Same Date	26 (63.4%)	2 (4.9%)	12 (29.3%)	1 (2.4%)		
AA Later	4 (57.1%)	1 (14.3%)	2 (28.6%)	0		
	Childhood linguistic environment (country)					
	A1	A2	Both	Other/Unclear		
AA Same Date	11 (26.8%)	11 (26.8%)	16 (39.0%)	3 (7.3%)		
AA Later	2 (28.6%)	1 (14.3%)	2 (28.6%)	2 (28.6%)		
	Language(s) of primary education					
	A1	A2	Both	Other		
AA Same Date	17 (43.6%)	10 (25.6%)	12 (30.8%)	0		
AA Later	3 (42.9%)	2 (28.6%)	2 (28.6%)	0		
		Language(s)	of secondary education	on		
	A1	A2	Both	Other		
AA Same Date	11 (28.2%)	11 (28.2%)	17 (43.6%)	0		
AA Later	0	4 (57.1%)	3 (42.9%)	0		

Table III.4. Comparison of responses of the AA Later and AA Same Date groups to the question of whether they relocated to a different country in childhood

Group	Did you relocate to a different country during childhood? (freq and		
	Yes	No	
AA Same Date	27 (65.9%)	14 (34.1%)	
AA Later	4 (57.1%)	3 (42.9%)	

#### (4) Language(s) of higher education:

Table III.5. Comparison of language(s) of higher education for the AA Later and AA Same Date groups

Group	Language(s) of higher education (freq and %)			
	A1	A2	Both	Other
AA Same Date	6 (15.4%)	18 (46.2%)	13 (33.3%)	2 (5.1%)
AA Later	1 (14.3%)	1 (14.3%)	5 (71.4%)	0

#### (5) Present linguistic environment and language use:

Table III.6. Comparison of present linguistic environment for the AA Later and AA Same Date groups

Group	Present linguistic environment (freq and %)  Dominant language(s) of country of residence					
	A1	A2	Both	Other		
AA Same Date	9 (23.1%)	23 (59.0%)	2 (5.1%)	5 (12.8%)		
AA Later	1 (16.7%)	4 (66.7%)	1 (16.7%)	0		
	Language(s) exposed to the most on a daily basis					
	A1	A2	Both	Other		
AA Same Date	6 (14.6%)	3 (7.3%)	30 (73.2%)	2 (4.9%)		
AA Later	1 (14.3%)	0	5 (71.4%)	1 (14.3%)		

Table III.7. Comparison of self-reported daily language use (by category) for the AA Later and AA Same Date groups

Group		Language use catego	ry and responses (fre	q and %)	
	Reading				
	A1	A2	Both	Neither	
AA Same Date	8 (19.5%)	2 (4.9%)	30 (73.2%)	1 (2.4%)	
AA Later	2 (28.6%)	0	5 (71.4%)	0	
		,	Writing	,	
	A1	A2	Both	Neither	
AA Same Date	1 (2.4%)	3 (7.3%)	36 (87.8%)	1 (2.4%)	
AA Later	1 (14.3%)	0	6 (85.7%)	0	
	Watching TV				
	A1	A2	Both	Neither	
AA Same Date	6 (14.6%)	7 (17.1%)	19 (46.3%)	9 (22.0%)	
AA Later	2 (28.6%)	0	4 (57.1%)	1 (14.3%)	
	Listening to the radio				
	A1	A2	Both	Neither	
AA Same Date	6 (14.6%)	15 (36.6%)	14 (34.1%)	6 (14.6%)	
AA Later	2 (28.6%)	3 (42.9%)	2 (28.6%)	0	
	Socialising				
	A1	A2	Both	Neither	
AA Same Date	4 (9.8%)	8 (19.5%)	29 (70.7%)	0	
AA Later	1 (14.3%)	0	6 (85.7%)	0	

		Using the internet				
	A1	A2	Both	Neither		
AA Same Date	6 (14.6%)	3 (7.3%)	30 (73.2%)	2 (4.9%)		
AA Later	1 (14.3%)	0	5 (71.4%)	1 (14.3%)		

#### (6) Interpreting and translation:

Table III.8. Comparison of responses for the AA Later and AA Same Date groups to questions pertaining to interpreting

Group	Responses to questions concerning interpreting (freq and %)					
	L	Language(s) you mostly interpret into:				
	A1	A2	Both			
AA Same Date	12 (29.3%)	11 (26.8%)	18 (43.9%)			
AA Later	3 (42.9%)	3 (42.9%)	1 (14.3%)			
		Preferred interpreting direction				
	A1	A2	No preference			
AA Same Date	7 (18.4%)	7 (18.4%)	24 (63.2%)			
AA Later	2 (40.0%)	2 (40.0%)	1 (20.0%)			
	What is more important: interpreting skill or linguistic skill?					
	Interpreting skill	Linguistic skill	Both equally important			
AA Same Date	15 (37.5%)	4 (10.0%)	21 (52.5%)			
AA Later	1 (14.3%)	0	6 (85.7%)			

Table III.9. Number of respondents from the AA Later and AA Same Date groups who also carry out translation work

Group	Do you do any translation work? (freq and %)		
	Yes	No	
AA Same Date	20 (48.8%)	21 (51.2%)	
AA Later	3 (42.9%)	4 (57.1%)	

Table III.10. Languages respondents from the AA Later and AA Same Date groups mostly translate into

Group	Language(s) you mostly translate into (freq and %):		
	A1	A2	Both
AA Same Date	5 (25.0%)	3 (15.0%)	12 (60.0%)
AA Later	1 (33.3%)	0	2 (66.7%)

#### (7) Language maintenance effort and motivation:

Table III.11. Comparison of responses for the AA Later and AA Same Date groups to questions pertaining to language maintenance effort and motivation

Group	Responses to ques	tions on language maintenanc	e effort and motivation (freq and %)			
	Do you feel that maintaining your two A languages at their present high standard requires a degree of conscious effort?					
	Yes	No	Unsure			
AA Same Date	29 (70.7%)	10 (24.4%)	2 (4.9%)			
AA Later	4 (57.1%)	3 (42.9%)	0			
		he effort to maintain both of y ndard if not working as a pro	our A languages at their present high fessional interpreter?			
	Yes No Unsure					
AA Same Date	23 (57.5%)	6 (15.0%)	11 (27.5%)			
AA Later	5 (71.4%)	1 (14.3%)	1 (14.3%)			
		sible to maintain both of your ndard if not working as a pro	A languages at their present high fessional interpreter?			
	Yes	No	Unsure			
AA Same Date	18 (43.9%)	12 (29.3%)	11 (26.8%)			
AA Later	4 (57.1%)	0	3 (42.9%)			
	Do you seek to fill gaps in your linguistic knowledge when you notice them?					
	Yes	No	Unsure			
AA Same Date	39 (95.1%)	1 (2.4%)	1 (2.4%)			
AA Later	7 (100.0%)	0	0			
	Is your motivat	ion for filling gaps in linguisti	c knowledge purely professional?			
	Yes	No	Unsure			
AA Same Date	5 (12.5%)	33 (82.5%)	2 (5.0%)			
AA Later	2 (28.6%)	5 (71.4%)	0			

<sup>(8)</sup> Awareness of fluctuations in language dominance and cross-linguistic interference:

Table III.12. Responses of the AA Later and AA Same Date groups to the question of whether language dominance is perceived to fluctuate between their A languages

Group	Do you feel that language dominance fluctuates between your two A languages? (freq and %)				
	Yes	No	Unsure		
AA Same Date	22 (53.7%)	14 (34.1%)	5 (12.2%)		
AA Later	2 (28.6%)	5 (71.4%)	0		

Table III.13. Comparison of responses for the AA Later and AA Same Date groups to the question pertaining to perceptions of non-linguistic interference between the A languages

Group	Do you notice yourself making unnatural utterances in either of your two A languages due to the influence of the other language? (freq and %)					
	In neither	Mostly in A1	Mostly in A2	In both equally	Unsure	
AA Same Date	12 (29.3%)	7 (17.1%)	5 (12.2%)	12 (29.3%)	5 (12.2%)	
AA Later	2 (28.6%)	0	0	3 (42.9%)	2 (28.6%)	

#### (9) Emotional response and attachment to the languages:

Table III.14. Comparison of responses for the AA Later and AA Same Date groups to questions pertaining to emotional response and attachment to the languages

Group	Responses to questions on emotional response and attachment to the languages (freq and %)  If possible, would you revert to one of your two A languages when tired/upset?					
	No	A1	A2	Unsure		
AA Same Date	15 (36.6%)	15 (36.6%)	4 (9.8%)	7 (17.1%)		
AA Later	5 (71.4%)	1 (14.3%)	1 (14.3%)	0		
	Do you feel tha	t words carry the same v swear words and words				
	Equal	A1 more weight	A2 more weight	Unsure		
AA Same Date	23 (56.1%)	8 (19.5%)	3 (7.3%)	7 (17.1%)		
AA Later	6 (85.7%)	1 (14.3%)	0	0		
	Do you feel a	greater attachment towa	rds one of your A lang	uages over the other?		
	No	To A1	To A2	Unsure		
AA Same Date	19 (46.3%)	15 (36.6%)	2 (4.9%)	5 (12.2%)		
AA Later	5 (71.4%)	1 (14.3%)	0	1 (14.3%)		

#### (10) Self-ratings of proficiency in the two strongest languages:

Table III.15. Self-ratings of proficiency in the A languages relative to each other (by category) of the AA Later and AA Same Date groups

Self-ratings in the two languages relative to each other across difference categories (freq and %)					
Do you feel that your command of registers is equal in both of your A languages, or better in one language than the other?					
Equal	Better in A1	Better in A2	Unsure		
27 (65.9%)	7 (17.1%)	2 (4.9%)	5 (12.2%)		
4 (57.1%)	1 (14.3%)	1 (14.3%)	1 (14.3%)		
Do you have few	ver gaps in vocabulary	in one of your A lang other?	uages compared with the		
Equal	Fewer in A1	Fewer in A2	Unsure		
23 (56.1%)	6 (14.6%)	5 (12.2%)	7 (17.1%)		
2 (28.6%)	4 (57.1%)	0	1 (14.3%)		
Do you	write equally well/jus	t as easily in both of yo	our A languages?		
Equal	Better in A1	Better in A2	Unsure		
27 (65.9%)	7 (17.1%)	4 (9.8%)	3 (7.3%)		
2 (28.6%)	3 (42.9%)	2 (28.6%)	0		
Are you indisting	guishable from a mon	olingual in writing in b	ooth of your A languages?		
Yes – in both	In A1 only	In A2 only	Unsure		
30 (73.2%)	3 (7.3%)	1 (2.4%)	7 (17.1%)		
5 (71.4%)	2 (28.6%)	0	0		
	Do you feel tha  Equal 27 (65.9%) 4 (57.1%) Do you have feel  Equal 23 (56.1%) 2 (28.6%) Do you  Equal 27 (65.9%) 2 (28.6%) Are you indisting  Yes – in both 30 (73.2%)	Do you feel that your command of rebetter in one late	Do you feel that your command of registers is equal in both better in one language than the other		

Table III.16. Self-ratings of proficiency relative to an average monolingual standard in each language (by category) of the AA Later and AA Same Date groups

Group	Self-ratings in	each language relative to a different categories (fi	monolingual standard across req and %)			
		Richness of vocabula	ary in A1			
	Weaker	Same	Stronger			
AA Same Date	3 (7.3%)	17 (41.5%)	21 (51.2%)			
AA Later	0	6 (85.7%)	1 (14.3%)			
		Richness of vocabula	ary in A2			
	Weaker	Same	Stronger			
AA Same Date	2 (2.4%)	19 (46.3%)	21 (51.2%)			
AA Later	0	6 (85.7%)	1 (14.3%)			
		ehension in A1				
	Weaker	Same	Stronger			
AA Same Date	2 (4.9%)	25 (61.0%)	14 (34.1%)			
AA Later	0	7 (100.0%)	0			
	Instinctive/native comprehension in A2					
	Weaker	Same	Stronger			
AA Same Date	0	26 (63.4%)	15 (36.6%)			
AA Later	0	7 (100.0%)	0			
		Pragmatic/cultural comp	petence in A1			
	Weaker	Same	Stronger			
AA Same Date	1 (2.5%)	31 (77.5%)	8 (20.0%)			
AA Later	1 (14.3%)	4 (57.1%)	2 (28.6%)			
	Pragmatic/cultural competence in A2					
	Weaker	Same	Stronger			
AA Later	1 (2.4%)	30 (73.2%)	10 (24.4%)			
AA Same Date	0	5 (71.4%)	2 (28.6%)			

#### Appendix IV

Comparison of responses given by the subgroup of AHC AB interpreters who consider both of their languages to be mother tongues ("AB Two MTs" group) with those given by the rest of the AB group ("AB Not Two MTs" group).

#### (1) Age of SLA:

Table IV.1. Comparison of mean age of SLA in years for the AB Two MTs and AB Not Two MTs groups

Group	No. in group	Mean age SLA (years)	Std. Deviation
AB Two MTs	25	4.76	4.772
AB Not Two MTs	175	9.19	4.672

An independent samples t-test showed that the difference in mean ages of SLA between the two groups is significant and represents a medium sized effect: t(198) = 4.428, p = .000 (SE = 1.001); r = .300.

#### (2) Childhood linguistic environment and languages of primary and secondary education:

Table IV.2. Comparison of childhood linguistic environment (by category) and language(s) of primary and secondary education for the AB Two MTs and AB Not Two MTs groups

Group	Childhood lin	guistic environment/e	education category an	d responses (freq and		
		First la	anguage(s) spoken			
	A	В	Both	Other		
AB Two MTs	14 (50.0%)	10 (35.7%)	4 (14.3%)	0		
AB Not Two MTs	169 (94.9%)	4 (2.2%)	0	5 (2.8%)		
		Moth	ner's language(s)			
	A	В	Both	Other		
AB Two MTs	14 (50.0%)	14 (50.0%)	0	0		
AB Not Two MTs	156 (87.2%)	6 (3.4%)	4 (2.2%)	13 (7.3%)		
	Father's language(s)					
	A	В	Both	Other		
AB Two MTs	17 (63.0%)	8 (29.6%)	1 (3.7%)	1 (3.7%)		
AB Not Two MTs	161 (90.4%)	3 (1.7%)	6 (3.4%)	8 (4.5%)		
	Language(s) mostly spoken in the home during childhood					
	A	В	Both	Other		
AB Two MTs	14 (50.0%)	6 (21.4%)	8 (28.6%)	0		
AB Not Two MTs	170 (95.0%)	1 (0.6%)	5 (2.8%)	3 (1.7%)		
		Childhood lingu	istic environment (co	untry)		
	A	В	Both	Other/Unclear		
AB Two MTs	12 (42.9%)	1 (3.6%)	2 (7.1%)	13 (46.4%)		
AB Not Two MTs	137 (77.0%)	3 (1.7%)	12 (6.7%)	26 (14.6%)		
	Language(s) of primary education					
	A	В	Both	Other		
AB Two MTs	16 (57.1%)	7 (25.0%)	5 (17.9%)	0		
AB Not Two MTs	144 (81.8%)	6 (3.4%)	21 (11.9%)	5 (2.8%)		

	Language(s) of secondary education				
	A	В	Both	Other	
AB Two MTs	16 (57.1%)	7 (25.0%)	4 (14.3%)	1 (3.6%)	
AB Not Two MTs	119 (67.6%)	19 (10.8%)	37 (21.0%)	1 (0.6%)	

Table IV.3. Comparison of responses of the AB Two MTs and AB Not Two MTs groups to the question of whether they relocated to a different country in childhood

Group	Did you relocate to a different country during childhood? (freq and %)		
	Yes	No	
AB Two MTs	15 (53.6%)	13 (46.4%)	
AB Not Two MTs	40 (22.3%)	139 (77.7%)	

#### (3) Language(s) of higher education:

Table IV.4. Comparison of language(s) of higher education for the AB Two MTs and AB Not Two MTs groups

Group	Language(s) of higher education (freq and %)				
	A	В	Both	Other	
AB Two MTs	9 (34.6%)	9 (34.6%)	7 (26.9%)	1 (3.8%)	
AB Not Two MTs	54 (31.0%)	24 (13.8%)	94 (54.0%)	2 (1.1%)	

#### (4) Present linguistic environment and language use:

Table IV.5. Comparison of present linguistic environment for the AB Two MTs and AB Not Two MTs groups

Group		Present linguistic	environment (freq a	nd %)	
	Dominant language(s) of country of residence				
	A	В	Both	Other	
AB Two MTs	11 (39.3%)	7 (25.0%)	6 (21.4%)	4 (14.3%)	
AB Not Two MTs	85 (51.5%)	42 (25.5%)	14 (8.5%)	24 (14.5%)	
	Language(s) exposed to the most on a daily basis				
	A	В	Both	Other	
AB Two MTs	12 (42.9%)	3 (10.7%)	12 (42.9%)	1 (3.6%)	
AB Not Two MTs	84 (46.9%)	38 (21.2%)	51 (28.5%)	6 (3.4%)	

 ${\it Table~IV.6.~Comparison~of~self-reported~daily~language~use~(by~category)~for~the~AB~Two~MTs~and~AB~Not~Two~MTs~groups}$ 

Group		Language use catego	ry and responses (fre	q and %)		
			Reading			
	A	В	Both	Neither		
AB Two MTs	3 (10.7%)	3 (10.7%)	22 (78.6%)	0		
AB Not Two MTs	14 (7.8%)	15 (8.4%)	148 (82.7%)	2 (1.1%)		
			Writing	'		
	A	В	Both	Neither		
AB Two MTs	7 (25.0%)	1 (3.6%)	19 (67.9%)	1 (3.6%)		
AB Not Two MTs	33 (18.4%)	8 (4.5%)	136 (76.0%)	2 (1.1%)		
		W	atching TV			
	A	В	Both	Neither		
AB Two MTs	5 (17.9%)	9 (32.1%)	11 (39.3%)	3 (10.7%)		
AB Not Two MTs	26 (14.5%)	25 (14.0%)	108 (60.3%)	20 (11.2%)		
	Listening to the radio					
	A	В	Both	Neither		
AB Two MTs	5 (17.9%)	3 (10.7%)	12 (42.9%)	8 (28.6%)		
AB Not Two MTs	53 (29.6%)	25 (14.0%)	73 (40.8%)	28 (15.6%)		
	Socialising					
	A	В	Both	Neither		
AB Two MTs	3 (10.7%)	1 (3.6%)	21 (75.0%)	3 (10.7%)		
AB Not Two MTs	43 (24.0%)	19 (10.6%)	111 (62.0%)	6 (3.4%)		
		Usin	g the internet			
	A	В	Both	Neither		
AB Two MTs	3 (10.7%)	3 (10.7%)	19 (67.9%)	3 (10.7%)		
AB Not Two MTs	11 (6.1%)	25 (14.0%)	133 (74.3%)	3 (10.7%)		

#### (5) Interpreting and translation:

Table IV.7. Comparison of responses for the AB Two MTs and AB Not Two MTs groups to questions pertaining to interpreting

Group	Responses to questions concerning interpreting (freq and %)  Language(s) you mostly interpret into:				
	A	В	Both		
AB Two MTs	13 (46.4%)	10 (35.7%)	5 (17.9%)		
AB Not Two MTs	110 (61.5%)	45 (25.1%)	24 (13.4%)		
	Preferred interpreting direction				
	A	В	No preference		
AB Two MTs	16 (59.3%)	6 (22.2%)	5 (18.5%)		
AB Not Two MTs	106 (61.6%)	26 (15.1%)	40 (23.3%)		
	What is more important: interpreting skill or linguistic skill?				
	Interpreting skill	Linguistic skill	Both equally important		
AB Two MTs	8 (28.6%)	6 (21.4%)	14 (50.0%)		
AB Not Two MTs	79 (44.9%)	16 (9.1%)	81 (46.0%)		

Table IV.8. Number of respondents from the AB Two MTs and AB Not Two MTs groups who also carry out translation work

Group	Do you do any translation work? (freq and %)		
	Yes	No	
AB Two MTs	22 (78.6%)	6 (21.4%)	
AB Not Two MTs	125 (70.2%)	53 (29.8%)	

 ${\it Table~IV.9.}~Languages~respondents~from~the~AB~Two~MTs~and~AB~Not~Two~MTs~groups~mostly~translate~into$ 

Group	Language(s) you mostly translate into (freq and %):			
	A	В	Both	
AB Two MTs	6 (27.3%)	1 (4.5%)	15 (68.2%)	
AB Not Two MTs	61 (49.2%)	3 (2.4%)	60 (48.4%)	

#### (6) Language maintenance effort and motivation:

Table IV.10. Comparison of responses for the AB Two MTs and AB Not Two MTs groups to questions pertaining to language maintenance effort and motivation

Group	Responses to question	ons on language maintenanc	e effort and motivation (freq and o		
	Do you feel that maintaining your two strongest languages at their present high standard requires a degree of conscious effort?				
	Yes	No	Unsure		
AB Two MTs	19 (67.9%)	6 (21.4%)	3 (10.7%)		
AB Not Two MTs	131 (74.4%)	35 (19.9%)	10 (5.7%)		
	Would you make the effort to maintain both of your strongest languages at their present high standard if not working as a professional interpreter?				
	Yes	No	Unsure		
AB Two MTs	12 (44.4%)	8 (29.6%)	7 (25.9%)		
AB Not Two MTs	91 (51.7%)	46 (26.1%)	39 (22.2%)		
	Would it be possible to maintain both of your strongest languages at their present high standard if not working as a professional interpreter?				
	Yes	No	Unsure		
AB Two MTs	12 (42.9%)	8 (28.6%)	8 (28.6%)		
AB Not Two MTs	51 (29.0%)	85 (48.3%)	40 (22.7%)		
	Do you seek to fill gaps in your linguistic knowledge when you notice them?				
	Yes	No	Unsure		
AB Two MTs	27 (100.0%)	0	0		
AB Not Two MTs	174 (97.8%)	1 (0.6%)	3 (1.7%)		
	Is your motivation for filling gaps in linguistic knowledge purely professional?				
	Yes	No	Unsure		
AB Two MTs	4 (14.3%)	22 (78.6%)	2 (7.1%)		
AB Not Two MTs	33 (18 6%)	123 (69.5%)	21 (11.9%)		

<sup>(7)</sup> Awareness of fluctuations in language dominance and cross-linguistic interference:

Table IV.11. Responses of the AB Two MTs and AB Not Two MTs groups to the question of whether language dominance is perceived to fluctuate between their strongest languages

Group	Do you feel that language dominance fluctuates between your two strongest languages? (freq and %)			
	Yes	No	Unsure	
AB Two MTs	12 (42.9%)	15 (53.6%)	1 (3.6%)	
AB Not Two MTs	66 (36.9%)	101 (56.4%)	12 (6.7%)	

Table IV.12. Comparison of responses for the AB Two MTs and AB Two MTs groups to the question pertaining to perceptions of non-linguistic interference between their strongest languages

Group	Do you notice yourself making unnatural utterances in either of your two strongest languages due to the influence of the other language? (freq and %)					
	In neither	Mostly in A	Mostly in B	In both equally	Unsure	
AB Two MTs	8 (28.6%)	2 (7.1%)	8 (28.6%)	8 (28.6%)	2 (7.1%)	
AB Not Two MTs	42 (23.7%)	31 (17.5%)	53 (29.9%)	35 (19.8%)	16 (9.0%)	

#### (8) Emotional response and attachment to the languages:

Table IV.13. Comparison of responses for the AB Two MTs and AB Two MTs groups to questions pertaining to emotional response and attachment to the languages

Group	Responses to questions on emotional response and attachment to the languages (freq and %)  If possible, would you revert to one of your two strongest languages when tired/upset?					
AB Two MTs	9 (32.1%)	10 (35.7%)	5 (17.9%)	4 (14.3%)		
AB Not Two MTs	55 (31.3%)	93 (52.8%)	13 (7.4%)	15 (8.5%)		
	Do you feel that words carry the same weight in your two strongest languages (in particular, swear words and words such as "love" and "hate")?					
	Equal	A more weight	B more weight	Unsure		
AB Two MTs	15 (53.6%)	4 (14.3%)	2 (7.1%)	7 (25.0%)		
AB Not Two MTs	60 (34.1%)	71 (40.3%)	16 (9.1%)	29 (16.5%)		
	Do you feel a g	reater attachment towa	rds one of your strong other?	gest languages over the		
	No	То А	То В	Unsure		
AB Two MTs	12 (44.4%)	5 (18.5%)	7 (25.9%)	3 (11.1%)		
AB Not Two MTs	65 (36.7%)	78 (44.1%)	18 (10.2%)	16 (9.0%)		

<sup>(9)</sup> Self-ratings of proficiency in the two strongest languages:

Table IV.14. Self-ratings of proficiency in the strongest languages relative to each other (by category) of the AB Two MTs and AB Not Two MTs groups

Group	Self-ratings in the two languages relative to each other across difference categories (freq and %)  Do you feel that your command of registers is equal in both of your strongest languages, or better in one language than the other?					
	Equal	Better in A	Better in B	Unsure		
AB Two MTs	10 (35.7%)	16 (57.1%)	1 (3.6%)	1 (3.6%)		
AB Not Two MTs	48 (27.1%)	119 (67.2%)	1 (0.6%)	9 (5.1%)		
	Do you have fewer gaps in vocabulary in one of your strongest languages compared with the other?					
	Equal	Fewer in A	Fewer in B	Unsure		
AB Two MTs	11 (39.3%)	13 (46.4%)	2 (7.1%)	2 (7.1%)		
AB Not Two MTs	45 (25.6%)	110 (62.5%)	9 (5.1%)	12 (6.8%)		
	Do you wri	te equally well/just as	easily in both of your	strongest languages?		
	Equal	Better in A	Better in B	Unsure		
AB Two MTs	11 (39.3%)	13 (46.4%)	2 (7.1%)	2 (7.1%)		
AB Not Two MTs	60 (34.3%)	107 (61.1%)	3 (1.7%)	5 (2.9%)		

Table IV.15. Self-ratings of proficiency relative to an average monolingual standard in each language (by category) of the AB Two MTs and AB Not Two MTs groups

Group	Self-ratings in each language relative to a monolingual standard across different categories (freq and %)  Richness of vocabulary in A				
	Weaker	Same	Stronger		
AB Two MTs	1 (3.6%)	11 (39.3%)	16 (57.1%)		
AB Not Two MTs	2 (1.1%)	42 (24.0%)	131 (74.9%)		
	Richness of vocabulary in B				
	Weaker	Same	Stronger		
AB Two MTs	7 (25.9%)	13 (48.1%)	7 (25.9%)		
AB Not Two MTs	23 (13.9%)	87 (52.4%)	56 (33.7%)		
	Instinctive/native comprehension in A				
	Weaker	Same	Stronger		
AB Two MTs	0	12 (42.9%)	16 (57.1%)		
AB Not Two MTs	1 (0.6%)	58 (33.9%)	112 (65.5%)		
	Instinctive/native comprehension in B				
	Weaker	Same	Stronger		
AB Two MTs	2 (7.4%)	16 (59.3%)	9 (33.3%)		
AB Not Two MTs	30 (18.4%)	89 (54.6%)	44 (27.0%)		
	Pragmatic/cultural competence in A				
	Weaker	Same	Stronger		
AB Two MTs	0	8 (28.6%)	20 (71.4%)		
AB Not Two MTs	3 (1.7%)	58 (33.1%)	114 (65.1%)		
	Pragmatic/cultural competence in B				
	Weaker	Same	Stronger		
AB Two MTs	3 (11.5%)	9 (34.6%)	14 (53.8%)		
AB Not Two MTs	33 (19.8%)	75 (44.9%)	59 (35.3%)		